WORLD TRADE ORGANIZATION

WT/DS146/R WT/DS175/R

21 December 2001

(01-6327)

Original: English

INDIA – MEASURES AFFECTING THE AUTOMOTIVE SECTOR

Report of the Panel

The report of the Panel on *India – Measures Affecting the Automotive Sector* is being circulated to all Members, pursuant to the DSU. The report is being circulated as an unrestricted document from 21 December 2001 pursuant to the Procedures for the Circulation and Derestriction of WTO Documents (WT/L/160/Rev.1). Members are reminded that in accordance with the DSU only parties to the dispute may appeal a panel report. An appeal shall be limited to issues of law covered in the Panel Report and legal interpretations developed by the Panel. There shall be no *ex parte* communications with the Panel or Appellate Body concerning matters under consideration by the Panel or Appellate Body.

Note by the Secretariat: This Panel Report shall be adopted by the Dispute Settlement Body (DSB) within 60 days after the date of its circulation unless a party to the dispute decides to appeal or the DSB decides by consensus not to adopt the report. If the Panel Report is appealed to the Appellate Body, it shall not be considered for adoption by the DSB until after the completion of the appeal. Information on the current status of the Panel Report is available from the WTO Secretariat.



TABLE OF CONTENTS

I.	INTRODUCTION	Page
ı. II.	FACTUAL ASPECTS	
	THE LICENSING REGIME	
A.	PUBLIC NOTICE NO. 60 AND THE MOUS	
В.		
III.	PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS	
A.	US CLAIMS AND REQUESTS FOR FINDINGS	
B.	EC CLAIMS AND REQUESTS FOR FINDINGS	
C.	India's Requests for Findings	
IV.	ARGUMENTS OF THE PARTIES	
A.	FACTUAL ARGUMENTS	
1.	The licensing regime	
2.	Public Notice No. 60 and the MOUs	8
(a)	The "indigenization" requirement (paragraph 3 (iii) of Public Notice No. 60) and paragraph III(IV) of the MOU)	9
(b)	The "trade balancing" requirement (paragraph 3 (iv) of Public Notice No. 60 and paragraph III (vi) of the MOU)	10
B.	HAS THE MATTER BEEN SETTLED AND ADJUDICATED?	14
1.	Has the matter been settled?	16
2.	Has the matter already been adjudicated?	23
(a)	India – Quantitative Restrictions	23
(i)	The notion of res judicata	28
(b)	The provisions are an inherent part of the licensing scheme that must be eliminated	29
(i)	The notion of abusive splitting	32
C.	ARE THE MEASURES WITHIN THE TERMS OF REFERENCE OF THE PANEL?	36
1.	Do the measures applied after 1 April 2001 fall outside the terms of reference?	36
(a)	The method of enforcement before and after 1 April 2001	39
D.	CLAIMS UNDER GATT 1994 AND TRIMS	53
1.	Article III:4	53
(a)	The indigenization requirement	54
(b)	The trade balancing requirement	58
2.	Article XI:1	61
(i)	The indigenization requirement	
(ii)	The trade balancing requirement	
(a)	Are the measures justified by Article XVIII:B of the GATT and Articles 3 and 4 of the TRIMs Agreement?	

<i>(i)</i>	Procedural requirements of the balance of payments provisions	73
(ii)	Substantive Requirements of the balance of payments provisions	74
(iii)	The burden of proof	77
(iv)	Relevance of the India – Quantitative Restrictions case	81
3.	TRIMs Agreement	83
V.	ARGUMENTS OF THE THIRD PARTIES	88
1.	The Agreements regarding the Indian measures must be respected and enforced	88
2.	The measures that India may take by 1 April 2001 are not, and as a matter of logic cannot be, properly before the Panel.	89
3.	Prospective measures	90
VI.	INTERIM REVIEW	91
A.	COMMENTS ON THE DESCRIPTIVE PART	91
1.	Comments by the United States	91
2.	Comments by India	91
3.	Comments by the European Communities	92
B.	COMMENTS ON THE FINDINGS AND CONCLUSIONS	92
1.	Comments by the United States	92
2.	Comments by the European Communities	93
3.	Comments by India	94
(a)	General comment in respect of India's procedural rights in the proceedings	94
(b)	India's argument that the matters addressed in the Recommendations section should be left to a compliance panel.	96
(c)	India's argument that the Panel could not "properly" conclude that the violation is still in existence	98
(d)	India's argument that the Panel inappropriately prescribes a "retrospective" remedy	99
VII.	FINDINGS	100
A.	CLARIFICATION OF THE CLAIMS IN THE COURSE OF THE PROCEEDINGS AND REQUESTS FOR PRELIMINARY RULINGS	101
B.	CLARIFICATION OF THE MEASURES WITHIN THE TERMS OF REFERENCE OF THIS PANEL	104
1.	Measures identified in the complainants' requests for establishment of a panel	104
2.	Evolution of the measures in the course of the proceedings	106
3.	Measures not within our terms of reference	108
C.	THE PANEL'S COMPETENCE TO CONSIDER THE MATTER BEFORE IT	109
1.	Are any of the United States' claims barred by the principle of res judicata?	110
(a)	Arguments of the parties and scope of India's invocation of the doctrine of res judicata	110
(b)	Role of India's import licensing system in the United States' claims	111

(c)	General approach to the <i>res judicata</i> arguments	112
<i>(i)</i>	The matter before this Panel	116
(ii)	The matter ruled on by the India – Quantitative Restrictions panel	117
(iii)	Comparison between the matter ruled on by the India – QRs panel and the matter before this Panel	119
	A comparison of the measures expressly considered	120
	A comparison of the express claims	121
	Are the matters the same?	122
	Conclusion	124
1.	Has any part of the matter before this Panel been settled through a mutually agreed solution?	125
(a)	The mutually agreed solution in the <i>India – Quantitative Restrictions</i> dispute with the EC	125
(b)	Relevance of mutually agreed solutions in subsequent proceedings	126
(c)	The scope of the mutually agreed solution between the European Communities and India	127
(d)	Conclusion	131
2.	Measures to be eliminated in the course of implementing the results of the <i>India - Quantitative Restrictions</i> dispute and abusive splitting of a dispute	132
(a)	Abusive splitting	132
(b)	Unnecessary litigation	133
3.	Conclusion	134
D.	ORDER OF EXAMINATION OF THE CLAIMS	134
E.	CONSISTENCY OF THE INDIGENIZATION CONDITION WITH THE GATT 1994	137
1.	Claims under Article III:4 of GATT 1994	138
(a)	Like products	139
(b)	"Laws, regulations or requirements"	140
<i>(i)</i>	The notion of "requirement" within the meaning of Article III:4	140
(ii)	The indigenization condition as a "requirement"	142
(c)	affecting the internal sale, offering for sale, purchase () of the products	143
(d)	Whether imported products are accorded less favourable treatment	144
(e)	Conclusion	145
2.	Claim under Article XI	146
F.	GATT CLAIMS CONCERNING THE TRADE BALANCING CONDITION	146
1.	Factual aspects of the measure	146
2.	Order of analysis of the claims	147
3.	Relationship between Articles III and XI	148
4.	Claims under Article XI:1	149

(a)	Scope of the European Communities' claim	149
(i)	The scope of the trade balancing obligation	150
(ii)	The continued enforceability of the MOUs after 1 April 2001	151
(b)	Arguments of the parties	152
(c)	Scope of Article XI:1	153
<i>(i)</i>	The notion of "measures" within the meaning of Article XI:1	153
(ii)	"Restriction on importation" within the meaning of Article XI:1 and "border measures"	154
(d)	The trade balancing condition as a "restriction on importation"	156
(i)	The notion of "restriction on importation"	156
(ii)	Analysis of the trade balancing condition	159
5.	Balance-of-payments defense	160
6.	Claims under Article III:4	162
(a)	Purchases of restricted imported kits and components on the domestic market	162
<i>(i)</i>	Like products	163
(ii)	Laws, regulations or requirements	163
(iii)	affecting internal sale, use,	164
(iv)	Whether the measure affords less favourable treatment to imported products	165
7.	Ruling on the consistency of Public Notice No. 60 with the GATT 1994	165
(a)	Consistency of Public Notice No. 60 with Article III:4 of GATT 1994	165
(b)	Consistency of Public Notice No. 60 with Article XI:1 of the GATT 1994 in so far as it contains the trade balancing condition	166
G.	CLAIMS UNDER THE TRIMS AGREEMENT	167
VIII.	CONCLUSIONS AND RECOMMENDATIONS	167
A.	CONCLUSIONS	167
B.	CONSEQUENCES OF EVENTS HAVING TAKEN PLACE IN THE COURSE OF THE PROCEEDINGS	168
1.	Presentation of the issue: arguments of the parties	168
2.	Approach of the Panel	170
3.	Analysis	173
(a)	Indigenization provisions	174
<i>(i)</i>	The cessation of the application of Public Notice No. 60 and the MOUs as private contracts	174
(ii)	Achievement of the required levels of indigenization by MOU signatories	176
(b)	Trade balancing provisions	177
<i>(i)</i>	Accrued obligations as independent from imports	177

TABLE 2		182
TABLE 1180		
C.	RECOMMENDATIONS	179
,	contracts	178
(ii)	Cessation of the application of Public Notice No. 60 and the MOUs as private	



I. INTRODUCTION

- 1.1 In a communication dated 12 October 1998 (WT/DS146/1), the European Communities requested consultations with India pursuant to Article 4 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXII:1 of the General Agreement on Tariffs and Trade 1994 (GATT 1994) and Article 8 of the Agreement on Trade-Related Investment Measures (TRIMs Agreement), concerning certain measures affecting the automotive sector. On 21 and 22 October 1998, Japan and the United States, respectively, lodged requests to join in the consultations. The European Communities and India held consultations in Geneva on 2 December 1998. The United States and Japan participated in the consultations under DSU Article 4:11.
- 1.2 In a communication dated 2 June 1999 (WT/DS175/1), the United States requested consultations with India pursuant to Articles 1 and 4 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXII:1 of the General Agreement on Tariffs and Trade 1994 (GATT 1994) and Article 8 of the Agreement on Trade-Related Investment Measures (to the extent it incorporates by reference Article XXII of the GATT 1994) regarding certain measures affecting trade and investment in the motor vehicle sector. The United States and India held such consultations, in which Japan and the European Communities joined, on 20 July 1999. The consultations provided some helpful clarifications but failed to settle the dispute.
- 1.3 On 15 May 2000, the United States (WT/DS175/4) requested the Dispute Settlement Body to establish a panel with standard terms of reference as set out in Article 7.1 of the DSU. The Panel was established on 27 July 2000. On 12 October 2000, the European Communities (WT/DS146/4) also requested the DSB to establish a panel pursuant to Article 4.7 and 6 of the DSU.
- 1.4 At its meeting of 17 November 2000, the DSB agreed that, in accordance with Article 9.1 of the DSU, the Panel established on 27 July 2000 to examine the complaint by the United States should also examine the complaint by the European Communities contained in WT/DS146/4 and agreed that the Panel should have standard terms of reference. The terms of reference of the Panel were therefore as follows:

"To examine, in the light of the relevant provisions of the covered agreements cited by the United States in WT/DSB/175/4 and by the European Communities in WT/DS146/4, the matters referred to the DSB by the United States and the European Communities in those documents and to make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements."

1.5 On 24 November, in accordance with Article 8.7 of the DSU, the Director-General composed the Panel as follows:

Chairman: Mr. John Weekes

Members: Ms. Gloria Peña

Mr. Jeffrey Waincymer

1.6 Japan and Korea reserved their third party rights in the dispute.

¹ WT/DS146/2 and WT/DS146/3.

1.7 The Panel met with the Parties on 21 and 22 March 2001 and 17 and 18 May 2001. The Panel held a third-party session on 22 March. Korea presented its submission. Japan did not make a submission.

II. FACTUAL ASPECTS

A. THE LICENSING REGIME

- 2.1 For many years, India applied import restrictions for balance of payment reasons. On 24 July 1997 the European Communities requested consultations with respect to all the import restrictions maintained by India on balance-of-payment grounds, including those on passenger cars, and chassis and bodies therefore.² As a result of those consultations, the European Communities and India reached a mutually agreed solution on 25 November 1997 (the "1997 Agreement").³ The 1997 Agreement stipulated that the import restrictions on passenger cars, and on chassis and bodies therefor, were to be eliminated no later than 31 March 2003. The 1997 Agreement, nevertheless, had to a large extent been superseded by the agreement concluded between India and the United States on 24 December 1999⁴ referred to below.
- 2.2 On 22 July 1997 the United States requested consultations under the DSU with respect to quantitative restrictions maintained by India for balance-of-payments reasons on 2,714 agricultural and industrial product tariff lines.⁵ A panel was established on 18 November 1997.⁶ The panel concluded that the restrictions applied by India, violated GATT Article XI:1 of GATT and were not justified by GATT Article XVIII:B.⁷ The Appellate Body upheld those findings.⁸
- 2.3 On 24 December 1999 India and the United States agreed, pursuant to Article 21.3(b) of the DSU, that the reasonable period of time for complying with the recommendations and rulings of the DSB would expire on 1 April 2001. Accordingly, India was required to eliminate the current system of non-automatic licenses for imports of passenger cars, and chassis and bodies therefor, no later than 1 April 2001. At the time when the United States (15 May 2000) and the European Communities (12 October 200) submitted their requests for the establishment of this panel, India applied discretionary import licensing to 715 tariff line items, including cars imported in the form of completely and semi-knocked down ("CKD/SKD") kits. The import licensing scheme for cars imported in the form of CKD/SKD kits was abolished on 1 April 2001 in accordance with the

² WT/DS96/1. The United States (WT/DS90/1), Australia (WT/DS91/1), Canada (WT/DS92/1), New Zealand (WT/DS93/1) and Switzerland (WT/DS94/1) also requested consultations with respect to the same measures.

³ The Agreement was notified under Article 3.6 DSU (WT/DS96/8). India reached similar Agreements with Australia, Canada, New Zealand, Switzerland and Japan, but not with the United States.

⁴ Paragraph 3 of the 1997 Agreement provides that:

[&]quot;India shall grant to the EC treatment no less favourable than that granted by India to any other country with respect to the elimination or modification of import restrictions on the products in the Annex and those in Annex II of document WT/BOP/N/24, under any form, either autonomously or pursuant to agreement or understanding with that country, including pursuant to the settlement of any outstanding dispute under the WTO Understanding on Rules and Procedures Governing the Settlement of Disputes".

⁵ WT/DS90/1.

⁶ WT/DS90/8 and WT/DS90/9.

⁷ Panel Report on *India – Quantitative Restrictions on Imports of Agricultural, Textile and Industrial Products*, (hereinafter "*India – Quantitative Restrictions*"), WT/DS/90/R, adopted 22 September 1999, at para. 6.1. The report addresses the compatibility with GATT Article XI:1 of the licenses required for the importation of "restricted" items at paras. 5.125 *et seq.*

⁸ Report of the Appellate Body in *India – Quantitative Restrictions*, WT/DS90/AB/R, adopted on 22 September 1999.

⁹ WT/DS90/15.

agreements reached with the European Communities and the United States under Articles 3.6 and 21.3(b) of the Dispute Settlement Understanding ("DSU").

B. PUBLIC NOTICE NO. 60 AND THE MOUS

- 2.4 On 12 December 1997, the Ministry of Commerce adopted Public Notice No. 60, the auto components licensing policy, issued under the Foreign Trade (Development and Regulation) Act of 1992 (the "FTDR Act").
- 2.5 The Notice, effective from the date of issue, requires any passenger car manufacturer wishing to import SKD/CKD kits to sign a Memorandum of Understanding (MOU) with the Director General of Foreign Trade ("DGFT") (see Annex Tables 1 and 2). A car manufacturer that does not sign an MOU or does not perform the obligations assumed under an MOU may therefore be denied a license for the importation of SKD/CKD kits. Subparagraphs 3(i) through (iv) of Public Notice No. 60 set out four requirements which an MOU must impose on the manufacturing company:
 - (i) Establishment of actual production facilities for manufacture of cars, and not for mere assembly.
 - (ii) A minimum of foreign equity of US\$50 million to be brought in by the foreign partner within the first three years of the start of operations, if the firm is a joint venture that involves majority foreign equity ownership.¹⁰
 - (iii) Indigenization (i.e. local content) of components up to a minimum level of 50% in the third year or earlier from the date of first import consignment of CKD/SKD kits/components, and 70% in the fifth year or earlier.
 - (iv) broad trade balancing of foreign exchange over the entire period of the MOU, in terms of balancing between the actual CIF value of imports of CKD/SKD kits/components and the FOB value of exports of cars and auto components over that period. While a firm that signs an MOU has an export obligation equivalent to the total CIF value of the imports made by the firm over the period of the MOU, there is a two-year moratorium during which the firm does not need to fulfill that commitment. The period of export obligation therefore begins from the third year of commencement of production. However, imports made during the moratorium count towards the firm's total export obligation under the MOU.

III. PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS

A. US CLAIMS AND REQUESTS FOR FINDINGS

3.1 The **United States** contended that the local content and trade balancing requirements in Public Notice No. 60 and the MOUs, together with the Indian domestic legislation under which they had come into force, were inconsistent with the obligations of India under the General Agreement on Tariffs and Trade 1994 ("GATT 1994") and the Agreement on Trade-Related Investment Measures ("TRIMs Agreement"). The United States requested the Panel to find that the measures at issue in this

¹⁰Paragraph 3(ii) also provides that this condition applies to new joint venture companies only. In response to a question from Japan, India stated that "this condition has been imposed on new joint venture companies because the existing companies have already invested more than their minimum stipulation." *Replies by India to Questions Posed by Japan*, G/TRIMS/W/15, circulated 30 October 1998, response to Question 18.

dispute, the indigenization and the trade balancing requirements, imposed by Public Notice No. 60 and the MOUs, were inconsistent with Article III:4 and XI:1 of GATT 1994 and Articles 2.1. and 2.2 of the TRIMs Agreement. The United States requested this Panel to make findings to this effect, and to recommend that India bring all such measures into conformity with its obligations.

3.2 The United States considered it would be appropriate for the Panel to rule on Public Notice No. 60 given that the requirements were still in place through the MOUs. The United States said it was not requesting the Panel to make findings on Public Notice No. 60 that were separate from its findings on the MOUs executed thereunder. The United States was, however, requesting that the Panel's findings encompass both Public Notice No. 60 and the individual MOUs. A comprehensive ruling by the Panel would apply to both Public Notice No. 60 (the source of the WTO-inconsistent requirements) and the MOUs (by which individual companies were bound to those WTO-inconsistent requirements), especially since Public Notice No. 60 was, as all the parties to this dispute agreed, in force at the time this Panel was established. However, the Panel need not analyze Public Notice No. 60 separately from the MOUs in order to make its findings. 12

B. EC CLAIMS AND REQUESTS FOR FINDINGS

- 3.3 The European Communities said that its complaint was directed primarily against the indigenization and balancing requirements contained in the MOUs concluded under Public Notice No. 60. 13
 - The European Communities claimed that the indigenization requirement violated Article III:4. It had not claimed that this requirement violates also Article XI:1.
 - The European Communities claimed that the balancing requirement was inconsistent with both Article XI:1 and Article III:4 because the signatories were required to balance the value of the automotive products which they import themselves, as well as the value of any imported automotive products which they purchase from vendors within India¹⁴.
 - To the extent that the balancing requirements required the signatories to balance the value of the imported products which they purchase within India they were inconsistent with Article III:4.
 - To the extent that the balancing requirements required the signatories to balance the value of the automotive products which they imported themselves, they restricted the importation of such products, thereby violating Article XI:1.
- 3.4 It also requested the Panel to make an express factual finding regarding the product scope of the trade balancing requirement.
- 3.5 The European Communities reiterated its request to the Panel to find that the trade balancing and indigenization requirements contained in Public Notice No. 60 and in the MOUs concluded thereunder were inconsistent with Articles III:4 and XI:1 of the GATT and Article 2.1 of the TRIMs

¹¹ See also Questions 1 and 17 of the Panel.

¹² See also US response to Question 72 of the Panel.

¹³ The EC's complaint did not extent to the MOUs "as a whole". More precisely, the European Communities did not challenge the other two requirements imposed by the MOUs (i.e. the minimum foreign equity requirement and the manufacturing requirement).

¹⁴ See EC First Submission, at paras. 32 and 74.

Agreement as of the date of establishment of this Panel and had remained so after 1 April 2001. Furthermore, the European Communities requested the Panel to find that, at the time when the Panel was established, Public Notice No. 60 was inconsistent with Article XI:1 and III:4 of GATT and Article 2.1 of the TRIMs Agreement. It clarified that the measures applied after 1 April 2001 continued to be inconsistent with the GATT and the TRIMs Agreement because they were the same as the measures applied before that date. 15

C. INDIA'S REQUESTS FOR FINDINGS

- 3.6 As to the claim that Public Notice No. 60 and the trade balancing provisions of the MOUs are inconsistent with Articles III and XI of the GATT, India requested the Panel to make the following findings and recommendations:
- 3.7 If the Panel were to conclude that the matter before it is the operation of Public Notice No. 60 and the trade balancing provisions of the MOUs under India's former licensing regime, it should reject the complaints as inadmissible because this matter had already been satisfactorily resolved and settled as a result of the previous invocations of the DSU by the complainants. Specifically, India requested the Panel to find that:
- the matter raised by the European Communities had already been settled through a mutually agreed solution jointly notified by the European Communities and India to the DSB in accordance with Article 3.6 of the DSU,
- the matter raised by the United States had already been adjudicated by the DSB in the proceedings on *India Quantitative Restrictions on Imports of Agricultural, Textile and Industrial Products* (hereinafter "*India Quantitative Restrictions*") (WT/DS90), and
- the re-invocation of the DSU in respect of the matters raised by the European Communities and the United States was inadmissible.
- 3.8 If the Panel were to conclude that the matter before it was the operation of Public Notice No. 60 and the trade balancing provisions under India's former licensing regime and were to regard the complaints as admissible, it should examine India's invocation of Article XVIII:B of the GATT and, in accordance with the precedent set by the Panel in *India Quantitative Restrictions*, consult with experts under Article 13 of the DSU, including the IMF. 16
- 3.9 If the Panel were to conclude that the operation of Public Notice No. 60 and the trade balancing provisions under India's former licensing regime was inconsistent with the GATT, it should following the practice of previous panels¹⁷ note that the licensing regime had been abolished on 1 April 2001, that the Public Notice and the trade balancing provisions no longer applied since 1 April 2001 and that it was therefore not necessary for the Panel to recommend to the DSB in accordance with Article 19:1 of the DSU that it request India to bring these measures into conformity with the GATT.
- 3.10 If the Panel were to conclude that the matter before it is the operation of Public Notice No. 60 and the trade balancing provisions of the MOUs outside India's former import licensing regime, the

¹⁵ See Question 87 of the Panel.

¹⁶ See the Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, paras 5.11 through 5.13.

¹⁷See the Panel Report on *Indonesia – Certain Measures Affecting the Automobile Industry*, (hereinafter "*Indonesia – Autos*"), WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted 23 July 1998, para. 14.9 and the reports cited therein (DSR 1998:VI, 2201).

Panel should reject the complainants' claims as factually baseless. Specifically, it should find that, in the absence of an import licensing regime, Public Notice No. 60 is not operational because it merely states how the former restrictions were to be administered and the trade balancing provisions do not entail any export obligation and hence do not restrict imports or favour domestic over imported products because they do not apply to freely importable products.

- As to the claim that the indigenization provisions set out in the MOUs are inconsistent with Article III of the GATT, India requested the Panel to find that the Government of India had, since the abolition of the import licensing regime the discretion whether or not to apply its contractual rights under the MOUs and that this raised new legal issues not covered by the Panel's terms-of-reference. If the Panel were to rule that the indigenization requirements in the MOUs are inconsistent with Article III:4, it should find that, according to the evidence submitted by the parties, there remained as of May 2001 only one company that was still bound by the indigenization requirement with respect to one car model. India therefore requested the Panel to limit any recommendation under Article 19:1 of the DSU to the indigenization requirement that remains to be performed by that particular company with respect to that model.
- 3.12 <u>As to the claims related to the TRIMs Agreement</u>, India requested the Panel to find that the provisions on trade balancing and indigenization set out in Public Notice No. 60 and the MOUs are trade measures and not investment measures and therefore do not fall within the framework of the TRIMs Agreement.

IV. ARGUMENTS OF THE PARTIES

A. FACTUAL ARGUMENTS

1. The licensing regime

4.1 The **European Communities** noted that, at the time the Panel was established, India regulated the imports of goods by means of a Negative List contained in the Export-Import Policy (the "EXIM Policy)¹⁸. The Negative List¹⁹ consisted of "prohibited"²⁰ items, "restricted" items and "canalised"²¹ items. "Restricted" items were permitted to be imported only against a specific import license issued by the Director General of Foreign Trade in India's Ministry of Commerce ("DGFT"²²). The granting of import licenses for "restricted" items was not automatic²³. Licenses were granted on a discretionary basis or, in some cases, subject to the conditions set out in advance in a Public Notice.

¹⁸ The EXIM Policy is formulated and announced from time to time by the Indian Government on the basis of the authority conferred by the Foreign Trade Development and Regulation Act, 1992 (hereinafter the "FTDR" Act) (cf. Section 5 of the FDTR). The policy currently in effect is the Export and Import Policy 1 April 1997 – 31 March 2002

^{1997 – 31} March 2002.

19 Chapter 16 of the EXIM Policy. The Negative List is further specified in the "ITC (HS) Classifications of Export and Import Items", published and notified by the DGFT (cf. Section 4.1 of the EXIM Policy).

Section 4.4 of the EXIM Policy ("Prohibited items in the Negative List of imports shall not be imported [...]").

imported [...]").

²¹ Section 4.8 of the EXIM Policy ("Any goods, the import or export of which is canalised, may be imported by the canalising agency specified in the Negative Lists [...]").

²² Section 4.5 of the EXIM Policy ("Any goods, the export or import of which is restricted through licensing, may be exported or imported only in accordance with a license issued in this behalf").

²³ Section 4.7 of the EXIM Policy ("No person may claim a license as a right and the Director General of Foreign Trade or the licensing authority shall have the power to refuse to grant or renew a license in accordance with provisions of the Act and the Rules made thereunder").

- 4.2 Under the EXIM Policy 1997-2002, passenger cars (HS²⁴ 87.03), and chassis (HS 87.06) and bodies (HS 87.07) therefor, were classified as "restricted" items in the Negative List. In contrast, parts and accessories for passenger cars (including *inter alia* those falling within HS 87.08, 40.11, 40.12, 40.13, 70.07, 70.09, 90.29, and 94.01) were not included in the Negative List and could, therefore, be freely imported.
- 4.3 In accordance with Rule 2(a) of the General Rules for the Interpretation of the HS nomenclature²⁵, the Indian customs authorities classify within HS 87.03 complete passenger cars that are presented unassembled ("CKD" kits), as well as incomplete passenger cars, whether assembled or disassembled ("SKD" kits), provided that they have the "essential character" of a complete car²⁶. According to the Indian Government, imports of parts and components were considered to have the "essential character" of a passenger car where, at least, all the following seven items were imported: engine, gear box, chassis, transmission assembly system, body/cab, suspension system and axles front and rear²⁷.
- 4.4 The **United States** stated that India applied licensing requirements both to finished vehicles as such and to SKD/CKD kits/components for such vehicles. Consequently, when Public Notice No. 60 was issued, no one was permitted to import finished motor vehicles, or kits and components in SKD/CKD form for such vehicles, into India without a license. At the time Public Notice No. 60 was issued, 90 tariff line items in Chapter 87 ("vehicles other than railway or tramway rolling-stock, and parts and accessories thereof") of India's tariff schedule were "restricted" and therefore subject to import licensing requirements.²⁸ At the time the panel was established, these import licensing requirements still applied to approximately 75 tariff line items in Chapter 87.
- 4.5 **India** clarified that some tariff lines covering cars in the form of CKD/SKD kits that were identified as restricted for "AUTO/BOP-XVIII-B" reasons in India's notification to the BOPs Committee (WT/BOP/N/24) required licenses under the MOUs. All tariff lines covering components for cars that were identified in the notification as restricted, including for "BOP-XVIII :B" reasons, also required licenses under the MOUs. Apart from CKD/SKD kits of cars, chassis with engines mounted for passenger motor cars and bodies (initial equipment and replacement parts) were also under Restricted Category requiring an import licence. As per Chapter 87, parts and accessories of motor vehicles like bumpers, safety seat belts, mounted brake linings, gear boxes, drive-axles, non driving axles, road wheels, suspension shock absorbers, radiators, silencers and exhaust pipes, clutches, steering wheels, steering columns and steering boxes, etc. were under free category.²⁹

 $^{^{24}}$ The term "HS" refers to the Harmonized Commodity Description and Coding System evolved by the World Customs Organisation.

²⁵ Rule 2(a) of the General Rules for the Interpretation of the HS reads as follows:

[&]quot;Any reference in a heading to an article shall be taken to include a reference to that article incomplete or unfinished, provided that, as presented, the incomplete or unfinished article has the essential character of the complete or finished article. It shall also be taken to include a reference to that Article complete or finished (or falling to be classified as complete or finished by virtue of this rule), presented unassembled or disassembled"

 $^{^{26}}$ India's responses to Questions by the EC (hereinafter "EC Questions"), dated 9 May 1999, Response to Question 1(a).

²⁷ See Public Notice No. 3/98 issued by the Delhi Customs House on 6 January 1998.

 $^{^{28}}E.g$, tariff heading 870321.01 ("Other vehicles, with spark-ignition internal combustion reciprocating piston engine; Of a cylinder capacity not exceeding 1,000 cc; Motor car, new, assembled") is classified as "Restricted: Not permitted to be imported except against a licence or in accordance with a Public Notice issued in this behalf."

²⁹ It should be noted that any part/component of a car e.g., gear box, radiator, silencer, etc. which were under free category could be imported without any licence as long as these were imported individually.

2. Public Notice No. 60 and the MOUs

- India explained that Public Notice No. 60 clarified policies first adopted in 1995. The 4.6 revision of the MOU scheme in 1997 was designed to render the existing policies more uniform and transparent. The parameters applied in 1995 were individually negotiated with each car manufacturer wishing to import CKD/SKD kits. As from 1997, the parameters followed a common format; the product coverage was the same.³⁰ Public Notice No. 60 improved upon the 1995 MOUs by removing the discretionary element from licensing, by stipulating uniform commitments for all MOU signatories, and by making the granting of licenses "automatic" upon fulfilment of the conditions stipulated therein.31
- 4.7 India went on to explain that Public Notice No. 60 was adopted to ensure the uniform and transparent administration of these restrictions. It declared that only car manufacturers that had signed an MOU with provisions on foreign equity, assembly operations, indigenization and trade balancing would obtain a license and that a license could be denied if the MOU provisions were not observed. India explained that only the car manufacturers that wished to import SKD/CKD kits were required to sign an MOU. This requirement did not apply to any manufacturer that did not wish to import such kits. By accepting the commitments set out in the MOUs, a car manufacturer obtained a significant economic advantage, namely the quota rents associated with the importation of SKD/CKD kits within the framework of India's restrictive licensing scheme. The quota rents enjoyed by the manufacturers, on the one hand, and the commitments to take actions to mitigate the adverse balance-of-payments impact of their imports, on the other hand, constituted a balanced whole.
- The European Communities pointed out that Public Notice No. 60 had no termination date.³³ 4.8 Nevertheless, according to the Indian Government, "once the quantitative restrictions which cover autos and auto parts in the phase out plan are removed, the Public Notice. 60 shall cease to operate". 34 By its own terms, Public Notice No. 60 applied only to joint ventures between Indian and foreign investors.³⁵ Nevertheless, the Indian Government had indicated that companies wholly owned by Indians are subject to the same requirements.³⁶ Public Notice No. 60 applied to both "existing" (i.e. those established before 12 December 1997) and "new" joint ventures.³
- 4.9 The United States noted that Paragraph 4 of Public Notice No. 60 provided that the MOU scheme was to be enforced through the import licensing mechanism. Paragraph 5 provided for MOU signatories to submit annual reports and for Indian officials to conduct an annual review of signatories' progress towards meeting their MOU obligations. The United States noted that Paragraph 8 of Public Notice No. 60 required MOUs to be signed in a standard format, which was appended to the Notice.³⁸ The MOU was to be signed by the Government of India acting through the

However, in case, these parts/components formed a part of a CKD/SKD kit of a car itself was under restricted category and required a licence for its import into the country prior to 31.3.2001.

Response to Question 43 of the Panel.

³¹ Response to Question 45 of the Panel.

³² In practice, the Director-General of Foreign Trade never denied a license to a signatory for failing to meet the indigenization or trade balancing requirements. (Panel Question 51).

³³ India's responses to Questions from the United States dated 10 May 1999 (hereinafter "US Questions"), response to Question 10. ³⁴ *Ibid*.

³⁵ Public Notice No. 60, at paras. 1 and 2.

³⁶ US Questions. In response to Question 3, India stated that "Indian companies importing CKD/SKD kits are also being subjected to the same requirements of indigenization as well as foreign exchange trade balancing". See also India's response to Question 16.

37 Public Notice No. 60, at para. 2.

³⁸ See Table 2.

Director General of Foreign Trade and by the managing director of the manufacturing company on its behalf. Paragraph III of the standard MOU format reconfirmed each of the requirements of Public Notice No. 60. The requirement to establish actual manufacturing facilities, and not mere assembly facilities, appeared in MOU paragraph III, clause (iii); the US\$50 million equity requirement appeared in MOU paragraph III, clause (ii); the indigenization obligation appeared in MOU paragraph III, clause (iv); and the trade balancing obligation appeared in MOU paragraph III, clause (vi). All or virtually all companies manufacturing automobiles in India had signed an MOU in accordance with the standard format, and it was easy to see why. Because companies were not granted an import license for CKD/SKD kits until an MOU was signed, India's policy effectively prohibited any company from importing CKD/SKD kits and components unless it abided by the terms and conditions of Public Notice No. 60 and the MOU.

- (a) The "indigenization" requirement (paragraph 3 (iii) of Public Notice No. 60) and paragraph III(IV) of the MOU)
- 4.10 The **European Communities** described its understanding of the "indigenization" requirement as follows: the MOU signatories must achieve a level of indigenization of 50% in the third year from the first import or earlier, and of 70% no later than the fifth year.³⁹ The Indian Government had confirmed that "indigenization" meant "local content" The European Communities understood that in practice the percentage of indigenization was computed in accordance with the following formula:

(CIF value of imported parts and materials assembled into the passenger car / ex works value of the passenger car) x 100

- 4.11 The European Communities further understood that the numerator included not only the value of the parts and materials imported directly by the MOU signatory but also the value of any imported parts and materials purchased in India by the joint ventures from local vendors. Public Notice No. 60 did not prescribe whether the percentage of indigenization was to be measured on a model, category or manufacturer basis. It appeared that, in practice, the Indian authorities might allow the manufacturers to meet the requirement on any of those bases.⁴¹
- 4.12 The United States noted that once an MOU signing firm had reached an indigenization level of 70%, there would be no need for further import licenses. By implication, licenses would be withheld if the targets were not met. India had confirmed that the MOU limited the importation of SKD/CKD kits/components when a firm failed to meet the indigenization requirement. The standard MOU format also imposed other requirements on the manufacturing firm (e.g.,, it required the firm to state its intended level of indigenization year by year). Significantly, the MOU also required that the firm "shall aggressively pursue and achieve as soon as possible the development of the local supply base and increased local content". This requirement highlighted the intentions that underlay the MOU scheme: to induce manufacturing firms to abandon imported goods and to favor domestic goods instead. With respect to automotive components, the Secretariat's report for the Indian Trade Policy Review noted that the value of their production rose from Rs33.6 million in 1990/91 to Rs67.5 billion by 1994/95 and Rs91 billion in 1995/96. Indian authorities estimated the level of indigenization in the automobile components sector at around 90%. This high level of indigenization was due in part to a number of previous requirements that companies investing in the automobile sector in India must agree to increase the level of indigenization in their units within a certain period of time. Public Notice No. 60 and the MOUs continued in this tradition of nurturing the domestic auto parts and components industry. Indeed, that was the very reason that Indian officials gave for the adoption of

 41 Ibid.

³⁹ *Ibid.*, para (iii).

⁴⁰ US Questions, response to Question 6.

Public Notice No. 60: "The objective of the new policy is to encourage local production of auto-components and thus, bring in modern technology and develop this key segment, explain ministry officials"⁴².

India pointed out that the signatories of the MOUs were required to reach a level of 70 per cent of indigenization. An MOU was effectively terminated on the date of the achievement of an indigenization level of 70 per cent, except for the export obligations that accrued before that date. In this regard, it was pertinent to note that, according to the WTO Secretariat report prepared for the 1998 Trade Policy Review of India, the level of indigenization in the automobile components sector was estimated to be around 90 per cent. 43 Moreover, according to as yet unverified reports received from car manufacturers in the autumn of 2000, most of the signatories had almost reached the 70 per cent threshold. By 1 April 2001, there might consequently no longer be any signatory bound by an indigenization requirement. India was examining this situation. India was committed to resolving it in a manner that did not entail any inconsistencies with its WTO obligations. In late April and early May of 2001, officials of the Ministry of Commerce and Industry met with the most important car manufacturers from the EC, Japan and the United States to verify their performance under the MOU. Subsequent to these meetings, these signatories confirmed in writing that they had all achieved or exceeded the 70 per cent threshold. The only exception was General Motors India, which had achieved only a 60 per cent level of indigenization in respect of one of its car models. However, this company wrote in a letter dated 4 May 2001 to the Commerce Secretary:

Localisation for any player is imperative in order to be competitive in the market. Keeping this in mind, we have . . crossed 70% location in the case of Astra much before the target date as envisaged in the MOU. . . In the case of Corsa, although launched only about a year back, we had already localised 60.31 % and intend to cross 70 % localisation during this fiscal year . . .

- 4.14 The results of the survey clearly confirmed that car manufacturers had had no difficulty meeting the 70 per cent threshold and that, except in respect of one recently launched car model, all surveyed manufacturers reached the threshold in the meantime. Indeed, the survey showed that indigenization was part of the manufacturers drive towards greater competitiveness.
- The "trade balancing" requirement (paragraph 3 (iv) of Public Notice No. 60 and (b) paragraph III (vi) of the MOU)
- 4.15 The European Communities understood that the requirement to achieve "broad trade balancing of foreign exchange" translated into the obligation to balance the CIF value of imports of "CKD/SKD kits/Components" and the FOB value of exports of "cars and auto components" over the entire period of duration of the MOU.⁴⁴ The MOU signatories were not required to start exporting until the third year from the commencement of production. ⁴⁵ But the value of imports made during the first two years must be balanced in subsequent years. 46 Public Notice No. 60 further specified that from the fourth year onwards "the value of imports of CKD/SKD may be regulated with reference to the export obligation fulfilled in the previous years as per the MOU". The precise meaning of this provision, however, remained unclear. ⁴⁷ The wording of Public Notice No. 60 suggested that the

⁴⁶ Ibid.

⁴² "Car Makers Have to Sign New MOUs in Accordance with New Automobile Policy", Business Standard, December 11, 1997.

⁴³ Trade Policy Review, India: Report by the Secretariat (WT/TPR/S/33), 5 March 1998, para. 109.

⁴⁴ Public Notice No. 60, para 3 (iv). See also Japan's Questions, response to Question 23.

⁴⁷ See Japan's Questions, response to Question 24, and US Questions, response to Questions 7 and 8.

balancing obligation extended not only to imports of "CKD/SKD kits" but also to imports of "components". That interpretation had been corroborated to the European Communities by some signatories of MOUs. During the consultations, however, India asserted that "the signatories have to neutralise only the value of imported CKD/SKD kits and not of those components which are freely imported". 48 It was also unclear whether the balancing requirement applied to imported kits and components purchased by the signatories within India, or only to those which they import themselves. Before the TRIMs Committee India explained that "[b]road trade balancing of foreign exchange ... includes the purchases in India of the imported components or CKD/SKD kits"⁴⁹. Yet, during the consultations, India answered to a question from the United States that "the signatories of the MOU have to neutralise only the value of CKD/SKD imported by them⁵⁰

- The United States added that Public Notice No. 60 also made clear that the trade balancing obligation remained in place independently of the indigenization obligation: [i.e. firms that have achieved 70% indigenization] will discharge the export obligation corresponding to the import made by them till that time." India had confirmed that a company's imports could be limited under this provision if it had not met its export obligations. India did not intend to terminate the requirements of Public Notice No. 60 or the MOUs on 1 April 2001. India had told the United States directly that the MOUs remained effective even after 1 April 2001. Moreover, Indian officials had told the press that automobile manufacturers who had entered into MOUs in the past would have to honour their commitments in terms of exports, localisation level, and minimum capital invested in the venture. India's Director-General of Foreign Trade himself had confirmed India's intentions. According to published reports, in response to "domestic auto-ancillary units, who felt threatened by the removal of QRs [and] have appealed to the government to protect their interests", the Director-General had "hinted that manufacturers using a large percentage of local components are likely to benefit from several incentives to be announced as part of the new policy". He added that "under all circumstances, the auto industry will have to fulfill the outstanding export commitments even after quantitative restrictions are removed".⁵¹
- India explained that imports of cars in the form of CKD/SKD kits and of components subject to import restrictions were taken into account for calculating the value of imports for trade balancing purposes.⁵² This included purchases in India of imported components or CKD/SKD kits.⁵³ That is, if an MOU signatory purchased a component that was subject to import restrictions in India from either a trading company or another MOU signatory that had imported such a component on the basis of an import license, the value of such components would be taken into account for purposes of the trade balancing requirement. In every case, the license clearly specified that it was subject to the trade balancing requirement. Components that were not subject to import restrictions under the EXIM Policy 1997-2002 were not taken into account for purposes of the trade balancing requirement.
- India considered that it should ensure that all car manufacturers that had accepted MOUs were treated equally. It was therefore not the intention of the Government of India to release the car manufacturers from the commitments they had assumed under the MOUs as a result of imports of

 $^{^{\}rm 48}$ EC Questions, response to Question 8. See also US Questions, response to Questions 7 and 8.

⁴⁹ Japan's Questions, response to Question 23.

⁵⁰ US Questions, response to Questions 7 and 8.

⁵¹ "India's New Automobiles Policy Will Protect Domestic Industry Using Tariffs, Officials Say," BNA Daily Report for Executives, 22 August 2000.

⁵² The following tariff lines in Chapter 87 that were notified as restricted for AUTO/BOP reasons cover cars in the form of CKD/SKD kits: 870321.04; 870322.04; 870323.04; 870331.04; 870332.04; and 870333.03. The tariff lines covering components for cars that were identified as restricted for BOP-XVIII-B reasons such as 870600.01, 870600.09, 870710.01 and 870710.02 would also have been subject to MOU signatory license privileges. 53 WT/G/TRIMS/W/15, p. 9, 30 October 1998.

SKD/CKD kits prior to 1 April 2001. India intended to require the signatories of the MOUs to continue to perform after 1 April 2001 those provisions of the MOUs that were consistent with India's WTO obligations. India was also of the view that to require the signatories of the MOUs to perform the obligations they had assumed as a result of imports of SKD/CKD kits prior to 1 April 2001 was perfectly consistent with India's obligations under WTO law. The trade balancing requirement did not remain in place after 1 April, 2001. As a result of the abolition of the import licensing regime, there remained only a WTO-consistent obligation to discharge export obligations that accrued as a result of imports prior to 1 April 2001. As from 1 April 2001, the signatories of the MOUs would continue to be required to discharge the export obligations corresponding to the imports made by them before that date. They would thus be required to discharge the export obligations they had already assumed but would no longer incur any new export obligations as a result of the further importation of SKD/CKD kits. From 1 April 2001, the right to import SKD/CKD kits would consequently no longer in any way depend on, or vary with, the level of exports achieved. For this reason, the export requirements would no longer create any incentive to purchase local products. Since 1 April 2001, the import restrictions for SKD/CKD kits and car components had been abolished and no licenses referring to the trade balancing requirement had been issued. The signatories of the MOUs had therefore no longer incurred any new export obligations as a result of the importation of SKD/CKD kits or components.

4.19 The **European Communities** noted that India had declared that, as from 1 April 2001,

The signatories of the MOUs will be required to discharge the export obligations they have already assumed but will no longer incur any new export obligations as a result of the further importation of SKD/CKD kits. Since 1 April 2001, the right to import SKD/CKD kits will consequently no longer in any way depend on, or vary with, the level of exports achieved.⁵⁴

4.20 The interpretation of the obligations imposed by the balancing requirements after 1 April 2001 made by India in the reply to Panel Question 33 could not be reconciled with the wording of paragraph 3(iv) of Public Notice No. 60 and paragraph III(iv) of the MOUs. The European Communities's interpretation of those two provisions had been confirmed by India's answer to Question 50(c), when India explained that

Under paragraph 3(iv) of Public Notice 60 and paragraph III (iv) of the MOUs, the MOU signatory goes "outside the ambit of the MOU" and will not require any import licenses once it achieves a 70% indigenization level. However the MOU signatory will be required to achieve trade balancing of the foreign exchange obligation incurred on imports made up to that date. 55

In the European Communities's view, the wording of the MOUs did not support the interpretation that the signatories were not required to balance the imports of CKD/SKD/components made after 1 April 2001. Even if it were correct that the parties agreed in the MOUs that imports of components not subject to licensing at that time would not be covered by the balancing requirements, it would not follow necessarily that the elimination of the import licensing as of 1 April 2001 entailed automatically the elimination of all the balancing requirements agreed in the MOUs. As regarded the different issue of whether imports of "CKD/SKD kits/components" made after 1 April 2001 had to be balanced, the European Communities had shown that the MOUs contained no provision stipulating that the balancing requirements would cease to apply in respect of any product which ceased to be subject to import licensing. The European Communities understood that in the MOUs the signatories assumed the commitment to export an amount equivalent to the value of imports estimated for the

⁵⁴ India's response to Question 33.

⁵⁵ India's response to Ouestion 50 (c). [Underlining added].

entire duration of the MOU. Those estimates included imports of "CKD/SKD kits/components" to be made after 1 April 2001, since at the time when the MOUs were concluded it was not anticipated that the import licensing would have to be eliminated as of that date. ⁵⁶

- 4.21 Paragraph III(vi) of the standard format for MOU attached to Public Notice No. 60 stipulated in relevant part that
 - "... the party shall achieve broad trade balancing of foreign exchange *over the entire period* of the MOU in terms of balancing between the actual CIF value of imports of CKD/SKD Kits/components and the FOB value of export of cars and auto components over the said period."⁵⁷
- 4.22 Thus, the signatories must neutralise all the imports made "over the entire period of the MOU", and not just those made before 1 April 2001. Paragraph 3(iii) of Public Notice No. 60 further provided that
 - "... as and when the time firms achieve 70% indigenization, they would go outside the ambit of the MOU automatically. However, they will discharge the export obligation corresponding to the imports made by them till that time."
- 4.23 This indicated clearly that the signatories must balance all the imports made until they achieved an indigenization level of 70%. Thus, any signatory which, as of 1 April 2001, had not achieved that threshold would be required to balance imports made after that date. In other words, the signatories were required to balance all the imports made by them until they achieved a 70% indigenization level. Since some signatories had not achieved that level as of 1 April 2001 (and had not achieved it yet), the imports made by those signatories after 1 April 2001 must still be balanced with exports, contrary to India's contention. India recognised expressly that, after 1 April 2001, the signatories continued to be required to meet the same indigenization commitments.⁵⁸ Nevertheless, it argued once again that
 - ... according to as yet unverified reports received from the car manufacturers in the autumn of last year, most of the signatories have almost reached the 70 per cent threshold. At present, there may consequently no longer be any signatory bound by the indigenization requirements.⁵⁹
- 4.24 This was mere speculation, which could neither form the basis for an informed panel decision nor give satisfaction to the complainants. The European Communities encouraged India to verify, once and for all, the reports in question, instead of complaining about the supposed "inappropriateness" of bringing this dispute. India had also stated that the balancing requirements applied not only with respect to imports of CKD/SKD kits but also of those "components" which were subject to discretionary import licensing until 1 April 2001. The European Communities would note, nevertheless, that the relevant provisions of Public Notice No. 60 and of the MOUs referred to all "components", without any further qualification. Should the Panel accept India's explanations regarding the product scope of the balancing requirements, the European Communities would reiterate its request that the Panel make an express factual finding to that effect in its report.

⁵⁸ India's response to Question 33, at p. 3, para. 2.

⁵⁶ EC's response to Questions 93 and 94 of the Panel.

⁵⁷ Emphasis added.

⁵⁹ India's response to Question 33, p. 4 *in fine* and p. 5, para. 1.

⁶⁰ *Ibid.*, at p. 5, para. 2.

⁶¹ India's response to Questions 47-49.

- 4.25 The **United States** continued to see some difficulties with India's position. First, paragraph III, clause (vi) of the MOU required MOU signatories to balance their imports of SKD and CKD kits and components "over the entire period of the MOU". India had said that it did not apply a balancing obligation to kits and components importable without a license but India did not address the phrase "over the entire period of the MOU", even when the Panel expressly asked it to do so. ⁶² India had yet to explain how this phrase functioned under its domestic law, and how India ensured that a car manufacturer whose MOU was still in force was not accruing further export obligations for current imports. Second, even if it were true that no new or additional export obligations were accruing after 1 April 2001, the United States disagreed with India's contention that the level of an MOU signatory's imports was not restricted by the level of its exports. It was still doubtful whether an MOU signatory was permitted to import CKD/SKD kits in excess of the signatory's accrued but not yet discharged export obligations.
- India confirmed that, in respect of freely importable products, the MOU signatory did not assume any export obligations that might limit its right to import or create an incentive to purchase domestic components. The European Communities and the United States interpreted the trade balancing provisions in the MOUs differently. Their reading of the MOUs was that they entitled the Government of India to require the signatory to offset through exports the foreign exchange expenditures for all imports of CKD/SKD kits and components whether subject to licensing or not.⁶³ This was not, and had in practice never been, the interpretation of the Government of India. The Government of India had never requested any of the MOU signatories to apply the trade balancing requirement to freely importable car components and had ceased applying the requirement in respect of all imports of cars and car components after 1 April 2001. This practice confirmed that the trade balancing requirement in the MOUs had always been understood by the Government of India to apply only to imports subject to licensing. India could not be required to prove the negative. India could not, therefore, be asked to prove that it had not required trade balancing in respect of freely importable products. It was consequently for the complainants to demonstrate that the Government of India had requested MOU signatories to apply the trade balancing requirements in respect of freely importable products. The MOUs had now been in place since 1996 and car companies had during this period imported freely importable components. If the complainants were correct, they should be able to indicate at least one instance in which the Government of India had required trade balancing for freely importable components. So far, they had failed to do so and had therefore not met their burden of presenting a *prima facie* case of violation.⁶⁴

B. HAS THE MATTER BEEN SETTLED AND ADJUDICATED?

4.27 **India** considered that the matter that the complainants had presented to the Panel in their first submissions concerned the application of the discretionary licensing system which had already been the subject of complaints brought by each of the complainants under the DSU. Both complaints concerned the application of India's discretionary licensing system to automobiles imported in the form of semi-knocked down or completely knocked down components or kits (SKD/CKD kits). This licensing system had already been the subject of complaints brought by each of the complainants under the Dispute Settlement Understanding (DSU). India noted that the European Communities⁶⁵ and the United States⁶⁶, together with other developed country Members,⁶⁷ brought complaints in July

⁶² India's response to Question 50(a) of the Panel.

⁶³ Responses to the Questions of the Panel to the European Communities after the First Meeting with the Parties, paras. 10-14. US response to Panel Questions, paras. 31-33.

⁶⁴ India's response to Question 111 of the Panel.

⁶⁵ WT/DS96/1.

⁶⁶ WT/DS90/1.

1997 under the DSU against all of India's import restrictions notified under Article XVIII:B, including cars. Basing themselves on a determination by the International Monetary Fund that India's foreign exchange reserves were adequate, the European Communities and the United States claimed that India's import restrictions were no longer justified by Article XVIII:B.

- 4.28 According to Article 21.3(b) of the DSU, the recommendations and rulings of the DSB must be implemented within a reasonable period of time, the length of which may be determined by agreement between the parties to the dispute. India and the United States agreed that India would remove its restrictions in stages and that the reasonable period of time would expire on 1 April 2001.⁶⁸ India must therefore bring its import restrictions, including those applied to SKD/CKD kits, into conformity with its WTO obligations by that date. According to the mutually agreed solution with the European Communities, India must accord to the EC treatment no less favourable than that accorded to any other country in respect of the elimination of the restrictions.⁶⁹ As a consequence, the implementation period effectively ended on 1 April 2001, both under the solution agreed with the European Communities under Article 3.6 of the DSU and under the agreement reached with the United States pursuant to Article 21.3(b) of the DSU.
- 4.29 In the complaint brought by the European Communities, a mutually agreed solution was found which was notified to the Dispute Settlement Body (DSB). In the complaint brought by the United States, the DSB issued rulings on the basis of panel and Appellate Body findings. Under Articles 3.6 and 21.3(b) of the DSU, India agreed with both the European Communities and the United States to bring the licensing system into conformity with its WTO obligations as of 1 April 2001. This commitment covered also the application of India's discretionary licensing system to SKD/CKD kits. In India's view, the matter raised by the European Communities and the United States in the case before this Panel had therefore already been settled and adjudicated in accordance with the DSU.
- 4.30 India considered that the dispute with the European Communities had been settled through an exchange of letters notified to the DSB on 7 April 1998 as a mutually agreed solution under Article 3.6 of the DSU. The exchange of letters stipulated that the import restrictions applied to specified products would be phased out in stages by 31 March 2003. It further stipulated that, as long as India complied with its phase-out commitments, "the European Communities will refrain from action under GATT Article XXIII or Article XXIII" as regards the restrictions notified by India under Article XVIII:B. The joint notification of this settlement to the DSB also made clear that the European Communities and India had reached a mutually agreed solution with respect to the whole of the dispute on which the European Communities had requested consultations. India had fully complied with every aspect of the mutually agreed solution notified in accordance with Article 3.6 of the DSU. Notwithstanding the fact that India had fully complied with the notified solution, the European Communities had re-invoked the DSU with respect to the application of the discretionary licensing to one of the items notified under Article XVIII:B.

 $^{^{67}}$ Australia (WT/DS91/1), Canada (WT/DS92/1), New Zealand (WT/DS93/1) and Switzerland (WT/DS94/1).

⁶⁸ WT/DS90/15

⁶⁹ Paragraph 3 of the mutually agreed solution provides that: "India shall grant to the EC treatment no less favourable than that granted by India to any other country with respect to the elimination or modification of import restrictions on the products in the Annex and those in Annex II of document WT/BOP/N/24, under any form, either autonomously or pursuant to agreement or understanding with that country, including pursuant to the settlement of any outstanding dispute under the WTO Understanding on Rules and Procedures Governing the Settlement of Disputes".

⁷⁰ WT/DS96/8, p. 2.

It was not clear from the submission what precisely the scope of measure at issue was. The complainants referred to the import licensing regime, Public Notice No. 60 and the MOUs. They acknowledged that the import licensing regime applied to CKD/SKD kits would not be applied after 1 April 2001. They also seemed to acknowledge that Public Notice No. 60, which implemented that licensing regime, would no longer be operational after that date. Their concern seemed to relate primarily to measures India might take after 1 April in respect of the MOUs outside the framework of the import licensing regime and Public Notice No. 60. India asked the Panel to request the complainants to clarify whether in their view the measure at issue in this proceeding was the enforcement of MOUs through the import licensing regime and Notice No. 60 or whether it was limited to the enforcement of MOUs outside the licensing regime and Notice No. 60. If the complainants were to respond that their complaints related to the enforcement of MOUs through the import licensing regime and Public Notice No. 60, India would ask the Panel to issue a preliminary ruling to the effect that both complaints were inadmissible because the DSU cannot be invoked in respect of the same matter twice. In this context, India requested the Panel to ask the complainants to clarify whether or not, in their view, India must cease the application of its import licensing regime to CKD/SKD kits after 1 April 2001 as a result of the agreements that India concluded with the complainants under Article 3.6 and 21:3(b) of the DSU. In particular, India asked the panel to request the United States to answer the following questions: Would the United States consider itself entitled to suspend concessions in accordance with Article 22 of the DSU if India were to maintain its licensing regime for CKD/SKD kits and Public Notice No. 60 after 1 April 2001 contrary to the findings of the panel on *India - Quantitative Restrictions*. If so, what additional rights did the United States expect to obtain from the present proceeding? If not, should India await the results of the present proceeding before acting on this matter?

1. Has the matter been settled?

4.32 From the start, the **European Communities** had made it clear that the measures in dispute were not allowed by the 1997 Agreement⁷¹: in the first place, the 1997Agreement only allowed India to maintain restrictions on imports and not on the internal purchase or use of already imported products; second, the 1997 Agreement only authorised India to maintain restrictions with respect to passenger cars (including in CKD or SKD form), and chassis and bodies therefor, but not with respect to other "components" for passenger cars; and third, the 1997 Agreement, read in conjunction with the 1999 Agreement, required India to eliminate all the import restrictions on passenger cars, and chassis and bodies therefor, no later than 1 April 2001. Yet, according to India's own explanations, the MOUs would nevertheless remain valid and legally enforceable after that date.

4.33 **India** responded to the European Communities' arguments as follows: first, the European Communities argued that the mutually agreed solution "only allows India to maintain restrictions on imports and not on the internal purchase or use of already imported products". This argument could not be reconciled with the fact that the European Communities was asking the Panel to find that the

The European Communities recalled that, in any event, the 1997 Agreement between the European Communities and India was not a "covered agreement" within the meaning of Articles 1 and 2 DSU Therefore, India could not invoke that Agreement in order to justify the violation of its obligations under the GATT and the TRIMs Agreement. See the Report of the Appellate Body on *European Communities – Measures Affecting the Importation of Certain Poultry Products*, WT/DS69/AB/R, adopted on 23 July 1998, paras. 79-80 (DSR 1998:V, 2031), where the Appellate Body concluded that a bilateral agreement concluded by Brazil and the European Communities under Article XXVIII of GATT was not a "covered agreement". See also the Panel Report on *Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef*, (hereinafter "*Korea – Various Measures on Beef*"), WT/DS161/R, WT/DS169/R, para. 539, where the Panel emphasised that it had examined the provisions of the "Records-of-Understanding" concluded by Korea with some other Members "not with a view to 'enforcing' the contents of these bilateral agreements, but strictly for the purposes of interpreting an ambiguous WTO provision, i.e. Note 6 to Korea's schedule".

measures at issue constituted import restrictions and that the alleged restrictions on the internal purchase and use of imported products were part and parcel of the import restrictions that India must in any case eliminate as a result of the mutually agreed solution. Second, the European Communities had argued that the mutually agreed solution "only authorises India to maintain restrictions with respect to passenger cars (including in CKD or SKD form), and chassis and bodies therefor, but not with respect to other "components" for passenger cars". In fact India confined its import restrictions to CKD/SKD kits. Therefore, India was in full compliance with the mutually agreed solution. Moreover, as noted above, the joint notification made to the DSB confirmed that the mutually agreed solution resolved the whole of the dispute on which the European Communities had requested consultations. Third, the European Communities had argued that the mutually agreed solution "was not a 'covered agreement' within the meaning of Articles 1 and 2 of the DSU; therefore, India could not invoke that Agreement in order to justify the violation of its obligations under the GATT and the TRIMs Agreement. In India's view, it was however completely irrelevant that the mutually agreed solution was not a covered agreement within the meaning of Article 1 of the DSU and that it could therefore not provide an exemption from the obligations under the GATT and the TRIMs Agreement. The issue was whether the DSU could be invoked again in respect of a matter formally raised under the DSU and settled through a mutually agreed solution notified under the DSU. This issue was related to the European Communities's procedural rights under the DSU, not India's substantive obligations under a covered agreement.

- 4.34 India went on to say that, under the DSU, mutually agreed solutions were clearly a formally recognised procedure for reaching settlement. The following paragraphs of Article 3 of the DSU left no doubt in that respect:
 - "3. The prompt settlement of situations in which a Member considers that any benefits accruing to it directly or indirectly under the covered agreements are being impaired by measures taken by another Member is essential to the effective functioning of the WTO and the maintenance of a proper balance between the rights and obligations of Members.
 - 6. Mutually agreed solutions to matters formally raised under the consultation and dispute settlement provisions of the covered agreements shall be notified to the DSB and the relevant Councils and Committees, where any Member may raise any point relating thereto.
 - 7.The aim of the dispute settlement mechanism is to secure a positive solution to a dispute. A solution mutually acceptable to the parties to a dispute and consistent with the covered agreements is clearly to be preferred.
 - 10.if a dispute arises, all Members will engage in these procedures in good faith in an effort to resolve the dispute. . . "

The above provisions made clear that mutually agreed solutions notified under Article 3.6 of the DSU were a formal part of the dispute settlement process set out in the DSU. In fact, they declared them to be the *preferred* method of resolving disputes. These provisions must be given effect and their objectives must be realised. This meant a mutually agreed solution to a matter formally raised under a covered agreement and jointly notified to the DSB as such by the disputant parties in accordance with Article 3.6 of the DSU must be regarded as a formal settlement of the dispute that made the re-submission of the same dispute inadmissible.

4.35 India wondered if the European Communities took the view that it could bring its complaint *notwithstanding* the solution agreed with India under Article 3.6 of the DSU or whether that solution

did not cover the matter raised before this Panel. If the European Communities agreed that a mutually agreed solution to a matter formally raised under a covered agreement and jointly notified to the DSB as such by the disputant parties in accordance with Article 3.6 of the DSU must be regarded as a formal settlement of the dispute that made the re-submission of the same dispute inadmissible then India wished to know from the European Communities which measures taken by India were not covered by the mutually agreed solution. If the European Communities took the view that it had the right to re-invoke the DSU in respect of a matter even if it had formally agreed under Article 3.6 of the DSU not to do so, India wondered if the European Communities could explain why India, or any other Member of the WTO for that matter, would ever want to negotiate a mutually agreed solution with the European Communities again and how this approach could be reconciled with the objectives of the DSU.

In response, the European Communities argued that the 1997 Agreement did not preclude 4.36 the European Communities from pursuing this dispute and India's defence was wrong already as a matter of fact. The 1997 Agreement dealt with different measures and different claims. At any rate, the European Communities considered that, even if the matter in dispute were the same, that would still not render the European Communities's complaint inadmissible. The 1997 Agreement settled a dispute concerning the system of discretionary import licenses applied by India with respect to a large number of products, including passenger cars, on alleged BOP grounds. In essence, the European Communities claimed that those measures were inconsistent with Article XI:1 of GATT and could not be justified under Article XVIII: B. Public Notice No. 60 and the MOUs were not the subject of that dispute and were not covered by the 1997 Agreement. Indeed, Public Notice No. 60 and the MOUs had not even been adopted by India when the 1997 Agreement was concluded. The "1997 Agreement" had authorised India to apply discretionary licensing with respect to imports of passenger cars until 1 April 2001; it did not authorise India to apply the balancing or indigenization requirements contained in the MOUs either before or after that date. Nevertheless, to the extent that the balancing requirement obligated the manufacturers to balance the value of the imports of passenger cars, its effects did not go beyond the potential restrictive effects on imports of passenger cars which could have resulted from the application of the discretionary licensing for those products permitted by the 1997 Agreement until 1 April 2001. For the above reason, the European Communities did not consider it necessary to challenge the application of the balancing requirement before 1 April 2001, in so far as imports of passenger cars were concerned. This was not because, as a matter of law, the 1997 Agreement allowed the application of that measure before 1 April 2001, but rather because the elimination of the balancing requirements in respect of import of passenger cars before 1 April 2001 would have provided no effective relief to the European Communities, since in any event India could have continued to restrict the importation of passenger cars through discretionary licensing.

4.37 The scope of the matter settled by the 1997 Agreement was defined in the request for consultations submitted by the European Communities on 18 July 1997⁷² and in the 1997 Agreement itself. Neither of those documents referred to the balancing and indigenization requirements. Moreover, those requirements were never discussed during the consultations or during the negotiations leading to the 1997 Agreement. According to the European Communities, there was, therefore, no basis for India's contention that the 1997 Agreement covered also those measures. The request for consultations of 18 July 1997 had made no reference to the balancing or the indigenization requirements. It described the measures in dispute as

"[the] quantitative restrictions maintained by India on importation of a large number of agricultural, textile and industrial products. The restrictions include those notified to the WTO in document WTO/BOP/N/24 of 22 May 1997."

⁷² WT/DS96/1.

Document WTO/BOP/N/24 contained a notification made by India under paragraph 9 of the Understanding on the Balance-of-Payments Provisions of the GATT 1994 (the Understanding). Annex I, Part B of document WTO/BOP/N/24 listed the import restrictions applied by India on BOP grounds as of the date of the notification. It described the import regime applicable to passenger cars falling within HS 8706⁷³ as "AUTO/BOP-XVIII:B", which means that

"Imports of passenger cars and automotive vehicles are permitted without a license on fulfilment of conditions specified in a Public Notice issued in this behalf, and restrictions on import through NAL are otherwise maintained."⁷⁴

As of 18 July 1997, when the European Communities made its request for consultations, no such Public Notice had been issued. Therefore, the conditions to be imposed by India were unknown to the European Communities. The import regime of chassis (HS 87.06) and bodies (HS 87.07) was described in document WTO/BOP/N/24 as "BOP-XVIII:B", which meant that imports were "regulated though NAL"⁷⁵ (i.e. through non-automatic licenses). There had been no mention in document WTO/BOP/N/24 that the importation of chassis or bodies was subject to balancing requirements or that their internal use or purchase was restricted through the application of indigenization or balancing requirements. Most parts and accessories for the assembly of passenger cars were not listed in Annex I, Part B⁷⁶. There had been no indication in document WTO/BOP/N/24 that the importation of these products or their internal use or purchase was restricted for BOP reasons through the application of balancing or indigenization requirements or in any other manner. As regarded the claims submitted by the European Communities, the request for consultations mentioned the following provisions: Articles XI, XIII, XVII and XVIII of GATT 1994; Article 4.2 of the Agreement on Agriculture; Articles 1 and 3 of the Agreement on Import Licensing Procedures; and Articles 2, 3 and 5 of the Agreement on the Application of Sanitary and Phytosanitary measures.⁷⁷ Thus, the request for consultations had not raised any claims under Article III:4 of GATT or under Article 2 of the TRIMs Agreement, contrary to what could have been expected if the request had been meant to cover also the balancing and indigenization requirements. Like the request for consultations, the 1997 Agreement made no reference to the balancing or to the indigenization requirements. The introductory paragraph of the 1997 Agreement specified that it was a settlement of

"... the difference regarding the quantitative restrictions maintained by India on import of industrial, agricultural and textile products, as notified by India to the WTO by Annex I, Part B (Notification on Quantitative Restrictions Maintained on Imports Under the Export and Import Policy (as on 1 April 1997) of document WT/BOP/N/24."

As explained above, Annex I, Part B of document WT/BOP/N/24 had not mentioned anywhere the balancing or indigenization requirements. Although that document referred to the conditions specified in a Public Notice (not adopted when the 1997 Agreement was concluded), the European Communities could not be assumed to have settled in advance all possible claims with respect to whatever conditions India might see fit to include subsequently in that Public Notice. It would be unreasonable to interpret the 1997 Agreement as giving a blank check to India.

⁷³ With the exception of ITC 870310.00 (snow and golf cars) and 870390.00 (a residual category covering "other passenger cars" not included in the preceding positions).

74 See the Coding Diagram included at the end of document WTO/BOP/N/24.

⁷⁶ See e.g., HS 87.08, 40.11, 40.12, 40.13, 70.07, 70.09, 90.29 and 94.01 in document WTO/BOP/N/24. ⁷⁷ The claims under the SPS Agreement reflect the fact that the request for consultations covered also the imports restrictions contained in Annex I, Part A of Document WTO/BOP/N/24, which were applied by India on alleged sanitary grounds. This part of the dispute is not covered by the 1997 Agreement.

- 4.38 The 1997 Agreement had not settled the matter currently in dispute and, at any rate, was not a covered agreement enforceable by this Panel. Consequently, the 1997 Agreement did not preclude the European Communities from bringing this dispute. Even if the 1997 Agreement had settled the matter in dispute in the present case, that would still not preclude the European Communities from bringing this dispute. The 1997 Agreement was not a "covered agreement" in the sense of Article 1.1 of the DSU. Therefore, the rights and obligations of the parties under the 1997 Agreement were not enforceable under the DSU. The 1997 Agreement only allowed India to restrict the importation of passenger cars until 1 April 2001, in derogation of Article XI:1 of GATT. Public Notice No. 60 and the MOUs went beyond this in a number of respects:
 - first, the indigenization requirements provided less favourable treatment to imported components and materials than to like domestic products with respect to their purchase and use for the assembly of passenger cars, contrary to Article III:4 of GATT and Article 2.1 of the TRIMs Agreement;
 - second, the balancing requirements provided less favourable treatment to the
 internal use and purchase of imported CKD/SKD kits and components for the
 assembly of passenger cars, contrary to Article III:4 of GATT and Article 2.1
 of the TRIMs Agreement; and
 - third, the MOUs had been concluded for a period of time that extended beyond 1 April 2001 (in principle, five years). As a result, the restrictions on imports of passenger cars imposed by the balancing and indigenization requirements would be maintained beyond that date in violation of Article XI:1 of GATT and Article 2.1 of the TRIMs Agreement.
- 4.39 The manufacturers acceded to sign the MOUs in order to obtain import licenses for passenger cars. But this did not mean that the MOUs were part of the licensing system temporarily authorised by the 1997 Agreement. The MOUs constituted separate measures and imposed additional trade restrictions that were not permitted by the 1997 Agreement. In addition, the balancing requirements appeared to require the manufacturers to balance not only the imports of passenger cars in CKD/SKD form but also of components, in violation of Article XI:1 and Article 2.1 of the TRIMs Agreement. The trade restrictive effects of the balancing and indigenization requirements went beyond those resulting from the application of the discretionary import licensing for passenger cars that India was authorised to maintain pursuant to the 1997 Agreement until 1 April 2001. In particular, India would have infringed the 1997 Agreement by continuing to apply the balancing and indigenization requirements beyond 1 April 2001. That violation would have arisen, and would have been open to challenge by the European Communities, from the moment that India concluded the MOUs, since the MOUs stipulated a period of duration for the balancing and indigenization requirements that went beyond 1 April 2001.⁷⁸
- 4.40 In the 1997 Agreement, the European Communities had expressly reserved its right to resort to dispute settlement in the event that India did not comply with the terms of that agreement. Therefore, even if the matter in dispute in this case had been settled by the 1997 Agreement (quod non), the European Communities would not be precluded by the terms of the 1997 Agreement from pursuing the present dispute with respect to those aspects of Public Notice No. 60 and the MOUs which were incompatible with that agreement. Furthermore, the European Communities recalled that the 1997 Agreement was not a "covered agreement" in the sense of Article 1.1 of the DSU. India did

⁷⁸ See EC's response to Question 18.

not dispute this. Therefore, this Panel could not rule on whether the measures in dispute were consistent with the 1997 Agreement. By the same token, this Panel could not rule on India's defence that the EC's complaint was "inadmissible" because it was a violation of the EC's obligation under the 1997 Agreement not to take dispute settlement action against India. India's defence essentially amounted to saying that the obligations imposed by the 1997 Agreement upon the European Communities were enforceable under the DSU, but those imposed upon India were not. India could not have it both ways: either the 1997 Agreement was a "covered agreement" enforceable by this Panel, or it was not. As recalled by India, the DSU encouraged Members to seek mutually agreed solutions. Nevertheless, the DSU did not attach to such solutions the legal effects alleged by India. It was nowhere stipulated in the DSU that where Members had notified a mutually agreed solution they could no longer request a Panel with respect to the same matter. This was only logical, given that the DSU provided no mechanisms to enforce mutually agreed solutions. Thus, if a complaining Member considered that the other party had failed to comply with a mutually agreed solution it did not have the option of requesting a "compliance" Panel under Article 21.5 of the DSU. Nor could the complaining Member request the establishment of a new Panel under Article 6.1 of the DSU to rule that the other party had violated the mutually agreed solution. The only remedy available to the complaining Member was to resume the original dispute settlement proceedings at the point where they were left. If that course of action were also barred to the complaining Member, the defending Member would enjoy complete impunity to violate the mutually agreed solution. Without the assurance that they could revert to dispute settlement under the DSU in the event that the other party failed to comply with a mutually agreed solution, no complaining Member would ever agree to any such solution. Thus, it was India's interpretation, and not the European Communities', that would defeat the DSU's objective to encourage mutually agreed solutions.

- 4.41 **India** expressed surprise that it was the considered position of the European Communities and its Member States that a WTO Member could request the establishment of a panel on a matter on which it had reached a settlement notified under Article 3.6 of the DSU even if the respondent had faithfully implemented the settlement. The European Communities claimed that this followed from the fact that the DSU provided for no mechanism to enforce the agreed solution. Of course, it did. If the solution was not faithfully implemented, the complaining Member could re-invoke the DSU to claim its rights under the WTO provisions that were not implemented. It was therefore simply incorrect to claim that the "defending Member would enjoy complete impunity to violate the mutually agreed solution". India believed that the European Communities was presenting arguments in this proceeding that if accepted would frustrate the objectives of the DSU, which explicitly favoured settlement over adjudication. Endorsing the EC's approach would seriously curtail the policy options of the European Communities and the other WTO Members under the DSU. India invited the European Communities and its Member States to reconsider their position on this important point in the light of their broader interests.
- 4.42 The **European Communities** noted that India's position had evolved during the course of the proceedings. At first, India had seemed to argue that, where a matter had been settled, the original complaint could not be re-submitted under any circumstances. Subsequently, India contended that a complaint could be re-submitted, if the Member complained against had infringed the settlement. Thus, on India's interpretation, the European Communities would have to show in this case, first, that India had violated the settlement and, second, that India had violated the WTO Agreement. In the view of the European Communities, India's more recent position was more reasonable. But, like the previous one, it had no basis whatsoever in the DSU. Moreover, India's new position could not be reconciled with the principle that it had asserted earlier, whereby a matter which had been settled could not be re-litigated again. If this were correct, the only remedy available to the complaining Member in the event of violation of the settlement would be to request a panel to find that violation, and not a violation of the WTO Agreement. As already explained by the European Communities, mutually agreed solutions were not "covered agreements" within the meaning of Article 1.1 of the

DSU. Therefore, the rights arising from a mutually agreed solution could not be enforced through the DSU. That meant that a complaining Member could not ask the establishment of a panel to find that the other party had violated the mutually agreed solution. It meant also that the Member complained against could not invoke the existence of a mutually agreed solution as a defence before a panel established to find a violation of the WTO Agreement. India's interpretation would create an unacceptable imbalance between the two parties to a settlement. Whereas panels could not enforce the rights of the complaining Member under the settlement (which can go beyond its rights under the WTO Agreement), they would be required to enforce the rights of the Member complained against (i.e. the correlative obligation of the complaining Member not to re-invoke the DSU).

4.43 The European Communities noted that India had asked why a Member complained against would agree to a settlement, if the other party was not prevented from re-invoking the DSU. The European Communities could ask why a complaining Member would agree to a settlement given that its rights under the settlement, even on India's interpretation, were not enforceable under the DSU. The answer was the same in both cases: the parties to an agreement might consider that enforceability was not indispensable, where it was in both parties' interest to comply with the agreement. India made the error of assuming that a settlement did not confer to the complaining Member any advantages and, therefore, that the complaining Member paid no price for re-invoking the DSU. India made much of the fact that the mutually agreed solutions was notified to the WTO. That notification, however, was made only for transparency purposes and did not confer any legal effects to the settlement. The notification allowed other Members to verify that settlements were consistent with the WTO Agreement⁷⁹, and in particular that any concessions negotiated with the complaining Member falling within the scope of the different MFN obligations provided in the covered agreements were duly extended to all Members. The European Communities continued that the status of mutually agreed solutions under the DSU could be usefully contrasted with that of the arbitration awards issued under Article 25 of the DSU. Article 25.4 of the DSU provided that Articles 21 and 22 of the DSU should apply mutatis mutandis to arbitration awards. No similar provision was made with respect to mutually agreed solutions. This proved the drafters' intention to maintain a clear distinction between those solutions which involved the adjudication of the parties' right under the WTO Agreement (panel reports and arbitration awards) and those which did not involve such adjudication. De lege ferenda, a case could perhaps be made that mutually agreed solutions should be made enforceable under the DSU. Indeed, some Members had already made proposals to that effect. As the law stood, however, mutually agreed solutions were not enforceable under the DSU. The Panel should reject India's request to re-write the DSU in order to make settlements partially enforceable (only the rights of the party complained against!) on the basis of vague and unsubstantiated legal principles. As recalled pertinently by the Panel in *India – Patent Protection*,

"the Panel is required to base its findings on the language of the DSU. [It] simply cannot make a ruling *ex aequo et bono* to address a systemic concern divorced from explicit language of the DSU". 80

4.44 In conclusion, the European Communities noted that India had taken the position that, in order to bring a dispute with respect to a matter settled by mutually agreed solution, the complaining Member must establish first that the other party had infringed that solution. Should the Panel agree with that view, and should the Panel conclude further that the 1997 Agreement settled the matter in dispute in this case, the European Communities submitted that the measures at issue would be incompatible with the 1997 Agreement for the reasons stated above.

⁷⁹ Cf. Article 3.5 of the DSU.

⁸⁰ India – Patents (EC), para. 7.23 (DSR 2998:VI, 2661).

⁸¹ See EC's response to Question 24 of the Panel.

2. Has the matter already been adjudicated?

(a) India – Quantitative Restrictions

India also argued that the dispute with the United States had been resolved by rulings adopted by the DSB on 22 September 1999 on the basis of the findings set out in the reports of the Panel and the WTO Appellate Body in the proceedings India - Quantitative Restrictions. The Panel, endorsed by the Appellate Body, had recommended that the DSB request India to bring the measures at issue into conformity with its obligations under the WTO Agreement.⁸² The Panel had specifically identified discretionary import licensing applied to all the products notified under Article XVIII:B, including SKD/CKD kits, as one of the measures at issue. 83 The dispute with the United States concerned all of the restrictions notified by India under Article XVIII:B, including those applied under the discretionary licensing scheme administered by the Ministry of Commerce. The panel declared this regime, to the extent that it was applied to the notified import restrictions, "a measure at issue" in the proceedings and found it to be inconsistent with Article XI:1 of the GATT. 84 The panel, endorsed by the Appellate Body, found that India no longer had any balance-of-payments justification for the "measures at issue" and that they were therefore not justified by Article XVIII:B of the GATT. Consequently, it recommended that the DSB request India to bring the measures at issue into conformity with its obligations under the WTO Agreement. The United States had thus obtained a ruling that the discretionary licensing scheme administered by the Ministry of Commerce was inconsistent with Article XI of the GATT. The United States was now requesting the present Panel to rule that also the application of this licensing scheme to imports of CKD/SKD kits was inconsistent with Article XI. However, the DSB ruling covered the operation and application of the discretionary licensing system in respect of all products that were notified under Article XVIII:B, including the products of which CKD/SKD kits were composed. The United States was thus requesting the present Panel to rule on a matter that had already been adjudicated by the DSB. If the Panel found that the inconsistency of India's licensing scheme with Article XI:1 of the GATT was not a settled and adjudicated matter, then India's invocation of Article XVIII:B of the GATT could also not be considered to be a settled and adjudicated matter. The Panel could not declare, with regard to the same matter, the complainant's claim of inconsistency to be a new issue and the respondent's defense to be settled and adjudicated. The two sides of the issue would have to receive the same treatment. Consequently, if the Panel were to decide to examine the consistency of the licensing scheme for SKD/CKD kits with Article XI, it would also have to examine its justification under Article XVIII:B. Before the Panel on *India - Quantitative Restrictions*, India had argued, *inter alia*, that:

- panels must exercise judicial restraint when examining measures subject to the jurisdiction of the Committee on Balance-of-Payments Restrictions,
- panels could not transfer the determination of the adequacy of foreign exchange reserves to the International Monetary Fund, and
- the provisions of the DSU on the implementation of DSB recommendations could not replace the mechanism for the phase-out of balance-of-payments measures set out in paragraph 13 of the Understanding on the Balance-of-Payments Provisions of the GATT 1994.⁸⁵

⁸² India – Quantitative Restrictions (WT/DS90/R), para. 6.2.

⁸³*Ibid.*, WT/DS90/R, paras. 6.1 and 5.117.

⁸⁴*Ibid.*, WT/DS90/R, para. 6.1.

⁸⁵ See Section III.D of the Panel Report on *India – Quantitative Restrictions* (WT/DS90/R).

All of these matters would have to be re-examined by this Panel if the inconsistency of India's licensing scheme with Article XI:1 of the GATT were not found to be a settled and adjudicated matter. Depending on the Panel's rulings on these issues, India's external financial positions in May and October 2000 would have to be examined, possibly with the advice of the International Monetary Fund, because they would be different from the India's external financial position in November 1997 which the earlier panel examined.⁸⁶ This panel correctly found that "the determination of whether balance-of-payments measures are justified is tied to a Member's reserve situation as of a certain date".⁸⁷

4.46 The United States responded that India had completely misconstrued the nature of its complaint. The measures at issue were not India's import licensing system; instead, the measures challenged were the indigenization and trade balancing requirements of Public Notice No. 60 and the MOUs. India could not avoid the complaint by defending measures that were not the subject of the case. Contrary to India's assertion, the rulings in the India – Quantitative Restrictions dispute did not adjudicate the matter that the United States had brought before this Panel. Second, India was inviting the Panel to analyze legal issues that were not presented by the facts of this case and therefore were not necessary to the resolution of this dispute; the Panel should decline that invitation. Third, because the measures and claims at issue in this dispute were different from those at issue in India -Quantitative Restrictions, this Panel's rulings would apply to discrimination and import restrictions that the India - Quantitative Restrictions panel could not and had not addressed. The United States argued that the India - Quantitative Restrictions dispute did not overlap this one; for that reason alone, the Panel should conclude that India's arguments lacked merit. India had argued that the measures at issue in this case had already been the subject of a ruling by the Dispute Settlement Body, and for that reason could not be examined by this Panel. The premise for this argument, of course, was that the measures at issue in this case were India's import licenses, and that premise was incorrect. It was also worth pointing out that the terms of reference of the India - Quantitative Restrictions Panel were established on 18 November 1997. 88 Public Notice No. 60, on the other hand, was issued on 12 December 1997. Therefore, even under India's logic, these measures could not have been a part of the India – Quantitative Restrictions dispute. In the United States' view, the India – Quantitative Restrictions panel did not examine the measures that had been referred to this panel; consequently, the DSB recommendations and rulings in that dispute would not have required India to bring these measures into compliance. The United States had asked India whether they shared this view, or whether, since these measures remained in force after the end of the reasonable period of time for compliance in that dispute, India believed instead that the United States was entitled to request authorization to suspend concessions in accordance with Article 22 of the DSU on the basis of the *India – Quantitative Restrictions* ruling.⁸⁹

4.47 In response to the Panel's Question 5, the United States had detailed the differences between the *India – Quantitative Restrictions* dispute and this one. In sum: in *India – Quantitative Restrictions*, the United States challenged the existence of India's non-automatic import licensing system as such, and not the application of specific criteria for licensing the importation of specific tariff line items. For example, the United States had noted in that case that "all that the United States knew was that the Indian licensing authority generally refused to grant import licences for 'restricted' items when it was considered prejudicial to the state's interest to do so." India had the same view of the case: "which implied that the scope of this proceeding did not include the administration of the import restrictions....India concluded that the United States invoked the dispute settlement procedures

⁸⁶ WT/DS90/R, para. 5.156.

⁸⁷ *Ibid.*, para. 5.162.

⁸⁸ *Ibid.*, para. 1.2.

⁸⁹ See Questions 4 and 5 of the United States to India.

⁹⁰ WT/DS90/R, *op.cit*, para. 3.20.

only with respect to the justification of the import restrictions and not in respect of matters arising from their application. India, therefore, reserved its position on the allegations of the United States with respect to the application of India's import restrictions, in terms of both their factual basis and their legal implications."91 Furthermore, the *India - Quantitative Restrictions* panel's findings confirmed that the panel too had that view: "we note that it is agreed that India's licensing system for goods in the Negative List of Imports is a discretionary import licensing system, in that licences are not granted in all cases, but rather on unspecified 'merits'. We noted also that India conceded this measure is an import restriction under Article XI:1."92 By contrast, the dispute before this Panel did not concern India's non-automatic licensing regime. It concerned instead different measures (those specified in the U.S. panel request, particularly Public Notice No. 60 and the MOUs) and different requirements (the indigenization requirement and the trade balancing requirement) that India imposed on car manufacturers and on automotive parts and components. Furthermore, this dispute raised different legal issues: first, did those requirements as such (apart from import licensing) impose prohibitions or restrictions on imports in violation of GATT Article XI:1? In the United States' view, those requirements did impose such import restrictions, but the particular restrictions at issue here (limitations correlated to projected and achieved export values and to the degree of local content) were different from the very existence of non-automatic licensing that was at issue in the India -Quantitative Restrictions dispute. Second, did those requirements as such (apart from import licensing) impose discrimination against imported goods in violation of GATT Article III:4? The US view was that these requirements treated imported goods less favorably than like domestic goods and that was a claim that even India realized was not and could not have been a part of the India -Ouantitative Restrictions dispute. The United States added that those measures and requirements came into force after the import licensing regime was put in place, and, as India had acknowledged, would remain in force after the import licensing regime expired. Moreover, those requirements existed because car manufacturers had undertaken to comply with them in order to obtain relief from the measures at issue in the India - Quantitative Restrictions dispute, but that point only confirmed that the measures at issue here were different from the ones India - Quantitative Restrictions considered. Furthermore, India's argumentation mixed up two quite separate concepts. On the one hand, India's licensing regime acted as a total ban on imports. So-called "restricted" items ordinarily could not be imported at all. That was the basis of the United States' complaint in the India -*Quantitative Restrictions* dispute. The present dispute, on the other hand, did not contest that import ban; instead, this dispute challenged the conditions that India attached to goods imported despite the ban. Simply put, the *India – Quantitative Restrictions* dispute was about India's efforts to keep goods out of India. This dispute was about what India did when goods actually got in.

4.48 In the view of the United States, this dispute was a logical successor to the *India – Quantitative Restrictions* case. After finally achieving access for their products into India by peeling away the import ban that India had been applying through its licensing regime, WTO Members were naturally concerned about the treatment their products would receive after importation - hence their desire to ensure national treatment for those products. And, Members were concerned about the durability of that access. They therefore wanted to ensure that India did not impede that access through an entirely new layer of barriers - such as, in this case, the independent import restrictions and prohibitions contained in Public Notice No. 60 and the MOUs.

4.49 The measures in this dispute were different, and the legal claims in this dispute were different. Consequently, in the terminology of DSU Article 7.1, the "matters" in the two disputes were different, ⁹³ and therefore this dispute was not even addressed, let alone "adjudicated" (to use India's

⁹¹ *Ibid.*, para. 3.41.

⁹² *Ibid.*, paras. 5.130 (footnotes omitted) (emphasis added).

^{93 &}quot;The 'matter referred to the DSB', therefore, consists of two elements: the specific measures at issue and the legal basis of the complaint (or the claims)." Report of the Appellate Body in Guatemala –

terms), by that prior panel. Therefore, the Panel should decline to undertake to clarify a legal issue that was neither presented by the facts before it nor necessary to the resolution of this dispute. The United States noted that in its responses to the Panel's question 36, India seemed to have accepted the distinction between its import licensing regime and the requirements at issue in this case:

The complainants thus appear to accept that a Member cannot resort to DSU twice with respect to the same matter but that their claims are different from that those that were the subject of the prior DSU proceedings. They asserted in particular that Public Notice No. 60 and the MOUs impose requirements and restrictions within the meaning of Articles III:4 and XI:1 of the GATT and the corresponding provisions of the TRIMs Agreement even in the absence of the discretionary licensing scheme that India was obliged to eliminate as a result of the prior proceedings. India therefore assumed that the issue before the Panel was no longer the scope of the principle of *res judicata* but rather the question of whether Public Notice No. 60 and the MOUs impose requirements and restrictions even in the absence of the licensing scheme for CKD/SKD kits.

According to the United States, the legal consequences of the rulings in the *India - Quantitative Restrictions* report entailed the elimination of the licensing requirements for a large number of goods, including SKD/CKD kits. The legal consequences of a ruling in favor of the United States in this dispute would entail the elimination of separate import restrictions imposed by Public Notice No. 60 and the MOUs. Without such a ruling in this dispute, those separate import restrictions would continue to apply to car manufacturers as long as the MOUs remained in place.

Of course, India's import licenses for SKD and CKD kits did play a role in this dispute. In the 4.50 context of this case, those licenses were, first and foremost, an "advantage" given to MOU signatories, as that term was used in the European Economic Community - Regulation on Imports of Parts and Components (hereinafter "EEC - Parts and Components") report. Companies willing to sign an MOU obtained an advantage not given to others - namely, the right to import SKD and CKD kits despite their "restricted" status. However, just as in the EEC – Parts and Components dispute, and in the Canada – FIRA dispute before it, it was not the "advantage" that was the subject of the complaint, but rather the conditions attached to the "advantage". In Canada - FIRA, for example, the "advantage" was authorization to invest - which, as India pointed out, was at the time not a subject of GATT disciplines at all. But that advantage was conditioned on local purchase undertakings, which in turn were examined by the Canada – FIRA Panel and were found to be contrary to GATT's national treatment disciplines. This case was no different: it was not the "advantage" - the import licenses that this panel needed to examine, but rather the conditions attached to those licenses. Those conditions included undertakings by MOU signatories to comply with indigenization and trade balancing requirements, and for the reasons already given, those conditions were inconsistent with India's WTO obligations. The **United States** considered that India's confusion about the role that import licenses played in this case should not deter the Panel from that conclusion. Elsewhere, the United States had discussed how the indigenization requirement damaged the competitive position of all automotive parts and components imported into India. That point alone should by itself put to rest India's assertion that this dispute concerned its import licenses. After all, India licensed the import of kits; the discrimination at issue in this dispute affected all automotive parts and components. Public Notice No. 60 and the MOUs were harming the export interests of all car part manufacturers around the world - not just the interests of manufacturers of the SKD and CKD kits to which India's import licensing regime applied.

Anti-Dumping Investigation Regarding Portland Cement from Mexico, (hereinafter "Guatemala – Cement I"), WT/DS60/AB/R, adopted on 25 November 1998, para. 72.

- 4.51 The **European Communities** pointed out that although the balancing and indigenization requirements were applied in conjunction with the discretionary import licenses at issue in *India Quantitative Restrictions* until 1 April 2001, they were legally distinct and separate measures, as evidenced by the following:
 - Public Notice No. 60 had not been issued when the United States requested the establishment of the panel in India QRs;
 - nothing in the EXIM Policy or in document WTO/BOP/N/24 required India to apply the balancing and indigenization requirements;⁹⁴
 - contrary to India's assertions, the balancing and indigenization requirements were not an "inherent part" of the discretionary import licensing scheme. It was perfectly possible to operate a discretionary import licensing scheme without applying any of those requirements. Most products subject to discretionary import licensing under the EXIM Policy were not subject to balancing and indigenization requirements;
 - by the same token, the balancing and indigenization requirements could be applied and enforced in the absence of discretionary import licensing; and
 - the balancing and indigenization requirements contained in the MOUs had remained binding and enforceable after the elimination of the import licenses, even though no additional action to achieve that effect had been taken by the Indian authorities after 1 April 2001.
- 4.52 Moreover, as conceded by India, the legal basis for the United States complaint in this case was also different, to the extent that it included also claims under Articles III:4 of GATT and Article 2 of the TRIMs Agreement.
- 4.53 In the course of the panel proceedings, **India** conceded that its argument that no WTO Member might resort to the DSU on the same matter twice was no longer relevant because the European Communities and the United States had made it clear that their case did not rest on the application of the discretionary licensing system. India assumed that the Panel would examine the operation of PN 60 and the MOUs as such, that is independently of the discretionary licensing system Specifically, India assumed that the panel would examine whether the trade balancing and indigenization provisions set out in PN 60 and the MOUs would have been inconsistent with Article III:4 and XI:1 even if India had not applied import licensing restrictions for cars imported in the form of CKD/SKD kits and for car components at the time when the request for the establishment of the panel was submitted. ⁹⁵ (See also III.C)

⁹⁴ Cf. Report of the Appellate Body on *United States – Import Measures on Certain Products from the European Communities*, (hereinafter "*US – Certain EC Products*") WT/DS165/AB/R, para. 60 *et seq.* Although this report concerns the different issue of whether a measure taken after the consultations was within panel's term of reference, it may offer guidance for the application of the *res judicata* principle in this case. The Appellate Body concluded that, although "related" to the 3 March Measure, the 19 April Measure was outside the terms of reference of the panel because it was a "separate and legally distinct measure". In reaching that conclusion, the Appellate Body considered as particularly significant that nothing in the 3 March Measure required the United States to take the 19 April Measure (*ibid.*, para. 76). Likewise, the import licenses challenged in *India – Quantitative Restrictions* did not require India to impose the balancing and indigenization requirements at issue in this dispute.

⁹⁵ See also India's response to Question 102 of the Panel.

- 4.54 The **European Communities** countered that the method of enforcement of the MOUs was not a constituent element to the violations claimed by the European Communities. The balancing and indigenization requirements were as such inconsistent with the GATT and the TRIMs Agreement, regardless of the method used to enforce them. The European Communities saw no merit in the hypothetical and speculative analysis proposed by India.
- (i) The notion of res judicata
- 4.55 **India** made reference to the principle of "res judicata", quoting:

Interest republicae ut sit finis litigium, is an old maxim deeply fixed in the law of fundamentals; that it concerns the state that there be an end to litigation. This maxim has wide application; it . . .is obviously based on common sense and sound policy. For if matters which have been solemnly decided are drawn again to controversy; if facts, once solemnly affirmed, are to again denied whenever the affirmant sees his opportunity, there can never be an end to litigation and confusion. ⁹⁶

For these very basic and fundamental reasons, all legal systems had developed principles to ensure that matters, once settled, could not be litigated again. India submitted that there were no legal procedures, national or international, under which a complainant would be entitled to do what the European Communities was doing in the present case, namely litigate a matter after having formally consented to a settlement of the matter. The Appellate Body had frequently resolved interpretative issues under the DSU by reference to generally applied principles of law. Thus, it ruled on the distribution of the burden of proof between the parties to a DSU proceeding on the basis of "a generally-accepted canon of evidence in civil law, common law and, in fact most jurisdictions" The Panel should do the same in the present case. According to Article 3.4 of the DSU, the "... rulings made by the DSB shall be aimed at achieving a satisfactory settlement of the matter", and, according to Article 3.7 of the DSU, "the aim of the dispute settlement mechanism is to secure a positive solution to a dispute". The generally applied principle of *res judicata* must therefore be considered to constitute an inherent part of the WTO dispute settlement process. To quote Henry M. Herman again:

The slightest reflection will show that if some point were not established at which judicial proceedings must stop, no one could ever feel secure in the enjoyment of his life, liberty and property; while unjust, obstinate and quarrelsome persons, especially such as are possessed of wealth or power, would have society at their mercy, and soon convert it into one vast scene of litigation, disturbance and ill will.⁹⁸

In fact, the principle of *res judicata* was firmly established in international law. In the case of the International Court of Justice (ICJ), the scope of this principle is defined in Articles 59 and 60 of its Statute. Article 59 provided as follows: "The decision of the Court has no binding force except between the parties and in respect of that particular case." Article 60 provided as follows: "The judgment is final and without appeal .." .The ICJ made clear that the principle of *res judicata* applied also outside the scope of Articles 59 and 60. Thus, in the advisory opinion of the ICJ on the *Effects of Awards of Compensation made by the United Nations Administrative Tribunal case*, the Court ruled that a judgment rendered by a judicial tribunal such as the Administrative Tribunal of the United

⁹⁶ Henry M. Herman, *Commentaries on the Law of Estoppel and Res Judicata*, F.D. Linn & Company, Jersey City, N. J. (1886), Vol. I, page 8.

⁹⁷ United States - Measures Affecting Imports of Woven Wool Shirts and Blouses from India, (hereinafter "US – Wool Shirts and Blouses"), (WT/DS33/AB/R), para. 4.6 (DSR 1997:I, 323).

⁹⁸ Henry M. Herman, *Commentaries on the Law of Estoppel and Res Judicata*, F.D. Linn & Company, Jersey City, N.J. (1886), Vol. I, page 8.

Nations would be *res judicata* between the parties based on general principles of law. ⁹⁹ This appeared to be also the view of the Appellate Body. It noted in a discussion of the status of panel reports adopted by the CONTRACTING PARTIES to the GATT 1947 that "the generally-accepted view under GATT 1947 was that the conclusions and recommendations in an adopted panel report bound the parties to the dispute in that particular case. ¹⁰⁰ The parties' and the Panel's efforts would lead to a ruling on a measure that would no longer exist when the Panel issued its report. The waste of resources that a failure to apply the *res judicata* principle would entail in the present case would therefore be enormous.

- (b) The provisions are an inherent part of the licensing scheme that must be eliminated
- India presented the argument that the provisions on which the European Communities and the United States were basing their claims were an "inherent" part of the licensing scheme. India noted that the European Communities and the United States claimed that the granting of import licences subject to indigenization and trade balancing was not only inconsistent with Article XI of the GATT but also with Article III:4 of the GATT and Article 2 of the TRIMs Agreement. Claims related to these provisions were not specifically mentioned in the mutually agreed solution negotiated with the European Communities nor in the DSB rulings obtained by the United States. However, the alleged violations of Article III:4 of the GATT and Article 2 of the TRIMs Agreement were an inherent part of the licensing scheme that India was obliged to eliminate as a result of the ruling on Article XI. By eliminating the licensing scheme, India would necessarily also eliminate those aspects of that scheme that the European Communities and the United States considered to be inconsistent with those provisions. India considered that it would be completely pointless to go through a whole new dispute settlement proceeding for the sole purpose of adding two further legal grounds why the scheme must be eliminated. In India's view, if the issue of the consistency of India's licensing scheme under Article III:4 of the GATT and Article 2 of the TRIMs Agreement had been raised before the panel on *India – Quantitative Restrictions*, the panel would probably have exercised judicial economy and declared a ruling on that issue unnecessary to resolve the dispute. India noted that Article 3:10 of the DSU provided that "if a dispute arises, all Members will engage in these procedures in good faith in an effort to resolve the dispute" and suggested that the Panel ask the complainants how they reconciled their approach with the principle of good faith set out in Article 3:10 of the DSU.
- 4.57 In this connection it was important to keep in mind that Article III:4 of the GATT could be violated only through "laws, regulations and requirements", and Article 2 of the TRIMs Agreement only through "measures" adopted by a Member. The MOUs, as such, and the conduct of firms under those MOUs, by itself, were therefore not covered by these provisions of the GATT and the TRIMs Agreement. The only "requirement" and "measure" that the Indian government {currently} imposed in respect of the MOUs were set out in paragraphs 2 and 4 of the Notice, which provided in relevant part:
 - ... import of components for motor vehicle in ckd/skd form, which is restricted for import under the current Export-Import Policy, shall be allowed for importation against a license and such a license will be issued only to joint venture automobile manufacturing companies on the basis of an MOU to be signed by these companies ...

⁹⁹ ICJ Reports (1954).

Japan – Taxes on Alcoholic Beverages, (hereinafter "Japan – Alcoholic Beverages II"), (WT/DS8/R, WT/DS10/R, WT/DS11/R), adopted 1 November 1996, as modified by the Appellate Body Report, para. 5.3 (DSR 1996:I, 125).

The MOU Scheme would be enforced through the import licensing mechanism and MOU signing firms would be granted import licenses by DGFT based on above parameters.

- 4.58 The MOUs were thus at the time when the requests for the establishment of a panel were submitted enforced exclusively through the denial of import licenses for CKD/SKD kits. The application of discretionary import licensing for CKD/SKD kits was consequently the only governmental measure related to the MOUs that was {currently} in place.
- 4.59 It followed from the above that, once the discretionary licensing system for CKD/SKD kits was eliminated, the only governmental measure currently in place in respect of the MOUs would be eliminated, and as a result also the factual basis on which the EC's and the US's claims of violation of Article III:4 of the GATT and Article 2 of the TRIMs Agreement rested. A separate legal finding of inconsistency with Article III:4 of the GATT and Article 2 of the TRIMs Agreement was for these reasons completely unnecessary. The new and different measures that India might take in the future in respect of the MOUs, including their enforcement through the courts, were not before the Panel.
- The **European Communities** responded that, although recognising that Public Notice No. 60 4.60 and the MOUs were not even mentioned in the 1997 Agreement¹⁰¹, India contended that they "were an inherent part of the licensing scheme that India was obliged to eliminate as a result of [the 1997 Agreement | 102. Until 1 April 2001 the balancing and indigenization requirements had been applied in conjunction with the licensing scheme for passenger cars, but this did not make them an inherent part of that scheme. The term "inherent" meant "existing in someone or something as a natural and inseparable quality, characteristic or right" 103. The balancing and indigenization requirements (those at issue in this dispute and, more generally, all similar requirements) were clearly distinct and separable from and, therefore, not "inherent" in discretionary import licensing: it was perfectly possible to operate a discretionary import licensing scheme without applying any balancing or local content requirements in conjunction therewith. Most products subject to discretionary import licensing under the EXIM Policy before 1 April 2001 were not subject to balancing and indigenization requirements; likewise, balancing and local content requirements could be operated in the absence of discretionary import licensing. That the balancing and indigenization requirements were not an "inherent part" of the import licensing scheme was further confirmed by the fact that the MOUs had remained binding and enforceable after the elimination of that scheme, even though no additional measure to achieve that effect had been taken by the Indian authorities either before or after 1 April 2001. Furthermore, assuming that the balancing and indigenization requirements were in fact an "inherent part" of the licensing scheme, they should have been eliminated automatically as of 1 April 2001.
- 4.61 The **United States** noted that India, despite acknowledging that the scope of "*res judicata*" was not an issue in this dispute once the United States claims were properly understood, nevertheless insisted that Public Notice No. 60 and the MOUs were "an inherent part of the licensing scheme" that India had now eliminated. India also maintained that it "fails to understand what interest protected by the DSU the European Communities and the United States are pursuing by requesting rulings on the specific conditions attached to the grant of licenses for SKD/CKD kits when they have already obtained an agreement and a DSB ruling according to which the whole licensing scheme for such kits must be eliminated on 1 April 2001 and the scheme was in fact eliminated on that date." For the United States too, there was nothing "inherent" about the indigenization and trade balancing

¹⁰³ Webster's New World Dictionary, Third College Edition.

¹⁰¹ India's response to Question 37 of the Panel.

¹⁰² Ibid

¹⁰⁴ India's response to Question 37 of the Panel.

¹⁰⁵ Response to Question 38.

requirements. Import licensing of the kind that India had maintained until 1 April 2001, could have easily been administered without imposing export requirements or requiring discrimination in favor of domestic goods. India had given the Panel no reason to believe otherwise. In any case, India had effectively confirmed this point by saying that "it would in any case have been totally impracticable for India to notify these details [i.e. the details of the licensing regime India applied to each restriction notified under Article XVIII:B] for 2,700 items and for the Committee on Balance-of-Payments Restriction to examine them." The fact that those details varied by item made clear that no one set of administrative details was inherent in the restrictions themselves. Moreover, the indigenization and trade balancing requirements were continuing in force even though the import licensing regime had been eliminated, as India had acknowledged. This too demonstrated that they were independent of, and not inherent in, that licensing regime. The fact that those requirements were continuing also answered India's second point. What the United States sought was a ruling whose legal consequences would entail the elimination of separate import restrictions imposed by Public Notice No. 60 and the MOUs, as well as the elimination of the less favorable treatment that Public Notice No. 60 and the MOUs accorded to imported automotive parts and components than to like domestic parts and By contrast, the legal consequences of the rulings in the India - Quantitative components. Restrictions report entailed the elimination of the import licensing requirement for a large number of goods, including SKD/CKD kits, a different matter entirely. Eliminating the latter would not eliminate the former.

- 4.62 In reply to a Panel question, the United States added that it had not been able to envision any situation in which a panel could have settled an issue on which it did not expressly make findings. Under DSU Article 11, the function of a panel was to "make such other findings as will assist the DSB in making the recommendations or in giving the rulings provided for in the covered agreements." The DSU did not give any legal status to findings that a panel does *not* make. Furthermore, GATT and WTO dispute settlement panels were authorized to exercise judicial economy and not to make a finding on a particular legal issue if the finding was not necessary to enable the Dispute Settlement Body to make sufficiently precise recommendations and rulings to allow for prompt compliance in order to ensure the effective resolution of the dispute. It was difficult to see how a finding that a panel chose not to make for reasons of judicial economy could settle the issue that would have been the subject of the finding if the panel had made it. It was equally difficult to see how a finding that a panel did not make *without* citing judicial economy could settle the issue that would have been the subject of the finding if the panel had made it.
- 4.63 **India** pointed out that the Appellate Body had ruled on the question of judicial economy in *United States Measure Affecting Imports of Woven Wool Shirts and Blouses from India* (hereinafter "US Wool Shirts and Blouses") as follows: 106
 - ... the basic aim of dispute settlement in the WTO is to settle disputes. This basic aim is affirmed elsewhere in the DSU. Article 3.4, for example, stipulates:

Recommendations or rulings made by the DSB shall be aimed at achieving a satisfactory settlement of the matter in accordance with the rights and obligations under this Understanding and under the covered agreements.

... Article 3.2 of the *DSU* states that the Members of the WTO "recognize" that the dispute settlement system "serves to preserve the rights and obligations of Members under the covered agreements, and *to clarify the existing provisions of those agreements* in accordance with customary rules of interpretation of public

¹⁰⁶ WT/DS33/AB/R, page 18 (DSR 1997:I, 323).

international law" (emphasis added). Given the explicit aim of dispute settlement that permeates the DSU, we do not consider that Article 3.2 of the DSU is meant to encourage either panels or the Appellate Body to "make law" by clarifying existing provisions of the WTO Agreement outside the context of resolving a particular dispute. A panel need only address those claims, which must be addressed in order to resolve the matter in issue in the dispute. 107

4.64 The ruling of the Appellate Body referred to the range of claims an individual panel should address. However, there was no reason why the principle that "a panel need only address those claims which must be addressed in order to resolve the matter in issue in the dispute" should not also apply when a panel was requested to rule on a matter already resolved or settled as a result of a prior invocation of the DSU. In the view of India, the Appellate Body ruling made clear that the matters "resolved" by a panel report were not only the matters on which the panel specifically ruled upon. The whole notion of judicial economy rested on the idea that a ruling on one matter could resolve another matter on which no ruling was made. For instance, if a panel ruled that a Member must eliminate its import licensing scheme, it had at the same time resolved all issues related to the administration of that scheme even if had not made rulings on the administration of the scheme.

(i) The notion of abusive splitting

India considered that there was no legal system in which a complainant was permitted to do what the European Communities and the United States were doing in the present case, namely, challenge without good cause one legal aspect of a situation in one proceeding and then another legal aspect of the same situation in a new proceeding. All legal systems imposed limits to such an abusive "splitting" of a subject matter into separate proceedings. Here again, India wished to invite the Panel to draw on generally applied principles of law and the objectives of the DSU. If an abusive splitting of legal challenges under the DSU were permitted, WTO Members would be deprived of their right to "a satisfactory settlement of the matter", to which they were entitled in accordance with Article 3.4 of the DSU. The principle of the prohibition of abusive splitting must therefore be considered to be an inherent part of the DSU procedures. India drew the Panel's attention to the importance that the Appellate Body had attached to the concept of good faith. In United States - Tax Treatment for "Foreign Sales Corporations" (hereinafter "US – FSC"), the Appellate Body described the concept of good faith as a "pervasive principle" of general international law that guides the way in which Members should interpret their rights and obligations. In *United States - Import Prohibition of Certain Shrimp and Shrimp Products* (hereinafter "US - Shrimp"), the Appellate Body explained as follows the way in which good faith constrains WTO Members' exercise of their rights under WTO law:

This principle [of good faith], at once a general principle of law and a general principle of international law, controls the exercise of rights by states. One application of this general principle, the application widely known as the doctrine of *abus de droit*, prohibits the abusive exercise of a state's rights and enjoins that whenever the assertion of a right "impinges on the field covered by [a] treaty obligation, it must be exercised bona fide, that is to say, reasonably." An abusive exercise by a Member of its own treaty right thus results in a breach of the treaty

¹⁰⁷ The "matter in issue" is the "matter referred to the DSB" pursuant to Article 7 of the DSU.

¹⁰⁸ India's responses to Questions 104 (b) and 108 of the Panel.

 $^{^{109}}$ US - FSC, WT/DS108/AB/R, adopted on 20 March 2000, as modified by the Appellate Body Report, para. 166.

rights of the other Members and, as well, a violation of the treaty obligation of the Member so acting. 110

The Appellate Body quoted an author who wrote: 4.66

A reasonable and bona fide exercise of a right in such a case is one which is appropriate and necessary for the purpose of the right (i.e. in furtherance of the interests which the right is intended to protect). It should at the same time be fair and equitable as between the parties and not one which is calculated to procure for one of them an unfair advantage in the light of the obligation assumed. A reasonable exercise of the right is regarded as compatible with the obligation. But the exercise of the right in such a manner as to prejudice the interests of the other contracting party arising out of the treaty is unreasonable and is considered as inconsistent with the bona fide execution of the treaty obligation, and a breach of the treaty.¹¹¹

India submitted that the European Communities' and the United States' exercise of their procedural rights under the DSU in the present case was abusive and did not meet the standards set by the Appellate Body. India urged the Panel to make the rulings necessary to protect India in this case, and other Members in future cases, against the use of the DSU procedures for the purpose of coercing developing countries into making concessions in policy areas not covered by WTO law.

The European Communities noted that India had invoked the existence of principles in all 4.67 legal systems whereby "matters, once settled, cannot be litigated again", but had offered no evidence of the existence of such principles, except a quotation from a Henry M. Herman dated 1886 concerning the maxim "Interest republicae ut sit finis litigium". That maxim meant simply that "there must be an end to litigation" (a platitude with which anybody would agree) and did not address specifically the issue in dispute. The European Communities agreed that where a matter had already been adjudicated in a panel report adopted by the Dispute Settlement Body, the complaining Member was bound by that report and could not request the establishment of another panel with respect to the same matter. 112 Nevertheless, the European Communities believed that panels should apply the principle of res judicata with great circumspection, especially where the complainant had been successful in the first dispute, as in the instant case. In those circumstances, the risk of unnecessary relitigation should be of less concern than that of incurring in a denial of justice by depriving a Member of its right to a panel under the DSU. The application of the res judicata principle was subject to strict requirements. There must be complete identity between the parties and between the "matters" in dispute. The identity of the "matter" required that both the measures in dispute and the claims submitted in connection with those measures be the same. It is not sufficient, therefore, that the matters be merely similar or related. In the present case, both the measures and the claims were different. Therefore, the conditions for considering that the matter in dispute was res judicata were clearly not met. As acknowledged by India 113, the Panel Report on India – Quantitative Restrictions was not binding upon the European Communities. Therefore, that report would not have the effect of res judicata vis-à-vis the European Communities even if this Panel were to conclude that the matter brought by the United States in these proceedings had already been adjudicated in that panel report. It

¹¹³ India's response to Question 42.

 $^{^{110}}$ US – Shrimp, WT/DS58/AB/R, adopted on 6 November 1998, para. 158 and fn. 156 (internal

footnote revised).

This quote in the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained for the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes from the Appellate Body report as explained in fn. 156 of *US - Shrimp*, comes 1953), Chapter 4, p.125.

See the Report of the Appellate Body on Japan –Alcoholic Beverages II, WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted on 1 November 1996, p. 24 (DSR 1996:I, 97).

was still unclear to the European Communities which were the "general principles of law" invoked by India. India had mentioned a "principle of the prohibition of abusive splitting". Later, India no longer referred to that principle, but instead to the "principle of good faith set out in Article 3:10 of the DSU". There was no reason why a complaining Member would ever wish to request a new panel with respect to a measure that had already been found to be inconsistent by another panel. Such situation might arise only as a result of an error on the part of the complaining Member regarding the scope of the earlier panel's findings, and not of bad faith. The United States would derive no advantage from "splitting" the dispute. Rather, the opposite was true. If India's view that the balancing and indigenization requirements were covered by the rulings of the panel on *India – QR*s were correct, the United States could have asked directly for a compliance panel under Article 21.5 of the DSU. That the United States had asked instead for the establishment of this panel evidence, if anything, the good faith of the United States. In contrast, India's decision to continue to apply the balancing and the indigenization requirements after 1 April 2001, notwithstanding its position that they were an inherent part of the measure that was found to be inconsistent with Article XI:1 of the GATT by the Panel on India – Quantitative Restrictions, was difficult to reconcile with the requirements of Article 3.10 of the DSU. (Not to speak of India's totally unsupported invocation of Article XVIII:B.)¹¹⁴

The **United States** hoped that India would clarify what it meant by its use of the term res 4.68 judicata, because while India referred to this "principle", India nowhere defined it. It was therefore not entirely clear what India had in mind when it used the term. Generally speaking, however, a principle of res judicata, if applicable, would presumably relate to the effect of a previously adopted panel report on a subsequent dispute involving the same matter between the same parties. India sought to support its "principle of res judicata" by reference to a decision of the International Court of Justice (ICJ) and a statement of the Appellate Body in the Japan – Taxes on Alcoholic Beverages (hereinafter "Japan – Alcoholic Beverages II") dispute. But neither of those cases provided any help for India's position. In Japan - Taxes, the Appellate Body rejected a claim that prior GATT panel decisions constituted binding precedent. 115 As for the ICJ's advisory opinion in *Effects of Awards of* Compensation Made by the United Nations Administrative Tribunal, while that opinion mentioned res judicata it did not define what it meant by that term. In any event, the question posed to the ICJ in that case was "Having regard to the Statute of the United Nations Administrative Tribunal and to any other relevant instruments and to the relevant records, has the General Assembly the right on any grounds to refuse to give effect to an award of compensation made by that Tribunal in favour of a staff member of the United Nations whose contract of service has been terminated without his assent?¹¹⁶ (i.e. was the General Assembly required to pay the award rendered by the tribunal?) The decision therefore did not address the question whether a tribunal hearing a subsequent dispute was in any sense bound by a previous tribunal decision, and therefore it is inapposite to India's contentions here.

4.69 The United States noted that India, despite acknowledging that the scope of "res judicata" was not an issue in this dispute once the US claims were properly understood, nevertheless insisted that Public Notice No. 60 and the MOUs were "an inherent part of the licensing scheme" that India had now eliminated. India also maintained that it "fails to understand what interest protected by the DSU the European Communities and the United States are pursuing by requesting rulings on the specific conditions attached to the grant of licenses for SKD/CKD kits when they have already obtained an agreement and a DSB ruling according to which the whole licensing scheme for such kits

¹¹⁴ Question 21 of the Panel.

¹¹⁵ Report of the Appellate Body in *Japan - Alcoholic Beverages II*, WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted on 1 November 1996, Part E (DSR 1996:I, 97).

¹¹⁶ (1954) I.C.J. Rep. 50.

¹¹⁷ India's response to Question 37 of the Panel.

must be eliminated on 1 April 2001 and the scheme was in fact eliminated on that date." Even if the Panel were inclined to pursue the legal issue that India had raised, it would have to conclude that India's analysis was without basis in the text of the WTO Agreement. India gave no citation at all for its so-called "principle of abusive splitting"; its one textual reference - namely, to the phrase "satisfactory settlement of the matter" in DSU Article 3.4 - did not support its position, for the straightforward reason that the "matter" at issue in this dispute was different from the "matter" at issue in the *India – Quantitative Restrictions* dispute. The measures were different; the claims were different. India's removal of the import licensing at issue in the earlier case would not end the discrimination against foreign goods or the additional import barriers that Public Notice No. 60 and the MOUs imposed. These separate violations were the ones that this Panel must address; the previous panel did not, and could not, address them.

4.70 India's "good faith" argument was equally unavailing. India referred to the provision in DSU Article 3.10 that "if a dispute arises, all Members will engage in these procedures in good faith in an effort to resolve the dispute." The Appellate Body's construction and application of that phrase, however, did not help India's position. In its report in the *FSC* dispute, the Appellate Body said:

By good faith compliance, complaining Members accord to the responding Members the full measure of protection and opportunity to defend, contemplated by the letter and spirit of the procedural rules. The same principle of good faith requires that responding Members seasonably and promptly bring claimed procedural deficiencies to the attention of the complaining Member, and to the DSB or the Panel, so that corrections, if needed, can be made to resolve disputes. The procedural rules of WTO dispute settlement are designed to promote, not the development of litigation techniques, but simply the fair, prompt and effective resolution of trade disputes. ¹¹⁹

The Appellate Body was considering purely a question of timing - whether or not an objection to a consultation request had been timely raised. In a subsequent case the Appellate Body considered the application of Article 3.10 to another question of timing - whether or not a Member had asked for clarification of a panel request in a timely manner. Nothing about Article 3.10 or the Appellate Body's construction of that provision had anything to do with "res judicata", "splitting", or the binding nature in a subsequent dispute of a ruling in an earlier one.

4.71 In effect, India was inviting this Panel to embark on an examination of "principles" that were not anchored in the text of the DSU. Previous panels had rejected similar attempts by India to obtain procedural rulings that were not based on the language of the DSU; as the United States had pointed out in its answers to the Panel's questions, the EC-Mailbox Panel properly concluded that a panel proceeding was not the appropriate forum to address systemic issues. The Panel explained that it was required to base its findings on the language of the DSU and could not make a ruling *ex aequo et bono* divorced from explicit language in the DSU. ¹²¹ In this case, not only had India failed to ground its argumentation in the text of the WTO Agreement, it had also not made any effort to deal with other DSU provisions that lead to the opposite conclusion. For example, India's position was at odds with DSU Article 3.7, which the Appellate Body noted in the *Bananas* report "suggests ... that a Member is

¹¹⁸ *Id.*, response to Question 38.

Report of the Appellate Body in *US – FSC*, WT/DS108/AB/R, adopted on 20 March 2000, para, 166.

para. 166.

Report of the Appellate Body in *Thailand – Anti-Dumping Duties on Angles, Shapes and Sections of Iron or Non-Alloy Steel and H-Beams from Poland*, WT/DS122/AB/R, adopted on 5 April 2001, para. 97.

¹²¹ Panel Report on *India – Patents (EC)*, WT/DS79/R, adopted on 22 September 1998, paras. 7.22-7.23 (DSR 1998:VI, 2661).

expected to be largely self-regulating in deciding whether any such action would be 'fruitful'. ¹²² For similar reasons, India's position was also at odds with DSU Articles 3.3 and 7.1: Much like Article 3.7, DSU Article 3.3 emphasized the importance of resolving the claims that a complaining Member "considers" to be impairing its benefits under the WTO Agreement, ¹²³ and DSU Article 7.1 allowed the complaining Member to define the measures challenged and the legal claims advanced. Furthermore, India's position was at odds with DSU Article 3.2, which, if India is correct, would be ignored in this case because violations of Article XI:1 and III:4 would go unaddressed. ¹²⁴ The same considerations apply here as applied in *EC – Mailbox*: the text of the DSU required the Panel to examine the matter that the United States had placed before it. Nothing in the language of the DSU supported the argument that India had made.

4.72 The United States noted that in its responses to the Panel's questions, India seemed to have accepted the distinction between its import licensing regime and the requirements at issue in this case. Moreover, as India noted, the issue before the Panel was not the scope of *res judicata* (or any of the other "principles" that India invoked in aid of its argument) –regardless of the precise meaning of India's "principles", this case did not involve the reprise of a dispute that had already been adjudicated, nor the "splitting" of a single matter in two. In light of this common ground on the issues before the Panel, the Panel should not tarry over India's assertions earlier in this dispute that its position was justified by a " principle of *res judicata*" or a "principle of abusive splitting"; or a principle of "good faith" There was no need for the Panel to consider whether such principles existed under the WTO Agreement, and if so what their scope might be -- those were legal issues that were neither presented by the facts before the Panel nor necessary to the resolution of this dispute.

C. ARE THE MEASURES WITHIN THE TERMS OF REFERENCE OF THE PANEL?

1. Do the measures applied after 1 April 2001 fall outside the terms of reference?

4.73 **India** said that it was not clear whether the measures on which the European Communities and the United States were requesting rulings were those that India was applying at the time when the requests for the establishment of a panel were submitted to the DSB or the measures that India might be applying as from 1 April 2001. It requested the Panel to request the complainants to clarify this issue. If the complainants were to respond that they considered India's future measures to be part of the present proceedings, India would request the Panel to issue a preliminary ruling to the effect that these measures fell outside of its mandate.

4.74 India argued that, under the DSU, Members may not bring complaints on proposed measures¹²⁵ nor may they request advisory rulings on potential future issues. This followed from Article 7.1 of the DSU, according to which the mandate of a panel was to examine "the matter" referred to the DSB in the document containing the request for establishment of the panel. According to Article 6.2 of the DSU, the panel request must identify (a) "the specific measures at issue" and (b) the "legal basis of the complaint". The Appellate Body clarified that the requirement to identify the

¹²² Report of the Appellate Body in *European Communities – Regime for the Importation, Sale and Distribution of Bananas*, (hereinafter "*EC – Bananas III*") WT/DS27/AB/R, adopted on 25 September 1997, para. 135 (DSR 1997:II, 591).

¹²³ Article 3.3 provides that "the prompt settlement of situations *in which a Member considers* that any benefits accruing to it directly or indirectly under the covered agreements are being impaired by measures taken by another Member is essential to the effective functioning of the WTO and the maintenance of a proper balance between the rights and obligations of Members" (emphasis added).

DSU Article 3.2 provides that "recommendations and rulings of the DSB cannot add to or diminish the rights and obligations provided in the covered agreements."

¹²⁵ This contrasts with the dispute settlement provisions of the North American Free Trade Agreement, which permit complaints on "any actual *or proposed* measure" (Article 2006:1).

legal basis of the complaint meant that the legal claims in respect of the measures identified must be specified in the request for the establishment of the panel. The reference of the DSU to "the matter" in the document requesting the establishment of the panel thus consisted of (a) the specific measures identified as the "measures at issue" and (b) the legal claims made in respect of those measures. It logically followed from this that both the measures at issue before a panel and the legal issues to which they gave rise must have existed at the time of the request for the establishment of a panel.

4.75 The Panel on *India – Quantitative Restrictions* had noted that:

... practice, both prior to the WTO and since its entry into force, limits the claims which panels address to those raised in the request for establishment of the panel, which is typically the basis of the panel's terms of reference (as is the case here). In our opinion, this has consequences for the determination of the facts that can be taken into account by the Panel, since the complainant obviously bases the claims contained in its request for establishment of the panel on a given set of facts existing when it presents its request to the DSB. . . . it would seem consistent with such a request and logical in the light of the constraints imposed by the Panel's terms of reference to limit our examination of the facts to those existing on the date the Panel was established. 128

In India's view, it followed from the above that the mandate of the present Panel was to examine the measures that existed at the time the panel requests were submitted by the European Communities and the United States and the legal claims that could be made in respect of those measures at that time. The relevant date for the United States was 15 May 2000 and 12 October 2000 for the European Communities. It was thus clear that the measures which India would apply as from 1 April 2001 to comply with the mutually agreed solution negotiated with the European Communities and the DSB ruling sought by the United States fell outside the Panel's terms of reference.

4.76 Although the European Communities and the United States alleged that India was currently violating its obligations under the GATT and the TRIMs Agreement, many of their arguments related to measures that India might take after 1 April 2001 and legal issues that *might* consequently arise at that time. It was unclear whether the legal arguments of the claimants related to the factual situation before or after the elimination of the discretionary licensing regime for SKD/CKD kits.

4.77 The European Communities claimed that the "trade balancing" requirements stipulated in the MOUs were inconsistent with GATT Article XI:1 in that they restricted imports of passenger cars, and of components therefor, by the signatories of the MOUs. The European Communities had recognised that the solution it reached with India under Article 3.6 of the DSU and the United States' agreement with India under Article 21.3 (b) of the DSU allowed India to maintain import restrictions on passenger cars until 1 April 2001. Therefore, the European Communities explained, it made its claim of violation of Article XI only in so far as: the MOUs require the "trade balancing" of imports

¹²⁶ See *India – Patent Protection for Pharmaceutical and Agricultural Chemical Products*, (hereinafter "*India – Patents (US)*") (WT/DS50/AB/R), adopted on 16 January 1998, para. 89 (DSR 1998:I, 9).

¹²⁷ Appellate Body Report on *EC – Bananas III*, (WT/DS27/AB/R), adopted on 25 September 1997, para. 143 (DSR 1997:II, 591), and Appellate Body Report on *India – Patents (US)*, (WT/DS50/AB/R), adopted on 16 January 1998, paras. 87-89.

¹²⁸ See *India – Quantitative Restrictions*, (WT/DS90/R), paras. 5.160-5.161.

¹²⁹ WT/DS146/4. In *US – Certain EC Products*, the Appellate Body found that a measure which did not exist at the time when the consultations required under Articles 4 and 6 of the DSU were held could not be deemed to be a measure falling within the panel's terms of reference (WT/DS165/AB/R, para. 70). This ruling suggested that the measures at issue in a panel proceeding must have existed even at the time of consultations.

of "components" other than chassis and bodies¹³⁰; and the MOUs will remain binding and enforceable after 1 April 2001, both with respect to passenger cars and components therefor. In India's view, this made it clear that the EC's complaint was really directed against measures which India *might* take to enforce the MOUs, but that were not yet in place, and legal issues that had therefore not yet arisen.

- 4.78 The United States' argumentation was equally contradictory. Thus, the United States "applauds the efforts that India is undertaking" in eliminating the licensing restrictions. However, the United States expressed regret that this would not resolve the dispute because "India intends to maintain and continue enforcing the requirements of Public Notice No. 60 and the MOUs after its import licensing regime ends". The United States explained that the dispute concerned discriminatory, trade-restricting conditions that India exacted from investors in the motor vehicle manufacturing sector and that it intends to continue to exact. Citing press reports of statements by Indian policy makers, the United States claimed that "India does not intend to terminate the requirements" of Public Notice No. 60 or the MOUs when it eliminates the balance-of-payments restrictions on imports of SKD/CKD kits and components on 1 April 2001.
- 4.79 The prospective nature of the United States' complaint had become particularly evident when it argued that India might deny import licenses under the general enforcement and confiscation provisions of the FTDR Act even after 1 April 2001. It claimed that:

<u>These additional import-restricting provisions</u> will evidently not disappear when India eliminates its balance-of-payments licenses on 1 April 2001, but <u>will instead</u>, <u>apparently, become the instruments</u> through which India carries out the import restrictions in Public Notice No. 60 and the MOUs (and thus prevents SKD/CKD kits/components from being brought into India to compete with domestic parts and components). ¹³¹

The United States thus recognised that India would eliminate the import restrictions at issue on 1 April 2001, but nevertheless argued that India would "apparently" re-introduce new restrictions under the general enforcement provisions of the FTDR Act. It was on the basis of this alleged possibility of a future measure covered by Article XI of the GATT that it was asking the Panel to rule that India was acting inconsistently with this provision.

4.80 The above made clear that both the European Communities and the United States were satisfied with the way in which India had so far implemented the mutually agreed solution and the DSB rulings. Their complaints related to measures that India *might* in their view adopt as from 1 April 2001 and legal issues that *might* arise in this connection. However, for the reasons explained above, these complaints – should events substantiate them – should appropriately be made in a dispute settlement proceeding initiated after 1 April 2001. India concluded that the European Communities and the United States had brought their complaints to the DSU before the end of the agreed implementation period, and thus prematurely, because they claimed to have reason to believe that India *might* not bring the application of its licensing system for automotive parts and components into conformity with its WTO obligations as from 1 April 2001. India would demonstrate that this concern was unjustified. In any event, according to established WTO jurisprudence the Panel's terms of reference did not extend to measures that had not yet been adopted and claims on legal issues that had not yet arisen. For these reasons, India would request the Panel to reject both complaints as inadmissible.

¹³¹ Underlining added.

 $^{^{130}}$ India wished to point out that the MOUs did not require the trade balancing of imports of components except when such imports were in the form of SKD/CKD kits.

- (a) The method of enforcement before and after 1 April 2001
- 4.81 **India** stated that both Public Notice No. 60 and the MOUs themselves clearly and explicitly stipulated that the MOUs were to be enforced through the import licensing mechanism. They both contained the following clause:

The MOU Scheme would be enforced through the import licensing mechanism and MOU signing firms would be granted import licenses by DGFT based on above parameters. 132

- As long as India maintained its import licensing regime for SKD/CKD kits, the only possible 4.82 legal consequence of the non-observation of the terms of a MOU was therefore the denial of an import license for such kits. An enforcement of the MOUs by other means would at that time have been contrary to the policy set out in Public Notice No. 60. Moreover, even if the Government of India had decided to enforce the MOUs through the courts, the automobile manufacturer concerned could have invoked the clause in the MOU according to which the licensing mechanism would be the exclusive mechanism of enforcement. If the European Communities and the United States were to state that their complaints related to the enforcement of the MOUs outside the licensing regime and Notice No. 60, India requested the Panel to ask them to clarify which enforcement measure India had taken outside the licensing regime and Notice No. 60 at the time when the panel requests were submitted was at issue in this proceeding. Specifically, the complainants should be asked to clarify what measure that India took outside the framework of the licensing regime and Notice No. 60 at that time constituted a "requirement" and "restriction" within the meaning of Articles III and XI of the GATT and the corresponding provisions of the TRIMs Agreement. In this context, India wished to emphasise that the MOUs were, prior to 1 April 2001, enforced exclusively through the denial of import licenses for CKD/SKD kits. The application of discretionary import licensing for CKD/SKD kits was consequently the only governmental measure related to the MOUs that was in place at that time. India was particularly interested in learning from the complainants how, in the absence of any import licensing regime for CKD/SKD kits, the MOUs could entail any restriction on the importation of such kits within the meaning of Article XI of the GATT. India continued by pointing out that compliance with the commitments made by car manufacturers under the MOUs was, under the regime applied by India prior to 1 April 2001, secured exclusively through the denial of import licenses for SKD/CKD kits. This method of securing compliance could of course no longer be applied when the restrictive licensing for such imports was abolished as from 1 April 2001. From then on, the MOUs would, if necessary, be enforced through the courts since they were binding contracts under Indian civil law. This method of enforcement would not entail any denial of import licenses. The MOUs potentially might now be enforced under the provisions of the Foreign Trade (Development and Regulation) Act, 1992 (the "FTDR Act") and the Foreign Trade (Regulation) Rules, 1993. However, none of the potential methods of enforcement would entail any import restrictions.
- 4.83 It followed from the above that the enforcement of the MOUs as contracts outside the import licensing scheme for SKD/CKD kits could not have arisen at the time when the European Communities submitted its request for the establishment of the Panel. The Panel would therefore be ruling on a prospective matter that is, a legal situation which did not exist at the time of its establishment if it were to examine the legal effects of MOUs after the abolition of the licensing regime. This prospective matter was outside its terms of reference. A panel may not rule: if the respondent were to do this or were to do that, then it would be acting inconsistently with this or that provision. Or: because the respondent has done this or that during the course of the proceedings, the respondent acted inconsistently with this or that provision. It must base its ruling on inconsistencies

 $^{^{132}}$ See para. 4 of Public Notice No. 50 and para. V of the standard format for MOUs annexed to the Notice.

that had already occurred at the time of the request for its establishment. After 1 April 2001, any measures that India might take to secure compliance with remaining obligations under the MOUs would no longer entail any limitation of the right of the signatories to import SKD/CKD kits. The enforcement of the MOUs after the abolition of the import licensing system would therefore be fully consistent with India's WTO obligations.

- 4.84 The Panel needed to take into account that the issue was the applicability of provisions of contracts to which the Government of India was a party. The Government of India had repeatedly stated, that the trade balancing provisions in Public Notice No. 60 and in the MOUs did not apply to freely importable products. This was, of course, also the position of the Government of India as the holder of contractual rights under the MOUs. The Panel should for this reason accept the position of the Government of India on the applicability of the MOUs. ¹³³ India wished to stress that the measures outlined above were those actively being considered at the present time, but that no final decision on this matter had as yet been taken. None of the measures to secure compliance with remaining obligations under the MOUs described above had thus far been adopted and none of the legal issues discussed above has therefore arisen as yet. Moreover, as far as the indigenization requirement under the MOUs was concerned, it was presently not clear whether there was any need to take any action at all.
- 4.85 The **United States** responded that India's argument that the United States was challenging measures that India would take in the future was based on flawed premises. First, the United States was actually challenging measures that India had already taken namely, the imposition of local content and trade balancing obligations on MOU signatories. And, while India said that import licensing would end on 1 April India had acknowledged that nothing about the MOUs would change after that date: they remained "binding" and "will, if necessary, be enforced". These measures were no different from the binding and enforceable purchase undertakings that the *Canada FIRA* Panel found inconsistent with the GATT.
- 4.86 The distinction that India tried to draw between the situation before 1 April 2001, and the situation after 1 April 2001, was contradicted by India's acknowledgement that the MOUs would remain in force even after 1 April 2001, and by India's further acknowledgement that the trade balancing and indigenization requirements in the MOUs remained binding and enforceable after that date. In fact, in addition to reconfirming that these requirements remained enforceable in Indian courts, India's answers to the Panel's questions also confirmed that they were enforceable through monetary penalties levied under Section 11 of the FT(DR) Act.

¹³³ India would like to note in this connection that the Panel on *United States – Sections 301-310 of the Trade Act of 1974* (hereinafter "*US – Section 301 Trade Act*") based its interpretation of some of the most controversial provisions of United States trade law, *inter alia*, " on representations and statements by the representatives of the United States . . . solemnly made, in a deliberative manner, for the record, repeated in writing and confirmed in the Panel's second hearing." (WT/DS152/R, para. 7.122, adopted on 27 January 2000) Since the assurances given by United States regarding the content of its domestic law were accepted as decisive by a WTO panel, the United States should accept by the same token the position stated by the Government of India on the applicability of the MOUs.

The position was less clear with respect to Public Notice No. 60; as far as the United States could tell, this notice had not been rescinded and thus remained in force. India had said only that "Public Notice No. 60 is no longer operational because the licensing scheme it was to administer no longer exists." India's response to Question 33 of the Panel.

¹³⁵ "As pointed out above, after the abolition of import licensing for CKD/SKD kits on 1 April 2001, MOUs potentially can be enforced as contracts through the domestic courts. The companies may also be liable to monetary penalties under Section 11 of the FTDR Act." India's response to Question 52(d) of the Panel.

Those acknowledgements necessarily meant that the trade balancing and indigenization requirements were legally independent of the licensing requirements that India previously imposed on imports of SKD/CKD kits and components. As the United States had made clear, it was not the licensing requirements but the indigenization and trade balancing requirements that were the subject of the US complaint in this dispute. Because the indigenization and trade balancing requirements did not change on 1 April 2001 (or on any other date), and because those requirements were independent of India's now-eliminated licensing regime, India's elimination of import licensing was not relevant to - and certainly did not resolve - this dispute. In its answers to the Panel's questions, India appeared to have finally recognized this point. India said, "there are requirements set out in the MOUs that apply also in the absence of any import restriction on SKD/CKD kits." The United States agreed: and it was those MOU requirements - and their inconsistencies with India's WTO obligations - that were the subject of this dispute. However, India regrettably persisted in arguing that the situation after 1 April 2001 (i.e. after elimination of the import licensing regime) was somehow different from the situation before that date. According to India, enforcement of the MOUs "could not have arisen at the time when the European Communities [or, presumably, the United States] submitted its request for the establishment of the Panel. The Panel would therefore be ruling on a prospective matter – that is, a legal situation which did not exist at the time of its establishment – if it were to examine the legal effects of MOUs after the abolition of the licensing regime. This prospective matter was outside its terms of reference." 137

India's argument relied on several misunderstandings. First, the US complaint was not directed at any particular method or means of enforcing the MOUs and Public Notice No. 60. The US complaint was instead directed to the indigenization and trade balancing requirements as such, and these requirements had not changed. These requirements had imposed the same obligations on car manufacturers since the date that the MOUs were signed, and it was the US position that those obligations were themselves inconsistent with India's WTO commitments. What mattered was that those obligations were binding and enforceable; what means India chose or would choose to enforce them was simply not relevant to the US legal claims in this dispute. 138 The text of the GATT 1994 made this point clear. Article III:4 of the GATT 1994 was addressed to "... laws, regulations and requirements" The indigenization and trade balancing requirements were indisputably "requirements"; and they remained "requirements" regardless of what methods India used to enforce them. 139 India's position – that the Panel should focus not on the requirements that the measures imposed but should look instead at changes in the means of enforcement – was simply not consistent with the GATT text. The situation was no different for Article XI:1: the scope of that article had always been interpreted to be broad, and it used the term " measures," the meaning of which was at least as broad as the term "requirement". 140

¹³⁶ India's response to Question 33 of the Panel.

¹³⁷ Ibid.

Thus, even if it were true that before 1 April 2001, "as long as India maintained its import licensing regime for SKD/CKD kits, the only possible legal consequence of the non-observation of the terms of a MOU was therefore the denial of an import license for such kits" and that a manufacturer could have resisted court enforcement of the MOUs on that basis (Indian Panel Answers, response to question 33), the fact remained that the enforceability as such of the requirements in the MOUs had not changed.

¹³⁹ Furthermore, Public Notice No. 60 was certainly also a "regulation".

¹⁴⁰ Cf. also Panel Report on Japan – Measures Affecting Consumer Photographic Film and Paper, (hereinafter "Japan – Film") WT/DS44/R, adopted on 29 April 1998, para. 10.51: "Given that the scope of the term requirement would seem to be narrower than that of measure, the broad reading given to the word requirement by the Canada - Administration of the Foreign Investment Review Act (hereinafter "Canada – FIRA") and European Economic Community – Regulation on Imports of Parts and Components (hereinafter "EEC – Parts and Components") panels supports an even broader reading of the word measure in Article XXIII:1(b)."

4.89 India's position was also inconsistent with the conclusions reached by the *Canada – FIRA* Panel. That panel had found as follows:

The Panel further noted that written purchase undertakings – leaving aside the manner in which they may have been arrived at (voluntary submission, encouragement, negotiation, etc.) – once they were accepted, became part of the conditions under which the investment proposals were approved, in which case compliance could be legally enforced. The Panel therefore found that the word "requirements" as used in Article III:4 could be considered a proper description of existing undertakings. ¹⁴¹

It was the enforceability of the undertakings – not the specific means of enforcement – that led the *Canada – FIRA* panel to conclude that the undertakings were "requirements" within the meaning of GATT Article III:4. Notably, none of the undertakings examined in the *Canada – FIRA* dispute had ever actually been enforced by the Canadian Government; performance of unfulfilled undertakings had always been either postponed or waived, or the undertakings had been replaced by revised undertakings. This fact did not change the Panel's legal conclusion that the undertakings imposed "requirements" that were inconsistent with the GATT.

4.90 India was also wrong to say that the situation after 1 April 2001, was outside the Panel's terms of reference. This Panel's terms of reference were:

To examine, in the light of the relevant provisions of the covered agreements cited by the United States in WT/DS/175/4 and by the European Communities in WT/DS146/4, the matters referred to the DSB by the United States and the European Communities in those documents and to make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements. 143

The "matters referred to the DSB by the United States" in document WT/DS175/4 included "Public Notice No. 60 ((PN)/97-02) of the Indian Ministry of Commerce, published in the Gazette of India Extraordinary, effective 12 December 1997; ... memoranda of understanding signed by the Government of India with manufacturing firms in the motor vehicle sector pursuant to Public Notice No. 60 ... " India's admission that "there are requirements set out in the MOUs that apply also in the absence of any import restriction on SKD/CKD kits " made clear that those requirements fell within the terms of reference of this panel. Furthermore, India's focus on the elimination of import licensing misapprehended the limited relevance of import licensing to this dispute. India used those licenses to induce car manufacturers in India to accept the indigenization and trade balancing requirements. Car manufacturers had to sign an MOU in order to receive those licenses and to import CKD/SKD kits. As a matter of factual background, therefore, it was relevant that import licensing existed in 1997 and 1998, when the MOUs were signed. With respect to the legal claims of the United States, the import licenses constituted an "advantage" that MOU signatories received in exchange for binding themselves to the indigenization and trade balancing requirements. The Canada – FIRA panel

-

¹⁴¹ Panel Report on *Canada - FIRA*, adopted on 7 February 1984, BISD 30S/140, para. 5.4.

¹⁴² *Ibid.*, para. 2.11.

¹⁴³ India – Measures Affecting the Automotive Sector: Constitution of the Panel Established at the Requests of the United States and the European Communities: Note by the Secretariat, WT/DS146/5, WT/DS175/5, 30 November 2000, para. 2. These are standard terms of reference drawn up in accordance with DSU Article 7.1.

¹⁴⁴ See, e.g., the Panel Report on EEC - Parts and Components, adopted on 16 May 1990, BISD 37S/132, para. 5.21 (those requirements "which an enterprise voluntarily accepts in order to obtain an advantage from the government constitute 'requirements' within the meaning of GATT Article III:4). And, in

considered an analogous situation: permission to establish an investment in Canada was conditioned on compliance after investment with various GATT-inconsistent undertakings. In this dispute, access to the import licenses played the same role as the permission to invest played in that one: as the one-time "advantage" provided by the government to induce companies to accept on-going WTO-inconsistent requirements.

In a related point, India also asserted that Public Notice No. 60 was no longer "operational" 145 4.91 As a matter of fact, the status of Public Notice No. 60 was not at all clear. To the best of the United States' knowledge, Public Notice No. 60 had not been rescinded. India had not explained what it meant for an Indian regulation not to be rescinded and yet not be "operational". In any event, even if Public Notice No. 60 were no longer in force, it remained in force for several months after the establishment of this Panel (27 July 2000, for the US complaint) and the fixing of the Panel's terms of reference on that date; the Panel could and should make findings concerning it. As the United States explained in answering the Panel's question 7, past GATT and WTO panels had ruled on measures that were discontinued during the panel's examination. 146 It would be particularly appropriate for the Panel to rule on Public Notice No. 60 in this case, given that the requirements of Public Notice No. 60 were still in place through the MOUs, which all parties agreed remained in effect, binding and enforceable. The Panel would thus achieve no economy by omitting Public Notice No. 60 from its findings; to the contrary, as long as there was uncertainty over the status of Public Notice No. 60, such a ruling could help ensure clear implementation of the DSB's rulings and recommendations. 147 The United States added that the MOUs were directly binding on individual signatories because, according to India, they were contracts enforceable under Indian domestic law. However, the indigenization and trade balancing requirements had originally been established by Public Notice No. 60. In fact, the terms of the MOUs were mandated by Public Notice No. 60, paragraph 8 of which provides that "a standard format for MOU is enclosed as appendix to this Public Notice and MOU is required to be signed as per this format." Consequently, a comprehensive ruling by the Panel would apply to both Public Notice No. 60 (the source of the WTO-inconsistent requirements) and the MOUs (by which individual companies were bound to those WTO-inconsistent requirements), especially since Public Notice No. 60 was, as all the parties this dispute agreed, in force at the time this Panel was established.

In summary: as long as the MOUs remained in force, the WTO-inconsistent indigenization and trade balancing requirements remained in force. Second, as India had confirmed, the MOUs, and therefore also those two requirements, were binding and enforceable independently of the import licensing requirements, and would remain so even after the import licensing requirements were removed. Therefore, the date of 1 April 2001, had no relevance to the legal issues before this Panel

the terms of the chapeaux to paragraphs 1 and 2 of the TRIMs Agreement Illustrative List, undertaking to comply with those two requirements was "necessary to obtain an advantage" - namely, the right to receive import licenses.

¹⁴⁵ India's response to Question 33 of the Panel.

Such cases include: the Panel Report on Argentina - Measures Affecting Imports of Footwear, Textiles, Apparel and Other Items, (hereinafter "Argentina - Textiles and Apparel") WT/DS56/R, adopted as modified by the Appellate Body on other issues on 22 April 1998, para. 6.12 (DSR 1998:III, 1033); the Report of the Appellate Body in US – Wool Shirts and Blouses, WT/DS33/AB/R, adopted on 23 May 1997, page 1 (DSR 1997:I, 323); and the Panel Report on Indonesia – Autos, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 25 September 1997, para. 14.9 (DSR 1998:VI, 2201).

As the Appellate Body has explained, "A panel has to address those claims on which a finding is necessary in order to enable the DSB to make sufficiently precise recommendations and rulings so as to allow for prompt compliance by a Member with those recommendations and rulings 'in order to ensure effective resolution of disputes to the benefit of all Members." Report of the Appellate Body in Australia – Measures Affecting Importation of Salmon, (hereinafter "Australia - Salmon"), WT/DS18/AB/R, adopted on 6 November 1998, para. 223 (quoting DSU Article 21.1).

and provided India no defense to the US claims. For the United States, what was relevant to the present dispute was that, both before and after 1 April 2001, those car manufacturers who signed MOUs before 1 April had been and would continue to be subject to the indigenization and trade balancing requirements. Both after and before 1 April 2001, compliance with these requirements had been necessary in order to comply with Indian law. Because those requirements were legally binding on MOU signatories and enforceable against them, they were measures that this Panel must rule upon. 148

- 4.93 The **European Communities** noted that India contended that the EC's complaint did not relate to measures that were in place when the panel request was submitted, but rather to the measures which India might take to enforce the MOUs as from 1 April 2001. Therefore, according to India, the measures in dispute would fall outside the terms of reference of this Panel. The European Communities contended that any measures taken after 1 April 2001 would be outside the terms of reference of this Panel to the extent that they changed the essence of the measures in dispute. It was common ground that the MOUs signed before the date of establishment of this Panel had remained in place after 1 April 2001 and continued to be binding upon the signatories. The only change was that the balancing and indigenization requirements had ceased to be enforceable though the denial of licenses. Nevertheless, those requirements had remained enforceable though other means. India's answers to the Panel's questions confirmed that:
 - the MOUs had not been eliminated as of 1 April 2001, despite India's claims that they were an "inherent part" of the import licensing scheme for passenger cars eliminated on that date;
 - after 1 April 2001, the signatories continued to be required to meet the commitments imposed by the MOUs¹⁵⁰;
 - after 1 April 2001, the MOUs were enforceable as contracts through the domestic courts¹⁵¹;
 - in addition, the MOUs were also enforceable under the FD(TR) Act¹⁵², in particular through the imposition of monetary penalties¹⁵³;
 - India had taken no action in order to achieve the effect that the MOUs remained legally binding and enforceable after 1 April 2001¹⁵⁴, either before or after that date.

4.94 The method of enforcement of the MOUs was not a constituent element of the violations claimed by the European Communities. The European Communities did not claim that the balancing and indigenization requirements were inconsistent with the GATT and the TRIMs Agreement because they were enforced through the denial of licenses. Those requirements were always inconsistent with the GATT and the TRIMs Agreement, irrespective of the precise method used to enforce them. Therefore, the change which occurred as of 1 April 2001 did not give rise to a new "legal situation" 155.

¹⁴⁸ Question 12 of the Panel.

¹⁴⁹ Cf. the Report of the Appellate Body on *Brazil – Export Financing Programme for Aircraft*, (hereinafter "*Brazil – Aircraft*"), WT/DS46/AB/R, adopted on 20 August 1999, at paras. 130 *et seq*.

¹⁵⁰ India's response to Question 33, para. 5.

¹⁵¹ India's response to Question 52 (d).

¹⁵² India's response to Question 52 (a).

¹⁵³ *Ibid*.

¹⁵⁴ India's response to Question 56.

¹⁵⁵ India's response to Question 33, at p. 4, para. 4.

It did not transform the MOUs into different measures and it did not bring them outside the scope of the terms of reference of the Panel, contrary to India's contentions. Any genuinely new measures taken by India after 10 October 2000 would be outside the terms of reference of this Panel. For the reasons explained elsewhere, a mere change of the method of enforcement of the balancing and indigenization requirements after that date would not transform those requirements into new measures beyond the terms of reference of this Panel. The EC's claims were addressed against the balancing and indigenization requirements as such, and not against the method used by India in order to enforce those requirements. Even if it were true that, as India argued now, before 1 April 2001 the MOUs could be enforced exclusively through the denial of import licenses, the use of a different means of enforcement as from 1 April 2001 would not change the obligations imposed by the balancing and indigenization requirements and would not transform them into different measures falling outside the terms of reference of this Panel. 157 The measures applied by India after 1 April 2001 were the same as those that were the subject of the panel request submitted by the European Communities on 12 October 2000. The balancing and indigenization requirements continued to impose the same obligations upon the signatories of the MOUs. The only change occurring on 1 April 2001 concerned the method of enforcement of those obligations. That change, however, was without relevance for this dispute. The EC's complaint was not directed against any of the specific methods available to the Indian authorities in order to enforce the balancing and indigenization requirements, but against the requirements themselves.

The European Communities agreed with India that the DSU did not allow a Member to launch a dispute settlement proceeding on "the-WTO consistency of implementation measures before those implementation measures are taken" 158. But that situation was not present here. The European Communities was quite simply not challenging any "implementation measures". As explained [repeatedly], the EC's complaint was not directed against the specific enforcement actions which India might be required to take in order to enforce the MOUs. In other words, the European Communities did not challenge the imposition of fines pursuant to the FT(DR) Act or the bringing of actions before the civil courts for breach of contract. Instead, the EC's complaint was addressed against the MOUs as such, regardless of the method used to enforce them. Contrary to India's assertion¹⁵⁹, the EC's complaint was not addressed against the enforcement measures which India might take after 1 April 2001. [The panel request and the EC's First Submission to the Panel made it perfectly clear that] the EC's complaint was directed against Public Notice No. 60 and the MOUs as such. Public Notice No. 60 and the MOUs were adopted by India well before the European Communities submitted its panel request and were unquestionably within the terms of reference of this Panel. The European Communities did not contest that Public Notice No. 60 ceased to be effective as of 1 April 2001. Yet, by now it was a well established practice that Panels should rule on all the measures within their terms of reference, even if they were discontinued during the proceedings. ¹⁶⁰ The European Communities saw no reason why this Panel should depart from that practice. The European Communities, therefore, reiterated its request to the Panel to find that, at the time when this Panel was established, Public Notice No. 60 was inconsistent with Articles XI:1 and III:4 of the GATT and Article 2.1 of the TRIMs Agreement. India's defence was premised on the mistaken assumption that the MOUs were not "measures", and that only the specific actions that India might take in order to enforce them could be subject to dispute settlement. That assumption was manifestly wrong, and so was India's defence. The MOUs were acts of the Indian Government. Moreover, they were binding upon the signatories

¹⁵⁶ EC's response to Question 18 of the Panel.

 $^{^{157}}$ Cf. the Report of the Appellate Body on Brazil - Aircraft, WT/DS46/AB/R, adopted on 20 August 1999, paras. 130 et seq.

¹⁵⁸ India's response to Question 37 of the Panel.

¹⁵⁹ Ibid.

 $^{^{160}}$ See e.g., the Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS64/R, adopted on 23 July 1998, para. 14.9 and the reports cited therein (DSR 1998:VI, 2201).

and legally enforceable. In the light of well established case law, that was more than sufficient to conclude that they were, as such, "measures" in the sense of GATT Article XI:1 as well as "requirements" in the sense of GATT Article III:4. Since the MOUs were "measures" and " requirements" on their own, they could be the subject of dispute settlement from the moment of their conclusion and for as long as they remained enforceable, regardless of whether the Indian authorities needed to take any specific actions to enforce them. Indeed, it was very likely that, after 1 April 2001, India would not have to take any enforcement actions at all, because, just as they had done until now, the signatories would comply with the terms of the MOUs in order to avoid the imposition by the Indian authorities of the penalties provided for in the FDTR or to be sued by the Indian Government before a civil court.

The European Communities argued that until 1 April 2001, the Indian authorities had had the possibility to enforce the MOUs by denying import licenses for passenger cars in SKD/CKD form to the signatories who failed to comply with the indigenization and balancing requirements set out in their MOUs. But this was just one of the various enforcement mechanisms available to the Indian authorities. The European Communities noted that the Indian Government had indicated that "the obligation under the MOUs already entered shall continue to be valid even after the termination of Public Notice as they shall continue to be enforceable under the FT (DR) Act¹⁶¹. The enforcement mechanisms provided for in the FDTR Act included the cancellation or suspension of the Importer-Exporter Code Number¹⁶² and of the import or export licenses¹⁶³, the imposition of fines¹⁶⁴ and the confiscation of the imported or exported goods¹⁶⁵. The terms of the MOU can be enforced in Indian courts as it is a legal contract between the Government of India and the joint venture car manufacturer. 166 During the consultations, India had said that the MOUs were enforceable under the Foreign Trade (Development and Regulation) Act of 1992 (the "FTDR Act"), and would remain so after 1 April 2001. Thus, according to India,

"... the obligation under the MOUs already entered shall continue to be valid even after the termination of Public Notice No. 60 as they shall continue to be enforceable under the FT(DR) Act."167

and

"The conditions mentioned on the MOUs signed by different joint venture automobile companies are enforceable as per the provisions of the Foreign Trade (Development & Regulation) Act, 1992."168

Yet, at the first meeting with the Panel, India seemed to take the position that the MOUs were not, and indeed would have never been, enforceable under the FT(DR) Act. (India explained away the above statements as being made by technicians without legal expertise!). The European Communities noted that the Panel has asked several questions to India in order to clarify this issue. 169

¹⁶⁶ India's responses to Questions by the United States, 13 July 2000, responses to Supplemental

 $^{^{161}}$ US Questions, TRIMs Committee, Response to Question 10.

¹⁶² Section 8 of the FTDR Act. Section 7 of the FTDR Act stipulates that "No person shall make any import or export except under an Importer-Exporter Code Number".

¹⁶³ Section 9 of the FTDR Act.

¹⁶⁴ Section 11 of the FTDR Act.

¹⁶⁷ India's responses to Questions by the United States, 20 May 1999, response to Question 10.

¹⁶⁸ India's responses to Supplemental Questions by the United States, 30 July 1999, response to Question 3. $\,^{169}$ Cf. Questions 52 (a) and (b) of the Panel.

consultations, India had also confirmed that the MOUs were binding contracts under India's civil law which could be enforced through the courts:

"The terms of the MOU already entered into can be enforced even after 2001 in the Indian courts as it is a legal contract between the Government of India and the joint venture car manufacturer." ¹⁷⁰

It must be emphasised that, even on India's interpretation, the enforceability of the MOUs as contracts after 1 April 2001 was not the result of any action taken by the Indian Government after the establishment of this Panel. India appeared to admit that the MOUs had been binding contracts from the moment of their signature. Nevertheless, it argued that before 1 April 2001 they could not be enforced through the courts, because of the clause contained in paragraph V of the standard format for MOUs. Following the elimination of the discretionary import licensing on 1 April 2001, the inhibitory effect of that clause would have ceased and the MOUs would have become enforceable through the courts automatically. In spite of the above, India now contended that, before 1 April 2001, the MOUs could be enforced exclusively through the denial of import licenses for CKD/SKD kit. According to India, this would result from paragraph V of the standard format for MOU, which provided that:

The MOU scheme would be enforced through the import licensing mechanism and MOU signing firms would be granted import licenses based on the progress made in respect of the parameter mentioned at para. III above.

The European Communities considered that the wording of the above clause did not exclude the possibility for the Indian authorities to enforce the MOUs through other methods generally available under Indian law. Indeed, if the interpretation of paragraph V now made by India were correct, it would mean that, as from 1 April 2001, the MOUs would have become unenforceable, since any attempt by the Indian authorities to enforce them as contracts through the courts would amount to a breach of paragraph V. The European Communities doubted that it was in the interest of India to maintain that interpretation. At any rate, whether or not the MOUs were enforceable exclusively through the denial of import licenses prior to 1 April 2001 was ultimately irrelevant. As explained, a mere change in the method used to enforce the balancing and indigenization requirements was without consequences for the legal characterisation of those requirements and did not transform them into new measures falling outside the terms of reference of this Panel.

4.97 The European Communities agreed with India that the Panel "must base its ruling on inconsistencies that had already occurred at the time of the request for [the establishment of the Panel]". However, the continued existence and enforceability of the MOUs after 1 April 2001 did not result from any additional measure taken by the Indian Government after that request, but exclusively from the terms of the MOUs concluded by the Indian Government before that date. Thus, the inconsistencies claimed by the European Communities existed from the moment of the conclusion of the MOUs. For the same reason, India's argument that the European Communities was requesting the Panel to rule on a "prospective matter" was also without merit. The European Communities was not asking the Panel to rule that India would infringe the GATT and the TRIMs Agreement if "India were to do this or were to do that" in order to enforce the MOUs. The European Communities claimed that already the existence of binding and enforceable MOUs was as such inconsistent with the GATT and the TRIMs Agreement, regardless of whether any specific enforcement action needed to be taken. It was well established in GATT/WTO law that mandatory measures could be challenged without

 $^{^{170}}$ India's responses to Supplemental Questions by the United States, 30 July 1999, response to Ouestion 4.

¹⁷¹ See India's responses to Supplemental Questions by the United States, 30 July 1999, response to Ouestion 4.

having to wait until they were actually enforced.¹⁷² Indeed, to hold the opposite, as India appeared to be suggesting, would mean that the MOUs could only be challenged in the WTO if and when the signatories failed to observe their terms.

4.98 The **United States** recalled that India had advised it as early as May of 1999 – some twelve months before it requested this Panel, and almost two years before India abolished import licensing for SKD and CKD kits – that the requirements of the MOUs could be enforced through the provisions of the FT(DR) Act and through Indian civil courts. India reconfirmed the point in July 2000. As the United States mentioned at the outset it was led to pursue this complaint precisely because India had told it about these additional means of enforcement, and because the trade balancing and indigenization requirements in the MOUs would not come to an end at the same time as the import licensing regime. Despite all of India's discussion of 1 April 2001, nothing of substance had actually changed with the MOUs: The requirements were still in place; they were still binding and enforceable; companies faced the prospects of an action against them in court, or enforcement under the FT(DR) Act, if they did not comply; and India – though it said it could waive compliance with the requirements if it so chose – had in fact not yet chosen to do so.

4.99 **India** disagreed with the assertions by the European Communities and the United States that the precise nature of the method used to enforce the balancing and the indigenization requirements was without relevance for the legal characterisation of those requirements. At issue was not merely a change in the method of enforcement of the MOUs. There was a substantial difference between a requirement that had to be fulfilled to obtain an import license under the Indian trade laws and a requirement that must be met under a private law contract with the Government of India. The Government of India was bound by its trade law. Once the restrictions on imports of cars had been imposed under the EXIM policy and conditions for the grant of import licenses were set out in Public Notice No. 60, the Government had no option but to act in accordance with that policy and that Notice. However, under the law of contract in India, it was always open to the Government of India, like any other party to a contract, to determine whether it would enforce or waive its rights under the MOUs. 173 With the removal of the restrictions on imports of cars in the form of SKD/CKD kits, the Government of India therefore acquired discretion with respect to the application of the indigenization requirements that it did not previously have. ¹⁷⁴ This created a new legal situation which did not exist when the requests for the establishment of a panel were submitted and could therefore not be examined by the panel. India was entitled to the presumption that it would exercise its discretion under the MOUs in a WTO-consistent manner. If the European Communities and the United States were to consider any future action in respect of the contractual indigenization requirements to be WTO-inconsistent, they had the right to invoke the DSU in respect of that matter. However, they could not ask this Panel to rule on that possible future action. India wished to recall in this context that, under the consistent jurisprudence of GATT and WTO panels, discretionary legislation as such could not be challenged, only the application of such legislation in a specific case. The practice of GATT panels was summed up in the report of the panel on United States - Measures Affecting the *Importation, Internal Sale and Use of Tobacco* as follows:

¹⁷² See e.g., the Panel Report on *United States – Taxes on Petroleum and Certain Imported Substances*, adopted on 17 June 1987, BISD 34S/136, 160, para. 5.2.2 (the objective of Article XI:1 "could not be attained if contracting parties could not challenge existing legislation mandating actions at variance with the General Agreement until the administrative acts implementing it had actually been applied to their trade.").

¹⁷³ The only limitation on actions by the Government of India in the realm of contract was the right to equality contained in Article 14 of the Constitution of India, which would require the Government to follow a uniform policy with respect to all contracts that fall in the same class or category.

¹⁷⁴ See also India's response to Question 127 of the Panel.

... panels had consistently ruled that legislation which mandated action inconsistent with the General Agreement could be challenged as such, whereas legislation which merely gave the discretion to the executive authority of a contracting party to act inconsistently with the General Agreement could not be challenged as such; only the actual application of such legislation inconsistent with the General Agreement could be subject to challenge.¹⁷⁵

If WTO Members must be presumed to apply their discretionary legislation WTO-consistently, they must also be presumed to exercise their discretion in exercising their private contractual rights WTOconsistently. Subsidiarily, India pointed out that there was no reason why the discretion that the executive branch of a WTO Member had under its trade law and the discretion it had under its civil law should be treated differently. In both instances, it was only fair if other Members were required to await the application of the domestic law before resorting to the WTO dispute settlement procedures. India noted that the United States had vigorously defended this principle when its domestic law was challenged by other WTO Members - most recently in the WTO proceedings on Sections 301-310¹⁷⁶ of the Trade Act of 1974 and on the Anti-dumping Act of 1916. ¹⁷⁷ The United States had in particular rejected the idea that the WTO obligations in respect of domestic discretionary law should be "interpreted by reference to a new-found obligation to avoid uncertainty and to ensure 'security and predictability'." ¹⁷⁸ India was entitled to the presumption that it would exercise its discretion under the MOUs in a WTO-consistent manner just as the United States was deemed entitled to the presumption that it would apply its trade laws WTO-consistently. However, these were matters that this Panel need not resolve because they arose only after the requests for the establishment of a panel had been submitted.

4.100 India said that the MOUs were merely "private contractual arrangements" between the Indian Government and car manufacturers, and that because of the elimination of import licensing, India now had discretion to waive its rights to enforce the indigenization requirement. As a consequence, India argued the requirement was a "discretionary" measure that could not be challenged. In the view of the United States, India's argument was mistaken, however, for the fundamental reason that at no time had the indigenization and trade balancing requirements ceased to be binding and enforceable on MOU signatories. The fact that India's enforcement options might have changed did not mean the requirements themselves had changed. Moreover, the fact that India had the option of revoking these requirements did not change their mandatory nature. The situation here was no different than that of any WTO-inconsistent measure: a WTO Member generally had the power to withdraw such a measure, but that did not render the measure discretionary. For example, mandatory legislation could typically be repealed; but that fact did not mean that legislation was inherently discretionary. To support its argument, India invoked the "consistent jurisprudence of GATT and WTO panels" that "discretionary legislation as such cannot be challenged". India misconstrued that jurisprudence, however. For example, India cited the Appellate Body report in the US – 1916 Act dispute, but that report did not support India's position. In that case, the Appellate Body drew particular attention to the reasoning in one paragraph of the GATT 1947 United States - Measures Affecting Alcoholic and

BISD 41S/131, para. 118. See also the jurisprudence cited in this report and the Report of the Appellate Body in *United States – Anti–Dumping Act of 1916*, (hereinafter "*US – 1916 Act*"), (WT/DS136/AB/R, WT/DS162/AB/R), adopted on 26 September 2000, paras. 84-102.

Panel Report on US – Section 301 Trade Act, (WT/DS152/R), adopted on 27 January 2000, paras. 4.301-4.317.

¹⁷⁷ Report of the Appellate Body in *US – 1916 Act*, (WT/DS136/AB/R, WT/DS162/AB/R), adopted on 26 September 2000, para. 13.

Panel Report on US – Section 301 Trade Act (WT/DS152/R), adopted on 27 January 2000, para. 4.301.

Malt Beverages Panel Report (hereinafter "US – Malt Beverages"). In that paragraph, the US - Malt Beverages Panel reasoned as follows:

In respect of the United States contention that the Massachusetts measure was not being enforced and that the Rhode Island measure was only nominally enforced, the Panel recalled its discussion of mandatory versus discretionary laws in the previous section. The Panel noted that the price affirmation measures in both Massachusetts and Rhode Island are mandatory legislation. Even if Massachusetts may not currently be using its police powers to enforce this mandatory legislation, the measure continues to be mandatory legislation which may influence the decisions of economic operators. Hence, a non-enforcement of a mandatory law in respect of imported products does not ensure that imported beer and wine are not treated less favourably than like domestic products to which the law does not apply. Similarly, the contention that Rhode Island only "nominally" enforces its mandatory legislation a fortiori does not immunize this measure from Article III:4. The mandatory laws in these two states by their terms treat imported beer and wine less favourably than the like domestic products. Accordingly, the Panel found that the mandatory price affirmation laws in Massachusetts and Rhode Island are inconsistent with Article III:4, irrespective of the extent to which they are being enforced. 180

Thus, even if India were actually not enforcing these requirements, the situation in this dispute would be the same as in *US - Malt Beverages*. As India had said, a car manufacturer that signed an MOU was legally obligated, under the contract law of India, to achieve 70% local content and to discharge its export obligations. The manufacturer could not do otherwise without breaching those contractual obligations; those obligations were mandatory. And, as in the *US - Malt Beverages* dispute, the fact that India might eventually decide not to enforce those obligations did not make them any less mandatory. This panel, like the *US - Malt Beverages* panel before it, should conclude that these requirements were inconsistent with India's WTO obligations. As a practical matter, if the Panel accepted India's argument, India's measures could never be examined unless manufacturers were willing to violate their contracts and expose themselves to the risk of enforcement by the Indian authorities – hardly a satisfactory outcome for them or for those WTO Members whose trade interests were undermined by these requirements.

4.101 A similar issue had also arisen in *Canada – FIRA*. That panel considered but rejected an argument that contractual undertakings like the MOUs in this case should be considered as private contractual obligations of particular foreign investors vis-à-vis the government. In particular, that panel concluded that such contractual obligations should not adversely affect the rights which GATT contracting parties possess under Article III:4.¹⁸¹ In fact, many of the undertakings had actually been waived by the Government¹⁸² but that did not change the Panel's legal conclusion that the undertakings were inconsistent with the GATT. The United States did not doubt that India could revoke the MOU requirements if it wished to do so. Of course, it had in fact *not* done so. Regrettably, India had reiterated that MOU signatories were still "bound" by those obligations. India's insistence on that point showed how far from "discretionary" the requirements of the MOUs really were and underscored the contradictory nature of India's arguments. If India actually took the welcome step of permanently rescinding the MOU requirements, it would indeed be changing the

¹⁸² *Ibid.*, para. 2.11.

 $^{^{179}}$ Report of the Appellate Body in US -1916 Act, WT/DS136/AB/R, WT/DS162/AB/R, adopted on 26 September 2000, para. 91, fn. 50.

¹⁸⁰ Panel Report on *United States – Measures Affecting Alcoholic and Malt Beverages*, (hereinafter "US – Malt Beverages") adopted on 19 June 1992, BISD 39S/206, para. 5.60.

Panel Report on Canada – FIRA, adopted on 7 February 1984, BISD 30S/140, para. 5.6.

MOU signatories' mandatory obligations. But the fact that India could in principle take that step did not change the legal analysis: the requirements themselves were binding and enforceable, and thus they were not discretionary. And, India's refusal actually to take this step – as opposed to stating that it had the power to do so – reinforced the need for Panel findings that would confirm India's obligation to do so.

4.102 The **European Communities** said that Panels were not bound by a Member's interpretation of its domestic law. Rather, Panels must treat the domestic law of a Member as a fact, the establishment of which was subject to the ordinary rules on the burden of proof. India could not ask this Panel to accept its interpretation of the MOUs simply because India was a party to the MOUs. Rather, India must prove that the MOUs actually said what India pretended now that they said. India made a similar argument in *India – Patents (US)*. The Appellate Body rejected it in categorical terms. According to the Appellate Body, India's position amounted

"... to say that only India can asses whether Indian law is consistent with India's obligations under the WTO Agreement. This clearly cannot be so." 185

India's reliance on *US – Section 301 Trade Act*¹⁸⁶ was inapposite. That case did not stand for the proposition that Panels were bound to accept whatever views were expressed by a Member with respect to the interpretation of its domestic legislation. In *US – Section 301 Trade Act*, the Panel found that the relevant statutory language (as interpreted by the Panel, and not by the United States) left a margin of discretion to the US Government. The Panel then relied upon the Statement of Administrative Action made to the US Congress at the time of the approval of the results of the Uruguay Round, and upon the statements made by the United States to the Panel to the effect that it would exercise that margin of discretion consistently with its WTO obligations in order to conclude that the measures in dispute were not inconsistent with the WTO Agreement.¹⁸⁷

4.103 This case concerned an entirely different situation. The MOUs left no margin of discretion to the Indian Government for making the interpretation which it had presented to the Panel. The MOUs required to balance all the imports made by the signatories until they achieved a 70% indigenization level. This left no scope for India's interpretation. The Panel could not accept an interpretation which was at odds with the plain meaning of the MOUs simply because it had been put forward by India.

4.104 India argued that, since 1 April 2001, the indigenization requirements had become "discretionary legislation" not subject to dispute settlement because, from that date, the MOUs were enforceable only as contracts and, under Indian contract law, the Indian Government was free to decide whether or not to enforce them. This argument reflected an erroneous understanding of the notion of "discretionary legislation", was refuted by well established precedent and, moreover, would lead to a manifestly absurd and unacceptable result. In WTO law the term "discretionary legislation" was used to designate legislation which allowed, but did not mandate, the executive branch of the Government to take measures that were inconsistent with the WTO Agreement. In the present case, the European Communities did not claim that the MOUs were inconsistent with the GATT because they «allow» the Indian Government to take other, GATT inconsistent measures. Rather, the European Communities claimed that the MOUs as such were GATT inconsistent, regardless of the measures that might be taken subsequently by the Indian Government in order to enforce them. There

 $^{^{183}}$ Panel Report on $\it US-Section~301~Trade~Act, WT/DS152/R,$ adopted on 27 January 2000, para. 7.19 184 $\it Ibid.,$ para. 7.18.

¹⁸⁵ Report of the Appellate Body in *India – Patents (US)*, WT/DS50/AB/R, adopted on 16 January 1998, at para. 66.

¹⁸⁶ Panel Report on *US – Section 301 Trade Act*, WT/DS152/R, adopted on 27 January 2000. ¹⁸⁷ *Ibid.*, para. 7.117.

was a fundamental difference between the «discretionary legislation» considered in previous cases and the MOUs. "Discretionary legislation" did not, as such, impose any legal obligations upon the economic operators. The obligations for the operators arose exclusively from the measures taken subsequently by the executive branch on the basis of the authority granted by the "discretionary legislation". In contrast, the MOUs imposed binding obligations upon the signatories, which they must fulfil in order to avoid the risk of sanctions. The measures which the Indian Government might take pursuant to the MOUs would not create any new legal obligations for the operators, but served merely to enforce those obligations already imposed by the MOUs. Previous cases confirmed that the discretion to enforce or not to enforce a measure did not transform such measure into "discretionary legislation" immune to dispute settlement action. In $US-1916\ Act$ the Appellate Body ruled that the discretion enjoyed by the US Department of Justice in order to initiate or not to initiate criminal proceedings was not

discretion of such nature or of such breadth as to transform the 1916 Act into discretionary legislation, as this term has been understood for purposes of distinguishing between mandatory and discretionary legislation¹⁸⁸.

By the same token, the discretion enjoyed by India's Government to bring or not an action before a civil court in order to enforce the MOUs did not transform the MOUs into discretionary legislation.

4.105 Further confirmation was provided by $US-Malt\ Beverages$. In that case the Panel found that the measures at issue were inconsistent with Article III:4, even though they were not being enforced in practice. The Panel reasoned that although the responsible authorities

may not be currently using its police powers ... the measures continue to be mandatory legislation which may influence the decisions of economic operators. ¹⁹⁰

Furthermore, by now it was well established that a measure might be inconsistent with Article III:4 even if it was not legally enforceable. Thus, for example, in Canada – Autos, the Panel held that certain value added requirements contained in the "Letters of Undertaking" submitted to the Canadian Government by some car manufacturers were inconsistent with Article III:4, even though the Letters were not enforceable through sanctions in the case of non-compliance 191. If non-enforceable local content requirements were inconsistent with Article III:4, then enforceable ones must be a fortiori inconsistent with that provision, even if the authorities had discretion not to enforce them. Moreover, India's interpretation would lead to an absurd result. On India's interpretation, the European Communities could not challenge the MOUs unless and until India brought an action against a signatory before an Indian civil court for breach of contract. Yet the Indian authorities would not have to take any such action unless the signatories failed to comply with the MOUs. Thus, India's interpretation would have the paradoxical result that the MOUs could be subject to WTO dispute settlement if they were not complied with (i.e. if they are ineffective), but not if they are complied with by the signatories (i.e. if they were effective in discriminating against imported products). Ultimately, India's interpretation would lead Members to encourage their economic operators to disobey the domestic laws of other Members which they considered to be WTO inconsistent, as a prerequisite for being able to challenge those laws in the WTO. The European Communities was

-

 $^{^{188}}$ Report of the Appellate Body in $\it US-1916$ $\it Act,$ WT/DS136/AB/R, WT/DS162/AB/R, adopted on 26 September 2000, at para. 91.

¹⁸⁹ Panel Report on *US – Malt Beverages*, BISD 39S/206.

¹⁹⁰ *Ibid.*, para 5.60.

Panel Report on *Canada – Certain Measures Affecting the Automotive Industry*, (hereinafter "*Canada – Autos*"), WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.120 *et seq*.

convinced that no Member, including India, could regard this as a desirable development of WTO law.

D. CLAIMS UNDER GATT 1994 AND TRIMS

1. Article III:4

4.106 The **United States** claimed that the indigenization and trade balancing obligations were "regulations" or "requirements" that "affect" the sale, use or purchase of automobile parts and components but that accord "less favourable treatment" to foreign products than to "like" domestic products.

- "Regulations" or "requirements": First, the ordinary meaning of "regulation" included its use as a generic term for governmental measures that implement statutes and other domestic legal provisions. Public Notice No. 60 was such a measure. Second, Public Notice No. 60 fell within the ordinary meaning of the term "regulation" because it regulated the conduct of manufacturing firms (they must meet local content and trade balancing targets) and the conduct of the Indian import licensing authorities (they may issue licenses for CKD/SKD components and kits if those targets are met). Third, Public Notice No. 60 and the MOUs both imposed "requirements" on firms manufacturing cars in India. A firm that signed an MOU was required to achieve 50% local content during the first three years of the MOU, and 70% before the end of the fifth year. That firm was also required to export cars and auto components with an FOB value at least equal to the CIF value of their importations of CKD/SKD kits and components. Previous panels had considered The Canada – FIRA panel¹⁹² recognized that the term similar situations. "requirements" properly described legally enforceable undertakings given to the Government of Canada by individual companies. In this case, as in Canada – FIRA, the local content and trade balancing commitments in the MOUs were part of the conditions under which signatories can receive import licenses. India had confirmed that those conditions were meant to be enforced. The analysis was not affected by the fact that firms could in theory choose not to sign an MOU. Any firm that did so would forfeit the right to import CKD/SKD components and kits. Accepting and complying with the terms of an MOU were requirements that a firm must fulfill to take advantage of the opportunity to import SKD/CKD kits and components. The term "requirement" encompassed such preconditions to obtaining an advantage from the government. Previous panels addressing this issue had reached the same conclusion. For instance, the EEC - Parts and Components Panel recognized that requirements that an enterprise voluntarily accepts to gain government-provided advantages are nonetheless "requirements".
- "Affecting": Ever since the Italian Discrimination Against Imported Agricultural Machinery report (hereinafter "Italy Agricultural Machinery"), panels had recognized that the term "affecting" in Article III:4 had a broad meaning, which extended not only to laws and regulations which directly govern the conditions of sale or purchase but also to any laws or regulations which might adversely modify the conditions of competition between the domestic and imported products on the internal market. The measures at issue here "affect" the sale, etc., of domestic and imported goods, because they required manufacturers in India to increase their purchases and use of Indian-made automotive parts and components at the expense of like foreign

¹⁹² Panel Report on *Canada –FIRA*, adopted on 7 February 1984, BISD 30S/140.

parts and components. A company manufacturing cars in India could not import SKD/CKD kits and components unless it used a decreasing percentage of imported goods in its production – no more than 50% in the first three years, no more than 30% by the fifth year. Moreover, the more a car manufacturer bought or used imported kits and components, the more it had an obligation to allocate a portion of its output to export, regardless of its business preferences. Both of these requirements made it less attractive for a manufacturer to purchase or use imported parts and components.

- "Like": Imported automotive parts and components were "like" automotive parts and components made in India. Domestic and imported components to be used in manufacturing a particular car share the same physical characteristics and commercial uses. Thus, while a clutch and a shock absorber differed from each other, a domestic clutch and an imported clutch to be incorporated in a particular car were "like" each other, just as a domestic shock absorber and an imported shock absorber to be incorporated into a particular car were "like" each other.
- "Less Favourable Treatment": The indigenization obligation and the trade balancing obligation accorded a competitive advantage to Indian goods and thus accord "less favorable treatment" to imported automotive components and kits as described in paras 4.107 and 4.110 and 4.111.

(a) The indigenization requirement

Manufacturers could meet the indigenization obligation only by purchasing and using Indian parts and components instead of imported ones. By the end of the third year, their production of finished vehicles could consist of 50%, 60%, or more Indian-origin parts; but that same production could include no more than 50% foreign-origin parts and components. By the end of the fifth year, Indian parts and components could comprise 70%, 80%, or more of a firm's production; but now that same firm's production could include no more than 30% imported parts and components. The indigenization requirement gave Indian goods a clear advantage: their imported counterparts will simply not be permitted to compete for the percentage of production that India expressly reserved for domestic goods. Such local content requirements had been systematically condemned. For instance, the Canada – FIRA panel recognized that undertakings that excluded the possibility of purchasing available imported products clearly treated such imported products less favorably than domestic products, and that such requirements were therefore not consistent with Article III:4. 193 The indigenization requirement did not disadvantage just kits and components imported in SKD/CKD form. The requirement discriminates against any imported part or component, from any non-Indian origin, that an Indian firm might use in manufacturing a motor vehicle. Any such foreign component, no matter how small or how separate from other parts, counts against a manufacturer trying to reach the mandated local content percentage. The indigenization requirement therefore accords imported parts and components treatment less favorable than it accords to like domestic items.

4.108 The **European Communities** claimed that the indigenization" requirements were inconsistent with GATT Article III:4 in that they provided less favourable treatment to imported parts and materials than to like domestic goods with respect to their internal use in the production of passenger cars. Article III:4 of GATT provided in relevant part that

"The products of the territory of any contracting party imported into the territory of any other contracting party shall be accorded treatment no less favourable than that

 $^{^{193}}$ Canada – FIRA, para. 5.8. To the same effect is the Panel Report on EEC - Parts and Components, para. 5.21.

accorded to like products of national origin in respect of laws, regulations and requirements affecting their internal sale, offering for sale, purchase, transportation, distribution or use [...]."

4.109 Accordingly, in order to rule on this claim the Panel would need to address the following issues:

- first, whether the measures at issue were "laws, regulations or requirements";
- second, whether domestic products were "like" imported products;
- third, whether the measures "affect" the "internal use" of the products concerned; and
- fourth, whether the measures afforded "less favourable treatment" to imported products than to domestic products.
- The measures in dispute were "requirements": By now it was firmly established that Government action need not be compulsory in order to qualify as a "requirement" for the purposes of GATT Article III:4. In Canada FIRA the Panel held that the legally enforceable undertakings given by some foreign investors to the Canadian Government constituted "requirements", even though the submission of such undertakings was voluntary. Similarly, in EEC Parts and Components the Panel concluded that the conditions accepted by a firm in order to obtain an "advantage" granted by the EC authorities also constituted "requirements". Subsequent cases

"The Panel first examined whether the purchase undertakings are to be considered 'laws, regulations or requirements' within the meaning of Article III:4. As both parties had agreed that the Foreign Investment Review Act and the Foreign Investment Review Regulations –whilst providing for the possibility of written undertakings- did not make their submission obligatory, the question remained whether the undertakings given in individual cases are to be considered 'requirements' within the meaning of Article III:4. In this respect the Panel noted that Section 9(c) of the Act refers to 'any written undertakings [...] relating to the proposed or actual investment given by any party thereto conditional upon the allowance of the investment' and that section 21 of the Act states that 'where a person who has given a written undertaking...fails or refuses to comply with the undertaking' a court order may be made 'directing that person to comply with the undertaking'. The Panel further noted that written purchase undertakings –leaving aside the manner in which they may have been arrived at (voluntary submission, encouragement, negotiation, etc)- once they were accepted, became part of the conditions under which the investment proposals were approved, in which case compliance could be legally enforced. The Panel therefore found that the word 'requirements' as used in Article III:4 could be considered a proper description of existing undertakings."

Panel Report on *Canada – FIRA*, adopted on 7 February 1984, BISD 30S/140, at para. 5.4.

Panel Report on *EEC – Parts and Components*, BISD 37S/132, adopted on 16 May 1990, at para 5.21:

".... Article III:4 refers to 'all laws, regulations or requirements affecting the internal sale, offering for sale, purchase, transportation, distribution or use'. The Panel considered that the comprehensive coverage of 'all laws, regulations or requirements *affecting* (emphasis added) the internal sale, etc.' of imported products suggests that not only requirements which an enterprise is legally bound to carry out, such as those examined by the *Canada – FIRA* Panel (BISD 30S/140), but also those which an enterprise voluntarily accepts in order to obtain an advantage from the government constitute 'requirements' within the meaning of that provision."

The same interpretation underlies the Report on *Italian Discrimination against Imported Agricultural Machinery*, (hereinafter "*Italy – Agricultural Machinery*"), adopted on 23 October 1958, BISD 7S/60, where the

¹⁹⁴ The Panel reasoned as follows:

had clarified that the "advantage" might consist of a benefit in respect of a border measure. More recently, in *Canada – Autos* the Panel went even further by holding that the "letters of undertaking" submitted by certain firms at the request of the Canadian Government were "requirements", even though they were neither legally enforceable nor a condition to obtain an advantage. Public Notice No. 60 did not impose upon the joint-ventures a legal obligation to conclude an MOU with the Indian Government. Nonetheless, the conclusion of an MOU was a necessary condition for obtaining an advantage: the grant of import licenses for CKD and SKD kits. Moreover, once it was signed, the MOU is binding upon the signatory and legally enforceable under the FTDR. The European Communities submitted that, in light of the precedents cited above, either of those two features in and by itself was more than sufficient to reach the conclusion that Public Notice No. 60 and the MOUs concluded thereunder constituted "requirements" within the meaning of Article III:4.

• Domestic parts and materials are "like" the imported goods: The distinctions operated by the "indigenization" requirements were based exclusively on the origin of the products: whereas parts and materials of Indian origin contributed to satisfy the "indigenization" percentage, imported parts and materials did not. Clearly, however, the mere fact of having Indian origin was not, as such, apt to confer upon parts and materials any characteristic, property or quality which made them, by definition, "unlike" any imported good. 198

Panel concluded that an Italian law providing especial credit terms to farmers for the purchase of agricultural machinery conditional upon the purchase by the farmers of Italian machinery was contrary to Article III:4 of GATT

¹⁹⁶ See e.g., Panel Report on *EC – Bananas III*, WT/DS27/R/USA, adopted 25 September 1997, as modified by the Appellate Body Report, at paras. 7.179 and 7.180, DSR 1997:II, 943, where the Panel found that a requirement to purchase domestic bananas in order to obtain the right to import bananas at a lower duty rate under a tariff quota was a requirement "affecting" the internal purchase of a product within the meaning of GATT Article III:4. On appeal, this finding was upheld by the Appellate Body, WT/DS27/AB/R, at paras. 208-211.

Similarly, in *Indonesia – Autos*, the Panel concluded that the granting of a duty exemption conditional upon compliance with a local content requirement was inconsistent with GATT Article III:4 and violated Article 2 of the TRIMs Agreement. In reaching this conclusion, the Panel rejected Indonesia's attempted defence that the measure was a "border" measure not covered by GATT Article III:4:

"We do not consider that the matter before us in connection with Indonesia's obligations under the TRIMs Agreement is the customs duty relief as such but rather the internal regulations, i.e. the provisions on purchase and use of domestic products, compliance with which is necessary in order to obtain an advantage, which advantage here is the customs duty relief. The lower duty rates are clearly 'advantages' in the meaning of the chapeau of the Illustrative List to the TRIMs Agreement and as such, we find that the Indonesian measures fall within the scope of the Item 1 of the Illustrative List of the TRIMs."

Panel Report on Indonesia-Autos, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 23 July 1998, at para. 14.89 (DSR 1998:VI, 2201).

197 The Panel cited the following circumstances in order to conclude that the "letters of undertaking" were "requirements":

"(i) in making the undertakings contained in the Letters, the companies acted at the request of the Government of Canada; (ii) the anticipated conclusion of the Auto Pact was a key factor in the decision of the companies to submit these undertakings; (iii) the companies accepted responsibility $vis-\dot{a}-vis$ the Government of Canada with respect to the implementation of the undertakings contained in the Letters, which they described as 'obligations' and in respect of which they undertook to provide information to the Government of Canada and indicated their understanding that the Government of Canada would conduct yearly audits; and (iv) at least until model year 1996, the Government of Canada gathered information on an annual basis concerning the implementation of the conditions provided for in the Letters."

Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, para. 10.122. ¹⁹⁸ In *Indonesia – Autos*, the Panel noted, para 14.113, that an • The measures "affect" the internal "use" of the products concerned: The term "affect" has been given a broad scope of application. According to the Panel Report in Italy – Agricultural Machinery,

"The selection of the word 'affecting' would imply [...] that the drafters of the Article intended to cover in [Article III:4] not only the laws and regulations which directly governed the conditions of sale and purchase but also any laws or regulations which might adversely modify the conditions of competition between the domestic and imported products on the internal market

In the present case, the "indigenization" requirements "affect" directly the "internal use" of parts and materials because, in order to satisfy the "indigenization" percentage, the signatories must incorporate into the vehicles that they manufactured a certain amount of domestic parts and materials. 199

- Imported goods are afforded "less favourable treatment": Finally, it was self-evident that, by requiring the use of a minimum amount of domestic parts and materials, the MOUs preclude the signatories from using an equivalent amount of imported parts and materials and, therefore, afford "less favourable treatment" to imported products.
- Precedent and the TRIMs Agreement confirmed that the "indigenization" requirements were inconsistent with Article III:4: Local content requirements constituted a clear-cut violation of the national treatment requirements imposed by GATT Article III:4, which had already been condemned by GATT/WTO panels in several occasions.
 - in *Canada FIRA*, the panel found that the undertakings given to the Canadian Government by some foreign investors to, *inter alia*, purchase goods of Canadian origin in specified amounts or proportions was contrary to Article III:4²⁰⁰:
 - similarly, in *EEC- Parts and Components*, the panel concluded that, by making the suspension of anti-circumvention proceedings conditional upon an undertaking to limit the use of Japanese parts and materials, the European Communities acted inconsistently with Article III:4²⁰¹;

[&]quot;... origin-based distinction in respect of internal taxes suffices in itself to violate Article III:2, without the need to demonstrate the existence of actually traded like products".

See also the Panel Report on *Canada – Autos*, para. 10.174, and the Panel Report on *Argentina – Measures Affecting the Export of Bovine Hides and the Import of Finished Leather*, (hereinafter "*Argentina – Hides and Leather*") WT/DS155/R, adopted on 16 February 2001, para.11.169.

Panel Report on *Italy – Agricultural Machinery*, para. 12. The Appellate Body confirmed that interpretation of the term "affect" in *EC – Bananas III*, WT/DS27/AB/R, para 220 (DSR 1997:II, 591). See also the Report of the Appellate Body in *Korea – Various Measures on Beef*, WT/DS161/AB/R, WT/DS169/AB/R, adopted on 10 January 2001, paras. 130 *et seq*.

²⁰⁰ Panel Report on *Canada – FIRA*, paras. 5.4-5.12, BISD 30S/140...

²⁰¹ Panel Report on *EEC – Parts and Components*, paras. 5.19-5.21, BISD 37S/132.

- in *Indonesia Autos*, the panel found that the grant of certain tax and import duty benefits to a so-called "National Car" manufacturer conditional upon meeting a certain local content percentage was in violation of Article III:4²⁰²;
- finally, in *Canada Autos*, the panel held that the grant of a customs duty exemption to certain manufacturers of motor vehicles subject to compliance with certain "Canadian Value Added" requirements was inconsistent with Article III:4²⁰³.

The TRIMs Agreement had confirmed beyond doubt that local content requirements were inconsistent with Article III:4 of GATT. Item 1(a) of the Illustrative List of TRIMs, included among the TRIMs that were inconsistent with Article III:4 those which require

"a) the [...] use by an enterprise of products of domestic origin [...], whether specified in terms of [...] value of products, or in terms of a proportion of [...] value of its local production"

The "indigenization" requirements at issue in this dispute fell squarely within the terms of Item 1(a) of the Illustrative List and were, therefore, inconsistent with GATT Article III:4.

(b) The trade balancing requirement

4.110 In the view of the **United States**, the trade balancing requirement also modified conditions of competition in favour of Indian goods, but its discrimination affected those kits and components that were imported in SKD or CKD form. All imported SKD/CKD kits/components carried with them an obligation to export from India goods (components or finished vehicles) in a value equal to the value of the imported goods. Moreover, this obligation attached not only to SKD/CKD kits/components that the manufacturer itself imported, but also to those SKD/CKD kits or components that the manufacturer purchased within India but were imported by someone else. Domestic goods, on the other hand, were free of this obligation.

4.111 Consequently, if a manufacturer built a car with components imported in SKD/CKD form, the manufacturer must either export that car or, if it wished to sell that car on the Indian market, the manufacturer must export some other finished vehicle or auto components whose value equalled that of the SKD/CKD importation. If, instead, the manufacturer built that same car without using components imported in SKD/CKD form, the manufacturer was free to sell the car in whatever market it chose; and, if the manufacturer chose the domestic market, it did not have any export obligation to discharge. Moreover, because the requirement attached to goods purchased within India, a number of other forms of discrimination arose as well. For example, a manufacturer seeking to meet its export obligation through the export of finished vehicles would purchase domestic components rather than imported kits -- because purchasing an imported kit would only further increase the manufacturer's export obligation, while purchasing the like domestic kit would not. Similarly, a manufacturer that needed to reduce an excess inventory of SKD/CKD kits or components would, all other things being equal, find a readier market for domestically produced kits than imported

 $^{^{202}}$ Panel Report on Indonesia-Autos, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 23 July 1998, para. 14.83 $et\ seq.$ (DSR 1998:VI, 2201).

²⁰³ Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.58 *et seq*.

ones, because the imported ones carried with them the additional burden of the trade balancing requirement and the domestic ones did not. In short, the trade balancing requirement added a burden to imported goods -- an interference with the distribution and other commercial choices of their user or purchaser -- that did not apply to like domestic goods. That additional burden was a disincentive to the use of imported SKD/CKD kits and components, and the trade balancing requirement therefore accorded less favorable treatment to them. A similar situation had been considered in *EEC - Measures on Animal Feed Proteins* (hereinafter "*EEC - Animal Feed Proteins*"). Both disputes involved a measure that imposed a burden on those who used imported goods but not on those who used like domestic goods (the obligation to buy milk powder from intervention agencies in that case, the obligation to export finished vehicles or auto parts in this one). As the *EEC - Animal Feed Proteins* recognized, such a measure was inconsistent with Article III:4.

4.112 According to the **European Communities**, the signatories of the MOUs seemed to be required to "neutralise" not only the value of the imports which they made themselves, but also the value of any imported kits and components which they purchased within India from local vendors. To that extent, the "trade balancing" requirements were inconsistent not only with Article XI:1, but also with Article III:4 in that they provided less favourable treatment to imported kits and components with respect to their internal purchase than to like domestic products. (see *supra* para. 4.38)

- The measures were "requirements": As demonstrated above (paragraph 4.109), the Public Notice No. 60 and the MOUs were "requirements" within the meaning of Article III:4.
- Imported products were "like" domestic product: The distinction operated by the trade balancing requirements was based exclusively on the origin of the products: on the one hand, imported kits and components purchased within India must be neutralised with exports; on the other hand, domestic kits and components need not be neutralised.
- The "trade balancing" requirements "affect" the internal "purchase" of imported goods: The trade balancing requirements "affect" directly the internal purchase of imported kits and components, because they subject the purchase of those products within India to the condition that their value must be neutralised with exports.
- The "trade balancing" requirements afforded "less favourable treatment" to imported products: The trade balancing requirements afford "less favourable treatment" to imported products because the internal purchase of domestic products is not subject to any similar requirement. All other circumstances being equal, this creates an incentive for the signatories of MOUs to purchase local inputs, thereby modifying adversely the competitive opportunities of imported products in relation to like domestic products.
- The TRIMs Agreement confirmed that the "trade balancing" requirements were inconsistent with Article III:4: Item 1(b) of the Illustrative List of Prohibited TRIMs confirmed that the "trade balancing" requirements, to the extent that they apply to products imported by other parties, was inconsistent with GATT Article III:4.

²⁰⁴ See also response to Question 10 of the Panel.

- 4.113 **India** argued that, in the absence of import licensing, Public Notice No. 60 and the trade balancing provisions of the MOUs were consistent with Article III:4. Public Notice No. 60 merely set out criteria for the automatic grant of import licenses for SKD/CKD kits and did not apply in the absence of any requirement to obtain such a license. The European Communities and the United States claimed that the trade balancing provisions set out in the MOUs were inconsistent with Article XI:1 of the GATT because they subjected the right to import SKD/CKD kits and components to the fulfilment of an export obligation. They further claimed that the trade balancing requirements were inconsistent with Article III:4 of the GATT because they accorded the signatory an advantage, namely the relief from an export obligation, on the condition that it purchased or used domestic components. India planned to demonstrate that the trade balancing provisions set out in the MOUs required the signatory to offset through exports only the foreign exchange expenditures for imports that had been subject to import licensing. In the absence of any requirement to obtain an import license for SKD/CKD kits and components, no export obligations accrued. The MOUs therefore did not limit in this situation the signatory's right to import SKD/CKD kits and components. Since no export obligations accrued when products not subject to licensing were imported, the signatory could also no longer acquire the advantage of reduced export obligations by purchasing or using domestic components.
- 4.114 The EC's claim that Public Notice No. 60 and the trade balancing provisions of the MOUs as such, that is independently of the discretionary licensing system for CKD/SKD kits, violated Articles III and XI of the GATT had no merit. The European Communities failed to explain how Public Notice No. 60, as such, restricted imports or favoured the purchase of domestic products over imported products. Public Notice No. 60 merely set out how the import restrictions on cars imposed until 1 April 2001 under the Export-Import Policy were to be administered. This was clear from paragraph 2 of this Notice which read in relevant part: " . . .import of components for motor vehicles in CKD/SKD form, which is restricted for import under the current Export-Import Policy, shall be allowed for importation against a license and such a license will be issued . . .on the basis of an MOU". Public Notice No. 60 was no longer operational because the licensing scheme it was to administer no longer existed. It was therefore not clear to India how the European Communities could claim that the Public Notice as such could violate WTO law in the absence of import restrictions on cars.
- 4.115 The United States noted that India had not actually attempted to defend the indigenization requirement. India had just said that car manufacturers might already be in compliance with it. If so, why was India continuing to insist that it needed to be able to enforce the requirement in its courts? Would such compliance mean anything other than that manufacturers were reluctant to test India's willingness to enforce the requirement? India also had not said what would happen to a manufacturer that slipped below the required percentage in the future. In any case, compliance by manufacturers with a WTO-inconsistent measure did not transform that measure into one that was compatible with India's WTO obligations. As India was well aware, GATT Articles III and XI protected conditions of competition, not trade flows; no demonstration of trade effects was necessary to establish a breach of those obligations. The United States added that the only response that India had advanced did not have merit. India claimed that export requirements as such were not prohibited by the GATT; India cited the Canada - FIRA panel report in support of its position. India overlooked, however, that the U.S. claims were directed at a discriminatory export obligation. Not all like products were subject to the export requirement; the requirement applied only to imported goods. An internal regulation that was otherwise consistent with the WTO Agreement became inconsistent if, like the trade balancing requirement in this case, it applied to imported products but not to like domestic ones. There was nothing to the contrary in the Canada – FIRA report; that dispute did not involve export requirements applied exclusively to imported goods.

2. Article XI:1

4.116 The **United States** claimed that the indigenization and trade balancing requirements were "prohibitions or restrictions ... on ... importation", as those terms are used in Article XI:1. Panels had recognized that the scope of Article XI:1 was comprehensive: it extended not only to outright prohibitions but also to discretionary or non-automatic import licensing and to conditional suspensions of an import prohibition. Moreover, the scope of the word "restriction" itself was broad, as seen in its ordinary meaning, which was "a limitation on action, a limiting condition or regulation".

(i) The indigenization requirement

4.117 The United States claimed that the indigenization requirement was a straightforward restriction or prohibition on imports. Failure to meet the 50% local content quota by the third year, or the 70% quota by the fifth year, led to the denial of the right to import parts and components in SKD/CKD form. Public Notice No. 60 was clear about this, and India had confirmed that the MOU was itself designed to limit imports of SKD/CKD kits/components when a firm failed to meet the indigenization requirement. Paragraph 3(iii) of Public Notice No. 60, and paragraph III, clause (iv) of the MOU, required that MOU signatories achieve specified levels of indigenization in order to be able to import SKD/CKD kits/components. Moreover, it was *India* that had said that the MOUs as such were intended to limit the importation of SKD/CKD kits/components when a firm failed to meet the indigenization requirement: "As all the companies have achieved the desired level of indigenization during the last two years (since issuance of Public Notice No. 60) the need to invoke MOUs to impose *limitation on them* has not arisen. ²⁰⁵ The indigenization requirement did not merely restrict the amount of kits that a manufacturer might import into India, it actually prohibited such imports outright. Any manufacturer that failed to achieve the local content targets in the MOU was forbidden from importing CKD/SKD kits. Thus, to the extent that the MOUs themselves prevented signatories from importing SKD/CKD kits if they did not meet the indigenization targets, the indigenization requirement was as such inconsistent with Article XI:1 of the GATT 1994 and with Article 2.1 of the TRIMs Agreement.

(ii) The trade balancing requirement

4.118 The trade balancing requirement was likewise a continuing restriction on imports of SKD/CKD kits and components. Starting in the fourth year the MOU imposed a quantitative limitation on imports -- and the quantity was correlated to the degree of compliance with the trade balancing requirement. The Government of India had also confirmed that denial of an import license was effectively mandatory if the trade balancing obligation was not met. Paragraph 3(iv) of Public Notice No. 60 provides that "From 4th year [i.e. of the MOU] onwards the value of import of CKD/SKD may be regulated with reference to the extent of export obligation fulfilled in the previous years as per the MOU." Paragraph III, clause (vi) of the MOU contained an identical provision. India had confirmed, that the MOUs were binding and enforceable. In addition, the first sentence of Paragraph III, clause (vi) provided that "the party [i.e. the MOU signatory] shall achieve a broad trade balancing of foreign exchange *over the entire period of the MOU* in terms of balancing between the actual CIF value of imports of CKD/SKD kits/components and the FOB value of exports of cars and

 205 India's responses to Questions by the United States, 13 July 2000, response to Question 5, (underlining in original, other emphasis added).

²⁰⁶ See also India's *Replies to Questions Posed by Japan*, G/TRIMS/W/15, circulated 30 October 1998, response to Question 24; Exhibit US-5: "CKD/SKD kits imports would be allowed with reference to the extent of export obligation fulfilled in the previous year. ... There is hardly any discretion involved in determining the extent of import of CKD/SKD kits except by way of considering any genuine problems the company may have faced in achieving the export levels."

auto components over the said period." The first sentence of paragraph 3(iv) of Public Notice No. 60 was an essentially identical provision. These provisions clearly stated that the trade balancing obligation extended over the entire period of the MOU. As India had confirmed, however, the MOUs and their requirements continued to remain in force even beyond the elimination of import licensing for SKD/CKD kits and components. Furthermore, the trade balancing requirement restricted imports because it placed a maximum limit on the value of an MOU signatory's imports that was equal to the value of the signatory's exports (which, pursuant to paragraph III, clause (vi), the MOU signatory is required to specify). In practical terms, there were limits to the amount of exports which a car manufacturer might be able or willing to make (whether related to its manufacturing capacity or to the demand for its products outside India). Thus, by limiting the amount of a manufacturer's imports to that of its exports, the trade balancing requirement restricted the amount of imports. India asserted means of restricting importations by MOU signatories other than just license denials. manufacturer's failure to comply with an MOU obligation could lead to loss of import privileges or to confiscation of the goods concerned pursuant to various provisions of the FTDR Act and the rules made thereunder. It appeared that after 1 April 2001, these additional provisions would be the instruments through which India carried out Public Notice No. 60 and the MOUs (and thus prevented SKD/CKD kits/components from being brought into India to compete with domestic parts and components). For all of these reasons, the trade balancing requirement in Public Notice No. 60 and the MOUs imposed import restrictions as such, and was therefore inconsistent with Article XI:1 of the GATT 1994 and with Article 2.1 of the TRIMs Agreement.

4.119 The **European Communities** also argued that the "trade balancing" requirements, stipulated in the MOUs, were inconsistent with GATT Article XI:1 in that they restricted imports of passenger cars, and of components therefor, by the signatories of the MOUs. The European Communities recalled, nevertheless, that the 1997 Agreement between India and the European Communities (as supplemented by the 1999 Agreement between India and the United States) allowed India to maintain import restrictions on passenger cars, and on chassis and bodies therefor (but not on imports of parts and components therefor) until 1 April 2001. Therefore, the European Communities made this claim only in so far as: the MOUs required the "trade balancing" of imports of "components" other than chassis and bodies; and the MOUs would remain binding and enforceable after 1 April 2001, both with respect to passenger cars and components therefor. Article XI:1 read as follows in pertinent part:

"No prohibitions or restrictions other than duties, taxes or other charges, whether made effective through quotas, import or export licenses or other measures, shall be instituted or maintained by any [Member] on the importation of any product of the territory of any other [Member] or on the exportation or sale for export of any product of the territory of any other [Member]"

As noted by the Panel Report in *Japan - Trade in Semi-conductors* (hereinafter "*Japan - Semi-conductors*"), the wording of Article XI:1 was "comprehensive": it applied to "all measures instituted or maintained by a [Member] prohibiting or restricting the importation [...] other than measures that take the form of duties, taxes or charges"²⁰⁷. The test for compliance with Article XI:1 was thus three-fold: first, was the action concerned a Government "measure" second, was the measure different from a "duty, tax or other charge"? and third, did the measure "restrict" imports? In the present case, the answer to the above three questions was clearly in the affirmative. The "trade balancing" requirements were "measures". The term "measure" had been given a broad definition, indeed even broader than that of "requirements" in Article III:4. ²⁰⁸ In *Japan - Semi-conductors* the

²⁰⁷ Japan – Semi-conductors, adopted on 4 May 1988, BISD 35S/116, para. 104.

²⁰⁸ See the Panel Report on *Japan – Photographic Film and Paper*, (hereinafter "*Japan – Film*") WT/DS44/R, adopted on 29 April 1998, para. 10.51 (DSR 1998:IV, 1179), where the Panel stated that

Panel found that non-mandatory Government action in the form of "administrative guidance" was a "measure" subject to Article XI:1 because it created sufficient incentives or disincentives for private parties to act. ²⁰⁹ In *Japan –Film* the Panel went even further by holding that, in certain circumstances, even purely hortatory wording in a statement of policy could qualify as a "measure". ²¹⁰ As shown above, Public Notice No. 60 and the MOUs were "requirements" in the sense of Article III:4. For the same reasons, and *a fortiori*, they were also "measures" in the sense of Article XI:1. The "trade balancing" requirements were not "duties, taxes or other charges" The "trade balancing" requirements did not involve any payment or transfer of money by or on the account of the signatories of MOUs and, therefore, could not be characterised as "duties, taxes or other charges". The "trade balancing" requirements "restricted" imports. The "trade balancing" requirements "restricted" imports because they placed a maximum limit on the value of the imports which the signatories were authorised to make equal to the value of their exports. In practice, there were limits to the amount of exports which a signatory might be able or willing to make (related both to its manufacturing capacity in India and to the demand for its products in foreign markets). Thus, by limiting the amount of a signatory's imports to that of its exports, the trade balancing requirements "restricted" the amount of imports. ²¹¹

4.120 **India** asserted that, with respect to manufacturing and equity requirements, the law of the WTO did not regulate foreign direct investments as such²¹² and therefore did not prevent India from imposing on foreign investors the obligation to manufacture (rather than merely assemble) automobiles and to make an equity investment of a specified amount. Article XI:1 of the GATT and Article 2 of the TRIMs Agreement merely prohibited the imposition of such requirements as a condition for the grant of an import license. However, after 1 April 2001, the requirement to manufacture (rather than merely assemble) automobiles, and to secure an equity investment of a specified amount, would have to be performed completely independently of any right to import. It would apply equally to all signatories of MOUs, whether they imported SKD/CKD kits or not. With respect to export requirements, the GATT panel which examined Canada's Foreign Investment Review Act in 1984 correctly found that "there is no provision in the General Agreement which forbids requirements to sell goods in foreign markets in preference to the domestic market.²¹⁴ There was also no such requirement in the TRIMs Agreement. Article XI of the GATT and Article 2 of the TRIMs Agreement merely prohibited export requirements that were imposed as a condition for the grant of an import license and export requirements that vary with the level of local purchases.²¹⁵ Nowhere in WTO law were export requirements prohibited as such. As from 1 April 2001, the signatories of the MOUs would continue to be required to discharge the export obligations corresponding to the imports made by them before that date. They would thus be required to discharge the export obligations they have already assumed but will no longer incur any new export

[&]quot;Given that the scope of the term requirement would seem to be narrower than that of measure, the broad reading given to the word requirement by the (Canada - FIRA) $(EEC - Parts\ and\ Components)$ panel supports an even broader reading of the word measure in Article XXIII: 1 b)."

²⁰⁹ Panel Report on *Japan – Semi-conductors*, BISD 35S/116, at pp. 154-155. See also the Panel Report on *Japan – Restrictions on Imports of Certain Agricultural Products*, (hereinafter "*Japan – Agricultural Products I*") adopted on 22 March 1988, BISD 35S/163, 242.

²¹⁰ Panel Report on *Japan – Film*, WT/DS44/R, para. 10.49 (DSR 1998:IV, 1179).

²¹¹ See also US response to Question 82 of the Panel.

²¹² See the finding of the GATT Panel on *Canada – FIRA*, which noted that "the General Agreement does not prevent Canada from exercising its sovereign right to regulate foreign direct investments" (BISD 30S/157).

²¹³ See paragraph 2(a) of the Annex to the TRIMS Agreement.

²¹⁴ Canada – FIRA (BISD 30S/140).

²¹⁵ See paragraph 2(a) of the Annex to the TRIMS Agreement. Article 3.1 (a) of the Agreement on Subsidies and Countervailing Duties prohibits export requirements that are tied to subsidies. However, this prohibition does not apply to India (see Article 27.2 (a) and Annex VII of the Agreement on Subsidies and Countervailing Duties).

obligations as a result of the further importation of SKD/CKD kits. From 1 April 2001, the right to import SKD/CKD kits would consequently no longer in any way depend on, or vary with, the level of exports achieved. For this reason, the export requirements would no longer create any incentive to purchase local products. As a result, the export requirements would have to be performed completely independently of any imports or local purchases and would consequently be entirely consistent with both the GATT and the TRIMs Agreement.

- 4.121 **India** argued that, in the absence of import licensing, Public Notice No. 60 and the trade balancing provisions of the MOUs were consistent with Article XI:1. As from 1 April 2001, India would consistently with its undertakings towards the European Communities and the United States no longer make the importation of SKD/CKD kits subject to the signing of an MOU. The right to import SKD/CKD kits would therefore no longer be subject to any conditions that could be deemed to constitute a restriction on the importation of a product within the meaning of Article XI of the GATT. The import liberalisation would thus benefit all car manufacturers, including those who had signed an MOU. Subsequently, India informed the Panel that, since 1 April 2001, the import restrictions for SKD/CKD kits and car components had been abolished and no licenses referring to the trade balancing requirement had been issued. Since then, the signatories of the MOUs had therefore no longer incurred any new export obligations as a result of the importation of SKD/CKD kits or components.
- 4.122 According to India, the European Communities and the United States claimed that the provisions of Public Notice No. 60, as such, restricted imports and favoured the purchase or use of domestic products over imported products and were therefore inconsistent with Article III:4 and XI:1 of the GATT. However, Public Notice No. 60 merely set out how the import restrictions on cars imposed until 1 April 2001 under the Export-Import Policy were to be administered. This was clear from paragraph 2 of this Notice which read in relevant part:
 - ... import of components for motor vehicles in CKD/SKD form, which is restricted for import under the current Export-Import Policy, shall be allowed for importation against a license and such a license will be issued . . .on the basis of an MOU.
- 4.123 After 1 April 2001, Public Notice No. 60 was no longer applicable because the import restrictions it was to administer no longer existed. In the absence of an import restriction on cars, this Notice could not be implemented and served no purpose. It was therefore not clear to India how the European Communities and the United States could claim that the Public Notice as such was capable of violating WTO law. India had requested both the European Communities and the United States to indicate which provision of the Public Notice No. 60 restricted imports or favoured domestic products in the absence of any licensing requirement for CKD/SKD kits and hence any requirement to sign an MOU. Neither the European Communities nor the United States had given a precise answer to this question. India requested the Panel to find that, in the absence of any requirement to obtain an import license for SKD/CKD kits and components, Public Notice No. 60 did not restrict imports inconsistently with Article XI:1 of the GATT. ²¹⁶ The United States claimed that Public Notice No. 60 "still remains in effect" because no public notice rescinding it has been issued.²¹⁷ India submitted a copy of Notification No. 2 (RE-2001)/1997-2002 dated 31 March 2001 issued by the Ministry of Commerce, which showed that India had removed the restrictions on imports of cars in the form of CKD/SKD kits and automobile components that were covered by Public Notice No. 60. In the absence of these restrictions, Public Notice No. 60 no longer applied. A new public notice rescinding Public Notice No. 60 was consequently not required to ensure the consistency of Public Notice No. 60 with Articles III:4 and XI:1 of the GATT.

²¹⁷ US response to Questions 2(b) of the Panel.

²¹⁶ See also India's response to Question 103 of the Panel.

4.124 The **United States** identified that India denied that Article XI:1 applied to these requirements, but India had misinterpreted that provision. India also denied that Public Notice No. 60 and the MOUs themselves imposed restrictions on imports; however, not only could that assertion not be reconciled with the actual language of the measures, but it did not appear even to be consistent with India's own statements. The United States had explained that the *trade balancing* requirement restricted imports because it limited the value of an MOU signatory's imports to the value of the signatory's exports (which, pursuant to paragraph III, clause (vi), the MOU signatory was required to specify at the time of signing). There were obviously limitations on the amount of exports which a car manufacturer might be able or willing to make. Thus, by limiting the amount of a manufacturer's imports to that of its exports, the trade balancing requirement itself restricted imports. India appeared not to accept that such a requirement was inconsistent with Article XI:1 of the GATT: India asserted that "Article XI of the GATT and Article 2 of the TRIMs Agreement merely prohibit export requirements that are imposed as a condition for the grant of an import license and export requirements that vary with the level of local purchases". This assertion was simply incorrect, as paragraph 2 of the TRIMs Agreement Illustrative List showed:

TRIMs that are inconsistent with the obligation of general elimination of quantitative restrictions provided for in paragraph 1 of Article XI of GATT 1994 include those which are ... enforceable under domestic law ... or compliance with which is necessary to obtain an advantage, and which restrict the importation by an enterprise of products used in ... its local production ... to an amount related to the ... value of local production that it exports

It was clear that the MOUs restricted importation to an amount related to the value of locally produced goods that a manufacturer exports, and it was clear that the MOU 's were "enforceable"; India had confirmed both points. The trade balancing requirement thus fell within the scope of this paragraph, and consequently it was "inconsistent with the obligation of general elimination of quantitative restrictions provided for in paragraph 1 of Article XI of GATT 1994". The Panel should explicitly reject India's assertion and should make an explicit finding that the trade balancing requirement was inconsistent with India's obligations under Article XI:1 of the GATT.

4.125 **India** contended that the MOU provisions on trade balancing did not apply in the absence of any requirement to obtain an import license. The United States claimed that the trade balancing provisions set out in the MOUs restricted imports inconsistently with Article XI:1 of the GATT for the following reasons:

The trade balancing requirement restricts imports because it places a maximum limit on the value of an MOU signatory's imports that is equal to the value of the signatory's exports (which, pursuant to paragraph III, clause (vi), the MOU signatory is required to specify). In practical terms, there are limits to the amount of exports which a car manufacturer may be able or willing to make (whether related to its manufacturing capacity or to the demand for its products outside India). Thus, by limiting the amount of a manufacturer's imports to that of its exports, the trade balancing requirement restricts the amount of imports.²¹⁹

²¹⁸ Replies by India to Questions Posed by Japan, G/TRIMS/W/15, circulated 30 October 1998, response to Question 24; Exhibit US-5: "CKD/SKD kits imports would be allowed with reference to the extent of export obligation fulfilled in the previous year." As discussed above, the MOUs were also "necessary to obtain an advantage".

²¹⁹ US response to Question 11 of the Panel.

The European Communities' and the United States' assertions were based on the incorrect assumption that the trade balancing requirement applied to all imports, irrespective of whether an import license was required or not. In fact, however, the trade balancing provisions did not apply, and had in practice never been applied, in respect of imports for which no license was required. (see IV A). Paragraph III (iv) of the MOUs declared that the MOU signatory shall discharge the export obligation corresponding to the imports made before the import licensing requirement ceased to apply. The signatories of the MOUs were therefore bound to discharge the export obligations they had already assumed before the abolition of the import restrictions on SKD/CKD kits on 1 April 2001. However, this residual export obligation would have to be performed completely independently of the level of imports or local purchases. Nowhere in WTO law were export requirements as such prohibited. The requirement to discharge the accrued export obligations would consequently be entirely consistent with both the GATT and the TRIMs Agreement. India noted that, so far, neither the European Communities nor the United States had presented any arguments challenging this position.

4.126 The issue was whether the trade balancing provisions in the MOUs gave rise to export obligations in respect of imports for which no import license was required. Contrary to the assertions of the United States, this issue was totally unrelated to the question of the duration of the MOU, the point in time at which the MOU signatory reached the 70 percent indigenization target and the enforceability of the MOUs. It was, of course, correct when the United States claimed that the MOU provisions "impose a trade balancing obligation on imports made over the full duration of the MOU" and require the trade balancing of the foreign exchange obligation incurred on imports made up to the date on which it achieves the 70 per cent indigenization target and that the MOUs are enforceable. However, it did not follow from these facts that the trade balancing requirement in the MOUs applied to imports for which no license was needed and that this requirement therefore continued to apply even after the import licensing regime was eliminated. The evidence presented by India demonstrated that, in respect of freely importable products, the MOU signatories had never assumed, and still did not assume, any export obligations that might limit their right to import or create an incentive for them to purchase domestic components. The provisions of the MOU regulating the point in time at which the signatory moved out of the ambit of the MOU and the enforceability of the MOUs did not change this fact. The conclusion that "the signatories are required to balance all the imports made by them until they achieve a 70% indigenization level" simply does not follow from the fact that the indigenization requirement ceased when the 70 per cent target had been reached. The fact that the trade balancing requirement must be performed only until a 70 per cent indigenization did not change the fact the trade balancing requirement only arose in respect of imports for which a license was **required.** It followed that, in the absence of a discretionary licensing system, the right to import SKD/CKD kits and components by MOU signatories did not depend on, or vary with, the level of exports or of purchases of domestic products. It was therefore plainly incorrect to claim that the trade balancing requirements in the MOUs as such restricted imports²²⁰ and favoured the purchase or use of

Article XI:1. The GATT made a clear distinction between measures discriminating against imports that were applied at the point and time of importation, which were covered by Articles II and XI, and those that were applied to imported products, which were covered by Article III. Article XI:1 applied to "restrictions . . . on the importation of any product". The dictionary meaning of "importation" was "the act of importing". The scope of application of Article XI:1 was therefore limited to acts affecting the process of entering products into the custom territory. By contrast, Article III:4 of the GATT applied to requirements applied to products that had already been imported into the customs territory. The exceptions applicable to Article XI:1 were wider than those applicable to Article III:4 (see for instance Article XI:2 of the GATT and Article 5 of the Agreement on Safeguards). A failure to maintain the distinction between measures applied at the border and internal measures would therefore broaden the scope of the exceptions to Article XI:1 in a manner not contemplated by the drafters of the GATT. Obviously, the provisions of the MOUs on trade balancing did not affect the process of entering products into India's customs territory and could also for this reason not constitute restrictions on importation within the meaning of Article XI:1. India invited the European Communities and the United States

domestic products over imported products. In respect of freely importable products, the MOU signatory did not assume any export obligations that might limit its right to import or create an incentive to purchase domestic components.

4.127 The United States understood that India had confirmed that the MOUs were measures independent of Public Notice No. 60. India also asserted, however, that any import restrictions imposed by the MOUs were eliminated when India eliminated its import licensing regime. The Panel should reject this additional argument as well. The first difficulty with India's argument came from the first sentence of Paragraph III, clause (vi) of the MOU. That sentence provided that "the party [i.e. the MOU signatory] shall achieve a broad trade balancing of foreign exchange over the entire period of the MOU in terms of balancing between the actual CIF value of imports of CKD/SKD kits/components and the FOB value of exports of cars and auto components over the said period." This provision imposed a trade balancing obligation on imports made over the full duration of the MOU; the trade balancing requirement in the MOU was thus independent of the elimination of the import licensing regime. 221 India had effectively confirmed this point in its answer to the Panel's questions. India said that "under paragraph 3(iv) of Public Notice 60 and paragraph III(iv) of the MOUs, the MOU signatory goes 'outside the ambit of the MOU' and will not require any import licenses once it achieves a 70% indigenization level. However, the MOU signatory will be required to achieve trade balancing of the foreign exchange obligation incurred on imports made up to that date". 222 Consequently, under the terms of the MOU, the export balancing requirement continued to attach to all imports made until the 70% indigenization level was reached - whenever that date arrived.²²³ While India said elsewhere that no further export obligations would accrue with respect to imports made after 1 April 2001, that statement seemed impossible to reconcile with India's own description of the meaning of the MOUs.

4.128 The second difficulty with India's assertion that the MOUs imposed no separate import restriction arose from India's acknowledgment that they remained enforceable under Section 11 of the FT(DR) Act.²²⁴ India said first that "failure to meet an export obligation after 1 April 2001 would not be an import or export prohibited by Section 11(1) of the FT(DR) Act." India added, however, that "the companies may also be liable to monetary penalties under Section 11 of the FTDR Act" and that after 1 April 2001, "MOU signatories would be treated as licensees in respect of the licences that they have utilized up to 31 March 2001." Section 11 of the FT(DR) Act contained only one provision imposing monetary penalties: Section 11(2), which provided that

... where any person makes or abets or attempts to make any export or import in contravention of any provision of this Act or any rules or orders made thereunder or the export and import policy, he shall be liable to a penalty not exceeding one thousand rupees or five times the value of the good in respect of which any contravention is made or attempted to be made, whichever is more.

to reconsider their argument that the MOUs as such violated Article XI:1 in the light of its broader systemic implications.

²²¹ In fact, the trade balancing requirement imposed restrictions above and beyond those imposed by India's import licensing regime even when that regime was in force, because the requirement limited the amount of licenses that an MOU signatory could use.

²²² India's response to Question 50 of the Panel (emphasis added).

²²³ Furthermore, according to India, at least some manufacturers have not yet reached that level; India's response to Question 33 of the Panel.

²²⁴ Question 52(d) of the Panel.

²²⁵ In the same Response India also confirmed once again that after 1 April 2001, "MOUs potentially can be enforced as contracts through the domestic courts".

It appeared, therefore, that the imposition of monetary penalties under Section 11 – as contemplated by India for companies that fail to abide by the provisions of the MOU – depended on the making of some export or import in contravention of Indian law. India had also said, however, that failure to meet an export obligation was not an export or import prohibited by Section 11(1). It was not at all clear how these two statements could be harmonized. One possible reconciliation of the provisions of Section 11 and India's statements was that while failing to export the mandated amount would not be a prohibited export or import (which was logical, since a *failure* to export hardly seemed like a *prohibited* export, let alone an import of any kind), a car manufacturer's *imports* in future years in excess of the amount of export obligation discharged would be a prohibited import to which a monetary penalty under Section 11(2) applied. For this reason as well, the export balancing requirement thus continued to impose an import restriction even after 1 April, 2001, and was inconsistent with GATT Article XI:1.

4.129 The **European Communities** noted that India contended that, in the absence of the import licenses on passenger cars, the MOUs could not restrict imports inconsistently with Article XI:1. Specifically, India argued that

... the MOUs as such do not in any way limit the right to import SKD/CKD kits or any other product. The non-observation of the terms of the MOUs also no longer leads to the automatic denial of an import license. Article XI of the GATT deals with 'restrictions . . . on . . . importation', that is with measures applied at the border in connection with the entry of products into the customs territory. The MOUs as such do not maintain or institute any such measure and can therefore not possibly violate Article XI. ²²⁶

That interpretation of Article XI was unduly narrow and found no support in the wording of that provision or in the GATT/WTO jurisprudence. As noted by the panel in *Japan – Semi–Conductors*²²⁷, the wording of Article XI:1 was "comprehensive". It applied to "all measures instituted or maintained by a [Member] prohibiting or restricting the importation, exportation or sale for export other than measures that take the form of duties, taxes or charges"²²⁸. Import restrictions were most commonly enforced at the border, since that was the easiest and most effective way of restricting imports. However, the scope of Article XI:1 was by no means limited to such restrictions. An agreement between the Government and the main importers of a product whereby the latter agreed to limit the value of their imports gave rise to an obvious restriction on the importation of that product. That restriction was clearly within the scope of Article XI:1, regardless of whether the agreement was enforced by means of measures taken at the border which prevented the actual entry of the goods or through other, less direct means, such as imposing fines on the signatories who import in excess of the agreed quantity or suing them for breach of contract. India's narrow interpretation of the scope of Article XI:1 was refuted by *Japan – Semi–Conductors*. In that case, the Panel concluded that:

The requests not to export semi-conductors ... addressed to Japanese producers and exporters of semi-conductors, combined with the statutory requirement for the exporters to submit information on export prices and the systematic monitoring of company and product-specific costs and export prices by the Government, backed up with the use of supply and demand forecasts to impress on manufacturers the need to align their production to appropriate levels, constituted a coherent system of

²²⁸ *Ibid*. at para. 104.

²²⁶ India's response to Question 33 of the Panel, at p. 3, para. 1 [footnotes omitted].

²²⁷ Panel Report on *Japan – Semi-conductors*, adopted on 4 May 1988, BISD 35S/116.

restricting the sale for export of monitored semi-conductors ... inconsistent with Article XI:1.²²⁹

None of the above measures were applied by Japan "at the border". Moreover, unlike the MOUs, the measures condemned in *Japan – Semi–conductors* were not even legally binding and enforceable. At any rate, India's argument that the MOUs as such could not be inconsistent with Article XI:1 because they were not border measures would not dispose of the EC's claims to the effect that both the balancing requirement and the indigenization requirements were inconsistent with Article III:4 of the GATT. More specifically, and contrary to India's assertions, import balancing requirements were not inconsistent with Article XI:1 only if they were enforced through the denial of import licenses. By its own terms, Article XI:1 prohibited import restrictions, whether made effective through import licenses or through any "other measures". As illustrated by *Canada – FIRA*, the obligations contained in legally binding contracts between the Government and individual firms could constitute a "requirement" prohibited by Article III:4. Likewise, a binding contract between the Government and a firm whereby the latter agreed to limit its imports fell clearly within the scope of Article XI:1. (Indeed, the European Communities recalled that for a Government action to be contrary to Article XI:1, it did not even have to be legally enforceable²³¹).

4.130 **India** argued that since Article XI of the GATT dealt with 'restrictions ... on ... importation', that is with measures applied at the border in connection with the entry of products into the customs territory, the MOUs as such did not maintain or institute any such measure and could therefore not possibly violate Article XI.²³² The European Communities had also failed to explain how the MOUs as such restricted imports. A failure to sign or observe the terms of an MOU could lead to a denial of an import license for SKD/CKD kits only as long as an import license for such kits was required. However, the MOUs as such did not in any way limit the right to import SKD/CKD kits or any other product. The non-observation of the terms of the MOUs also no longer led to the automatic denial of an import license.²³³ Article XI of the GATT dealt with "restrictions ...on ... importation", that is with measures applied at the border in connection with the entry of products into the customs territory. The MOUs as such did not maintain or institute any such measure and could therefore not possibly violate Article XI. India considered that the EC's argument that India's:

... interpretation of Article XI is unduly narrow and finds no support in the wording of that provision or in the GATT/WTO jurisprudence. As noted by the panel in *Japan – Semi-Conductors*²³⁴, the wording of Article XI:1 is "comprehensive". It applies to "all measures instituted or maintained by a [Member] prohibiting or restricting the importation, exportation or sale for export other than measures that take the form of duties, taxes or charges"²³⁵.

Import restrictions are most commonly enforced at the border, since that is the easiest and most effective way of restricting imports. However, the scope of Article XI:1 is by no means limited to such restrictions.

²³⁰ Panel Report on *Canada – FIRA*, adopted on 7 February 1984, BISD 30S/140, at para. 5.4.

²³⁵ *Ibid.*, para. 104.

²²⁹ *Ibid*. at para. 132.

²³¹ See the Panel Report on *Japan –Semi-conductors*, BISD 35S/116, pp. 154-155; and the Panel Report on *Japan – Agricultural Products I*, BISD 35S/163. The same is true with respect to Article III:4. See the Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, at para. 10.122 and the Panel Report on *Japan – Film*, WT/DS44/R, adopted on 22 April 1998, para. 10.49 (DSR 1998:IV, 1179).

²³² India's response to Question 33, p. 3.

²³³ See the response to Question 52(b).

²³⁴ Panel Report on *Japan – Semi-conductors*, adopted on 4 May 1988, BISD 35S/116.

was based on a misunderstanding of GATT law. The GATT made a clear distinction between measures that were applied in connection with or at the point or time of importation ("border measures"), which were covered mainly by Articles II and XI, and those that were applied internally to imported products, which were covered mainly by Article III. Article XI:1 applied to "restrictions ... on the importation of any product". The dictionary meaning of "importation" was "the act of importing". The scope of application of Article XI:1 was therefore limited to acts affecting the process of entering products into the custom territory. By contrast, Article III:4 of the GATT applied to requirements that were applied internally. Obviously, the provisions of the MOUs on trade balancing were not applied in connection with or at the point or time of importation and could for that reason alone not constitute restrictions on importation within the meaning of Article XI:1. The European Communities overlooked that, as far as measures affecting exports were concerned, the GATT did not distinguish between internal measures and border measures. Article XI:1 prohibited both restrictions "on the exportation or the sale for export". As to measures affecting exports, the drafters of the GATT had to extend the scope of Article XI to internal measures because the national treatment provisions of Article III did not cover measures favouring sales in the domestic market over sales for export. The panel report on Japan - Semi-conductors dealt with measures affecting exports. That panel did not have to distinguish between internal measures and border measures affecting exports because both were treated equally under Article XI. In fact, it made no ruling on this point at all. It merely found that the GATT jurisprudence on minimum import prices could be extended to minimum export prices. Its remark on the comprehensive nature of Article XI:1 provided the reason for this finding. 236 The panel report on Japan - Semi-conductors therefore did not lend any support to the EC's position. India considered that the European Communities might wish to reconsider the broader implications of its argumentation. The exceptions applicable to Article XI:1 were wider than those applicable to Article III:4 (see for instance, Articles XI:2, XII and XVIII:B of the GATT and Article 5 of the Agreement on Safeguards). A failure to maintain the distinction between measures applied at the border and internal measures would therefore broaden the scope of the exceptions to Article XI:1 in a manner not contemplated by the drafters of the GATT.

- 4.131 The **United States** replied that India still appeared to deny that Article XI:1 applied to the export balancing requirement. India advanced a very limited interpretation of Article XI:1; India said that "Article XI of the GATT and Article 2 of the TRIMs Agreement merely prohibited export requirements that were imposed as a condition for the grant of an import license and export requirements that vary with the level of local purchases. Now, India said that the scope of application of XI:1 was limited to measures "affecting the process of entering products into the customs territory". Neither of these formulations was correct. It was the text of the WTO Agreement that defined the scope of Article XI:1. And, the TRIMs Agreement Illustrative List made clear that any requirement which restricted importation to an amount related to the value of locally produced goods that a manufacturer exports such as the one in paragraph III, clause (vi) of the MOUs was inconsistent with GATT Article XI:1. The United States considered that the Panel should reject India's proposed reformulations of Article XI:1 and should find that the trade balancing requirement was inconsistent with India's obligations under that Article.
- (a) Are the measures justified by Article XVIII:B of the GATT and Articles 3 and 4 of the TRIMs Agreement?
- 4.132 **India** argued that it maintained the restrictive import licensing regime under which the Notice and the MOU scheme operated for balance-of-payments reasons. This licensing regime, although inconsistent with the general prohibition of quantitative restrictions set out in Article XI of the GATT, was justified under Article XVIII:B of the GATT, according to which the developing country Members of the WTO could impose import restrictions to safeguard their external financial position

²³⁶ BISD 35S/153, paras 104-105.

and to ensure a level of reserves adequate for the implementation of their programme of economic development. As stipulated in Article XVIII:B, India had notified its restrictions to the WTO.²³⁷ India had not yet disinvoked Article XVIII:B of the GATT. India noted that Article 3 of the Agreement on Trade-Related Investment Measures ("TRIMs Agreement") provided that all exceptions under the GATT should apply to the provisions of the TRIMs Agreement. Furthermore, under Article 4 of the TRIMs Agreement, in the case of developing country Members such as India, Article XVIII:B of the GATT constituted not only an exception from Article XI of the GATT but also from the obligations set out in Article 2 of the TRIMs Agreement.

- 4.133 India stated that the restrictions on the importation of SKD/CKD kits were imposed in order to safeguard India's external financial position. 238 When it notified the BOPs Committee on 27 May 1997 that it was maintaining a system of non-automatic licensing for import of cars in the form of CKD/SKD kits, it had been entering into MOUs with manufacturers that wished to import CKD/SKD kits since 1995. Any differences between the discretionary licensing scheme for import of cars in the form of CKD/SKD kits applied under the MOUs entered into since 1995 and that applied after 1997 were not legally significant from the standpoint of Article XI of the GATT or of the claim that they were justified under Article XVIII:B. 239
- 4.134 Prior to the establishment of the Panel, India had stated that "the main purpose of the MOU Policy is the management of balance-of-payments even while following an open door policy toward foreign investment which should not lead to a net outflow of foreign exchange" ²⁴⁰ The **United States** said it was not certain what India meant by that statement. For example, it was not clear how opening the door towards foreign investment might be related to a net outflow of foreign exchange. To the extent that India was concerned about capital outflows, the United States recalled that India had notified the WTO of a separate dividend balancing requirement, which required foreign-owned investments in several industries (including portions of the motor vehicle sector) to earn through exports the foreign exchange necessary to repatriate earnings to their foreign parents. India's assertion was also inconsistent with statements made when Public Notice No. 60 was adopted; at that time, Indian officials described a different purpose for the MOUs: "The objective of the new policy was to encourage local production of auto-components and thus, bring in modern technology and develop this key segment, explain ministry officials."241
- 4.135 The European Communities added that this statement had been made by India more than three months after the European Communities requested consultations with respect to the MOUs. The obvious purpose of the MOU policy was to promote the establishment and consolidation of a local automotive industry in India.
- 4.136 For the United States, India did not have a defense under Article XVIII:B for the measures at issue in this dispute. First, India had not complied with the procedural prerequisites for justifying these measures under the balance-of-payments provisions of the GATT Second, on 12 December 1997, when Public Notice No. 60 was adopted, India no longer had a balance-of-payments problem justifying the use of the balance-of-payments provisions of the GATT. The International Monetary Fund (IMF) had reached that conclusion several months earlier, in January of 1997²⁴², and the

²³⁷ WT/BOP/N/24.

²³⁸ Question 63 of the Panel.

²³⁹ Question 64 of the Panel.

²⁴⁰ Questions from Japan, Committee on TRIMs.

[&]quot;Car Makers Have to Sign New MOUs in Accordance with New Automobile Policy", Business Standard, 11 December 1997.

Panel Report on India - Quantitative Restrictions, WT/DS90/R, adopted on 22 September 1999, para. 3.225.

India - Quantitative Restrictions panel reached the conclusion that India had no valid balance-of-payments justification as of November 18, 1997 - less than a month before Public Notice No. 60 was issued. Third, India had in any case not met its burden of proof with respect to this issue.

4.137 In elaborating on the first of these three points, the United States considered that nowhere did India actually say that the indigenization and trade balancing requirements were justified by the balance-of-payments provisions. Instead, India had simply said that if the Panel attempted to examine the US complaint, the Panel would be obliged to embark on the complicated task of re-examining India's balance-of-payments situation. The Panel should not let itself be distracted by India's suggestion, because the balance-of-payments provisions of the WTO Agreements provided no defence to the violations at issue in this case. In the first place, India was prohibited from raising those provisions as a defence to the complaint because it had never notified the measures at issue to the BOPs Committee and had never asserted a balance-of-payments justification for them. Article XVIII:12(a) of the GATT 1994 and paragraph 6 of the Uruguay Round Understanding on the Balance-of-Payments Provisions of the GATT 1994 made clear that such notification was required. However, India had never brought the indigenization or trade balancing requirements before the Committee. India could not both say that it would have been impractical for the indigenization and trade balancing requirements to be notified to and examined by the BOPs Committee, and yet also say that the *India – Quantitative Restrictions* panel ruled on these details without either knowing about them or examining them.²⁴⁴

4.138 In the second place, even if these measures had been taken for balance-of-payments reasons (and India's failure to notify them reinforced the fact that they were not), these measures dated from late 1997. Pursuant to paragraphs 2 and 3 of the Understanding, even if India had wanted to take new measures for balance-of-payments purposes, India would have been required to give preference to price-based measures. Public Notice No. 60, however, imposed quantitative restrictions and local content requirements, not tariff surcharges. Third, the balance-of-payments provisions in any case could not justify a violation of the national treatment obligations of the GATT. Article XVIII: 9 made clear that the balance-of-payments provisions could be used to "control the general level of imports" - not to deny national treatment to foreign goods. Once a foreign good had been imported, the foreign exchange for that import had been expended; there was no balance-of-payments justification for permitting discrimination against that imported good. For all of these reasons, the Panel should reject India's balance-of-payments arguments.

4.139 **India** responded to the US argument that first, the measures at issue were not new to the extent that they represented only the conditions that India applied in the context of non-automatic licensing of imports of cars in the form of CKD/SKD kits. Second, the conditions in Public Notice No. 60 and the MOUs based on it were basically the same as those contained in 1995 MOUs. Third, they did not involve either a raising of the general level of restrictions or a substantial intensification of the restrictions applied by India to imports of cars in the form of CKD/SKD kits; if anything, they represented a relaxation of the restrictions. Fourth, Members did not have to notify the details of the licensing regime they applied to each of the restrictions notified under Article XVIII: B to benefit from the legal cover provided by this provision. In practice, the notifications under GATT's balance-of-payments provisions did not contain such details and Members were not expected to consult on them in the BOPs Committee. It would in any case have been totally impracticable for India to notify these details for 2,700 items and for the BOPs Committee to examine them. Finally, as from 1 April 2001, India no longer applied any balance-of-payments measures and therefore no longer invoked the balance-of-payments provisions of the GATT. The points made by the United States, expressed before

²⁴³ *Ibid.*, paras. 5.160-5.161 and 5.236.

²⁴⁴ US response to Question 74 of the Panel.

1 April 2001, were therefore no longer relevant. 245 The Panel asked us to confirm whether India was relying on Article XVIII:B of the GATT as a defence. The answer was that if the United States and the European Communities insisted on re-opening the issue of the WTO-consistency of measures which the earlier panel has already ruled upon, India also reserved its right to take up the defence that the measures were justified under Article XVIII:B of the GATT 1994. The United States argued that India could not invoke Article XVIII:B because India did not notify Public Notice No. 60 under Article XVIII:B of the GATT. However, Public Notice No. 60, as such, did not establish an import restriction for SKD/CKD kits that would have required notification under Article XVIII:B. The relevant import restriction was imposed under the Export-Import Policy. Public Notice No. 60 merely set out how this import restriction was to be administered. This was clear from paragraph 2 of this Notice which read in relevant part: " ...import of components for motor vehicle in CKD/SKD form, which is restricted for import under the current Export-Import Policy, shall be allowed for importation against a license and such a license will be issued ... on the basis of an MOU ..." Under Article XVIII:B, Members must notify the restrictions they are applying, not the import licensing schemes they use to administer those restrictions. India had explained again and again to its trading partners that Public Notice No. 60 implemented the import restrictions on SKD/CKD kits, which were notified under Article XVIII:B, and would cease to be operational once these restrictions were eliminated. It was simply not logical to tear the procedures for implementation of the import restriction on imports of CKD and SKD kits in the Public Notice out of the context of the import restriction and declare those implementing procedures in the Notice themselves to be restrictions that would have required notification under Article XVIII:B. For this reason, it was also incorrect to claim that the Public Notice No. 60 was a new measure in respect of which India was required to give preference to pricebased measures pursuant to paragraphs 2 and 3 of the Understanding on the balance-of-payments provisions of the GATT.

(i) Procedural requirements of the balance of payments provisions

The **United States** said that India had not met the procedural prerequisites for justifying these measures under the balance-of-payments provision of the GATT. GATT Article XVIII:B and the Understanding on the Balance-of-Payments Provisions of the GATT 1994 (the Understanding) established procedural prerequisites to any invocation of the balance-of-payments provisions of the GATT. 246 These prerequisites also applied to the TRIMs Agreement, Article 4 of which permitted developing country Members to deviate temporarily from the provisions of Article 2 to the extent and in such a manner as GATT Article XVIII and the Understanding permit These prerequisites included a requirement to consult with the BOPs Committee (the Committee) concerning any new measures immediately after applying them (or, where prior consultation was practicable, as it almost certainly was in this case, before doing so)²⁴⁷. In the view of the United States, if a Member changed an import licensing scheme that the Member justified for balance-of-payments reasons, the Member was required to notify the changes to the WTO. Significant changes must be notified within 30 days. These points flowed from paragraph 9 of the Understanding on the Balance-of-Payments Provisions of the GATT 1994 (the Understanding), which provides that "a Member shall notify to the General Council the introduction of or any changes in the application of restrictive import measures taken for balance-of-payments purposes, as well as any modifications in time-schedules for the removal of such measures as announced under paragraph 1. Significant changes shall be notified to the General Council prior to or not later than 30 days after their announcement." Paragraph 9 of the Understanding also obligated Members applying restrictions for balance-of-payments purposes to

²⁴⁵ Question 65 of the Panel.

²⁴⁶ Pursuant to GATT Article XVIII:9, a Member's right to maintain balance-of-payments measures is made subject to the provisions of Articles XVIII:10 to 12.

Article XVIII:12(a). Pursuant to paragraph 6 of the Understanding, consultations are to be held within four months after the institution or intensification of restrictions.

furnish to the WTO, on a yearly basis, a consolidated notification that included all changes in relevant laws, regulations, policy statements or public notices. India did not undertake such consultations; indeed, it did not even notify the measures challenged in this dispute to the Committee, despite the requirement that it do so within 30 days if they were significant or in India's annual notification if not.²⁴⁸ Because India had not complied with those requirements, it could not now assert a balance-of-payments justification for those measures, and for that reason as well the Panel should reject any such assertion.

- 4.141 A panel might consider but should reject an Article XVIII:B defense concerning measures that the defending Member had not notified to the BOPs Committee (the Committee). In the *India Quantitative Restrictions* dispute, the Appellate Body found that panels had the competence to review the justification of measures purportedly taken for balance-of-payments purposes, and therefore a panel was empowered to consider such a defense. It would rarely if ever be appropriate, however, for a panel to sustain such a defense if the defending Member had not complied with the procedural requirements of Article XVIII:B and the Understanding. In this case, Public Notice No. 60 had entered into effect over three years ago, and India's Article XVIII:B arguments should not be accepted.
- 4.142 The **European Communities** also considered that a new notification was required. Paragraph 9 of the Understanding provided in relevant part that
 - "A Member shall notify the General Council the introduction of or *any* changes in the application of restrictive measures taken for balance-of-payments purposes ... Significant changes shall be notified to the General Council prior to or no later than 30 days after their announcement".

Nor, in its view, could a Panel consider an Article XVIII:B defence concerning measures not specifically notified to the Committee. Paragraph 9 of Article XVIII:B authorised a Member to take BOP measures "subject to paragraphs 10 to 12". Paragraph 12 (a) of Article XVIII:B provided that any party applying new restrictions must consult with the Committee immediately. Paragraph 6 of the Understanding specified that consultations must take place within four months.²⁴⁹

- 4.143 **India** recalled that according to Article XVIII:12(a) of the GATT, a new consultation was required if the Member applied new restrictions or **raised** "the **general** level of its existing restrictions by a **substantial** intensification of the measures" applied under Article XVIII:B. India's notification covered restrictions applied in the form of non-automatic licensing on imports of cars, including cars imported as CKD/SKD kits. The changes in 1997 did not entail a substantial intensification of this measure. On the contrary, the changes in 1997 had the effect of relaxing the restriction on the importation of SKD/CKD kits. India was therefore not required to re-notify and hold new consultations on the changes introduced through Public Notice No. 60 in order to benefit from the legal cover of Article XVIII:B. It was furthermore neither required nor customary to notify under Article XVIII:B the details as to the administration of the restrictions. If the Panel were to create such a requirement, the consequences would be administratively unmanageable.
- (ii) Substantive Requirements of the balance of payments provisions
- 4.144 The **United States** argued that India did not meet the substantive requirements of the balance-of-payments provisions of the GATT. Even if India were able to overcome the legal objections described [in the foregoing paragraphs], India could not meet the substantive requirements of

²⁴⁹ Question 30 of the Panel.

²⁴⁸ Understanding, para. 9.

Article XVIII:B. The first substantive difficulty with India's assertion of a balance-of-payments justification for these measures arose from the fact that India adopted Public Notice No. 60 on 12 December 1997. Under GATT Article XVIII:9, a Member might institute balance-of-payments measures "provided that the import restrictions instituted, maintained or intensified shall not exceed those necessary: (a) to forestall the threat of, or to stop, a serious decline in its monetary reserves, or (b) in the case of a contracting party with inadequate monetary reserves, to achieve a reasonable rate of increase in its reserves."

4.145 However, the Panel in *India – Quantitative Restrictions* found that, as of 18 November 1997 (only 24 days before Public Notice No. 60 was issued) India's balance-of-payments situation met none of the requirements in Article XVIII:9. The Panel found that:

Overall, we are of the view that the quality and weight of evidence is strongly in favour of the proposition that India's reserves are not inadequate. In particular, this position is supported by the IMF, the Reserve Bank of India and three of the four methods suggested by India. Accordingly, we find that India's reserves were not inadequate as of 18 November 1997. ²⁵⁰

[...]

We find that as of the date of establishment of this Panel [18 November 1997], there was not a serious decline or a threat of a serious decline in India's monetary reserves, as those terms are used in Article XVIII:9(a).²⁵¹

[...]

We find that as of the date of establishment of this Panel, India's monetary reserves of US\$25.1 billion were not inadequate as that term is used in Article XVIII:9(b) and that India was therefore not entitled to implement balance-of-payments measures to achieve a reasonable rate of growth in its reserves.²⁵²

4.146 It strained all credulity to suggest that 24 days after 18 November 1997, India's foreign reserve situation had changed so dramatically as to justify new balance-of-payments measures. Certainly India had provided no facts to suggest that such a change occurred. And, as the United States explained in its answers to the Panel's questions, this Panel could and should be guided by those factual and legal findings of the *India – Quantitative Restrictions* panel.²⁵³ Even if – despite the foregoing points – these measures could permissibly have been introduced under GATT Article XVIII:B in December of 1997, India would still be required to explain the basis on which it was maintaining them on 15 May and 27 July 2000, the dates on which the United States requested this Panel and on which this Panel was established.²⁵⁴ Article XVIII:11 made clear that a Member

²⁵⁰ Panel Report on *India – Quantitative Restrictions*, para. 5.176. This finding is a predicate for the Panel's ultimate finding under Article XVIII:9(b).

²⁵¹ Panel Report on *India – Quantitative Restrictions*, para. 5.180.

²⁵² Panel Report on *India – Quantitative Restrictions*, para. 5.183.

²⁵³ US response to Panel Questions, paras. 48-50.

Throughout its submissions, India said that the relevant date for a balance-of-payments analysis was the date on which a panel request is submitted. The *India – Quantitative Restrictions* panel concluded instead that the relevant date was the date of panel establishment, and it made its findings as of that date (18 November 1997). Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, adopted 22 September 1999, paras. 5.160-5.161. In this dispute, it made no difference which date was chosen: as described below India's currency reserves on either date were more than adequate and not seriously declining, and consequently India lacked a balance-of-payments justification for its measures on both dates.

applying balance-of-payments measures "shall progressively relax any restrictions applied under this Section as conditions improve, *maintaining them only to the extent necessary* under the terms of paragraph 9 of this Article and *shall eliminate them when conditions no longer justify such maintenance*." In fact, however, India did not have balance-of-payments difficulties at all in the spring of 2000, and therefore, under Article XVIII:11 it lacked any basis for maintaining them.

4.147 Without prejudice to the United States' position that India had failed to advance evidence required to sustain its burden to put forward at least a *prima facie* defense, the United States provided materials from the Reserve Bank of India that demonstrated that India did not face balance-of-payments difficulties that would justify maintenance of the measures at issue in this dispute at the time when this Panel was requested and established. In its Annual Report for 1999-2000, which was published in August 2000, the Reserve Bank provided information about India's currency reserves over the preceding decade. In December of 1998, those reserves had risen to US\$30.1 billion (as compared to US\$25.1 billion in November of 1997, the date as of which the *India – Quantitative Restrictions* panel made its rulings). In December of 1999, the reserves had risen to US\$34.9 billion. According to the Reserve Bank's Weekly Statistical Supplement, on 12 May 2000, India's reserves stood at US\$37.6 billion. On 28 July 2000, India's reserves stood at US\$36.2 billion. In other words, between November of 1997 and the end of July 2000, India's reserves increased by more than 44%.

4.148 Furthermore, the Reserve Bank provided the following analysis of India's external position:

The movements in India's foreign exchange reserves, in recent years, have kept pace with the requirements on the trade as well as the capital accounts. As a matter of policy, foreign exchange reserves are kept at a level that is adequate to cover the liquidity needs in the event of both cyclical and unanticipated shocks. Particularly after the South-East Asian currency crises, there has been a growing opinion that the central banks need to hold reserves far in excess of the levels that were considered desirable going by the conventional indicators. The import cover of reserves improved to about 8.2 months as at end-March 2000 as against 6.5 months as at end-March 1997 while the ratio of short-term debt to reserves declined to 10.6 per cent as at end-March 2000 from 25.5 per cent as at end-March 1997. Even in relation to a broader measure of external liabilities, foreign exchange reserves provide adequate cover. For instance, short-term debt and cumulative portfolio investment inflows taken together were only 59.3 per cent of reserves as at end-March 2000. These ratios remain, by and large, unchanged even if unencumbered reserves (gross reserves net of forward liabilities) are taken into account, given the relatively small size of forward liabilities in the Indian context. The strength of the foreign exchange reserves has also been a positive factor in facilitating flow of portfolio investment by FIIs and in reducing the 'risk premia' on foreign borrowings and Global Depository Receipts (GDR) / American Depository Receipts (ADR) issued by the Indian corporates. It is, however, important to note that unanticipated domestic or external developments, including undue volatility in asset prices in equity/bond markets, can create disproportionate pressures in the foreign exchange market in emerging economies.²⁵⁵

The Reserve Bank's confidence in the strength of India's currency reserve position - as gauged not only by measures such as the total amount of reserves, but also by measures such as import cover, ratio of reserves to short-term debt, and ratio of reserves to short-term debt plus cumulative portfolio

²⁵⁵ Annual Report, para. 6.31. Exhibit US-27 includes the paragraphs of the Annual Report devoted to India's foreign exchange reserves, paras. 6.24-6.31.

investment inflows, among others - confirmed what the growth and absolute size of India's foreign currency reserves themselves make clear: that India was not facing inadequate reserves, seriously declining reserves or the threat of a serious decline in the spring of 2000.

- 4.149 Finally, there would be several other substantive difficulties with any Indian assertion of a balance-of-payments justification for these measures. First, Article XVIII:B authorized developing country Members only to "control the general level of imports". India had not explained, however, how an inconsistency with national treatment "controls the general level of imports." As the United States had noted, once an item has been imported, the foreign exchange for that import has been expended; there was no balance-of-payments justification for permitting discrimination against that imported good. Furthermore, India had also not explained how its introduction of these measures in late 1997 would be consistent with the requirement in paragraphs 2 and 3 of the Understanding to give preference to price-based measures. The indigenization and trade balancing requirements were not import surcharges, import deposit requirements or other measures with an impact on the price of imported goods. India, however, had not provided the justification for these measures that the Understanding would require, and India therefore could not maintain them as balance-of-payments measures.
- 4.150 In summary, India had not provided any argumentation that would overcome the serious legal difficulties with any assertion of a balance-of-payments justification for the indigenization and trade balancing requirements. Nor had India provided any evidence that -- even if those legal objections could be overcome its foreign currency reserve position met the requirements of Article XVIII:B either at the time those measures were introduced or at the time this Panel was requested and established. To the contrary, the evidence of India's own central bank proved the opposite. India's assertion of a balance-of-payments justification for these measures should be rejected.

(iii) The burden of proof

4.151 The **United States** went on to state that India had not met its burden of proof on the balance-of-payments defense; it could not surmount the procedural obstacles to any such defense; and India's external financial position had at all relevant times been so strong that it could not meet the substantive requirements for invoking Article XVIII:B. The Appellate Body had provided guidance on the allocation of burden of proof in several reports. Particularly relevant were its conclusions in the *US – Wool Shirts and Blouses* report:

In addressing this issue, we find it difficult, indeed, to see how any system of judicial settlement could work if it incorporated the proposition that the mere assertion of a claim might amount to proof. It is, thus, hardly surprising that various international tribunals, including the International Court of Justice, have generally and consistently accepted and applied the rule that the party who asserts a fact, whether the claimant or the respondent, is responsible for providing proof thereof. Also, it is a generally-accepted canon of evidence in civil law, common law and, in fact, most jurisdictions, that the burden of proof rests upon the party, whether complaining or defending, who asserts the affirmative of a particular claim or defence. If that party adduces evidence sufficient to raise a presumption that what is claimed is true, the

²⁵⁶ GATT Article XVIII:9 and Understanding, para. 4.

²⁵⁷ India did not respond to the Panel's invitation to address this point. India's response to Question 65 of the Panel.

burden then shifts to the other party, who will fail unless it adduces sufficient evidence to rebut the presumption. ²⁵⁸

In this case, India had provided no evidence whatsoever about its balance-of-payments situation. To the extent that India was asserting that its balance-of-payments position provided a legal justification for the measures at issue, it was responsible for providing proof of that assertion. It had done nothing of the sort. Having adduced no evidence of any kind on the point, India obviously had not adduced evidence sufficient to raise a presumption that it had a balance-of-payments problem justifying the measures in question. For that reason alone, any balance-of-payments defense must fail.

4.152 The **European Communities** pointed out that Article XVIII:B was an exception to other GATT provisions. Accordingly, it was for India to prove that the strict requirements of Article XVIII were met in this case. India had provided no evidence or argument whatsoever in support of its invocation of Article XVIII:B. The evidence provided by the European Communities showed that India's BOP situation had been even more buoyant when the European Communities requested the establishment of this Panel than in November 1997, the date considered by the panel in *India - Quantitative Restrictions*. In addition, India had not complied with the procedural requirements imposed by Article XVIII:B and by the Understanding. In view of that, the Panel should dismiss India's attempted defence outright. In particular, the European Communities considered that it would be totally unwarranted to request the advice of the IMF. To do so would serve only to reward India's delaying tactics. For these reasons, the Panel need not address the arguments outlined by India in paragraph 42 of its First Submission (see para. 4.45 above). These arguments had already been rejected by the Panel and the Appellate Body in *India – Quantitative Restrictions*. There was no reason for the Panel to depart from that precedent in this dispute.

4.153 **India** recalled that the Appellate Body Report on *US –Woven Wool Shirts and Blouses*, stated that:

the burden of proof rests upon the party, whether complaining or defending, who asserts the affirmative of a particular claim or defence. If that party adduces evidence sufficient to raise a presumption that what is claimed is true, the burden then shifts to the other party, who will fail unless it adduces sufficient evidence to rebut the presumption. ²⁵⁹

4.154 Furthermore, the Appellate Body, in its report on *European Communities – Measures concerning Meat and Meat Products (Hormones)* (hereinafter "EC – Hormones"), stated that:

The general rule in a dispute settlement proceeding requiring a complaining party to establish a *prima facie* case of inconsistency of a provision [...] before the burden of showing consistency with that provision is taken by the defending party is *not* avoided by simply describing the same provision as an exception. ²⁶⁰ (emphasis in original)

In the *India – Quantitative Restrictions* case, the Panel concluded from this jurisprudence:

This implies that the United States has to prove any of its claims in relation to the alleged violation of Article XI:1 and XVIII:11. Similarly, India has to support its

²⁵⁸ Report of the Appellate Body in *US - Wool Shirts and Blouses*, WT/DS33/AB/R, adopted on 23 May 1997, page 14 (footnotes omitted) (emphasis added) (DSR 1997:I, 323).

²⁵⁹WT/DS33/AB/R, p. 16.

²⁶⁰ WT/DS26/AB/R-WT/DS48/AB/R, para. 104 (DSR 1998:I, 135).

assertion that its measures are justified under Article XVIII:B. We also view the rules stated by the Appellate Body as requiring that the United States as the complainant cannot limit itself to stating its claim. It must present a *prima facie* case that the Indian balance-of-payments measures are not justified by reference to Articles XI:1 and XVIII:11 of GATT 1994.²⁶¹ Should the United States do so, India would have to respond in order to rebut the claim. ²⁶²

In India's view, it was thus for the United States and the European Communities to submit a *prima facie* case that the restrictions notified by India under Article XVIII:B were inconsistent with that provision. It was only after they had done so that India was required to present a rebuttal. The complainants would have had to show in particular that India's reserves were sufficient to: safeguard its external financial position, and ensure a level of reserves adequate for the implementation of its programme of economic development²⁶³ and that India therefore violated Article XVIII:B

4.155 India noted that, in fact, the panel in *India – Quantitative Restrictions* found (para. 5.119) that the United States had to make a *prima facie* case that India's import restrictions were not justified under Article XVIII:11. India recalled the conclusions of the Appellate Body on the burden of proof in the case of safeguard actions by the United States against imports of shirts and blouses from India:

India has argued that it is "customary GATT practice" that the party invoking a provision which is identified as an exception must offer proof that the conditions set out in that provision are met. We acknowledge that several GATT 1947 and WTO panels have required such proof of a party invoking a defence, such as those found in Article XX²⁶⁴ or Article XI:2(c)(i)²⁶⁵, to a claim of violation of a GATT obligation, such as those found in Articles I:1, II:1, III or XI:1. Articles XX and XI:(2)(c)(i) are limited exceptions from obligations under certain other provisions of the GATT 1994, not positive rules establishing obligations in themselves. They are in the nature of affirmative defences. It is only reasonable that the burden of establishing such a defence should rest on the party asserting it.²⁶⁶

The transitional safeguard mechanism provided in Article 6 of the ATC is a fundamental part of the rights and obligations of WTO Members concerning non-

²⁶³ See Article XVIII:9 of the GATT.

²⁶¹ Report of the Appellate Body in *Australia – Salmon*, adopted on 6 November 1998, WT/DS18/AB/R, paras. 257-259.

²⁶² WT/DS90/R, para. 5.119.

²⁶⁴ Canada – FIRA, adopted on 7 February 1984, BISD 30S/140, para. 5.20; United States – Section 337 of the Tariff Act of 1930, adopted 7 November 1989, BISD 36S/345, para. 5.27; US – Malt Beverages, adopted on 19 June 1992, BISD 39S/206, paras. 5.43 and 5.52; and Panel Report, United States – Standards for Reformulated and Conventional Gasoline, as modified by the Appellate Body Report, adopted on 20 May 1996, WT/DS2/9, para. 6.20.

²⁶⁵ Japan – Agricultural Products I, adopted on 22 March 1988, BISD 35S/163, para. 5.1.3.7; EEC –

²⁶⁵ Japan – Agricultural Products I, adopted on 22 March 1988, BISD 35S/163, para. 5.1.3.7; EEC – Restrictions on Imports of Dessert Apples, Complaint by Chile, (hereinafter "EEC – Apples (Chile)"), adopted on 22 June 1989, BISD 36S/93, para. 12:3; and Canada - Import Restrictions on Ice Cream and Yoghurt, adopted on 5 December 1989, BISD 36S/68, para. 59.

Furthermore, there were a few cases that were similar in that the defending party invoked, as a defence, certain provisions and the panel explicitly required the defending party to demonstrate the applicability of the provision it was asserting. See, for example, *United States – Customs User Fee*, adopted on 2 February 1988, BISD 35S/245, para. 98, concerning Article II:2 of the GATT 1947; *Canada – Import, Distribution and Sale of Certain Alcoholic Drinks by Provincial Marketing Agencies*, adopted on 22 March 1988, BISD 35S/37, para 4.34, concerning Article XXIV:12 of the GATT 1947; and *US – Malt Beverages*, adopted on 19 June 1992, BISD 39S/206, para. 5.44, concerning the Protocol of Provisional Application.

integrated textile and clothing products covered by the *ATC* during the transitional period. Consequently, a party claiming a violation of a provision of the *WTO Agreement* by another Member must assert and prove its claim. In this case, India claimed a violation by the United States of Article 6 of the *ATC*. We agree with the Panel that it, therefore, was up to India to put forward evidence and legal argument sufficient to demonstrate that the transitional safeguard action by the United States was inconsistent with the obligations assumed by the United States under Articles 2 and 6 of the *ATC*. India did so in this case. And, with India having done so, the onus then shifted to the United States to bring forward evidence and argument to disprove the claim. This, the United States was not able to do and, therefore, the Panel found that the transitional safeguard action by the United States "violated the provisions of Articles 2 and 6 of the ATC".²⁶⁷

In our view, the Panel did not err on this issue in this case.

The Appellate Body found that Article 6 of the ATC established not merely a limited exception but positive rules and that India therefore had to bear the burden of proving that the United States applied Article 6 inconsistently. What was true for the safeguard provisions of Article 6 of the ATC must also be true for balance-of-payments provisions of Article XVIII:B of the GATT. Just as Article 6 of the ATC was a fundamental part of the rights and obligations of WTO Members under the ATC, Article XVIII:B was a fundamental part of the rights and obligations of developing countries under the GATT. There could not be one distribution of the burden of proof for a safeguard provision that had been used extensively by the United States and another for a safeguard provision applicable to developing countries. It was consequently up to the European Communities and the United States to present a prima facie case of violation of Article XVIII:B and, once they had done so, for India to rebut this case. The Appellate Body ruled that "a prima facie case is one which, in the absence of effective refutation by the defending party, requires a panel, as a matter of law, to rule in favour of the complaining party presenting the *prima facie* case". ²⁶⁸ The scant and anecdotal information provided so far by the complainants would not permit the Panel to rule as a matter of law that India's reserves were adequate for the implementation of its programme of economic development. In fact, they had not even attempted to assess India's reserves in the light of its development programme. As to India's external financial position, the European Communities and the United States had merely cited statements that the Reserve Bank of India made without any reference to the requirements of Article XVIII:B. As to the relationship between India's reserves and India's programme of economic development, the complainants submitted no evidence at all. The evidence submitted by the complainants so far could not possibly be the basis of a ruling that India was violating Article XVIII:B. According to the consistent GATT and WTO practice, ²⁶⁹ the Panel would have to examine India's balance-of-payments situation on 15 May 2000 and 12 October 2000, the dates on which the complainants submitted their panel requests. India's balance-of-payments situation at the time of the adoption of Public Notice No. 60 was totally irrelevant. The findings of the panel on India - Quantitative Restrictions did not provide any evidence in respect of India's balance-ofpayments situation on 15 May 2000 and 12 October 2000 and could therefore not possibly guide the present Panel.

4.156 With respect to the role of the IMF, India wished to note that the position of the European Communities and the United States was inconsistent with the position that the United States took in

²⁶⁷ WT/DS33/R, para. 8.1.

²⁶⁸ Report of the Appellate Body in *EC – Hormones*, WT/DS26/AB/R, WT/DS48/AB/R, adopted on 13 February 1998, para. 104 (DSR 1998:I, 135).

²⁶⁹ See Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, adopted on 22 September 1999, as upheld by the Appellate Body Report, paras. 5.160-5.161.

the proceeding of the panel on *India – Quantitative Restrictions*. The United States then asserted that Article XV:2 expressly allocated to the IMF the authority for making certain specific determinations within the scope of its particular expertise. Allocation of that authority to the IMF had been the intent of the drafters of Article XV:2, and that intent was carried out in the practice of the CONTRACTING PARTIES. Since Article XV:2 required the WTO to consult the IMF, in all cases where the WTO is called on to consider matters concerning monetary reserves, balance of payments or foreign exchange arrangements, the Panel must also do so . . . Although Article XV:2 did not mention panels, per se, an interpretation of WTO would include panels.²⁷⁰ The position of the United States had been supported by the jurisprudence of the Appellate Body. Although the Appellate Body had found in Argentina – Measures Affecting Imports of Footwear, Textiles, Apparel and Other Items (hereinafter "Argentina - Textiles and Apparel") that the dispute did not involve "problems concerning monetary reserves, balance of payments or foreign exchange arrangements", it did find with respect to panel proceedings that Article XV:2 "requires the WTO to consult with the IMF when dealing with 'problems concerning monetary reserves, balance of payments or foreign exchange arrangements". Also in the *India – Quantitative Restrictions* case the Appellate Body noted with approval the decision of the panel to submit questions to the IMF on India's balance-of-payments situation on the basis of Article 13 of the DSU and Article XV:2 of the GATT.²⁷² It followed from the above that the Panel would have to consult the IMF in accordance with Article XV:2 of the GATT if it were to find that Public Notice No. 60 and the MOUs would have restricted imports in May and October 2000 even in the absence of any requirement to obtain a licence for imports of cars and components and an examination of India's Article XVIII:B invocation would therefore become necessary.

(iv) Relevance of the India – Quantitative Restrictions case

4.157 With respect to deciding what account to take of the findings and conclusions reached by the panel in the *India – Quantitative Restrictions* dispute when considering the justification under Article XVIII:B of the measures at issue in this dispute, the **United States** considered that this Panel should be guided by conclusions reached by the panel in the *India – Patents (EC)* dispute. That panel considered the extent to which it was bound by the findings of a previous panel on the same measure and found as follows:

It can thus be concluded that panels are not *bound* by previous decisions of panels or the Appellate Body even if the subject-matter is the same. In examining dispute WT/DS79 we are not legally bound by the conclusions of the Panel in dispute WT/DS50 as modified by the Appellate Body report. However, in the course of "normal dispute settlement procedures" required under Article 10.4 of the DSU, we will take into account the conclusions and reasoning in the Panel and Appellate Body reports in WT/DS50. Moreover, in our examination, we believe that we should give significant weight to both Article 3.2 of the DSU, which stresses the role of the WTO dispute settlement system in providing security and predictability to the multilateral trading system, and to the need to avoid inconsistent rulings (which concern has been referred to by both parties). ²⁷³

4.158 Therefore, if the Panel decided to consider the balance-of-payments justification of the measures at issue in this dispute, the Panel should bear in mind that both the panel and the Appellate

²⁷⁰ *Ibid.*, paras. 3.305 and 3.311.

²⁷¹ Report of the Appellate Body in *Argentina – Textiles and Apparel*, WT/DS56/AB/R, adopted on 22 April 1998, para. 84 (DSR 1998:III, 1003).

²⁷² WT/DS90/AB/R, para. 149.

²⁷³ WT/DS79/R, para. 7.30 (DSR 1998:VI, 2661).

Body thoroughly analyzed the legal issues in the *India - Quantitative Restrictions* case. Furthermore, this Panel should bear in mind that the *India - Quantitative Restrictions* panel made a factual finding, after extensive factual submissions, on India's balance-of-payments situation as of 18 November 1997 - less than thirty days before the adoption of Public Notice No. 60. It would be neither necessary nor appropriate for this Panel to repeat that work. The Panel should of course consider the arguments of the parties, but it should be guided by the panel's and the Appellate Body's factual findings and recent interpretation of the provisions at issue in that case.

- 4.159 Moreover, the *India Quantitative Restrictions* panel decided that, having regard to both DSU Article 13 and GATT Article XV:2, it would consult with the IMF. After giving both parties an opportunity to comment on a draft, the Panel sent the IMF a letter asking several questions about India's balance-of-payments situation as of 18 November 1997. The IMF's replies confirmed that, as of 18 November 1997, India's reserves were neither declining seriously nor threatened with serious decline; that India's level of foreign currency reserves was adequate; and that India's quantitative restrictions were not needed for balance-of-payments adjustment and should be removed over a relatively short period of time. The IMF also presented additional data for the period ending June 1998; the IMF indicated that this later data corroborated its views.²⁷⁴ It would be entirely unnecessary to ask the IMF once again for an opinion that it had already given.
- 4.160 It was obvious to the **European Communities** that this Panel could not rely on the factual finding of the panel in *India Quantitative Restrictions* to the effect that, as of the date of the establishment of that panel, India was not entitled to implement BOP measures. Rather, this Panel would have to make its own factual findings based on India's BOP situation as of the date of establishment of the panel. On the other hand, as confirmed by the Appellate Body in *Japan Taxes on Alcoholic Beverages*²⁷⁵, the Panel must take into account the findings reached by the panel (and the Appellate Body) in *India Quantitative Restrictions* with respect to issues of legal interpretation of Article XVIII:B, (and in particular with respect to those raised by India at paragraph 4.80).
- 4.161 **India** contended that as the United States had submitted its request for the establishment of a panel on 15 May 2000 and the European Communities on 12 October 2000. India's external financial positions on these dates would have to be examined because they would be different from India's external financial position in November 1997 which the earlier panel examined. The panel on *India Quantitative Restrictions* correctly found that "the determination of whether balance-of-payments measures are justified is tied to a Member's reserve situation as of a certain date". The present Panel therefore could not rely on the assessment of India's balance-of-payment situation by the panel on *India Quantitative Restrictions*.
- 4.162 The **European Communities** contended that Article XVIII:B was an exception to other GATT provisions. Accordingly, it was for India to prove that the strict requirements of Article XVIII:B were met in this case. So far, however, India had submitted no relevant evidence to this Panel. The European Communities recalled that the Panel on *India Quantitative Restrictions* concluded that, as of November 1997, India's reserve position did not justify India's invocation of Article XVIII:B. India had not claimed, let alone proved that, since November 1997, India's BOP situation had deteriorated to the point where resort to Article XVIII:B of GATT would have become justified. Indeed, any claims to that effect would be untenable. India's reserve position as of 12 October 2000, as well as its current position, seemed to be, if anything, even more buoyant than in

 274 Panel Report on India – Quantitative Restrictions, WT/DS90/R, adopted on 22 September 1999, as upheld by the Appellate Body Report, paras. 3.360, 3.361, 3.367, and 3.368.

²⁷⁵ Report of the Appellate Body in *Japan – Alcoholic Beverages II*, WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted on 1 November 1996, at p. 24 (DSR 1996:I, 97).

²⁷⁶ WT/DS90/R, para. 5.157.

November 1997. (As evidence of this, the European Communities is providing to the Panel a copy of the relevant excerpts of the "Economic Survey 2000-2001" published by India's Ministry of Finance).

3. TRIMs Agreement

4.163 The **United States** claimed that the indigenization and trade balancing requirements fell squarely within the scope of Paragraphs 1(a), 1(b) and 2(a) of the Illustrative List, and for that reason they violated Articles 2.1 and 2.2 of the TRIMs Agreement. Separately, they also violated Article 2.1 of the TRIMs Agreement because they were inconsistent with GATT Articles III:4 and XI:1.

Paragraph 1(a): First, firms manufacturing passenger cars in India were "enterprises"; because the ordinary meaning of "enterprise" included "a business firm, a company". Second, India had affirmed that indigenization meant "local content" and "use of local materials alone". Thus, the indigenization obligation required "enterprises" manufacturing automobiles to "use" local parts and components (and effectively also to "purchase" such local parts and components if they themselves did not import them). The indigenization requirement therefore fell within the terms of Paragraph 1(a).

Paragraph 1(b): First, the measures made clear that an MOU signatory's imports of CKD/SKD kits/components "may be regulated with reference to the export obligation fulfilled in the previous years as per the MOU." Second, the export obligation itself was expressed in terms of value -- namely, an FOB value equal to the CIF value of imported CKD/SKD kits and components. Thus, a firm manufacturing cars in India could only import and use a maximum amount of SKD/CKD components that was related to the value of its exports in previous years. If a firm wished to use a greater amount, it must purchase and use such components from an Indian source instead. The trade balancing obligation therefore fell within the terms of Paragraph 1(b).

Paragraph 2(a): Import licenses for SKD/CKD kits/components were not given to companies that do not sign and comply with an MOU. Because compliance was a condition to importation of SKD/CKD kits/components used by such companies in their motor vehicle production in India, those requirements "restrict" importation "generally" of "products used in ... local production". For that reason alone the measures fell within the scope of Paragraph 2(a). The trade balancing obligation also fell within the scope of Paragraph 2(a) for a second reason. Beginning in the fourth year, CKD/SKD kits imports were allowed with reference to the extent of export obligation fulfilled in the previous year. The trade balancing obligation thus imposed a "restriction" on a firm's imports that was "related to the ... value of local production that it [the firm] exports", as provided in Paragraph 2(a).

Illustrative list chapeaux: As required by the chapeaux to Paragraphs 1 and 2, the measures at issue either were "mandatory or enforceable under domestic law or under administrative rulings" or were measures "compliance with which is necessary to obtain an advantage." First, India has confirmed that the measures could be "enforced" at least under the provisions of the FTDR Act, through the import licensing mechanism, and under Indian contract law. Second, manufacturers must comply with the indigenization and trade balancing obligations to be permitted to import SKD/CKD kits/components. This permission constituted an "advantage" because components in SKD/CKD form were "restricted" under the Exim Policy and not generally importable into India.

Non-illustrative list violations of Article 2.1: As shown elsewhere, Public Notice No. 60 and the MOUs were inconsistent with Articles III:4 and XI:1 of the GATT 1994. Therefore, the measures were inconsistent with Article 2.1 of the TRIMs Agreement even if they discriminated against imported goods, or prohibited or restricted the importation of foreign goods, in ways that were not described in the provisions of the Illustrative List. For example, the trade balancing requirement violated Article III:4 because it obligated users and purchasers of imported SKD/CKD parts and components (but not users/purchasers of like domestic goods) to export an equal value of automobiles or automotive components. Even if this form of discrimination was not encompassed in the Illustrative List it was inconsistent with GATT Article III; and for that reason it was also inconsistent with Article 2.1 of the TRIMs Agreement.

4.164 Whether or not the TRIMs Agreement required a separate analysis of whether a measure was a "trade-related investment measure", the measures in this case definitely were such measures.

"Investment measure": First, Public Notice No. 60 and the MOUs required a minimum foreign equity of US\$50 million dollars and thus were designed to increase the amount of foreign investment. Second, these measures steered foreign investment in the Indian motor vehicle manufacturing sector towards establishment of actual production facilities for "manufacture" of cars and away from "mere assembly" of imported kits/components. Third, these measures were also aimed at encouraging investment in and the development of the Indian parts and components industry generally (from whom India plainly wants automobile companies to purchase the components and parts that they need to engage in "manufacture"). The MOU made that clear when it provided that the signatory "shall aggressively pursue and achieve as soon as possible the development of the local supply base " Moreover, Indian officials announcing Public Notice No. 60 said that the policy objective was "to encourage local production of auto-components and thus, bring in modern technology and develop this key segment". The indigenization and trade balancing requirements furthered that domestic investment objective both by favoring local automotive parts and components and by restricting the entry into India of foreign components.

"Trade-related": The measures favoured domestic goods over like imported goods, and they restricted or prohibited importation of certain foreign goods. As past panels had recognized, local content requirements were trade-related by definition. Quantitative import restrictions and prohibitions (the elimination of which was a cornerstone of the GATT 1994) were no less so.

4.165 Under Article 5.2 of the TRIMs Agreement, all WTO Members benefitted from a transition period to eliminate trade-related investment measures inconsistent with Article 2. Those transition provisions, however, did not apply to this case, for three reasons. First, India introduced these measures in December of 1997, nearly three years after the entry into force of the WTO Agreement, but the transition provisions did not apply to measures adopted less than 180 days before the date of entry into force of the WTO Agreement. Second, the only measures to which the transition provisions applied were those that were notified in accordance with the provisions of Article 5.1. While India did notify a dividend balancing requirement in various sectors, India did not notify local content or trade balancing requirements. Third, India's transition period had in any case ended. Pursuant to Article 5.2, India had five years from the entry into force of the WTO Agreement to eliminate all its inconsistent measures. That five-year period had expired on 1 January 2000.

4.166 The **European Communities** pointed out that Article 2.1 of the TRIMs Agreement provided that:

"Without prejudice to other rights and obligations under GATT 1994, no Member shall apply any TRIM that is inconsistent with the provisions of Articles III or XI of GATT."

The TRIMs Agreement did not define the notion of TRIM. Nevertheless, the Annex to that Agreement contained what Article 2.2 describes as:

"An illustrative list of TRIMs that are inconsistent with the obligation of national treatment provided for in paragraph 4 of Article III of GATT 1994 and the obligation of general elimination of quantitative restrictions provided for in paragraph 1 of Article XI of GATT 1994 ..."

The wording of Article 2.2 indicated that any measure which fell within the List constituted *per se* a TRIM inconsistent with Article 2.1, without it being necessary for the complainant to demonstrate in each particular case that the measure concerned was an "investment measure" and was "related to trade in goods". ²⁷⁷ The "indigenization" requirements and the "trade balancing" requirements were both "investment measures" and "related to trade in goods" and, consequently, constituted TRIMs within the meaning of Article 1 of the TRIMs Agreement. Since those requirements were inconsistent with GATT Articles III:4 and XI:1, it followed that they also infringed Article 2.1 of the TRIMs Agreement. Furthermore, the "indigenization" requirements fall within Items 1(a) of the Illustrative List, whereas the "trade balancing" requirements are caught by Items 1(b) and 2(a).

• The requirements are "investment measures": The term "investment measure" had been interpreted in *Indonesia – Autos*, where the panel found that the grant of tax and duty incentives upon compliance with certain local content requirements was an "investment measure". The Panel reasoned as follows:

"[T]hose measures are aimed at encouraging the development of a local manufacturing capability for finished motor vehicles and parts and components in Indonesia. Inherent to this objective is that these measures necessarily have a significant impact on investment in these sectors. For this reason, we consider that these measures fall within any reasonable interpretation of the term 'investment measures'."

Indonesia – Autos stood for the proposition that the characterisation of a measure as an "investment measure" must be based on the purpose of the measure, as discerned from the measure's structure and objectives, rather than on the measure's actual effects on investment. The measures in dispute had a similar structure and pursue the same objective as the measures at issue in Indonesia – Autos, namely to encourage the development of a local automotive industry. That objective is further evidenced by the other two "parameters" laid down in Public Notice No. 60, i.e. "the establishment of actual production facilities for manufacture of cars and not for mere assembly" and the minimum US\$50 million investment requirement. Inherent in the

²⁷⁸ *Ibid.*, para. 14.80.

²⁷⁷ This issue was left undecided by the Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS64/R, adopted on 23 July 1998, paras. 14.64-14.72 (DSR 1998:VI, 2201).

objective of developing a local automotive industry was that India's measures, like Indonesia's, had a "significant impact on investment" and, therefore, constituted "investment measures" for the purposes of the TRIMs Agreement.

- The "indigenization" requirements and the "trade balancing" requirements were "related to trade in goods": The "indigenization" requirements were "related to trade in goods" because they favoured the use of domestic inputs over imported inputs. In turn, the "trade balancing" requirements were "related to trade in goods" because they restricted imports, as well as the internal purchase of imported goods.
- The "indigenization" requirements and the "trade balancing" requirements fell within the Illustrative List of prohibited TRIMs: The "indigenization" requirements fell within Item 1(a) of the Illustrative List, which included among the TRIMs that were inconsistent with Article III:4 those which required:

"the [...] use by an enterprise of products of domestic origin [...], whether specified in terms [...] of [...] value of products, or in terms of a proportion of [...] value of its local production"

To the extent that they required the trade balancing of imported goods purchased by the signatories within India, the "trade balancing" requirements fell within Item 1(b), which covered those measures that require:

"that an enterprise's purchases [...] of imported products be limited to an amount related to the [...] value of local products that it exports"

Finally, to the extent that they required trade balancing of imports made directly by the signatories, the "trade balancing" requirements were caught by the terms of Item 2(a) which covered those measures which restricted

"the importation by an enterprise of products used in or related to its local product ion, [...] to an amount related to the [...] value of local production that it exports."

4.167 The "indigenization" and "trade balancing" requirements had not been notified under Article 5.1 of the TRIMs Agreement: The measures in dispute had not been notified by India under Article 5.1 of the TRIMs Agreement.²⁷⁹ Indeed, since all these measures were introduced by India well after the entry into force of the WTO Agreement, they could never have been validly notified under that provision. Therefore, India's measures could not benefit from the five-years transitional derogation provided for in Article 5.2 with respect to prohibited TRIMs duly notified by developing countries. The European Communities further recalled that, in any event, this derogation expired as of 1 January 2000, without prejudice to the possibility under Article 5.3 that a further period may be granted for those measures previously notified under Article 5.1.

4.168 **India** responded that, according to Article 2:1 of the TRIMs Agreement, "no Member shall apply any *TRIM that is inconsistent with the provisions of Article III or XI of GATT 1994*". An

India had confirmed that the TRIMs notified in G/TRIMS/N/1/IND/1/Add.1 do not cover the measures at issue in this dispute. See India's Replies to Questions posed by the United States in the TRIMs Committee, G/TRIMS/W/16.

inconsistency with the TRIMs Agreement could therefore arise only if there was an inconsistency with the GATT. Having demonstrated that the measures which the Panel was competent to rule upon were not inconsistent with Article III or XI of the GATT, India had therefore also demonstrated that these measures were consistent with the TRIMs Agreement. India considered that Public Notice No. 60 and the MOUs contained provisions that might possibly be regarded as "investment measures" within the meaning of Article 1 of the TRIMs Agreement because they imposed requirements relating to the investments that automobile manufacturers must make, e.g., the provisions relating to minimum foreign equity and the prohibition of mere assembly operations. However, these requirements were not at issue in these proceedings. In any case, they were no longer imposed as conditions for the grant of an import license and could therefore no longer be regarded as "related to trade" within the meaning of Article 1 of the TRIMs Agreement. The provisions of Public Notice No. 60 and the MOUs that were at issue in the present proceedings are clearly trade measures and not investment measures. As long as India applied its discretionary licensing system to SKD/CKD kits and car components, the trade balancing and indigenization provisions might have affected the decisions of car manufacturers as buyers of kits and components but they did not constrain the decisions of car manufacturers as investors. The purpose of the provisions was to minimise the foreign exchange expenditures of car manufacturers, not to regulate their investments. For these reasons, the trade balancing and indigenization provisions do not fall within the framework of the TRIMs Agreement.

4.169 The **United States** considered that India's suggestion that because the trade balancing and indigenization requirements were "trade measures", they were not "investment measures" and therefore were not within the scope of the TRIMs Agreement was a contention that should not be adopted by the Panel. The dichotomy that India tried to draw between "trade" and "investment" measures could not be sustained. Presumably any measure disciplined by GATT Article III:4 could be considered a "trade measure"; on India's logic, however, no such measure would be an "investment measure" – and therefore no measure disciplined by Article III:4 could ever fall within the scope of the TRIMs Agreement. India's hard-and-fast division between "trade" and "investment" measures was a manifestly impossible interpretation of the Agreement. Furthermore, India's position was not consistent with the conclusions of the Panel in *Indonesia – Autos*, which found as follows:

On the basis of our reading of these measures applied by Indonesia under the 1993 and the 1996 car programmes, which have investment objectives and investment features and which refer to investment programmes, we find that these measures are aimed at encouraging the development of a local manufacturing capability for finished motor vehicles and parts and components in Indonesia. Inherent to this objective is that these measures necessarily have a significant impact on investment in these sectors. For this reason, we consider that these measures fall within any reasonable interpretation of the term "investment measures". 280

The measures in question in this dispute were like those in *Indonesia – Autos*: they were aimed at encouraging local manufacturing capability, and they had a significant impact on investments (foreign and domestic) in that sector. As that panel said, the measures in this dispute "fall within any reasonable interpretation of the term 'investment measures'." India's only basis for its assertion appeared to be that the measures in question were "trade measures". But even if India was correct that these were "trade measures", there was no reason to think that they were not also "investment measures" (and therefore *a fortiori* "trade-related investment measures"). India certainly offered no

²⁸⁰ Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 23 July 1998, para. 14.80 (DSR 1998:VI, 2201). The panel emphasized that its characterization of the measures as "investment measures" was based on an examination of the manner in which the measures at issue in that case related to investment, and that there might be other measures that qualified as investment measures within the meaning of the TRIMs Agreement because they relate to investment in a different manner.

reason to believe that a measure could only be one or the other, and there was no support for that position in the text of the TRIMs Agreement.²⁸¹ The indigenization and trade balancing requirements were precisely the sort of trade-distorting measures applied to foreign investment that the TRIMs Agreement was intended to address. India's assertion that these measures were not "investment measures" should be rejected. India also made a second argument – namely, that the trade balancing and indigenization requirements affected the car manufacturers as "buyers of kits" but not as "investors". However, India gave the Panel no textual or other basis for adopting that distinction as the basis of deciding whether a measure was an " investment measure" either. Even if one were to accept India's proposal, however, India was mistaken about the effect of the indigenization and trade balancing requirements. For example, these measures imposed additional costs on car manufacturers that lowered the return on their investment in India. Moreover, car manufacturers in India had to be structured to meet the indigenization and trade balancing requirements; these requirements therefore necessarily affected their investment planning. Furthermore, India had not responded to the other reasons the United States had presented for considering these measures as "investment measures". For instance, these measures were clearly intended to encourage investment in the Indian parts and components manufacturing industry. Indian officials had made that point publicly when they introduced Public Notice No. 60. Furthermore, the MOUs required signatories to "aggressively pursue and achieve as soon as possible the development of the local supply base".

4.170 In short, the United States believed that the Panel should reject India's argument that the indigenization and trade balancing requirements were outside the scope of the TRIMs Agreement. And, given that they were inconsistent with Articles III:4 and XI:1 of the GATT for reasons already discussed, the Panel should further conclude that these measures were also inconsistent with Article 2 of the TRIMs Agreement.

V. ARGUMENTS OF THE THIRD PARTIES

Korea's submission to the Panel was as follows:

1. The Agreements regarding the Indian measures must be respected and enforced

- 5.1 This Panel should consider the years of negotiations and dispute settlement proceedings between India and the European Communities and India and the United States that preceded this dispute. With both the European Communities and the United States, India made due efforts to resolve the issues in a manner that met the concerns of the United States and the European Communities and provided India with the time and flexibility to restructure its automotive regime.
- 5.2 The United States concedes that the first proceeding regarding its claims ended with an agreement between the United States and India that the reasonable period of time for implementation "shall finally expire on 1 April 2001. Yet the United States seeks to readjudicate those same claims (and to challenge "measures" which do not yet exist). India correctly notes the subtle shift of focus by the United States in this proceeding. The earlier proceeding regarded, among other things, India's discretionary licensing scheme; this proceeding regards the application of the very same licensing scheme. Thus, the "distinction" upon which the United States relies does not appear to be a meaningful distinction. As India accurately notes, the DSB ruling covers both the text and the operation and application of the discretionary licensing system for all products notified under Article XVIII:B of the General Agreement on Tariffs and Trade (GATT 1994), including CKD and

 $^{^{281}}$ Cf. Panel Report on EC – Bananas III, WT/DS27/R/USA, adopted on 25 September 1997, as modified by the Appellate Body Report, para. 7.185 (DSR 1997:II, 943) ("the TRIMs Agreement does not add to or subtract from those GATT obligations, although it clarifies that Article III:4 may cover investment-related matters").

SKD kits. Therefore, the United States should have waited until the reasonable period had expired. Then, if the United States still believed that India had not fully implemented the recommendations and rulings of the DSB, it could have sought recourse under Article 21.5 of the DSU or requested consultations for the establishment of a new panel.

5.3 The above analysis of the US action applies also to the EC action. The European Communities and India reached a mutually agreed solution of the India - Quantitative Restrictions (WT/DS96) dispute that they notified under DSU Article 3.6.²⁸² As the European Communities concedes, the solution (as supplemented by the agreement on reasonable period of time between the United States and India) requires India to conform its Regime to its WTO commitments by 1 April 2001. 283 For its part, the European Communities agreed to "refrain from action under GATT" Article XXII or Article XXIII as regards those restrictions during the phasing-out period," i.e. for the Regime, until 1 April 2001.²⁸⁴ Thus, the solution reached in the earlier, broader dispute (WT/DS96) applied to the Regime – the subject matter of this dispute. India has complied with the agreed-upon schedule and the 1 April 2001 deadline has yet to pass. Yet, the European Communities nonetheless has brought this proceeding and asked the Panel to address measures covered by the mutually agreed solution. The EC claim that the subject matter of the current dispute is not within the scope of WT/DS96 is based primarily on three assertions. Korea notes in paragraphs 30-37 of India's First Submission where India rebuts the EC assertions. As a review of India's presentation indicates, the subject of this dispute – the Regime –seems to be covered by the mutually agreed solution. The EC's action raises a systemic question for the WTO dispute settlement system enshrined in the DSU. A review of the DSU indicates that the preferred goal in every dispute settlement proceeding is a mutually agreed solution:

... The aim of the dispute settlement mechanism is to secure a positive solution to a dispute. A solution mutually acceptable to the parties to a dispute and consistent with the covered agreements is clearly to be preferred.²⁸⁵

The EC action seems to undermine this goal by removing any incentive to agree to a solution.

5.4 In the cases of both the US action and the EC action against India, fundamental values are at stake. Where two Members reach an agreement (as here) and formalize the agreement by notifying it to the WTO (as here), the agreement <u>must</u> be respected by both Members. Any other resolution will undermine the entire dispute settlement mechanism. Members must be able to rely on agreements they reach with other Members. This is *a fortiori* the case when, as here, the dispute is between a developing and a developed Member of the WTO.

2. The measures that India may take by 1 April 2001 are not, and as a matter of logic cannot be, properly before the Panel.

5.5 The Panel should not consider any arguments regarding measures that India might propose. First, DSU Article 6.2 indicates that a complainant must "identify the specific measures at issue." Here, the European Communities and the United States could not possibly have identified post—1 April measures as "the <u>specific</u> measures at issue," because the measures do not yet exist. This is confirmed by a review of the panel requests. Second, under DSU Article 7.1, the Panel's terms of

²⁸² See WT/DS96/8, 6 May 1998. (Exhibit EC-8).

²⁸³ EC's First Submission, *India – Measures Affecting the Automotive Sector*, WT/DS146, WT/DS175, 16 January 2001, para. 61.

²⁸⁴ WT/DS96/8, 6 May 1998, p. 2.

²⁸⁵ DSU, Article 3.7 (emphasis added).

²⁸⁶ See WT/DS146/4, 13 October 2000; WT/DS175/4, 18 May 2000.

reference, which set the scope of the proceeding, generally are based (and, in this proceeding, are based) on the "specific measures" identified in the panel request(s). Here, not having been specifically identified in the panel requests, measures that may or may not compose the post—1 April Regime are not part of the Panel's terms of reference and, thus, are not properly before the Panel. Third, a panel cannot possibly examine a measure not in existence when the panel is established. Once the panel is established and its terms of reference are set, all subsequent occurrences are not subject to review. This is because the scope of the panel's authority is defined when the panel is established.²⁸⁷ In sum, all arguments by the United States and the European Communities regarding steps India may take after the date of each complaining party's panel request to comply with its mutually agreed solution with the European Communities and its agreement with the United States pursuant to DSU Article 21.3 have no subject matter (i.e. there is no actual measure to attack) and are untimely. If they wish, the United States and the European Communities may request consultations for a new panel or pursue a DSU Article 21.5 proceeding with India after 1 April 2001. But they cannot circumvent the procedures of the DSU and the terms of the understanding and the agreement with India by challenging in the current proceeding measures that do not yet exist.

3. Prospective measures

- As demonstrated above, the post– 1 April Regime is not within the scope of this proceeding. However, as an aside, Korea wishes to note that the post-1 April Regime described by India appears problematic. As India states, the exclusivity enjoyed by the manufacturers, on the one hand, and the commitments to mitigate the adverse impact of the imports on the balance of payment, on the other hand, constitute a "balanced whole." India has indicated that it does not intend to release the car manufacturers from the commitments they assumed under the memoranda of understanding (MOUs) in order to import SKD/CKD kits prior to 1 April 2001. But, India also has indicated that, from 1 April 2001, it no longer will make the importation of SKD/CKD kits subject to the signing of an MOU. The right to import SKD/CKD kits therefore no longer will be conditioned.
- 5.7 Thus, under India's plan, starting from 1 April 2001, companies already committed to the MOUs would be put in a situation where the MOU obligations are left, but most of the MOU benefits have been removed. The exclusivity granted under the MOUs has been the primary benefit given to MOU enterprises abolishing the exclusive right to import SKD/CKD kits while maintaining obligations under the MOUs would compromise the "balanced whole" for the MOU enterprises. As a result, the continued enforcement of the MOU requirements would undermine the conditions of competition by disadvantaging companies that already have committed to India's market, in comparison with new entrants not subject to the MOU requirements.
- 5.8 India submits that MOU companies should not complain about the modified regime because they already have obtained a comparable advantage. India further asserts that the MOU requirements imposed after 1 April 2001 are merely a *pure* "obligation to manufacture" or "requirements to sell goods in foreign markets in preference to the domestic market" and, thus, are consistent with the WTO agreements.²⁸⁸
- 5.9 Korea reminds India that the MOU companies undertook these obligations in exchange for the exclusive right to import for the duration of the MOU period. For India to remove this right but maintain the MOU obligations would be unfair to the MOU companies. India's continuation of the MOU requirements cannot be separated from the existing Regime and interpreted as merely an

²⁸⁷ See *India — Quantitative Restrictions*, WT/DS90/R, adopted on 22 September 1999, as upheld by the Appellate Body Report, paras. 5.160-5.161 (a panel may examine only those facts existing on the date the panel was established).

²⁸⁸ India's First Submission, paras. 10-12.

imposition of a *new* "obligation to manufacture" or "requirements to sell goods in foreign markets in preference to the domestic market." Rather, it is a *continuation* of the commitments made by the MOU companies as a reward of obtaining the exclusive right to import SKD/CKD kits prior to 1 April 2001. In other words, the post-1 April Regime sketched by India would establish not merely an "obligation to manufacture" or a "requirements to sell goods in foreign markets in preference to the domestic market" – it would establish an obligation and requirement, *compliance with which is necessary to obtain an advantage (which was already exhausted)*. This, of course, would be inconsistent with India's obligations under the TRIMs Agreement and under GATT 1994.

5.10 Moreover, no provision of the WTO agreements justifies imposing disadvantages on the basis of having provided advantages in the past. If such an "cross-temporal balancing" were allowed, the basic nature – the "immediacy and unconditionality" – of the MFN obligation²⁸⁹ would be compromised. Korea invites India to consider the above points as it decides how to alter the Regime.

VI. INTERIM REVIEW

- 6.1 The interim report of the Panel was issued to the parties on 12 October 2001. The parties submitted their comments on 26 October 2001. India expressed its request for a meeting in conditional terms, indicating that it requested a meeting "if the Panel were to decide to make in its final report any findings on whether the violations ceased to exist as a result of events that occurred during the panel proceedings".
- 6.2 In the view of the Panel, the purpose of an interim review meeting is not to provide the parties with an opportunity to introduce new legal issues and evidence or to enter into a debate with the Panel. In the view of the panel, the purpose of the interim review stage is to consider specific and particular aspects of the interim report as requested by the parties. In this instance, the Panel did not feel that it would have been appropriate to make a decision to hold a meeting on the basis of the *outcome* of the interim review process, of which the meeting is a part. In addition, the Panel was of the view that it would not have been in accordance with requirements of due process for the Panel to pronounce on any changes it may make to its findings and recommendations as suggested by India before hearing the views of the other parties on the issues raised by India. Consequently, the Panel decided to hold the meeting on the basis of India's request in order to allow the parties to exchange views on the specific and particular aspects of the interim report on which a review has been requested, without prejudice to the outcome of this interim review process.
- 6.3 An interim review meeting was held on 19 November 2001.
- A. COMMENTS ON THE DESCRIPTIVE PART

1. Comments by the United States

All of the comments made by the United States on the draft descriptive were accepted by the Panel. A number of points, not heretofore reflected in the descriptive part, were incorporated.

2. Comments by India

6.5 India suggested that the descriptive part of the report be restructured, in order to address the preliminary procedural issues first, and that subtitles be phrased either in terms of questions, claims or arguments but not a mix of all three. The Panel considered this acceptable and the descriptive part of the report was accordingly restructured and subtitles adjusted in accordance with India's suggestion.

²⁸⁹ See GATT 1994 Article I:1; General Agreement on Trade in Services, Article II.1.

- At the end of the factual section (II.A), India wished to add that "Since the removal of these restrictions, importers of cars no longer have to sign an MOU to obtain an import license and the signatories of the MOUs, including those with accrued export obligations, are now free to import any amount of cars and components". The United States objected, by letter, dated 13 August 2001, as it considered that this issue was actually a subject of debate during the proceedings and should not be included in the factual section. The Panel agreed with the United States that this point was clarified by India during the panel proceedings. The section of the report titled Factual Aspects (II) records all relevant and undisputed facts available to the Panel. Since, in this case, many of the facts required clarification, the report contains a first section under Arguments of the Parties, Factual Arguments (IV.A).
- 6.7 India also requested that its requests to the Panel be summarized in a more coherent fashion. The Panel accepted the redraft to India's requests for findings (III.C).
- 6.8 In section IV.B, concerned with whether the matter had been settled and adjudicated, India wished a redraft of a concluding paragraph regarding its argument. Since the additions suggested by India were a compilation and not found as a whole in the written evidence, the Panel has included similar language found in India's second submission. (para. 4.53)
- 6.9 In response to comments made by the United States, India submitted additional comments on 20 August. However, many of the comments made by India had either (a) been reflected elsewhere in the report, (b) could not be found reflected in the written evidence or (c) inferred the Panel's use of editorial license to a less than comfortable degree.

3. Comments by the European Communities

- 6.10 The European Communities informed the panel that it had no comments on the draft descriptive part of the report.
- B. COMMENTS ON THE FINDINGS AND CONCLUSIONS

1. Comments by the United States

- 6.11 The United States made only a limited number of comments on the Panel's findings.
- 6.12 It suggested clarification of the language in paragraph 7.86 concerning the reference to the panel *India Quantitative Restrictions*. The Panel found this clarification helpful and revised the paragraph accordingly. The United States also suggested clarifying paragraph 7.190 by expressly indicating that the reference to contract law concerned "Indian" contract law. The Panel agreed with this suggestion and adapted the paragraph accordingly.
- 6.13 The United States commented on a portion of paragraph 7.287 which, in its view, was questionable because it read it to assume a certain assignment of burden of proof to parties participating in domestic proceedings concerning safeguards, anti-dumping or countervailing determinations. While it had not been the intention of the Panel, in that paragraph, to draw any such conclusions or inferences, the Panel accepted to revise the paragraph in order to avoid any misunderstanding on this issue.
- 6.14 In paragraph 7.211, which describes one of the factual elements of the trade balancing obligation, the United States suggested adding a reference to the fact that the complainants had specifically claimed its existence, in order to make clear that the complainants presented a *prima facie* case on this point. The Panel added the reference suggested, because it found it to be factually correct

and a useful addition to the description of the facts of the case. However, the Panel wishes to stress that, in its view, the mere statement that the complainants asserted the existence of a certain fact or measure, as in this instance, in no way amounts to the demonstration of a *prima facie* case of violation in relation to the said measure. The addition of this reference to the complainants' factual assertion therefore cannot be read to constitute in itself an assertion by the Panel that the complainants had presented a *prima facie* case on this issue. The legal analysis of the relevant claims is conducted at a later stage.

6.15 The United States requested a reflection, in footnote 464 of the fact that the meaning of the notion of moving "out of the ambit of the MOU" had been disputed and suggested that the notion should either not be reflected in its current form or clarified in light of the parties' disagreements on its meaning. The Panel did not find it necessary, for the purposes of reaching its conclusions and recommendations, to make an express finding on the meaning of the notion. However, the footnote was clarified to reflect the differences of view between the parties as to its interpretation.

2. Comments by the European Communities

- 6.16 The European Communities commented on certain paragraphs where they felt that its position had not been adequately reflected, and pointed to two instances in which it questioned the appropriateness of the Panel's statements on specific issues.
- 6.17 The European Communities requested a rectification in the description of their claims in paragraph 7.21. The Panel rectified this paragraph accordingly, because it agreed that the rectification requested provided a more accurate description of the claims developed by the European Communities before the Panel.
- 6.18 In footnote 405, the Panel added the relevant reference to the European Communities' argument as requested by it.
- 6.19 The European Communities argued that the Panel confused, in certain paragraphs (7.213, 7.229, and 8.13) two distinct arguments it had presented, namely (1) an argument concerning the scope of the trade balancing requirement, and (2) an argument concerning the duration of the trade balancing obligation beyond 1 April 2001. The Panel accepted the changes suggested by the European Communities and revised paragraph 7.213 of the report to reflect the European Communities' two arguments. In paragraph 7.229, the Panel deleted the last sentence and added the European Communities' second argument in paragraph 7.235. The Panel then addressed the European Communities' second argument in paragraphs 7.278 and 8.57.
- 6.20 The European Communities noted that paragraphs 7.230 and 7.232, which contain an analysis of the scope of the trade balancing obligation, do not fully reflect the indication provided by the European Communities that it did not have the relevant information at its disposal, or the fact that India had not presented supporting evidence for its own assertions. For the sake of clarity and completeness, the Panel added further indications of the elements before it, as submitted by all parties. However, this did not lead to any alteration of its general conclusion on the issue at stake.
- 6.21 The European Communities suggested that footnote 364, which elaborates on some of the systemic issues which might come into play in assessing the role of mutually agreed solutions under the DSU in subsequent disputes, was not necessary to the Panel's analysis and thus inappropriately made. The Panel is certainly fully aware that its role is not to engage in general systemic discussions divorced from what is necessary to the resolution of the dispute at hand. However, this footnote plays a specific role in the Panel's reasoning, in explaining why the Panel is not convinced by the European Communities' argument relating to the fact that a mutually agreed solution is not a "covered

agreement" under the DSU as a sufficient reason for dismissing its potential relevance in subsequent proceedings. The Panel has therefore decided to maintain the essential substance of this footnote, while clarifying its language in order to explain why it is instrumental to its reasoning.

6.22 The European Communities found that a specific assertion by the Panel on the potential impact of its rulings in these proceedings in subsequent "inter-party" determinations concerning the trade balancing conditions was beyond its jurisdiction to assess. The Panel modified this part of paragraph 7.234 in order to clarify that the objective of the disputed statement was merely to assert that it was not this Panel's mandate or objective to make or prejudge, beyond its mandate under the DSU, any future determinations which might be made on the scope of obligations on individual manufacturers in other contexts.

3. Comments by India

- 6.23 India's comments concentrated exclusively on the Recommendations section of the report. In its view, the entire analysis conducted by the Panel under this section was inappropriate, and it suggested the deletion of the entire analysis conducted in paragraphs 8.6 to 8.26, where the Panel assessed, in light of the events argued by India to have occurred in the course of the proceedings, whether it could appropriately make recommendations under Article 19.1 of the DSU that India bring its measures into conformity with its obligations under the GATT 1994.
- 6.24 India suggested instead that the Panel, following its finding that India's measures were in violation of its obligations under GATT articles III and XI, simply make a recommendation as per Article 19.1 that India "bring its measures into conformity with the agreements", and "note that India claimed that it has taken measures to bring its measures into conformity with its obligations under the GATT 1994", and that since this assertion is being challenged by the complainants, it should be left to a compliance panel to assess whether that is the case.
- 6.25 The United States and the European Communities responded in detail to India's comments during the interim review meeting, and their arguments will be also reflected in this section. In light of all the parties' comments, and for the reasons explained below, the Panel decided to maintain a detailed analysis of the implications of the events which took place in the course of the proceedings, in particular as of 1 April 2001, but it modified its analysis in order to clarify further its approach in this section. Since India's comments affect the whole analysis under the Recommendations section, rather than merely individual paragraphs, the Panel will address here successively the legal arguments at stake rather than proceeding with a paragraph by paragraph analysis.
- 6.26 India raises three principal arguments in support of its position, and also raised more generally a concern of due process to the effect that the Panel's analysis of the post-April situation was entirely unexpected and that, had India realized that the Panel would engage in such an analysis, it would have presented relevant evidence. This procedural issue is addressed first.
- (a) General comment in respect of India's procedural rights in the proceedings
- 6.27 India argues that "the Panel embarked on a completely unprecedented approach not foreseen in the DSU without giving any prior indication to that effect. Neither the complainants nor India could therefore have possibly expected the Panel to make the rulings set out in the Recommendations Section. It is irreconcilable with the most basic principles of procedural justice for a tribunal to make a ruling on issues which the disputant parties could not reasonably expect any rulings and on which they could therefore not provide all the evidence available to them." ²⁹⁰ In particular, India argues that

²⁹⁰ Comments on Interim Report by India, para. 15.

if the parties to the dispute could have anticipated that the Panel would determine whether the measures at issue have remained in violation, they would no doubt have exchanged further evidence and arguments on this point. India argues in this context that the Panel's analysis is factually incomplete and, *inter alia*, that it "saw no reason to inform the Panel of the fact that Public Notice No. 60 no longer exists because this was not relevant to any of the issues that the Panel had been requested to address".

- 6.28 The Panel disagrees with India's assertion that it was unpredictable to it that the Panel would address the issue of whether the measures remained in violation after 1 April 2001, or more generally the impact of events having occurred after the requests for establishment of this Panel, including the effect of the removal of the import licensing regime as of 1 April 2001, on the claims presented to it.
- 6.29 In reality, it was clear in these proceedings from the very first exchange of submissions that the elimination of India's import licensing regime in the course of the proceedings, as described by India itself, was an important part of the issues in discussion throughout the proceedings. The issue of the effects of this event was brought to the Panel's attention from the beginning of the proceedings, in the context of various arguments.
- 6.30 In its first submission, India divided its legal arguments in two sections, one of which related exclusively to the measures to be applied by India after 1 April 2001. Clearly, India argued principally that the Panel should have refrained in the first instance from examining these measures as they stood as of the date of its establishment, that is, in the context of the import licensing regime, and India also argued that the Panel should have found any measures India may apply as of 1 April 2001 to be outside its terms of reference. The Panel recalls that it concluded, however, that it was competent to examine the measures as they existed at the time of its establishment and that it would, indeed, not be considering any subsequent measures that India might take for the enforcement of the existing measures before it. Nonetheless, India also presented extensive arguments and evidence concerning the situation after 1 April 2001 beyond these jurisdictional arguments. India expressly argued, as part of its legal argumentation in response to the claims, that the measures that it would be applying as from 1 April would be consistent with its obligations under the GATT and the TRIMs Agreement. Indeed, the very first section of India's legal argument in its first submission is entirely devoted to this issue.²⁹¹
- 6.31 Indeed, as reflected in Section II of this report (which was inserted using India's very own words), India's latest articulation of its requests for findings to the Panel prior to the issuance of the interim report, clearly suggests that it anticipated that these events were of potential relevance in determining the type of recommendations to be made under Article 19.1 of the DSU and requested the Panel to take them into account, were it to decide to examine the measures as they existed and operated under India's import licensing regime. It stated that:

If the Panel were to conclude that the operation of Public Notice No. 60 and the trade balancing provisions under India's former licensing regime was inconsistent with the GATT, it should – following the practice of other panels – note that the licensing regime had been abolished on 1 April 2001 and that it was *therefore* not necessary for the Panel to recommend to the DSB in accordance with Article 19:1 of the DSU that it request India to bring these measures into conformity with the GATT. (emphasis added)

6.32 In addition, India presented arguments in the course of the proceedings to the effect that only one manufacturer had yet to achieve the required level of indigenization and, as it highlights itself in

²⁹¹ See First submission of India, paras. 10 and following.

its interim comments, requested the Panel to "limit any recommendation under Article 19.1 of the DSU to the indigenization requirement that remains to be performed by that particular company with respect to that model"²⁹², thereby inviting a consideration of events having taken place in the course of the proceedings in the Panel's Recommendations under Article 19.1.

- 6.33 It is therefore clear, in the Panel's view, not only that India could reasonably have expected that the Panel would or could address the impact of the events which occurred in the course of the proceedings, but that it could also reasonably anticipate that the consideration of these issues might be specifically relevant to the Panel's recommendations under Article 19.1 of the DSU, as reflected in its own requests for findings.
- 6.34 It is worth noting in this respect also that the complainants, in addition to disagreeing with India's assertion that it was not provided with a full opportunity to present arguments on the events at issue, also do not deny that they themselves had the opportunity to respond to these arguments. Indeed, as will be noted below, the expression of their requests for findings to the Panel reflects this.
- 6.35 Paragraphs 8.4 to 8.30 of the Recommendations Section of the Report have been developed in order to present more fully the arguments of the parties to clarify further the issue before the Panel in the circumstances of this case.
- 6.36 Finally, with regard to India's allegation that it was not provided with adequate opportunity to present evidence and arguments in relation to the elements of relevance to the Panel's analysis in its recommendations section, the Panel notes that the only new piece of evidence presented by India was evidence of repeal of Public Notice No. 60, which it had previously stated was not a necessary action for it to take.²⁹³
- (b) India's argument that the matters addressed in the Recommendations section should be left to a compliance panel.
- 6.37 India argues that the analysis conducted by the Panel under its Recommendations section should have been left to a compliance panel. In this argument, India draws a distinction between a panel's obligation under Article 19.1 of the DSU to recommend that a violating measure be brought into conformity and the distinct question of whether and how the respondent has complied with this obligation, which is a matter for a compliance panel under Article 21.5.
- 6.38 India argues that Article 21.5 "does not distinguish between measures taken during the panel proceeding" and those taken after the completion of its work. In the view of India, a disagreement on this issue, even occurring before the end of the initial panel proceedings, should be left to a compliance panel, and the role of the initial panel should only be to take into account the situation existing as of the request for establishment of the panel and record the parties' disagreement as to subsequent evolutions.
- 6.39 The Panel first wishes to clarify the scope of the issue at stake here: contrary to India's assertion, the Panel did not aim to analyse, under its Recommendations Section, any measures which it had previously found to be outside its terms of reference: the analysis relates *not* to any new measure which India might have taken in the course of the proceedings. Its exclusive objective was to examine the status of those measures which the Panel *had* identified to be covered by its terms of reference in order to determine the nature and extent of the Recommendations it should make to the DSB in relation to *those* measures.

²⁹³ See First Submission of India, para. 14.

²⁹² See Comments on the Interim Report by India, para. 16.

- 6.40 The issue at stake here therefore relates exclusively to the extent to which it was appropriate for this Panel to seek to address any evolutions of the matter squarely within its terms of reference which might have occurred in the course of the proceedings and the form in which it may address them.
- 6.41 In the particular circumstances of this case, and for the reasons further elaborated in paragraphs 8.31 to 8.63, the Panel found that the most appropriate way to discharge its mandate under the DSU was to conduct an analysis of the issues presented to it regarding evolutions having arguably affected the matter before it in the course of the proceedings in the context of its section on Recommendations.
- 6.42 However, the Panel wishes to answer here the systemic argument raised by India that the matter it addresses in this section should have been left to a compliance panel and that "Article 21.5 does not distinguish between measures taken during the panel proceeding and those taken after the panel's work is completed", and that the issue of "how" a violating Member might bring its measures into conformity was not within the competence of this Panel under Article 19.1.
- 6.43 In response to India's arguments concerning the competence of an Article 21.5 panel on the matters addressed by this Panel in its Recommendations section, both complainants noted that a 21.5 compliance panel in any case only comes into play following recommendations by the DSB to bring certain measures into conformity with the relevant Agreements. The role of a compliance panel is precisely to address "disagreement as to the existence or consistency with a covered agreement of measures taken to comply with the recommendations and rulings" of a previous panel. The Panel notes, in this regard, that the Appellate Body in *US Shrimp* (21.5) stated that Article 21.5 of the DSU involves "in principle, not the original measures, but a new and different measure that was not before the original panel".
- 6.44 The Panel agrees that, reading the terms of Article 21.5 in their ordinary meaning in accordance with the customary rules of interpretation reflected in Article 31 of the Vienna Convention, the formal conditions of a compliance panel are in any case manifestly not met, since no ruling or recommendation had yet been made, let alone adopted in this instance. Furthermore, the Panel in this instance is no way attempting to pre-empt, in its analysis, what a compliance panel might be led to examine in the future. The definition of a compliance panel's role actually assists in distinguishing it from the analysis conducted by this Panel: as previously noted, the sole aim of the analysis conducted in the Panel's recommendations section is to determine whether it was still appropriate to make a recommendation notwithstanding India's assertion that the legal situation had fundamentally changed and that there was nothing left to bring into compliance. In concluding that such a recommendation was appropriate, the Panel did not attempt to indicate *how* India was to bring the relevant measures into compliance, or to prejudge on any future developments which may later occur following such recommendations, which might, if need be, be referred to a compliance panel.
- 6.45 The Panel also notes that India disputes the reference made to the Appellate Body's report on US Certain EC Products, to the effect that where a panel has found a measure no longer to exist, it is logically inconsistent for it to make a recommendation that that measure be brought into conformity with the concerned Member's obligations under the covered Agreements. In India's view, this reference is inappropriate because it concerned a case where both parties agreed that the measure had ceased to apply, which is not the case here. In response to India's arguments, both complainants argued that the facts of that case did not suggest, as argued by India, that the parties were simply in agreement on discontinuation of the measure. Indeed, the Panel notes that there were some disputed

²⁹⁴ See Oral Statement by the European Communities at the interim review meeting, para. 3, and Oral Statement by the United States at the interim review meeting, para. 11.

factual contentions in that case concerning the definition of the measure at issue, which were addressed explicitly by the panel and the Appellate Body. These suggest that the Appellate Body's observation concerning the appropriateness of the recommendation foreseen in Article 19.1 of the DSU was not a mere suggestion that the panel should have simply "endorsed" an agreement of the parties on the cessation of the measures. It also suggests that, contrary to India's suggestion, the agreement of the parties on the fact that the measure has ceased to apply was not the primary source of the Appellate Body's conclusion.

- 6.46 More significantly perhaps, the Panel notes also that the Appellate Body's motivation in that case does not rely on the alleged fact that the *parties* had agreed that the measure no longer existed but rather on the fact that "the <u>Panel has found</u>" that the measure no longer exists. That panel was therefore not considered by the Appellate Body to be simply "endorsing" a fact agreed on by the parties, but rather had made its own finding as to the continued existence of the measure at issue. Accordingly, the Panel has not modified its reference to this report of the Appellate Body.
- (c) India's argument that the Panel could not "properly" conclude that the violation is still in existence
- 6.47 While India does not contest that there can be no sense in making a recommendation to bring into conformity a measure of which a panel is properly of the view that the violation has ceased to exist, it challenges that the Panel can, in this instance, properly have <u>any</u> view on the matter. One of India's arguments in this respect is that the Panel cannot reach this conclusion because the facts are in dispute. The Panel will merely note in this respect that it is the essence of any litigation process to reach conclusions, where appropriate, on disputed issues. What is more pertinent is whether there was a proper legal basis for the Panel's analysis.
- 6.48 India argues in this respect that the terms of reference of a panel are limited to the matter referred to it by the DSB, which excludes, in the view of India, any matters arising in the course of the proceedings. At this point, the Panel merely wishes to recall its earlier observation that it did *not* examine any new measures adopted subsequent to the requests for its establishment, having found that such measures were not within its terms of reference and indeed, had not been the object of any claims by the complainants. The Panel's Recommendations section concerns exclusively those measures that *were in existence* at the time of both complainants' requests for establishment of panels, and which were within its terms of reference. The detailed reasons which led the Panel, in the circumstances of this case, to consider events subsequent to the Panel's establishment in the context of its recommendations are reflected in paragraphs 8.31 to 8.63 of the revised Recommendations section.
- 6.49 India also argues that neither the United States nor the European Communities asked the Panel to make any of the rulings set out in the Recommendations section of the Report. However, as will be also further reflected in the revised Recommendations section, both complainants, either spontaneously or in reaction to India's arguments and the Panel's questions, had clarified the nature of the findings they were requesting, and this included a request for findings that the measures were inconsistent and a recommendation that they be brought into conformity with India's obligations. While it is true that both complainants argued that the changes subsequent to 1 April 2001 did not affect the nature of their legal claims, they both consequently also argued that the measures should be found to have *remained* inconsistent with the same provisions.
- 6.50 While they did not necessarily request that such findings be made separately in the context of the Recommendations section as was done by the Panel²⁹⁵, the complainants certainly envisaged that the Panel may be led to consider the continued existence of violating measures, since they were

²⁹⁵ See Oral Statement of the United States at the interim review meeting, para. 22.

arguing that these measures continued to be in violation and expected the Panel to make a finding to that effect. The fact that the complainants may not have requested the Panel's assessment of this matter in a particular form does not alter the Panel's general obligation, under Article 11 of the DSU, to make an objective assessment of the matter before it and to assess how best to fulfil its mandate in this regard within the limits of its terms of reference. This matter included, in this instance, the measures and claims identified in the requests for establishment of a panel, and the Panel was required, under its terms of reference, to make such findings as will assist the DSB in making recommendations or in giving the rulings provided for in the relevant agreements concerning this matter. For these reasons, and those further elaborated in paragraphs 8.15 and following, the Panel disagrees with India that its Recommendations section was outside the scope of what was submitted to it under its terms of reference.

- 6.51 India also argued that the Panel's recommendations were based on incomplete evidence, in particular because it did not, in India's view, take account of the arguments produced by India concerning the levels of indigenization reached by individual MOU signatories. The Panel notes that it had in fact reflected India's statement in this regard in paragraph 8.22 of its original recommendations. However, this had not altered the Panel's general legal conclusion on the matter. The relevant paragraphs have been expanded to further clarify the Panel's analysis in this respect and reflect the parties' most recent exchanges on this matter (see paragraphs 8.46 to 8.49).
- 6.52 The Panel notes also that India produced new evidence to the effect that Public Notice No. 60 has been repealed, indicating that it previously had no reason to suspect that this could be relevant to the Panel's analysis and therefore did not produce it before. The Panel cannot agree with India's argument for not producing this piece of evidence earlier, namely that it did not suspect that the issue would be of any relevance to the Panel's analysis. In fact, as noted earlier, both India and the complainants addressed this issue expressly in the course of the proceedings, and India had devoted a section of one of its earlier submissions to the argument that Public Notice No. 60 did not need to be repealed.
- 6.53 Leaving aside the issue that this last argument raises as to the actual relevance of the new piece of evidence produced, even in India's own view, the Panel considered whether it was appropriate to accept it at this late stage of the proceedings. The complainants had diverging views on the matter, since the European Communities expressed no opposition while the United States thought it inadmissible.²⁹⁶
- 6.54 Certainly, the Panel can only regret that a piece of evidence concerning a matter which was, in its view, clearly within the purview of matters discussed in the proceedings, was produced at such a late stage in the proceedings. In the interest of completeness, and in keeping with its general approach in its recommendations, the Panel nonetheless decided to accept to consider the new evidence, in light also of the fact that this evidence only sought to confirm the official status of the measure as it had already been argued and discussed during the proceedings. It therefore did not bring into the debates any fundamentally new issue that would have required extensive additional discussion among the parties. This new evidence is taken into account in paragraphs 8.35 and following and 8.60 and following.
- (d) India's argument that the Panel inappropriately prescribes a "retrospective" remedy
- 6.55 India challenges the Panel's analysis of "residual" export obligations as having accrued in relation to past imports and its conclusion that they are therefore not of a different nature from any of the previously accrued export obligations, even if no import licenses are currently being generated. In

²⁹⁶ See ibid, para. 19.

India's view, this implies that the Panel is requesting it to "remove the consequences of a past illegal act" and thus providing a retrospective remedy not foreseen by the DSU, whose remedies are limited to "the cessation of the illegal act".

- 6.56 While India's systemic concern that the DSU generally does not aim at providing "retrospective" remedies may be valid, this is not, in the Panel's view, what the Panel has done here.
- 6.57 The Panel has not sought, in its analysis, to determine the nature or modalities of remedies to be provided by India, beyond determining *whether* there was still a need to make a recommendation to the DSB in order to remedy an identified violation. What the Panel has sought to address in this section is what remains as of *today* of a measure found to be illegal. The Panel has said nothing of any *past* fulfilment of export obligations or of any need to compensate manufacturers for any such *past executions* of illegal obligations.
- 6.58 To the extent that this issue is even one of remedy, it is one of prospective remedy, not retrospective as argued by India. Indeed, the Panel's analysis corresponds to what India itself identifies as the correct interpretation of remedies under the WTO: the "cessation of an illegal act", for the future. By contrast, the implementation of the rulings in the *Australia Automotive Leather II* case referred to by India in its interim comments²⁹⁷, involved repayment of subsidies already paid over past years, which is clearly "retrospective".
- 6.59 Overall, including for the reasons above, the Panel was not convinced by India's arguments that it inappropriately performed the role of a compliance panel, or made rulings on issues not within its terms of reference, or provided any retrospective remedy in its Recommendations Section. However, the Panel is fully aware that the approach it has chosen to adopt here as a result of the particular circumstances of the case is not one that will necessarily be appropriate in other cases. In revising the Section in light of the parties' comments, the Panel was mindful to fully explain the purpose of its analysis and the specific circumstances which led it to adopt this approach in order to fulfill its mandate under the DSU, and to adapt as appropriate its whole analysis with a view to clarifying its exact nature and scope.
- 6.60 The Panel notes in this respect the concern expressed by the United States that while it was understandable in the circumstances of this case that the Panel chose to introduce this analysis, in particular in light of India's arguments, this should not be seen to suggest that there is a general obligation for panels to systematically "reassess", at the time of their recommendations, the existence of previously established violations. As is explained in detail in the revised Recommendations section, the Panel agrees that the approach it has chosen to follow here does not, indeed, imply such a general obligation. However, in conclusion, the Panel must point out to parties that the way in which they plead a case and develop their arguments cannot but have repercussions on the reasoning and conclusions of the Panel; if it were otherwise, the Panel would be derelict in its duty under Article 11 of the DSU to make an objective assessment of the matter before it.

VII. FINDINGS

7.1 This dispute is based on the facts described in Section II above. The parties' claims and requests for findings are summarized in Section III of our Report.²⁹⁸

²⁹⁷ See Comments on the Interim Report by India, para. 28.

²⁹⁸ As will be apparent from the following section, some of the facts of the case were a subject of debate.

- 7.2 In this case, the United States and European Communities successively requested the establishment of panels concerning certain matters. The DSB decided, in accordance with Article 9 of the DSU, to entrust a single panel with the examination of both complaints.²⁹⁹
- 7.3 Both the United States and European Communities allege violations of Articles III:4 and XI:1 of GATT and of Article 2 of the TRIMs Agreement in relation to certain Indian measures affecting the automotive industry. Although these claims related to the same measures, they partly differed in their specific details. As a consequence, the Panel has sought to carefully reflect these differences in its report. It has also been mindful to organize its examination and present its findings to the DSB in such a manner as to in no way impair the rights which the parties to the dispute would have enjoyed had separate panels examined the complaints. This is in conformity with the requirements of Article 9.2 of the DSU. Neither party having requested the submission of separate reports as foreseen in Article 9.2 of the DSU, this report addresses both the United States and the European Communities complaints in a single report.
- 7.4 India has raised defenses relating to the exact nature of the matter before the Panel and the extent of its jurisdiction. These jurisdictional arguments, which condition the Panel's ability to proceed further with the analysis of the claims, are addressed first. Before doing so, this Report begins with an examination of preliminary issues that arose in the course of the proceedings relating to the clarification of the claims. The Report then clarifies the measures within the Panel's terms of reference. This will assist the Panel in subsequently addressing the detail of India's procedural arguments.
- A. CLARIFICATION OF THE CLAIMS IN THE COURSE OF THE PROCEEDINGS AND REQUESTS FOR PRELIMINARY RULINGS
- 7.5 During the first meeting with the parties, India requested the Panel to ask the complainants a number of questions in order to clarify the claims. India indicated that, depending on the complainants' answers to these questions, it would seek preliminary rulings from the Panel concerning the precise scope of the complaints. These requests for clarification and the Panel's approach to them will be detailed in this section.
- 7.6 At the first meeting,
 - (1) India requested the Panel to ask the complainants whether the measures on which they were requesting rulings were those that India was applying at the time when the requests for the establishment of a panel were submitted to the DSB or conversely, whether the measures were those that India might be applying as from 1 April 2001. It advised that "[i]f the complainants were to respond that they consider India's future measures to be part of the present proceedings, [India] would request the Panel to issue a preliminary ruling to the effect that these measures fall outside of its mandate". 300
 - (2) India also requested the Panel to ask the complainants to clarify whether in their view, the measures at issue in this proceeding involved the enforcement of MOUs through the import licensing regime and PN60, or conversely, whether the claims were limited to the enforcement of MOUs outside the licensing regime and PN60. It advised that "[i]f the complainants were to respond that their complaints relate to the enforcement of the MOUs through the import licensing regime and Notice N° 60,

²⁹⁹ See paragraphs 1.3 and 1.4 of this Report for details.

³⁰⁰ See Statement of India at the First substantive meeting, para. 8.

India would ask the Panel to issue a preliminary ruling to the effect that both complaints are inadmissible because the DSU cannot be invoked in respect of the same matter twice."³⁰¹

- 7.7 In response to this request, the Panel invited the parties to ask each other any questions that they felt might be useful. It indicated its own intention to ask the parties a range of questions, including questions relating to the preliminary issues raised by India. The Panel indicated that, if necessary, it would revert to the issue of preliminary rulings depending upon the parties' responses to these questions. This approach was in line with the "conditional" character of India's request for preliminary rulings. In the view of the Panel, it was also appropriate to seek further clarifications on the nature of the matter before it. 302
- 7.8 In light of the complainants' responses to questions, India indicated in its second submission that:

"the European Communities and the United States appear[ed] to accept the validity of [India's] arguments because they made two matters clear. First, the European Communities and the United States clarified that their complaints are not directed against the application of discretionary licensing system to cars and components but against Public Notice No. 60 and the MOUs as such. Second, the European Communities and the United States clarified that their complaints relate to the measures applied by India at the time when they submitted their requests for the establishment to the DSB, and not to any measures that India may take in the future. With these clarifications as to the scope of the rulings requested from this Panel, two central arguments that India had advanced in its First Submission are no longer relevant. India's argument that no WTO Member may resort to the DSU on the same matter twice is no longer relevant because the European Communities and the United States have made clear that their case does not rest on the application of the discretionary licensing system that had been the subject of their prior complaints. India's argument that the Panel is not competent to rule on measures that India may take after 1 April 2001 is also no longer relevant."

- 7.9 At that stage therefore, India did not pursue its initial request for a preliminary ruling, and the Panel did not revert to it.
- 7.10 However, India made other comments at the same stage that led the Panel to believe that there was still some confusion between the parties as to the exact scope of the claims and that India's comments above were conditional. In particular, India indicated two paragraphs below the above quotation that it would:

_

 $^{^{301}}$ Ibid

We recall in this respect the observation made by the panel in *Canada – Measures Affecting the Export of Civilian Aircraft*, (hereinafter "*Canada – Aircraft*"), WT/DS70/R, that "in our view, there is no requirement in the DSU for panels to rule on preliminary issues prior to the parties' first written submissions. Nor is there any established practice to this effect, for there are numerous panel reports where rulings on preliminary issues have been reserved until the final report. Furthermore, there may be cases where the panel wishes to seek further clarification from the parties before providing a preliminary ruling." para 9.15. In this case, the request for a preliminary ruling came at a later stage in the proceedings, and was by its own terms conditional upon further clarifications. In the view of the Panel, these clarifications helped in identifying the matter sufficiently to make it possible, to proceed without a preliminary ruling while taking care to ensure full respect for due process requirements for all parties.

"assume throughout this submission that the Panel will examine the operation of Public Notice No. 60 and the MOUs as such, that is independently of the application of the discretionary licensing system. Specifically, India will assume that the Panel will examine whether the trade balancing and indigenization provisions set out in Public Notice No. 60 and the MOUs would have been inconsistent with Articles III:4 and XI:1 of the GATT even if India had not applied any import licensing restrictions for cars imported in the form of CKD/SKD kits and for car components at the time when the requests for the establishment of a panel were submitted. If the Panel does not share this understanding of the scope of the complaints, India would appreciate it if the Panel were to alert India to that effect before the next meeting and give it the opportunity to make a supplementary written submission." (emphasis in original)³⁰³

- 7.11 The Panel did not consider it appropriate to simply endorse or reject India's reading of the claims before it as suggested. Rather, it believed it was appropriate, and in conformity with the requirements of due process, to ensure that the complainants were given an opportunity to clarify the claims further in order to ensure India a full opportunity to respond comprehensively to the claims as presented.
- 7.12 In a letter sent to the parties prior to the second meeting, the Panel therefore declined to make the interpretation of the claims before it as suggested by India. Instead, it invited the parties, including India, to present all the arguments that they felt may be of relevance in the course of the second meeting so as to allow the Panel to make an objective assessment of the matter before it, as required by Article 11 of the DSU.³⁰⁴
- 7.13 The Panel then put a number of additional questions to the parties after the second meeting. These included questions to India concerning its understanding of the claims following exchanges between the parties in the course of the second meeting. The Panel wished to further ensure that India was in a position to present the arguments and defenses it wished to rely on, in light of the

The Panel does not intend to put forward or endorse, at this stage in the proceedings, any interpretation of the "scope of the complaints", unless the parties were to reach a common understanding in this matter. In this light, we would strongly urge all parties to present to the Panel, during the second meeting, the full range of arguments which they feel are or may be of relevance to the resolution of the dispute, so as to enable the Panel to make an objective assessment of the matter before it in accordance with the terms of Article 11 of the DSU. In particular, India is not constrained in bringing forward arguments in defense in response to the claims submitted."

³⁰³ Second Submission of India, para. 4.

³⁰⁴ The Panel addressed the following message to the parties prior to the second meeting:

[&]quot;The Panel wishes to respond to a point raised in the Indian submission, in order to ensure that all parties are in a position to prepare themselves adequately for the second substantive meeting.

In paragraph 4 of its second submission, India indicates that "[i]n light of the clarifications provided by the complainants, India will assume throughout this submission that the Panel will examine the operation of Public Notice No. 60 and the MOUs as such, that is independently of the application of the discretionary licensing system. Specifically, India will assume that the Panel will examine whether the trade balancing and indigenization provisions set out in Public Notice No. 60 and the MOUs would have been inconsistent with Articles III:4 and Article XI:1 of the GATT even if India had not applied any import restrictions for cars imported in the form of SKD/CKD kits and for car components at the time when the requests for the establishment of a panel were submitted". India then invites the Panel to alert India before the meeting if it does not share its understanding of the scope of the complaint and give it the opportunity to make a supplementary written submission.

complainants' latest clarifications. In the Panel's view, this procedure adequately preserved the interests of all parties, including India's opportunities to respond fully to the claims presented.

7.14 In light of the evolving factual circumstances and successive clarifications in the course of the proceedings, the Panel sought to give the parties adequate opportunity at each stage to pursue the possibility of reaching a mutually satisfactory solution, as required by the final sentence of Article 11 of the DSU. The Panel thus drew the parties' attention to this provision at each of its formal meetings with them and when so requested, afforded them additional time to present their submissions to the Panel.

B. CLARIFICATION OF THE MEASURES WITHIN THE TERMS OF REFERENCE OF THIS PANEL

- 7.15 India has raised a number of arguments relating to the competence of this Panel to examine the claims before it. Some of these arguments relate to the identification of the measures properly within the Panel's terms of reference. Since these terms of reference determine a panel's jurisdiction³⁰⁵, the Panel will start its analysis with a clarification of the measures articulated within those terms. In this context, it addresses India's arguments relating specifically to the ambit of the measures under review.
- 7.16 This initial clarification of the measures before the Panel will also assist in addressing, in the next section of the Report, India's other arguments relating to the Panel's competence to examine the matter before it.

1. Measures identified in the complainants' requests for establishment of a panel

7.17 This Panel was established with standard terms of reference in accordance with Article 7 of the DSU. The measures and claims within those terms of reference are therefore those that have been identified in the complainants' requests for establishment of a panel.

7.18 The United States' request for establishment of a panel indicated:

"The United States considers that certain Indian measures affecting trade and investment in the motor vehicle sector are inconsistent with India's obligations under the WTO Agreement. The measures in question require manufacturing firms in the motor vehicle sector to achieve specified levels of purchase or use of domestic content; to achieve a neutralization of foreign exchange and to balance the value of certain imports with the value of exports of cars and components over a stated period; and to limit imports to a value based on previous exports. These requirements are enforceable under Indian law and rulings (including under the Foreign Trade (Development and Regulation) Act 1992). In addition, manufacturing firms in the motor vehicle sector must comply with these requirements in order to obtain certain Indian import licenses. The United States believes that the Indian measures in question are therefore inconsistent with India's obligations under Article III:4 and XI:1 of the General Agreement on Tariffs and Trade 1994 (GATT 1994) and Articles 2.1 and 2.2 of the Agreement on Trade-Related Investment Measures.

The measures in question are Public Notice No. 60 ((PN)/97-02) of the Indian Ministry of Commerce, published in the Gazette of India Extraordinary, effective

³⁰⁵ See Report of the Appellate Body in *India – Patents* "The jurisdiction of a panel is established by that panel's terms of reference, which are governed by Article 7 of the DSU", WT/DS50/AB/R, para 92.

12 December 1997; the Foreign Trade (Development and Regulation) Act 1992; the Export and Import Policy, 1997-2002; memoranda of understanding signed by the Government of India with manufacturing firms in the motor vehicle sector pursuant to Public Notice No. 60; as well as amendments thereto, other legislative and administrative provisions implemented thereby or consolidated therein, and implementing measures or associated administrative actions taken thereunder." (emphases added)³⁰⁶

7.19 The European Communities indicated in its request for establishment of a panel:

"The measures concerned are contained in:

- Public Notice No. 60 (PN 97-02) of the Indian Ministry of Commerce, effective 12 December 1997; and
- the Memorandums of Understanding (MOUs) entered into by certain manufacturers of automobiles with the Government of India pursuant to Public Notice No. 60.

The above measures require manufacturers of automobiles to sign an MoU as a condition for obtaining licenses for importing automotive products that are currently subject to import restrictions. According to the Indian authorities, the MOUs are "binding" and "enforceable" instruments, which shall remain valid after the date when those restrictions are eliminated. The MOUs require (i) to establish "actual production facilities" in India; (ii) to make a minimum investment; (iii) to achieve a certain level of "indigenisation"; and (iv) to export a certain amount of automotive products.

The European Communities considers that the requirements imposed by the above measures are in violation of Articles III:4 and XI:1 of GATT and of Article 2.1 of the TRIMs Agreement."

- 7.20 Both requests thus included express reference to Public Notice No. 60 and the MOUs signed thereunder and to the specific conditions provided for under those measures. As will be detailed below, the complainants explained in their submissions that they were specifically challenging two of these conditions, namely, the "indigenization" and "trade balancing" conditions.
- 7.21 The United States requested the Panel to find that the indigenization and the trade balancing requirements imposed by Public Notice No. 60 and the MOUs, were inconsistent with Article III:4 and XI:1 of GATT 1994 and Articles 2.1 and 2.2 of the TRIMs Agreement and to request India to bring these measures into conformity with its obligations. The European Communities, similarly, requested the Panel to find that the trade balancing requirements contained in Public Notice No. 60 and in the MOUs concluded thereunder were inconsistent with Articles III and XI of the GATT 1994 and Article 2 of the TRIMs Agreement and that the indigenization requirements contained in these measures are inconsistent with Articles III and XI of the GATT 1994 and with Article 2 of the TRIMs Agreement. 308

 $^{^{306}}$ WT/DS175/4.

³⁰⁷ See para. 3.1 above.

³⁰⁸ Both complainants also referred in their requests to related measures, concerning which they did not make distinct claims in the course of the proceedings.

7.22 It is not disputed that Public Notice No. 60 was in existence and in operation at the time of both requests for establishment of a panel. It is also not disputed that most major car manufacturers had already, at that time, entered into MOUs as foreseen in Public Notice No. 60, which included, *inter alia*, the two conditions specifically referred to by the complainants. These measures as they existed at the time of the complainants' requests for establishment of this Panel are thus expressly contained in the Panel's terms of reference and are consequently within its jurisdiction in accordance with the terms of Articles 6.2 and 7 of the DSU.³⁰⁹

2. Evolution of the measures in the course of the proceedings

- 7.23 India indicated in the course of the proceedings that import licensing requirements were to be removed on the products at issue in this dispute as of 1 April 2001, in order to comply with the implementation obligations agreed with the United States and the European Communities in prior disputes concerning quantitative restrictions maintained by India for balance-of-payments purposes. Nevertheless, India also explained that the provisions of existing MOUs would continue to apply, to the degree that they were operative as of 1 April 2001.
- 7.24 India advised that the implications of these changes would be that as of 1 April 2001, no new MOUs would be required under Public Notice No. 60 and imports of previously restricted kits and components could be made freely as from that date. However, signatories of existing MOUs under Public Notice No. 60 would continue to be required to discharge outstanding obligations under the MOUs they had entered into. In particular, they would continue to be required to meet the indigenization condition foreseen in Public Notice No. 60 and their MOU and to discharge export obligations accrued in relation to previously restricted imports under the "trade balancing" condition.
- 7.25 In India's view, the events of 1 April 2001 modify the legal situation before the Panel. It argues in particular that any measures which India might apply as of 1 April 2001 would fall outside the terms of reference of this Panel.³¹¹ It also argues that the measures before the Panel "as applied after 1 April 2001" (i.e. the remaining obligations under the MOUs), will in any case be consistent with the GATT and TRIMs Agreements.³¹²
- 7.26 A WTO panel is generally competent to consider measures in existence at the time of its establishment. This power is not necessarily adversely affected simply because a measure under review may have been subsequently removed or rendered less effective. Panels in the past have examined discontinued measures where there was no agreement of the parties to discontinue the proceedings.³¹³

³⁰⁹ This observation on the basis of the terms of reference is without prejudice to the examination, in the next section of this report, of India's further challenges to the Panel's competence to examine the case on distinct grounds.

distinct grounds.

310 India – Quantitative Restrictions (complaint by the United States) (WT/DS90), and India – Quantitative Restrictions on Imports of Agricultural, Textile and Industrial Products (complaint by the European Communities) (WT/DS96). The agreed period for implementation of the DSB recommendations in the case concerning the United States is contained in document WT/DS90/15, circulated on 17 January 2000 (US Exhibit – 9). The mutually agreed solution reached between India and the European Communities in their dispute is contained in document WT/DS96/8, circulated on 6 May 1998 (European Communities Exhibit – 8).

³¹¹ See paras. 4.73 to 4.80 above for a more complete description of India's arguments.

See for example paras. 4.113 and 4.121 above.

See for instance the Panel Report on US - Wool Shirts and Blouses, WT/DS33/R, adopted on 23 May 1997, as upheld by the Appellate Body Report, para. 6.2 (DSR 1997:I, 343), where the measure was withdrawn following the issuance of the interim report, and the panel nonetheless issued a complete report. See also the Panel Report on *Indonesia – Autos* where the panel proceeded with its examination of the claims despite

- 7.27 In any event, the complainants dispute India's contentions about the ramifications of the events of 1 April 2001. The complainants have clearly indicated that they are requesting the Panel to find that the measures were inconsistent with the cited provisions as of the date of this Panel's establishment and that they have *remained* so subsequently. The complainants consider that the events of 1 April 2001 have no effect on their legal claims, because these do not rest on the continued existence of import licensing after 1 April 2001.
- 7.28 While India has indicated that Public Notice No. 60 has ceased to apply as of 1 April 2001, it is not disputed that the MOUs signed in accordance with its terms did not *ipso facto* cease to exist at the same time. India has in particular confirmed that "it is not the intention of India to release the car manufacturers from the commitments they have assumed under the MOUs as a result of imports of SKD/CKD kits prior to 1 April 2001"³¹⁵ and that signatories were still required to fulfill the indigenization condition, although India indicated in the later stages of the proceedings that as a matter of fact, all but one of the signatories had achieved the required level of indigenization. Therefore, while the framework governmental measure which initially led to the signature of the MOUs may have ceased to "operate", the MOUs themselves have remained in existence. Indeed, the practical implications of this dispute rest primarily on their *continued* existence after 1 April 2001.
- 7.29 In light of the foregoing, the Panel is of the view that it is not prevented by subsequent events which impacted upon the measures before it, from examining these measures as they stood at the date of establishment of the Panel.
- 7.30 However, the Panel also recognizes that it may need to consider whether these developments affect the continued relevance in the post-April 2001 period, of any findings and conclusions it might make as to the consistency of the measures as at the date of its establishment. Article 19.1 requires a panel that has made a finding that a measure is in violation of a covered Agreement, to recommend that the Member concerned bring the measure into compliance with that Agreement. The Panel's ability to make meaningful recommendations to the DSB under Article 19.1 of the DSU may be affected if any changes in the measures had occurred which would affect the continued existence of any violations that might be identified as at the date of establishment. However, the Panel does not believe that these issues relate to its initial competence to examine the measures as they stood as of that date, but rather, to the nature and scope of any recommendations it might make. It will therefore address these aspects more fully at a later stage in the context of its conclusions and recommendations. ³¹⁶

a notification in the course of the proceedings by the respondent that the programme in issue had expired: "(...) In any event, taking into account our terms of reference, and noting that any revocation of a challenged measure could be relevant to the implementation stage of the dispute settlement process, we consider that is appropriate for us to make findings in respect of the National Car Programme. In this connection, we note that in previous GATT/WTO cases, where a measure included in the terms of reference was otherwise terminated or amended after the commencement of the panel proceedings, panels have nevertheless made findings in respect of such a measures" (WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, para. 14.9, DSR 1998:VI, 2201). As mentioned by that panel, there have also been such instances of continued proceedings despite expiry or partial disappearance of the measures at issue under the GATT: see for instance *EEC -Apples I (Chile)* (BISD 27S/98) paras 2.2 and 2.4; *United States - Prohibition of Imports of Tuna and Tuna Products from Canada* (BISD 29S/91), paras. 2.8, 4.2 and 4.3, where despite some evolution in the measures in the course of the proceedings and encouragement from the Panel to reach a mutually agreed solution, there was no agreement among the parties that such a solution had been found and the panel issued a complete report.

³¹⁴ See paras. 3.1, 3.5 and 4.92 to 4.94 above.

³¹⁵ First Submission, para. 11

³¹⁶ See Section VIII:B below.

3. Measures not within our terms of reference

- 7.31 India has also raised a distinct argument to the effect that any future measures which India may take after 1 April cannot be within the terms of reference of this Panel. This argument requires a separate analysis.
- India indicated in its First Submission that it was unclear from the complainants' submissions whether they were challenging measures that India was applying as of the date of their requests for establishment of a panel, or measures that India "might" take in the future.
- 7.33 India argued that a valid complaint required that "both the measures at issue before a panel and the legal issues to which they give rise must have existed at the time of the request for establishment of a panel". It indicated that "the measures which India will apply as from 1 April 2001 to comply with the mutually agreed solution negotiated with the European Communities and the DSB ruling sought by the United States fall outside the Panel's terms of reference". India highlighted that "[a]lthough the European Communities and the United States allege that India is currently violating its obligations under the GATT and the TRIMs Agreement, many of their arguments relate to measures India might take after 1 April 2001 and legal issues that might consequently arise at that time." 317 India argued that any future enforcement measures which India might take after 1 April to enforce the MOUs are not within the terms of reference of this Panel, because "[u]nder the DSU, Members may not bring complaints on proposed measures nor may they request advisory rulings on potential future issues". The also argued that "... these complaints - should events substantiate them - should appropriately be made in a dispute settlement proceeding initiated after 1 April 2001."319
- 7.34 The Panel agrees with India's contention that hypothetical future measures which it could, but might not take, including any future action that India might take for the enforcement of Public Notice No. 60 or the MOUs, would not fall within the scope of this Panel's terms of reference.
- 7.35 The complainants subsequently clarified that they were not seeking rulings on any measures which India might take after 1 April 2001 for the future enforcement of the conditions being challenged. Their claims concerned the measures that India was applying at the date of establishment of the Panel and would continue to apply subsequently.³²⁰

³¹⁷ First Submission, para. 20 (emphases in original).

³¹⁸ First Submission, para. 17. In India's view,

[&]quot;although the European Communities and the United States allege that India is currently violating its obligations under the GATT and the TRIMs Agreement, many of their arguments relate to measures that India might take after 1 April 2001 and legal issues that might consequently arise at that time. Both submissions often leave the reader confused as to whether the legal arguments relate to the factual situation before or after the elimination of the discretionary licensing regime for SKD/CKD kits." First Submission, para 26.

³²⁰ The European Communities thus explained that it was

[&]quot;not asking the Panel to rule that India would infringe the GATT and the TRIMs Agreement if 'India were to do this or do that' in order to enforce the MOUs. The European Communities claims that already the existence of binding and enforceable MOUs is as such inconsistent with the GATT and the TRIMs Agreement, regardless of whether any specific action needs to be taken. (...) 'Finally, the European Communities agrees with India that the Panel 'must base its rulings on inconsistencies that had already occurred at the time of the request for the [establishment of the Panel]. However, the continued existence and enforceability of the MOUs after 1 April 2001 does not result from any additional measure taken by the Indian Government after that request, but exclusively from the terms of the MOUs concluded by the

- 7.36 Further to the complainants' clarifications, India indicated in its second submission that
 - "(...) the European Communities and the United States clarified that their complaints relate to the measures applied by India at the time when they submitted their requests for the establishment to the DSB, and not to any measures that India may take in the future. With these clarifications as to the scope of the rulings requested from this Panel, two central arguments that India had advanced in its First Submission are no longer relevant. India's argument that no WTO Member may resort to the DSU on the same matter twice is no longer relevant because the European Communities and the United States have made clear that their case does not rest on the application of the discretionary licensing system that had been the subject of their prior complaints. India's argument that the Panel is not competent to rule on measures that India may take after 1 April 2001 is also no longer relevant." (emphasis added). 321
- 7.37 The Panel will thus not be considering any future measures which India might take after 1 April 2001.
- C. THE PANEL'S COMPETENCE TO CONSIDER THE MATTER BEFORE IT
- 7.38 The Panel recalls its conclusion that both complainants have specifically identified two conditions under Public Notice No. 60 and the MOUs signed thereunder in their requests for establishment of a panel in this case: the "indigenization" and "trade balancing" conditions. It is against these particular conditions that their claims are directed. The Panel has also concluded that these measures are expressly within the Panel's terms of reference and thus within its jurisdiction in accordance with the terms of Articles 6.2 and 7 of the DSU. As explained in paragraph 7.32, the claims should be considered as of the date of establishment of the Panel, as was requested by both complainants and as was recognized to be appropriate by India. 322
- 7.39 India has not disputed that Public Notice No. 60 and the MOUs, including the trade balancing and indigenization conditions thereunder, are properly within the Panel's terms of reference. However, it has raised a number of additional arguments to those addressed above, to the effect that this Panel still cannot validly examine these claims. The Panel's observations as to its express terms of reference were without prejudice to the consideration of these arguments of India.
- 7.40 India has expressly raised three distinct challenges in this respect:
 - (a) India argued that certain claims have already been decided on between it and the United States through the rulings adopted by the DSB in the *India Quantitative*

Indian Government before that date. Thus, the inconsistencies claimed by the European Communities existed from the moment of the conclusion of the MOUs."

The United States, for its part, clarified that:

"the US complaint is not directed at any particular method or means of enforcing the MOUs and Public Notice No. 60. The US complaint is instead directed to the indigenization and trade balancing requirements as such, and these requirements have not changed. These requirements have imposed the same obligations on car manufacturers since the date that the MOUs were signed, and it is the US position that those obligations are themselves inconsistent with India's WTO commitments. What matters is that those obligations are binding and enforceable; what means India chose or will choose to enforce them is simply not relevant to the US legal claims in this dispute."

³²¹ Second Submission, para. 2.

³²² First Submission, para. 19.

- *Restrictions* dispute, and that the principle of *res judicata* applies to prevent the relitigation of these claims;
- (b) It also asserts that these claims have already been resolved between India and the European Communities through the notification of a mutually agreed solution (hereafter "MAS"), and that the terms of the MAS bind the European Communities and this Panel:
- (c) India also argued in its First Submission that the claimants may not raise new legal claims with respect to a measure that must be eliminated in any case as a result of the abovementioned DSB ruling and MAS. It supports this proposition in two ways, by alluding to a suggested principle of "abusive splitting" in the context of different legal claims brought at different stages in relation to the same measures, and by separately arguing that it would be needless to consider new grounds to determine the validity of measures that are to be brought into compliance in any event.
- 7.41 The Panel addresses each of these arguments in turn in order to ascertain its competence to examine the claims before proceeding further.

1. Are any of the United States' claims barred by the principle of res judicata?

- 7.42 In India's view, the United States' initial submission suggested that it was attempting to obtain a new ruling on a matter which had already been decided on between India and the United States through the recommendations of the DSB in their *India Quantitative Restrictions* dispute, namely, the inconsistency of India's discretionary import licensing system under its Export Import Policy of 1997-2002 (hereafter "EXIM Policy") with Article XI of GATT 1994.
- 7.43 India's reliance on the principle of *res judicata* evolved in the course of the proceedings. This has coincided with its evolving view as to the nature and ambit of the United States claims as they were further clarified. India's reliance on the principle was conditional upon the way the claims were clarified and upon the way the Panel would choose to view the measures before it. The Panel must therefore first clarify whether the conditions on which India considered *res judicata* to be raised as a defense are met, to determine whether there is a need to proceed further with the examination of this argument.
- (a) Arguments of the parties and scope of India's invocation of the doctrine of res judicata
- 7.44 As noted above, India argued in its First Submission that the matter before the Panel was *res judicata* with respect to at least some of the United States claims. India argued that *res judicata* is a firmly established principle in international law which must be considered to constitute an inherent part of the WTO dispute settlement process. ³²³ In India's view, the principle prevents the resubmission of the same dispute between the same parties to a new panel. It considered that the United States had already obtained a ruling that "the discretionary licensing scheme administered by the Ministry of Commerce is inconsistent with Article XI" in the *India Quantitative Restrictions* ruling. In India's view, that ruling also covers the "operation and application of the discretionary licensing system in respect of all products that were notified under Article XVIII:B", including the products of which CKD/SKD kits are composed.
- 7.45 The United States, on the contrary, considered that India has not demonstrated the relevance of the principle of *res judicata* to WTO dispute settlement. It argued that in any case, the measures

³²³ See First Submission, para. 40.

and claims in this dispute are different from those at issue in *India – Quantitative Restrictions*. Consequently, it asserted that the *India – Quantitative Restrictions* panel could not and did not address the matter brought before this Panel and that this Panel should not engage in an analysis of legal notions not called for by the circumstances of the case.

- 7.46 In its second written submission, India considered that through their clarifications, the claimants had confirmed that their assertion was essentially that the measures as identified in the terms of reference, namely, conditions contained in Public Notice No. 60 and the MOUs executed under Public Notice No. 60, impose requirements and restrictions within the meaning of Articles III:4 and XI:1 of the GATT and the corresponding provisions of the TRIMs Agreement independently of the application of the discretionary licensing scheme that India was obliged to eliminate as a result of the prior proceedings. Based on that perception of the claims, it indicated that it "assumed that the issue before the Panel is no longer the scope of the principle of *res judicata* but rather, the question of whether Public Notice No. 60 and the MOUs impose requirements and restrictions even in the absence of the licensing scheme for CKD/SKD kits" and confirmed in its second submission that its argument on *res judicata* was no longer relevant to the case. 325
- 7.47 Nevertheless, India also indicated in its second submission that in its view, the Panel was required to examine whether the measures would have been inconsistent with the covered Agreements "even if" there had been no import licensing. It later confirmed that if the Panel were to examine these measures "in the presence" of import licensing, then India was continuing to argue that the matter had been adjudicated upon already and could not be examined by the Panel. Thus by that stage, it seemed that India's invocation of the doctrine of *res judicata* was conditional, depending on whether the claimant was calling for its claims to be considered "in the presence of" import licensing or as "if there had been no import licensing in place" at the date of the Panel's establishment.
- 7.48 In order to determine whether the Panel needs to examine India's *res judicata* argument, it therefore needs to give further attention to the clarifications made by the United States as to the exact role of the import licensing system in its claims and determine whether the existence of import licensing at the time of establishment of the Panel is a factor to be taken into account in the Panel's analysis.
- (b) Role of India's import licensing system in the United States' claims
- 7.49 Further questions between the parties and from the Panel sought to resolve this issue. Paragraphs 7.78 to 7.80 below outline the clarification of the claims between the parties in order to accurately identify the matter before this Panel in the context of a *res judicata* analysis. For present purposes, it merely needs to be noted that those clarifications confirmed that the United States was asking for a ruling on the identified conditions in the context of India's discretionary licensing regime.
- 7.50 In response to a question by India, the United States indicated that it would not be a correct reflection of the facts for the Panel to assume in its legal analysis that there was no import licensing in place at the time of establishment of the Panel. It further clarified the role of import licensing in its claims as follows: "the limited relevance of these licenses is that India used them to induce car manufacturers in India to accept the indigenization and trade balancing requirements (...) What is

³²⁴ See response to Question 36 of the Panel.

³²⁵ See Second Submission para. 2.

³²⁶ See Second Submission para. 4.

³²⁷ See response to Question 109 of the Panel, where India indicated that "[t]he principle of *res judicata* would only be relevant if the Panel were to decide to examine the operation of Public Notice No. 60 and the trade balancing provisions in the MOUs *under the licensing scheme* that India was obliged to eliminate as a result of the complainants prior invocations of the DSU (emphasis added)".

relevant, therefore, is that import licensing existed in 1997 and 1998, when the MOUs were signed. (...)". 328 The United States thus confirmed that it was asking the Panel to rule on the conditions in Public Notice No. 60 and the MOUs in the context of that licensing regime, i.e. neither a ruling on the licensing regime per se nor a ruling "in abstraction" of the undisputed fact that import licensing existed for these products at the time of establishment of the Panel.³²

- In the Panel's view, the normal role of a panel under the DSU does not contemplate consideration of measures in isolation from their circumstances. To do so would be to rule on a hypothetical abstraction divorced from the circumstances of the particular matter that is within the Panel's terms of reference. In this case, import licensing was part of the factual circumstances present at the time of establishment of this Panel. The Panel is required to take this fact into account in order to make the objective assessment of the matter required by Article 11 of the DSU.
- The Panel therefore agrees that the United States' articulation correctly reflects the way it should approach the examination of the matter before it. The Panel is requested and required to examine these measures as they existed as of the date of establishment of the Panel in their actual context, which includes the existence of import licensing. The Panel therefore concludes that it is necessary to address India's res judicata argument.
- 7.53 While the Panel has concluded that the presence of import licensing is something that must be taken into account as a factual matter, the extent to which this will be relevant to its deliberations will depend upon the nature of the claims. In turn, this will impact upon the Panel's analysis of the relationship between this dispute and the *India – Quantitative Restrictions* dispute.
- General approach to the res judicata arguments (c)
- As a preliminary matter, the Panel first notes that the conclusions reached in the India -Quantitative Restrictions dispute - namely that the discretionary import licensing system applied by India to certain restricted products under the EXIM policy was inconsistent with Article XI - were not disputed in these proceedings. The issue before this Panel is rather whether the scope of the India – Quantitative Restrictions dispute and the rulings adopted in it can be such as to preclude any further litigation concerning the claims and measures in *these* proceedings between the same parties.
- 7.55 This issue has two aspects. It first raises a general question as to the applicability of the doctrine of res judicata in the WTO. Secondly, it raises a specific question as to whether the facts in this dispute and in the *India – Quantitative Restrictions* dispute are such as to satisfy the requirements of the doctrine, if it were applicable to WTO dispute settlement.
- India refers to a number of general provisions in the DSU highlighting the objective of settlement of disputes through the DSU, in support of its contention that res judicata may apply. India has referred to the jurisprudence of the International Court of Justice and to two specific provisions of the Court's statute concerning the status of its own decisions. India has also invoked res judicata as a general principle of law applicable in WTO dispute settlement. 330 The United States, however, does

³²⁸ See response to Question 9(a) of the Panel, para. 27.

The European Communities, although not directly concerned by the *res judicata* issue, confirmed a similar understanding of how the Panel was being asked to assess the claims. 330 In its responses to the second set of questions from the Panel, India also suggests that the

res judicata principle is "but one facet of the broader principle of judicial economy". It suggests that the Appellate Body's rulings to the effect that a panel need only address those claims which must be addressed in order to resolve the matter at issue should also apply when a panel is being asked to rule on a matter already resolved in prior proceedings (see response to Question 104 (b) from the Panel). The Panel is not convinced, however, that the application of the principle of res judicata could appropriately be reduced or equated to the

not believe that India has demonstrated the relevance of this doctrine to WTO dispute settlement or that it is necessary or desirable for the Panel to consider this systemic issue.

- 7.57 The general question as to the applicability of the doctrine of *res judicata* to WTO dispute settlement is of systemic importance. It does not appear to have been explicitly considered in WTO dispute settlement.³³¹ A general principle of *res judicata* has also not been otherwise referred to or endorsed by any WTO panel or by the Appellate Body, although it is certainly true that certain widely recognized principles of international law have been found to be applicable in WTO dispute settlement, particularly concerning fundamental procedural matters.³³²
- 7.58 The text of the DSU does not directly address this issue. More generally, the DSU does not contain any express provision concerning the status of adopted panel or Appellate Body reports, or concerning their potential impact in separate proceedings under the DSU concerning the same matter. Nevertheless, when considering the status of adopted panel reports, the Appellate Body has indicated that they are binding on the parties "with respect to that particular dispute". This statement was not directly addressed at the notion of *res judicata*, but rather, appears to have been made primarily in the context of determining the status of adopted reports *per se* and their relevance to subsequent panels as a matter of precedent. It is thus not necessarily determinative of the status of the doctrine of *res judicata* in the WTO as a bar to re-litigation of the same issues, although it may be open to debate whether this could form the basis, or a basis, for the application of the principle of *res judicata*.

guidance given to panels to address, within the matter in their jurisdiction, only those issues which are necessary to the resolution of the dispute at issue. The issue of *res judicata* involves an argument of quite a distinct nature concerning the very competence of the adjudicatory body to address the issue submitted to it at all. The exercise by panels of judicial economy as recalled by India is a totally different concept allowing panels to address the matter properly within their jurisdiction in the most efficient way to achieve the objectives of WTO dispute settlement.

settlement.

331 The doctrine does not appear to have been directly raised as a defense in any dispute to date, although it has been referred to by India before the panel in *India – Patents (EC)*, WT/DS79/R.

³³² See for instance *US – Shrimp*, Report of the Appellate Body, WT/DS58/AB/R, para. 158, on the principle of good faith as a general principle of law and general principle of international law, or *Indonesia - Autos*, Panel Report, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, para. 14.28 concerning the "presumption against conflict" in international law.

Two adopted panel reports under the WTO have considered claims in situations involving the establishment of "successive" panels concerning the same matter. However, this was in circumstances where the complainants were distinct or where the first panel was not pursued, so that in either case *res judicata* would not be directly relevant, in the first case for lack of identity of parties, and in the second instance for lack of a prior decision on the same issue between the parties.

The first of these cases was *India – Patents (EC)*, where the second panel, established shortly after a first report had been circulated concerning the same measures, but at the request of a different complainant, found that it was competent to examine in full the claims, taking account of the reasoning and conclusions of the previous panel. In another case, *Australia – Subsidies Provided to Producers and Exporters of Automotive Leather* (hereinafter "*Australia – Automotive Leather II*"), a second panel was established at the request of the same complainant, in circumstances where the composition of the panel established a short time before was not pursued and the first proceedings never led to any further developments. The responding party having challenged the validity of the establishment of the second panel in these circumstances, the panel found that Australia was asking this panel to read into the DSU an implicit prohibition on multiple panels between the same parties regarding the same matter that does not exist in the text of the DSU, noting that this was not a case where a complainant was actively pursuing two procedures with respect to the same matter, since the first panel was never composed and then never began its work. (Cf. Panel Report, WT/DS126/R, adopted on 16 June 1999, para. 9.14).

³³⁴ *Japan – Alcoholic Beverages II*, Report of the Appellate Body, WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, p. 14, DSR 1996:I, 97.

- 7.59 In the absence of any clear guidance, many important interpretative issues would thus need to be considered in determining whether the doctrine of *res judicata* applies in WTO dispute settlement in circumstances where the facts would support such an assertion. However, the potential relevance of the notion of *res judicata* to this case would only arise if its commonly understood conditions of application were met on the facts.
- 7.60 The Panel recalls in this regard that its mandate under the DSU requires it only to address those issues which need to be addressed in order to resolve the dispute. Thus the Panel finds it appropriate, in this instance, to first consider whether the factual circumstances for the application of res judicata could be met in the circumstances of this case. If the basis of this dispute is sufficiently similar to that of *India Quantitative Restrictions* so as to come within accepted notions of the doctrine, the Panel would need to rule on its applicability as a doctrine. Conversely, if the Panel were to find, as a matter of fact, that the matter ruled on by the *India Quantitative Restrictions* panel is distinct from that submitted to it in this dispute, it would not be necessary to make a general ruling on the role of res judicata in WTO dispute settlement.
- 7.61 To conduct such an analysis, the Panel must at least identify a benchmark by which disputes might be seen as distinct or similar for the purposes of rejecting or applying *res judicata*.
- 7.62 In international jurisdictions where it is applicable, the doctrine is generally understood to mean that an issue that has been decided on in a final adjudication, that is, after exhaustion of any available appeal rights, must be considered as a settled matter between the parties to the dispute. Consequently the issue previously resolved cannot be re-opened in subsequent proceedings. This doctrine has found application in the jurisprudence of the International Court of Justice, whose Statute contains express provisions concerning the binding and final character of its judgments. 336

³³⁵ While there may be variations in the exact understanding of the doctrine depending on the jurisdiction concerned or commentators' interpretation of its scope, it can be noted that *res judicata* is broadly understood to encompass three elements: a final decision, on a given issue, between the same parties. Black's Law Dictionary defines *res judicata* as follows:

[&]quot;A matter adjudged; a thing judicially acted upon or decided; a thing or matter settled by judgment. Rule that a final judgment rendered by a court of competent jurisdiction on the merits is conclusive as to the rights of the parties and their privies, and, as to them, constitutes an absolute bar to a subsequent action involving the same claim, demand or cause of action. (...) And to be applicable, requires identity in thing sued for as well as identity of cause of action, of persons and parties to action, and of quality in persons for or against whom claim is made. The sum and substance of the whole rule is that a matter once judicially decided is finally decided (...)".

³³⁶ In the context of the International Court of Justice, two provisions of its Statute, Articles 59 and 60, as referred to by India, are frequently cited as the source of the principle of *res judicata* concerning its own decisions, although some divergences appear to emerge among commentators as to the exact role of either of the two provisions or even the need to consider them as the source of this principle since it is arguably a general principle of law in any case (see Collier and Lowe, *The Settlement of Disputes in International Law. Institutions and Procedures*, Oxford University Press, p. 177).

In the words of Sir Gerald Fitzmaurice,

[&]quot;A judgment of the [International] Court [of Justice] has, internationally, the authority of res judicata, and this covers all matters which are actually the subject of decision in the judgment. Thus in the *Haya de la Torre* case, the Court (*I.C.J.*, 1951, 77) referred to

^{&#}x27;... questions which the Judgment of November 20th, 1950, had already decided with the authority of *res judicata*'

^(...) Unless, however, the point is duly covered by the previous decision, there is no res judicata, and therefore nothing (on that particular score) to prevent it being raised in later proceedings. Thus, again in the *Haya* case, the Court said (ibid, 80):

- 7.63 In this instance, while the United States was of the view that India had not demonstrated the relevance of this principle to WTO dispute settlement, the submissions did not display any fundamentally different view of either party as to the nature of the concept. On the other hand, the parties have generally provided limited guidance as to how they would expect a panel to assess the similarity of two cases to determine the relevance of *res judicata* in the context of WTO dispute settlement.
- 7.64 Because the policy underlying *res judicata* is to bring litigation of a particular nature to an end at an appropriate stage, the key to its application should be to compare what has already been ruled on to what is being brought before the adjudicating body in the subsequent proceedings. Both India and the United States have used a comparison between the "matter" ruled on in the *India Quantitative Restrictions* case and the "matter" brought before this Panel in identifying the similarities or dissimilarities between the two disputes for the purposes of assessing whether the issues before this Panel can be considered to be *res judicata*.³³⁸
- 7.65 In the context of WTO dispute settlement, the notion of "matter", as referred to in Article 7.1 of the DSU, determines the scope of what is submitted, and what can be ruled upon, by a panel. As confirmed by the Appellate Body in the *Guatemala Cement* case, the matter referred to the DSB consists of two elements: the specific measures at issue and the legal basis of the complaint (or the claims). This appears to the Panel to be the most appropriate minimal benchmark by which to assess whether the conditions of *res judicata* could conceivably be met, if such a notion was of relevance. 339
- 7.66 The Panel therefore considers that for *res judicata* to have any possible role in WTO dispute settlement, there should, at the very least, be in essence identity between the matter previously ruled on and that submitted to the subsequent panel. This requires identity between both the measures and the claims pertaining to them. There is also, for the purposes of *res judicata*, a requirement of identity of parties which is clearly met with regard to the United States in this instance.

'As mentioned above, the question of the surrender of the refugee was not decided by the judgment of November 20th. This question is new; it was raised by Peru in its Note to Colombia of November 28th, 1950, and was submitted to the Court by the Application of Colombia of December 13th, 1950. There is consequently no *res judicata* upon the question of surrender.'"

(in The Law and Procedure of the International Court of Justice, Sir Gerald Fitzmaurice, Volume Two, Cambridge, Grotius Publications Limited, 1986, pp. 584-585).

The United States thus indicated that "[g]enerally speaking, a principle of *res judicata*, if applicable, would presumably relate to the effect of a previously adopted panel report on a subsequent dispute involving the same matter between the same parties".

The European Communities, which is not directly concerned by this issue of *res judicata*, has also used this benchmark in explaining the contours of the notion and its relevance to this case.

This might only be a minimal benchmark. Where respondents have made allegations as to a lack of specificity of the matter articulated in panel requests, some cases have upheld very general claims. The more broadly claims are stated in two distinct cases, the more it might superficially appear that the issues are similar. Nevertheless, the policy behind such considerations in cases challenging specificity under Article 6.2 of the DSU differs from the policy behind the *res judicata* concept. Thus to finally determine the applicability of the *res judicata* concept to the facts of a particular case, a tribunal may need to identify a more detailed articulation of the claims in each case and might even need to integrate this with a consideration of which arguments were addressed in each dispute. Whether this is so or not, it remains the case that if even the "matters" are not the same, the doctrine could not apply. The Panel will therefore proceed with an examination of the scope of each dispute through a comparison between the matter before this Panel and the matter ruled on by the *India - Quantitative Restrictions* panel. If they are the same, the Panel would then need to consider this subsidiary question of whether further refinement of the notion is necessary.

(i) The matter before this Panel

7.67 While the measures at issue and the legal basis of the complaint are identified in the United States request for establishment of a panel, it was perhaps not fully apparent from its terms what the precise role of India's discretionary import licensing scheme was intended to be in the matter submitted to the Panel. The Panel has already resolved that this dispute cannot be conducted on a hypothetical basis as if the licensing scheme did not exist. The Panel believes it is appropriate, in considering the potential application of *res judicata*, to consider in more detail the subsequent elaboration by the United States of its claims, in order to clarify the exact scope of the complaint identified in our terms of reference. The panel basis are identified in our terms of reference.

7.68 In its First Submission, the United States explained that:

"The present dispute (...) concerns not the import licenses themselves; maintaining them beyond April 1 would in any case be inconsistent with India's existing obligation to remove them. Instead, this dispute concerns discriminatory, traderestricting conditions that India exacts from investors in the motor vehicle manufacturing sector and that it intends to continue to exact.

The United States contends that the local content and trade balancing requirements in Public Notice No. 60 and the MOUs, together with the Indian domestic legislation under which they have come into force, are inconsistent with the obligations of India under the General Agreement on Tariffs and Trade 1994 ("GATT 1994") and the Agreement on Trade-Related Investment Measures ("TRIMs Agreement"). The United States respectfully requests this Panel to make findings to this effect, and to recommend that India bring all such measures into conformity with its obligations."

7.69 This explanation made it clear that the United States did not seek to challenge the import licensing regime or its application to the products at issue, as such. Nevertheless, it referred in a number of its arguments to the import licensing system and to the possible denial of import licenses as a consequence of failing to comply with the requirements at issue.³⁴² In view of these arguments of the United States and India's assertions as to the nature of this claim, the Panel sought further clarification as to the scope of this dispute and the exact extent to which the claimants saw the licensing scheme as relevant to these proceedings, but still distinct from the issues considered in *India – Quantitative Restrictions*.

7.70 Subsequent exchanges of submissions, as well as answers to questions from the Panel, further confirmed that the United States in this case was not seeking a ruling from this Panel on the consistency with the WTO agreements of the import licensing regime *per se*. Instead, its claims related to the "indigenization and trade balancing requirements" in Public Notice No. 60 and in the

³⁴⁰ Thus it was not necessarily clear on the face of the document that the measures were alleged to be in violation *per se*, or because they were enforceable, or because failure to comply would lead to the refusal of import licences. The use of the term "therefore" after such statements in the sentence which articulates the provisions in dispute does not make clear whether these are distinct claims or whether each feature is a necessary element of the alleged violation.

necessary element of the alleged violation.

341 In the view of the Panel, such recourse to subsequent clarifications of the claims is appropriate in order to determine exactly the scope of the matter that the complainant intended to refer to the Panel. This is with the clear understanding that the full scope of the matter referred to the panel is determined by its terms of reference, and that subsequent elaborations can only serve to clarify the meaning of a matter already within such terms of reference. See, in this respect the Report of the Appellate Body in Guatemala – Cement I.

³⁴² See for example paragraph 101 of the First US Submission, which refers to the possible denial of import licenses as a source of restriction on importation.

MOUs. 343 It has requested the Panel to find that these requirements "are inconsistent with Article III:4 and XI:1 of the GATT 1994 and Articles 2.1 and 2.2 of the TRIMs Agreement." Nevertheless, as noted above, the United States considered that the Panel must address these measures in the context of the discretionary licensing scheme that applied at the time of the Panel's establishment, and that this licensing scheme constituted an advantage used to induce the manufacturers to sign the MOUs. It contended that the measures at issue in these proceedings were in violation of those provisions, independently of the violation of the licensing scheme *per se* and requests the Panel to recommend that India bring these measures into conformity with its obligations under the GATT 1994 and the TRIMs Agreement. 344

(ii) The matter ruled on by the India – Quantitative Restrictions panel

- 7.71 The Panel is mindful that in seeking to consider the ambit of the *India Quantitative Restrictions* case, it is only analysing the nature and scope of that case and its resultant rulings and recommendations to consider the relevance of India's jurisdictional arguments. This analysis is without prejudice to any determination an Article 21.5 compliance panel in the context of the *India Quantitative Restrictions* case might make as to the scope of that case and its rulings and their potential impact on India's subsequent behaviour.
- 7.72 A consideration of the matter resolved before the *India Quantitative Restrictions* panel should also begin with an examination of its terms of reference, which in turn, incorporate the United States request for establishment of a panel in that case. That request provided that:

However, we also recall that it is normal for claims to be clarified in the course of proceedings. While the clarification process might have been more lengthy in this case than would normally be expected, we also find that the United States has sought to clarify its claims in good faith in a manner that enabled India to properly understand the nature of the claims before this Panel. These claims remain within the Panel's terms of reference. The Panel also feels that India's ability to defend itself was not jeopardised, in particular because at all stages of the proceedings it was accorded ample opportunity to develop its arguments in light of the complainants' clarifications.

³⁴³ The United States thus further clarified that "[t]he measures at issue in this dispute are those set forth in the US panel request: (...). The US claims relate to the indigenization requirement and the trade balancing requirement found in these measures" and that the limited relevance of the licenses is that India used them to induce manufacturers to sign the MOUs and they constitute the advantage gained in signing the MOUs.

In its Second Submission, the United States further indicated:

[&]quot;it is not the licensing requirements but the indigenization and trade balancing requirements that are the subject of the US complaint in this dispute. Because the indigenization and trade balancing requirements did not change on April 1, 2001 (or on any other date), and because those requirements are independent of India's now-eliminated licensing regime, India's elimination of import licensing is not relevant to - and certainly does not resolve - this dispute."

In its final set of responses to the Panel's questions, India argued that while it was legitimate for complainants to narrow the focus of their claims in the course of the proceedings, they cannot legitimately successively narrow or broaden them at will as India alleges they did here (see response to Question 102 of the Panel). This is a distinct argument from the *res judicata* claim and is distinct from any assertion that the claim as finally relied upon is outside the terms of reference.

The Panel agrees with India that parties in a dispute should avoid successive shifts or modifications in the presentation of their arguments. In some circumstances this could jeopardise the ability of the other parties to present their arguments in sufficient time and prejudice the fulfilment of requirements of due process. In this case we believe that the successive exchanges between the United States and India constituted a clarification of the United States claim rather than a shifting back and forth as alleged, although the Panel notes that the United States' articulation of the relevance of the licensing scheme to its claims in its first written submission was at the very least confusing (as reflected for instance in paragraphs 101 and 103 of that submission).

"quantitative restrictions maintained by India, including, but not limited to, the more than 2700 agricultural and industrial product tariff lines notified to the WTO in Annex I, Part B of WT/BOP/N/24 dated 22 May 1997, appear to be inconsistent with India's obligations under Articles XI:1 and XVIII:11 of GATT 1994 and Article 4.2 of the Agreement on Agriculture. Furthermore, the import licensing procedures and practices of the Government of India are inconsistent with fundamental WTO requirements as provided in Article XIII of GATT 1994 and Article 3 of the Agreement on Import Licensing Procedures.

The quantitative restrictions referred to above include all import prohibitions, bans, restrictions, import licenses, special import licenses and the prohibition of non-commercial (sample) quantities as well as the procedures to implement and administer these measures".³⁴⁵

7.73 The factual section of the *India – Quantitative Restrictions* panel report indicates that "(a)t the time the panel was established, India maintained quantitative restrictions on imports of products falling in 2,714 tariff lines at the eight-digit level of HS96 for which it claimed balance-of-payments justification. These restrictions had been notified to the BOPs Committee in May 1997 in the course of consultations being held with India. The restrictions that are within the scope of the dispute appear in Annex 1, Part B of WT/BOP/N/24".

7.74 During the *India – Quantitative Restrictions* proceedings, the United States requested the panel to find that "the quantitative restrictions at issue violate Article XI:1 and XVIII:11 of the GATT 1994 and Article 4.2 of the Agriculture Agreement". The United States identified four specific measures that it asked the panel to rule upon.

7.75 The panel then articulated the measures at issue in that case as follows:

"India regulates the importation of goods found in a 'Negative List of Imports', contained in Chapter 15 of India's Export and Import Policy 1997-2002. The United States identifies four measures which are implemented under India's Export and Import Policy and which it claims constitute quantitative restrictions within the meaning of Article XI:1: (a) discretionary import licensing; (b) canalization of imports through government agencies; (c) the Special Import Licensing (SIL) system; (d) the 'Actual User' condition on import licensing. (...) To the extent that India applies these four measures as balance-of-payments restrictions on the products specified in Annex I, Part B of WT/BOP/N/24, we refer to these measures as the 'measures at issue' in this dispute."

7.76 The measures which the panel ruled on were therefore those specific measures, to the extent that India applied them as balance-of-payments restrictions on the products specified in Annex 1 part B of its notification WT/BOP/N/24. In stating its conclusions, the panel referred to the "measures at issue", as described in this paragraph. 348

³⁴⁵ WT/DS90/8.

 $^{^{346}}$ Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, adopted on 22 September 1999, as upheld by the Appellate Body Report, para. 5.122.

heading "8703", at the 8 digit level, which relate to automotive vehicles. These are marked as restricted on grounds of "AUTO/BOP-XVIII-B", and under the QR symbol "NAL", "AL/NAL" or "SIL/NAL". The reference "AUTO/BOP-XVIII-B" is explained in a note as follows: "When imports of passenger cars and automotive vehicles are permitted without a license on fulfilment of conditions specified in a Public Notice

- 7.77 The report further explains that imports of goods are regulated in the EXIM policy through the Negative List and explains the general rules applicable to the issuance of import licenses by the DGFT. In this context, none of the specific criteria for various products are mentioned, other than the "Actual user" condition, which required that only the "actual users" of the product could obtain an import license. With regard to the "Special Import Licensing" (SIL), which applied to a limited number of goods identified with the symbol SIL in the notification, it is explained that these licenses were granted on the basis of export performance criteria.
- 7.78 The panel went on to analyse the four measures it had identified in the light of Article XI of GATT 1994. In respect of India's import licensing system for products on the "Negative List of Imports", the panel considered the United States claim that "imports of these products into India are subject to an arbitrary, non-transparent and discretionary import licensing system, under which licenses are granted "on merit" and only to a category of operators called "actual users". The panel noted that it was agreed that India's licensing system for goods in the Negative List of Imports is a discretionary import licensing system, in that licenses are not granted in all cases, but rather on unspecified "merits". The panel concluded that this licensing system, to the extent that it applied to the products specified in WT/BOP/N/24, Annex 1, Part B, operated as a restriction on imports within the meaning of Article XI:1.
- 7.79 The Actual User condition and SIL, having been identified as distinct and specific measures in the request, were examined separately and were the basis of distinct findings of violation of Article XI.
- (iii) Comparison between the matter ruled on by the India QRs panel and the matter before this Panel
- 7.80 As noted above, the notion of "matter" in the context of the DSU encompasses both the specific measures at issue and the claims, that is, the legal basis on which the complaint is made with respect to those measures. If the application of *res judicata* at the very least requires that the same matter which has already been adjudicated on be the object of a new complaint, then it seems to

issued in this behalf, and restrictions on imports through NAL are otherwise maintained". "SIL/AUTO/BOP" is defined as "when imports of passenger cars and automotive vehicles are permitted without a license on fulfilment of conditions specified in a Public Notice issued in this behalf, and imports are otherwise permitted against Special Import Licenses." See also responses to Questions 48 and 49 of the Panel.

In response to a question by this Panel, India confirmed that the following tariff lines in Chapter 87 notified as restricted for Auto/BOP reasons cover cars in the form of CKD/SKD kits: 870321.04, 870322.04, 870323.04, 870331.04, 870332.04, 870333.03 (sic). In addition a number of tariff lines covering components restricted for balance-of-payments reasons as considered in *India – Quantitative Restrictions* were also subject to the MOU privileges (870600.01, 870600.09, 870710.01 and 870710.02). All of these tariff lines were identified in India's BOP notification as restricted under "AUTO/BOP" grounds and subject to non-automatic licensing ("NAL").

licensing ("NAL").

348 With regard to the legal basis for import restrictions and import licensing, the descriptive part of the report notes the following factual aspects. "Indian domestic legislation governing import licensing can be found in: (i) section 11 of the Customs Act 1962, (ii) the Foreign Trade (Development and Regulation) Act, 1992, (iii) the rules and orders promulgated under the Foreign Trade (Development and Regulation) Act, 1992, and (iv) the Export and Import Policy 1997-2002." It is further explained that the Customs Act and the FT(DR) Act foresee the possibility for the Indian Government to regulate imports through rules and orders. It is further explained that the FTDR Act authorizes the Central Government to formulate and announce by notification in the Official Gazette the export and import policy, which is done on a five yearly basis, the current EXIM being the 1997-2002 policy, which "includes, *inter alia*, the Negative List of Imports ("Negative List") found in Chapter 15 of the Export and Import Policy. The list sets forth various prescribed procedures or conditions for imports, and the eligibility requirements including export performance that must be met to qualify for Special Import Licenses".

require as a minimum, that both the measures and the legal basis of the claims be essentially the same. The Panel will consider both aspects in turn, although there may be some artificiality in separating the analysis. This is because it may be difficult to know how to describe and characterise the measures without understanding the nature of the claims and vice versa. For this reason the Panel will first analyse each separately and then consider them together in drawing its conclusions.

- 7.81 India, in its second submission, seemed to concede that if the complaint concerned the consistency with the WTO Agreements of these specific conditions under Public Notice No. 60 and the MOUs rather than the consistency of the application of discretionary import licensing to the products at issue *per se*, then its *res judicata* argument was no longer relevant. Nonetheless, it later indicated that any examination of these conditions as they operated *in the presence of* import licensing was also covered by *res judicata*. This suggests that India considers that even if the application of the discretionary import licensing system to restricted kits and components is not the direct object of this claim, *res judicata* would apply to preclude the litigation of this claim concerning any administration or application measure of the import licensing regime considered in the *India Quantitative Restrictions* ruling. This seems to imply that even an "indirect" coverage of these measures would, in the view of India, make them *res judicata*.
- 7.82 The Panel would only need to rule on the role of indirect coverage in the doctrine of *res judicata* if the facts supported India's contention. The Panel will therefore consider this factual question also. The Panel will first compare the measures submitted to it with those expressly ruled on in the context of the *India Quantitative Restrictions* dispute, before comparing the express claims in both disputes. The Panel will then assess overall whether the matters ruled on in *India Quantitative Restrictions* can be considered to include, directly or indirectly, the matter it is requested to rule on in these proceedings.

A comparison of the measures expressly considered

- 7.83 The Panel has determined that the specific measures in dispute before it are the indigenization and trade balancing conditions contained in Public Notice No. 60 and the MOUs entered into under it. Although the products at issue in this case fell within the scope of restricted items within the meaning of the EXIM policy at issue in the *India Quantitative Restrictions* case, it is clear from an analysis of the terms of reference and panel report in *India Quantitative Restrictions* that *these* specific conditions were not expressly considered by that panel.
- 7.84 In any event, neither Public Notice No. 60 nor the MOUs signed thereunder could have been expressly within the terms of reference of the *India Quantitative Restrictions* panel, since they were only adopted after its establishment.³⁴⁹ It would only have been possible to expressly refer to measures that preceded Public Notice No. 60 in respect of these products. India has explained that prior to the existence of Public Notice No. 60, a number of MOUs had been signed with automotive manufacturers that called for some indications of indigenization and export intent.³⁵⁰ These, however, were also not expressly referred to or considered by the *India Quantitative Restrictions* panel.
- 7.85 In fact that panel never evaluated, or made any ruling on, any element of the administration of the discretionary licensing system, other than an Actual User condition provided for in the EXIM policy. This was specifically identified by the complainant in the *India Quantitative Restrictions* case and was the object of a distinct ruling of violation of Article XI. The parties' arguments in that case further document that none of the specific criteria or conditions for the granting of non-automatic

³⁴⁹ The *India – Quantitative Restrictions* panel was established on 18 November 1997. Public Notice No. 60 was adopted on 12 December 1997.

³⁵⁰ See India's First Submission, para. 4 and its responses to Questions 43 and 110 of the Panel.

licenses were otherwise considered or directly ruled on. By contrast, the "Special Import Licensing" system, or "SIL", was analyzed in light of its export obligation component.

7.86 The Panel therefore concludes that the measures before it were not expressly considered by the *India – Quantitative Restrictions* panel. Nor did it expressly consider the earlier MOUs applicable to restricted kits and components at the time of establishment of that panel. The panel based its finding of violation in relation to the discretionary import licensing on the more general observation that licenses were granted on the basis of "unspecified 'merits". India's BOPs notification, which served as the basis for the definition of the measures before that panel, included some general indication of the type of measure that could apply to specific products, but not in any detail, including for the products at issue in this proceeding. However, the *India – Quantitative Restrictions* panel, more generally, did not consider directly any of the specific application measures or criteria relating to specific product categories subjected to discretionary non-automatic import licensing.

A comparison of the express claims

- 7.87 India does not appear to argue that *res judicata* applies to the claims under Articles III:4 of GATT and 2 of the TRIMs Agreement.³⁵¹ These claims could not constitute the "same" claims as those examined by the previous panel, which did not address either of these provisions. The issue therefore only arises with respect to the claims based on Article XI of GATT.
- 7.88 Leaving aside the references to provisions other than Article XI:1, in a minimal sense at least, the legal bases for the claims have some superficial similarity: a violation of Article XI:1 is claimed in both cases. Nevertheless, to argue that the claims are the same merely because the same provision is in issue would be a strained usage of the notion of claim, at least in the context of an analysis of the potential application of *res judicata*.
- 7.89 The Panel believes that in considering the factual potential for a *res judicata* defense, it must go beyond a mere identification of the provision claimed to have been violated.³⁵² It must at the very least accurately identify the precise legal basis for the claimed violation in relation to the measure at issue.
- 7.90 As was noted above, the consistency with Article XI:1 of the trade balancing and indigenization conditions was not expressly examined by the *India Quantitative Restrictions* panel. They were not expressly examined in any context. The *India Quantitative Restrictions* panel addressed the claim that India's discretionary licensing system under the EXIM policy was

³⁵¹ In its First Submission, India argued exclusively in its section on *res judicata* in relation to a possible "re-examination" of claims under Article XI (see for instance paragraphs 41 and 43). On the other hand, it made distinct arguments in a subsequent section concerning the reasons why this Panel should not, in its view, be examining the claims based on other legal bases invoked by the complainants in this case (see paragraphs 45 to 49). Despite a specific question from the Panel at a later stage inviting India to clarify whether it was only invoking the principle of *res judicata* in respect of violations of Article XI or was also invoking it with respect to claims of violation of Article III:4 of GATT 1994 or the TRIMS Agreement, it declined to answer this point expressly (see response to Question 109 of the Panel).

³⁵² The Panel is mindful of the fact that in the context of procedural challenges as to the specificity of

The Panel is mindful of the fact that in the context of procedural challenges as to the specificity of claims, the Appellate Body has indicated that a mere identification of the provision of the covered Agreement at issue may be sufficient for the purposes of identifying the matter within the meaning of Article 6.2 of the DSU. This was in the context of determining whether the nature of the claim was sufficiently identified to put the defendant on notice of the challenge faced. Even in this context the jurisprudence makes it clear that this assessment should be made on a case by case basis. This Panel is comparing the claims in the cases for a different reason and should look to see if there has previously been a final determination of a specific issue that is now before it.

inconsistent with Article XI. The United States in this case is claiming, *inter alia*, that the trade balancing and indigenization conditions subsequently introduced through Public Notice No. 60, in and of themselves, are distinctly in violation of Article XI. In the Panel's view, this is a distinct question. While the *India – Quantitative Restrictions* panel considered whether it was consistent with the terms of Article XI for a member to apply discretionary or non-automatic licensing to certain products, this panel is being asked to consider whether two specific obligations are consistent with the same provision. The Panel concludes that the *India – Quantitative Restrictions* panel did not rule on the specific claims before this Panel.

Are the matters the same?

- 7.91 The Panel has identified that neither the specific measures in this case, nor any comparable measures in existence at the time of that panel's establishment, were expressly considered in the *India Quantitative Restrictions* case. The ruling on India's discretionary import licensing scheme in that case was in the context of a very broad-ranging challenge relating to measures taken by India on products covering over 2,700 tariff lines in the context of India's general EXIM policy, which included quantitative restrictions instituted for balance-of-payments purposes. The ruling on discretionary import licensing was accordingly broad-ranging and sought to address the existence of discretionary licensing rather than any of the specific conditions foreseen for the granting of any individual licenses. Where specific conditions were identified within the policy itself, such as the Actual User condition, or the export requirements under the "Special Import Licenses" (SIL), these were specifically addressed. Under the ruling concerning discretionary import licensing, no specific conditions were even identified, and certainly, the consistency of specific "trade balancing" and "indigenization" conditions was never considered.
- 7.92 As was noted above, India, in its second submission, considered that even if the application of the discretionary import licensing system to restricted kits and components was not the direct object of the claim, any examination of these conditions as they operated *in the presence of* import licensing was also covered by *res judicata*.
- 7.93 In India's view, such claims would be covered because the *India Quantitative Restrictions* ruling "covers the operation and application of the discretionary licensing system in respect of all products that were notified under Article XVIII:B".³⁵³ In India's view, the specific measures now referred to this Panel are an "inherent part" of the licensing scheme that India is required to eliminate as a result of the previous ruling.³⁵⁴ The Panel previously noted that it would look to see if the facts supported this contention.
- 7.94 It might be expected that a ruling that a particular broad-ranging scheme is in violation of provisions of the covered Agreements and must be brought into conformity may have implications for a number of administrative features of the system. This is not determinative however. In the context of a *res judicata* analysis, the question to determine is whether a particular issue was ruled on and decided upon, and not merely whether the implementation of a previous ruling may have practical implications for particular measures in the later dispute.
- 7.95 Furthermore, it cannot be assumed that all measures related to the application or administration of such a broad system would necessarily be so closely connected to it that they could not form the object of a discrete challenge. This is illustrated by the fact that the "Actual user" condition for import was discretely challenged and ruled on in the *India Quantitative Restrictions* dispute.

³⁵³ First Submission, para 39.

³⁵⁴ See response to Question 37 of the Panel.

- 7.96 In this regard, it is notable that the specific measures challenged before us were not even in existence at the time of establishment of the previous panel. These measures could thus hardly be argued to be in themselves "inherent" in the licensing scheme as considered by the *India Quantitative Restrictions* panel. Although another MOU scheme preceded that foreseen in Public Notice No. 60, that scheme itself was not envisaged in the *India Quantitative Restrictions* panel.
- 7.97 Furthermore, the 1997 scheme of which Public Notice No. 60 and its ensuing MOUs were part, was a new and legally distinct measure taken at Cabinet level that introduced a new auto licensing policy. This replaced an ad-hoc voluntary scheme. Public Notice No. 60 established a very significant, more permanent and binding regime. Companies had to have been prepared to invest US\$50 million and undertake a range of other binding contractual commitments. These commitments included the two elements of indigenization and trade balancing which are the subject of this dispute. These were long-term contracts. The two schemes thus seem to imply significant commercial differences for the companies concerned and the Indian government. While the policy change may have been motivated by issues of transparency and objectivity as argued by India, the Panel finds that it does significantly more than that.
- 7.98 Had the United States sought to challenge in these proceedings the mere existence or application of discretionary import licensing to the specific products at issue here, such a request may indeed have merely duplicated one part of the broader ruling of the *India Quantitative Restrictions* panel. In this instance, however, the challenge before this Panel is different and concerns discrete measures which in themselves constitute an "Auto Licensing Policy". This Panel is being asked to rule on conditions which are specific to this policy, and the legal issue to consider here, i.e. whether the trade balancing and indigenization conditions are compatible with certain provisions of the covered Agreements, was neither expressly nor implicitly asked of, or addressed by, the *India Quantitative Restrictions* panel.
- 7.99 In this instance, the most that can be said in aid of India's argument is that the existence of the import licensing scheme is a significant element of the factual context of the case, since the provisions of the MOUs were entered into with a view to obtaining the right to import some of the products restricted under India's EXIM policy and subject to import licensing. In a broad sense, the MOUs and the conditions foreseen in Public Notice No. 60 contributed to the continued application of one part of the discretionary licensing system examined by the *India Quantitative Restrictions* dispute. However, this does not mean that the specific terms and conditions of the MOUs in themselves cannot be of such a nature as to be inconsistent with the same or other provisions of the Agreements, beyond the import restriction generated by the import licensing itself.
- 7.100 Complex measures or sets of measures may involve different aspects, which generate distinct violations of the WTO Agreements. This has been recognized through the acknowledgement in dispute settlement proceedings that different aspects of a single measure may be covered by different WTO Agreements. In this respect, the Panel also notes the comment by the *Korea Measures*

³⁵⁵ On the basis of the facts available to the Panel, it seems that there may have been other significant differences between both schemes. There was also evidence presented that no one had been denied a license under the 1995 regime. Even if the same were true for the 1997 scheme, that would have depended upon a corporations agreement to the investment and other commitments. In addition, as is evident in this dispute, under the 1997 regime, India maintains that the MOUs continue to have a life even after discretionary import licensing has been removed. This would not have been likely with the 1995 regime if the earlier MOUs involved voluntary projections leading to annual licenses. If the projections were not met in one year under the earlier scheme, they would more likely than not, have simply been taken into account in the discretionary decision about the following years licence. If the licensing regime ceased at that stage, it is not shown on balance that any other repercussions would have been likely to ensue.

Affecting Imports of Fresh, Chilled and Frozen Beef panel (hereinafter "Korea – Various Measures on Beef"), that "where a discretionary licensing system is implemented in conjunction with other restrictions, such as in the present dispute, the manner in which the discretionary licensing system is operated may create additional restrictions independent of those imposed by the principal restriction". The converse must also be possible. This led to the finding in India – Quantitative Restrictions as to the Actual User condition.

7.101 The question put to this Panel relates to aspects of Public Notice No. 60 and the MOUs that are specific to the auto policy expressed in Public Notice No. 60, and which are not, in the Panel's view, "inherent" components of the import licensing system. To put it differently, knowing that the discretionary import licensing system is inconsistent with Article XI does not necessarily answer the question of whether indigenization and trade balancing obligations constitute restrictions on imports within the terms of Article XI. Indeed, the terms of India's BOPs notification suggest that with regard to these products, measures existed not only for balance-of-payments purposes but also for the purposes of an "auto policy". It is some specific aspects of that policy that this panel is being asked to review. Whether they needed to be altered as a result of the ruling in *India – Quantitative Restrictions* is not a matter for this Panel's determination.

Conclusion

7.102 In view of the foregoing, the Panel concludes that the matter before it concerns

new claims never brought before the original panel,

concerning *specific conditions* applicable to automotive manufacturers wishing to import restricted kits and components

(1) never envisaged by the *India - Quantitative Restrictions* panel, and

 $^{^{356}}$ WT/DS161/R, para. 782. Another illustration of the diverse legal consequences which a complex licensing scheme can have is to be found in the EC – Bananas III case. In that case, an issue arose as to whether the EC's procedures and requirements concerning the allocation of import licenses among operators who fulfilled the requirements in the European Communities fell within the scope of Article III:4. The European Communities having appealed the Panel's finding that these measures fell within the scope of Article III, the Appellate Body found as follows:

[&]quot;At issue in this appeal is not whether *any* import licensing requirement, as such, is within the scope of Article III:4, but whether the European Communities procedures and requirements for the *distribution* of import licences for imported bananas among eligible operators *within* the European Communities are within the scope of this provision. The European Communities licensing procedures and requirements include the operator category rules, under which 30 per cent of the import licences for third-country and non-traditional ACP bananas are allocated to operators that market European Communities or traditional ACP bananas, and the activity function rules, under which Category A and B licences are distributed among operators on the basis of their economic activities as importers, customs clearers or ripeners. These rules go far beyond the mere import licence requirements needed to administer the tariff quota for third-country and non-traditional ACP bananas or Lomé Convention requirements for the importation of bananas. These rules are intended, among other things, to cross-subsidize distributors of European Communities (and ACP) bananas and to ensure that European Communities banana ripeners obtain a share of the quota rents. As such, these rules affect "the internal sale, offering for sale, purchase, ..." within the meaning of Article III:4, and therefore fall within the scope of this provision. Therefore, we agree with the conclusion of the Panel on this point."

³⁵⁷ The terms of India's notification to the BOPs Committee, in referring to "AUTO/BOP" as the grounds for restrictions for the products at issue, suggested that measures existed concerning these products *both* in the context of quantitative restrictions for balance-of-payments purposes and in specific "Auto" purposes.

- (2) contained in <u>legally distinct measures</u>, adopted subsequent to the initiation of the *India Quantitative Restrictions* panel, which constitute, in their own terms a "new auto licensing policy".
- 7.103 On the basis of the foregoing, the Panel further concludes that the matter referred to it in these proceedings is not the same as the matter adjudicated upon by the *India Quantitative Restrictions* dispute, to the extent that it relates to violations resulting from the trade balancing and indigenization provisions in the MOUs under Public Notice No. 60, and not from the discretionary import licensing as such. For this reason the Panel considers that the doctrine of *res judicata* could not apply to the facts in this dispute. The Panel does not seek to rule on whether the doctrine could potentially apply to WTO dispute settlement.
- 7.104 India has also asked that, in the event the Panel rules that the United States claim is not barred by *res judicata*, it rule that India's balance-of-payments defense in this dispute is also not barred. Given that an analysis of a balance-of-payments defense considers the evidence in existence at the date of a particular panel's establishment, ³⁵⁸ such defense cannot be precluded by another panel's consideration of economic factors on a wholly distinct date. Thus, regardless of the relevance of the *res judicata* doctrine to WTO dispute settlement, India is not prevented from raising a balance-of-payments defense in this case simply as a result of the panel in *India Quantitative Restrictions* having held that the defense was not made out in that dispute.

1. Has any part of the matter before this Panel been settled through a mutually agreed solution?

7.105 India has argued that with regard to the EC claim, the mutually agreed solution concluded in 1997 in settlement of their dispute on India's quantitative restrictions, (hereafter the "MAS"), prevents the European Communities from bringing this dispute to this Panel. 359

7.106 This argument by India, if successful, would lead to the same result as a successful *res judicata* argument, in that it aims at excluding from the scope of these proceedings issues alleged to have already been resolved. Nevertheless, its legal basis is quite distinct. Here it is necessary to consider the terms of the agreement and the express provisions of the DSU dealing with such agreements.

(a) The mutually agreed solution in the *India – Quantitative Restrictions* dispute with the EC

7.107 The MAS reached with the European Communities was notified to the DSB in accordance with Article 3.6 of the DSU. This agreement, which was in the form of an exchange of letters, provides for three stages for the phasing out of import restrictions maintained on a number of products as notified by India to the Committee on Balance-of-Payments Restrictions (hereafter "BOP Committee"). It provided that

"the European Communities will refrain from action under GATT Article XXII or Article XXIII as regards those restrictions during the phasing-out period (...) as long as India complies with its obligations [under their agreement]."³⁶⁰

³⁶⁰ WT/DS96/8, p. 2 (EC exhibit EC-8).

 $^{^{358}}$ See Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, adopted on 22 September 1999, as upheld by the Appellate Body Report, para. 5.159.

India's First Submission, para. 29, citing WT/DS96/8.

- 7.108 The phasing out period ended on 1 April 2001, a point in time after the European Communities made its panel request in this matter.
- 7.109 In its First Submission, the European Communities argued that, not being a "covered Agreement" under the DSU, the MAS cannot be invoked by India "in order to justify the violation of its obligations under the GATT and the TRIMs Agreement". India argued that the issue is not whether the MAS is a covered agreement. The issue is whether the DSU may be invoked again in respect of a matter formally raised under the DSU and settled through a mutually agreed solution notified under the DSU. It asserts that this issue is related to the EC's procedural rights under the DSU, not India's substantive obligations under a covered agreement. ³⁶²
- 7.110 India further argues that a mutually agreed solution to a matter formally raised under a covered agreement and jointly notified to the DSB as such "must be regarded as a formal settlement of the dispute that makes the re-submission of the same dispute inadmissible". 363
- 7.111 In addition to arguing that the agreed solution is not a covered agreement, the European Communities has also argued that the MAS did not settle this dispute, since both the measures and the claims are different. In particular, the European Communities stresses that the MOUs and Public Notice No. 60 were introduced after the negotiation of the MAS and are not an "inherent part" of the licensing scheme, which could have been administered in many different ways.
- 7.112 Like the issue of *res judicata* concerning the *India Quantitative Restrictions* ruling, the issue before the Panel involves two questions: the first, and more general question, is that of the legal effect of a previously agreed and notified solution under Article 3.6 of the DSU in relation to subsequent proceedings concerning the same matter. The second question is, to the extent that such agreements may have legal implications for a panel's jurisdiction, whether the facts support its application in this case. The Panel will consider each question in turn.
- (b) Relevance of mutually agreed solutions in subsequent proceedings
- 7.113 The status of mutually agreed solutions under the DSU and their impact in subsequent dispute settlement proceedings is not expressly indicated in the DSU and has not been previously addressed in WTO dispute settlement proceedings. As was the case with the issue of *res judicata*, the Panel is thus unaided by any jurisprudence. Unlike that issue, however, such agreements are expressly referred to and supported by the DSU. It is certainly reasonable to assume, particularly on the basis of Article 3 of the DSU, as cited by India, that these agreed solutions are intended to reflect a settlement of the dispute in question, which both parties expect will bring a final conclusion to the relevant proceedings.
- 7.114 This does not necessarily resolve the issue of what can be done if, despite the agreed solution, a subsequent disagreement emerges relating to the scope of the solution or to compliance with it. This is not an issue expressly addressed in the DSU, nor has it been previously examined by another panel established under the DSU.
- 7.115 Without clear guidance in the DSU, this question raises an important systemic issue.³⁶⁴ On the one hand, the Panel recognizes that the right for any WTO Member to bring a dispute to the DSB

_

³⁶¹ EC First Submission, para. 98 (in footnote 73).

³⁶² India's First Submission, para. 33.

³⁶³ *Ibid*, para. 35.

³⁶⁴ In the absence of any express mechanism foreseen in the DSU for the consideration of such agreements, there would be a need to determine whether such a power can be implied from various provisions of

is one of the fundamental tenets of the DSU, and that it could not be lightly assumed in what particular circumstances the drafters of the DSU might have intended such right to be foregone. On the other hand, it may also be the case that it could not be lightly assumed that those drafters intended mutually agreed solutions, expressly promoted by the DSU, to have no meaningful legal effect in subsequent proceedings. There may be significant differences between the provisions of mutually agreed solutions from case to case, which may also make it difficult to draw general conclusions as to the relevance of such solutions to subsequent proceedings other than on a case by case basis.

7.116 At the very least, the Panel sees merit in India's argument that the issue in this respect is not solely whether the mutually agreed solution is a covered agreement, but rather, what effects it may have on the exercise of procedural rights under the DSU in subsequent proceedings. As was the case with the question of *res judicata*, this systemic legal question of the effect of a mutually agreed solution in subsequent proceedings concerning the same matter only needs to be answered if the facts support the argument that the matter before us is covered by the MAS. In light of the European Communities' argument that the matter referred to this Panel has not been settled by the MAS, the Panel will proceed first with an analysis of whether the MAS can be considered to relate to the same matter. If that is the case, the Panel must then rule on the legal effect of such agreements. If it is not the case, there is no need for it to do so.

(c) The scope of the mutually agreed solution between the European Communities and India

7.117 To determine the coverage of the MAS the Panel needs to consider its provisions and also the ambit of the previous dispute to which it was addressed. In that dispute, the European Communities called for formal consultations but did not proceed to a panel request. Thus the Panel needs to consider the terms of the consultation request as well as the terms of the MAS. Ultimately it is the terms of the MAS that are the only possible source of any restriction on our jurisdiction, since it is the scope of the matter that has been resolved through this solution that is at stake in the current determination. The consultation request may merely aid in an understanding of those terms. Because an MAS does not involve a specific adjudicatory ruling that would allow a precise identification of the matter actually "decided on", the Panel must rely on the articulation within the solution of the scope of what the parties intended to settle. 365

7.118 As the MAS is not a covered agreement, it is not expressly subject to the DSU requirement to utilise customary rules of interpretation of international law. Nevertheless, since this is an agreement among States, the Panel finds it appropriate to address the terms of this agreement in accordance with the customary rules of interpretation of international law. It will therefore consider the ordinary

the DSU. Alternatively, there may be an argument that a general principle such as *estoppel* may apply to WTO dispute settlement. These possibilities suggest that the issue cannot necessarily be resolved simply through an acknowledgement that an MAS is not a covered agreement as was argued by the EC. That argument simply is another way of noting that the DSU does not expressly give a panel a mandate to consider whether a "violation" of such an agreement might exist as a distinct basis for a dispute under the DSU. It does not necessarily prove that a panel may not in some circumstances need to consider the terms of such agreed solutions in order to fulfill its duties under the DSU. Here the Panel notes that disputes concerning the application of the DSU itself can be the object of proceedings under the DSU. This might possibly include disputes concerning mutually agreed solutions, since these are expressly referred to in the DSU.

³⁶⁵ In this analysis, the Panel again adopts the notion of "matter", as referred to in the DSU, as a minimal benchmark by which to assess the respective scopes of the MAS and the matter brought before it. The nature of the comparison, however, varies *mutatis mutandis* in considering an MAS rather than a ruling argued to be *res judicata*. In this instance, what should be the basis for the comparison is the scope of the MAS, as defined essentially by the parties themselves, rather than any legal issues actually ruled on in previous proceedings.

meaning of the terms of the MAS in light of their context and taking into account their object and purpose.

7.119 The EC's initial request for consultations in the dispute which gave rise to the MAS with India (WT/DS96) identified the following measures:

"quantitative restrictions maintained by India on importation of a large number of agricultural, textile and industrial products. The restrictions include those notified to the WTO in document WT/BOP/N/24 of 22 May. Annex 1 of this document lists more than 3,000 quantitative restrictions maintained as per 1 April 1997."³⁶⁶

These measures were alleged to be inconsistent with, *inter alia*, Articles XI, XIII, XVII and XVIII of GATT 1994.

7.120 The MAS reached by exchange of letters on 12 November 1997 was notified in accordance with Article 3.6 of the DSU, and circulated on 6 May 1998.³⁶⁷ This agreement begins with the following provisions:

"Desiring to reach a satisfactory settlement of the difference regarding quantitative restrictions maintained by India on import of industrial, agricultural and textile products, as notified by India to the WTO by Annex 1, Part B (Notification on Quantitative Restrictions Maintained on Imports Under the Export and Import Policy (as on 1 April 1997)) of document WT/BOP/N/24 of 22 May, and for which the European Communities requested consultations with India under GATT Article XXII on 18 July 1997, India, on the understanding that nothing in this Agreement prejudices the respective rights and obligations of the European Communities and India under the WTO Agreement and that the European Communities will refrain from action under GATT Article XXII or Article XXIII as regards those restrictions during the phasing-out period as defined below, as long as India complies with its obligations under this exchange of letters, agrees as follows:

- 1. India will eliminate all quantitative restrictions on imports maintained by reference to GATT Article XVIII and notified to the WTO in Annex I, Part B of document WT/BOP/N/24 of 22 May, in accordance with the time schedules contained in Annex III of that document, as modified by the attached Annex. The modifications in the attached Annex concern the duration of each of the three phases of the period for the elimination of quantitative restrictions on imports imposed by India for balance-of-payments purposes and the shifting of certain products from Phase II to Phase I or Phase III to Phase I or II. This is without prejudice to India's rights and obligations under the WTO Agreement." (emphasis added)
- 7.121 The scope of the solution is identified in this initial part of the MAS as relating to "quantitative restrictions maintained by India on import of industrial, agricultural and textile products, as notified by India to the WTO by Annex 1, Part B (Notification on Quantitative Restrictions Maintained on Imports Under the Export and Import Policy (as on 1 April 1997)) of document WT/BOP/N/24 of 22 May, and for which the European Communities requested consultations with India under GATT Article XXII on 18 July 1997". These are thus in essence the same measures as

_

 $^{^{366}}$ WT/DS96/1.

³⁶⁷ WT/DS96/8.

were brought to the panel on *India – Quantitative Restrictions* by the United States at approximately the same period.³⁶⁸

- 7.122 The MAS further provides a timetable for "the elimination of quantitative restrictions imposed by India and notified to the WTO in Annex I, Part B of document WT/BOP/N/24 over three phases". While the period for elimination of the restrictions foreseen for the products at issue in the instant proceedings extended beyond 1 April 2001, an MFN clause in the solution allowed the European Communities to benefit from the more favourable solution subsequently reached with the United States. This meant that the date of 1 April 2001 was the relevant date for elimination of the quantitative restrictions maintained by India on kits and components as contemplated in the MAS with the EC.³⁶⁹
- 7.123 India refers, in its argument, to the final part of the above introductory paragraph, and more particularly, to the European Communities' commitment not to initiate proceedings under Articles XXII or XXIII of GATT "as long as it complied with its phase-out obligations".
- 7.124 In considering that argument, the Panel notes that both the consultation request and the MAS refer exclusively to the "quantitative restrictions" maintained by India on a number of products as notified to the BOP Committee. The European Communities promise is merely to "refrain from action ... with regard to those measures." There is no express reference in the MAS to any such promise with regard to supplementary or subsidiary measures. Indigenization and trade balancing are not mentioned. Where the MAS is concerned, there is no direct or indirect reference in the definition of the scope of the solution to any specific criteria, Public Notices or other instruments used for the administration of these restrictions, nor to any specific criteria applied in conjunction with the restrictions maintained.
- 7.125 As in the case of the dispute between the United States and India, the Panel also notes that the specific measures before it could not have been expressly dealt with in the consultations or the MAS as negotiated. This is because Public Notice No. 60 was adopted exactly one month after the exchange of letters which sealed the MAS, albeit several months before its notification to the DSB.³⁷⁰
- 7.126 The Panel also notes that there is no express reference in the MAS to the indigenization or trade balancing elements of the MOUs which were utilised prior to the 1997 changes. Thus those changes cannot be seen to be a mere continuation of similar measures expressly covered.
- 7.127 The Panel again considers that a claim that contractual conditions as to indigenization and trade balancing in and of themselves constitute a violation of Article XI, is different from a claim that a general quantitative restrictions regime is itself in violation of that Article. The MAS only expressly precluded resort to formal dispute settlement in respect of the quantitative restrictions, although it did address other provisions as well. Indeed, India's argument in relation to the MAS only refers to the European Communities' promise relating to the obligation to "phase-out" these restrictions. The phase-out obligation is contained in paragraph 1 of the MAS. This leads us to a preliminary conclusion that the ordinary meaning of the European Communities' promise in the MAS concerning the phase-out of the quantitative restrictions referred to, does not expressly preclude the initiation of a

³⁶⁸ The panel on *India – Quantitative Restrictions* was established on 18 November 1997, while the mutually agreed solution between India and the European Communities, although only notified in May 1998, was reached on 12 November 1997.

³⁶⁹ See MAS paragraph 3, in WT/DS96/8.

As previously noted, the MAS was reached on 12 November 1997. Public Notice No. 60 was adopted on 12 December 1997.

dispute aimed at questioning the validity *per se* of subsequently introduced specific conditions employed in the administration of a part of India's then applicable quantitative restrictions.

7.128 The MAS also contains additional provisions that address the relations of the parties in the course of the implementation period. The Panel feels that it must consider these as part of the context of the provisions of the MAS principally raised by India, in order to ensure a complete interpretation of the scope of the MAS. These are as follows:

"Without prejudice to the rights and obligations of the European Communities and India under the WTO Agreement, India will refrain during the phase-out period from making more restrictive the import arrangements for the products in the attached Annex and in Annex III of document WT/BOP/N/24.

During the transition period India will operate the SIL and the NAL arrangements in a transparent and non-discriminatory manner so as to ensure that they do not have trade-restrictive or distortive effects additional to those caused by the restrictions themselves. This is without prejudice to India's rights and obligations under the Agreement on Import Licensing Procedures. India will notify licensing procedures, and changes in these procedures, in accordance with Article 5 of the Agreement on Import Licensing Procedures.

(...)

This agreement shall be reviewed annually by India and the EC. For the purposes of this review, India will provide information on the implementation of the phasing-out schedule. In this framework, there will be an opportunity for India and the European Communities to discuss the implementation of the time schedules and for the European Communities to raise questions regarding the functioning of licensing procedures during the phasing-out period."

7.129 These clauses suggest that the parties contemplated that modifications might be made to the arrangements the subject of the MAS. They imply that during the phase-out period, if India met the interim phase-out targets it would only be in breach of the MAS if such modifications were "more restrictive" or caused "additional distortive effects."

7.130 The express promise to refrain from bringing a challenge against the quantitative restrictions refers to the condition of India complying with its obligations under the agreement. This could be argued to include the obligations as to changes in the licensing "arrangements". These clauses however do not address the separate question of whether a change which raises a distinct claim can be separately challenged. The plain meaning of the European Communities' promise with respect to "those restrictions" is thus not, in the view of the Panel, affected by these clauses. This is the interpretation that the Panel adopts.³⁷¹

³⁷¹ If the Panel were wrong in that view, the next question would be whether the facts support the application of those provisions in a way which precludes the European Communities' express DSU rights to bring an action where it deems it appropriate. Those clauses refer to modifications and alterations. It is not in dispute that some changes were made. For the same reasons as articulated with respect to the *res judicata* issue, the Panel believes that the changes were significant and were more than merely measures that made existing provisions more transparent and objective. That conclusion does not however lead to the conclusion that they are necessarily more restrictive or distortive. Nor is it clear what evidentiary elements would be appropriate to rely on in considering whether the 1997 changes are more restrictive or distortive. Nevertheless, as India seeks to rely on the MAS as a defense, it would have been for it to address this issue and provide a prima facie case

- 7.131 In addition, the Panel also notes that the claims invoked in the request for consultations in the earlier dispute, which formed the basis of the subsequent agreed solution, referred to a number of provisions of the GATT, including Article XI, but not to any inconsistency with Article III. In addition, while other covered Agreements were identified, the TRIMs Agreement was not one of those. From these observations, the Panel cannot conclude that the scope of the agreed solution with regard to the measures actually covered, encompasses any claims of violation of either Article III or the TRIMs Agreement.
- 7.132 The Panel now turns to consider whether it can be successfully argued that the matter raised before it by the European Communities is indirectly or by implication precluded from further dispute by reason of the terms of the MAS. As in the case of the *res judicata* analysis, the Panel acknowledges that such a broad-ranging dispute as that which gave rise to the MAS could not be reasonably expected to have such a solution list every single measure which may have some bearing on the application of the quantitative restrictions at issue. The Panel again acknowledges that compliance by India with the terms of the MAS would no doubt have had implications for many such administrative elements. Again, that does not determine whether those elements are covered by the promise not to resort to dispute settlement. In much the same way as it held that the specific trade balancing and indigenization requirements were not implicitly covered by the *India Quantitative Restrictions* panel ruling, the Panel concludes on balance that the mutually agreed solution concerning India's quantitative restrictions for balance-of-payments purposes does not indirectly encompass the specific conditions before it.
- 7.133 The Panel considers that the reference to "those restrictions" in the introductory paragraph of the MAS is, as far as discretionary import licensing is concerned, a reference to the licensing scheme *per se* and not to future elements of it that are argued to violate provisions of the covered agreements in and of themselves. Thus, the Panel does not consider that the plain meaning of the promise relating to challenges to quantitative restrictions, impliedly deals with the two conditions now in dispute. Nor does it consider that it should be interpreted to cover distinct challenges to all administrative features. The Panel does not read in the terms of the MAS an all-encompassing promise barring litigation of any subsequent administrative aspect of the measures expressed to come within the covered solution. The Panel would therefore not interpret a promise by the European Communities to "refrain from action ... as regards those restrictions" as impliedly applying to the measures in dispute before it. This plain meaning and contextual interpretation is sufficiently clear but in any event does not, in the Panel's view frustrate the object and purpose of the MAS, which from its terms is to bring to an end the particular dispute it dealt with: i.e. the European Communities' concerns with a broad range of quantitative restrictions taken by India for balance-of-payments purposes.

(d) Conclusion

- 7.134 In view of the foregoing, the Panel concludes that the terms of the mutually agreed solution notified by the European Communities and India do not preclude it from examining the matter brought before it by the European Communities. In view of that factual finding the Panel does not intend to rule on the legal question as to whether a notified mutually agreed solution can ever operate as a bar to a panel's express mandate from the DSB.
- 7.135 The European Communities also argues, as a subsidiary matter, that if the Panel were to find that the issues raised in this dispute are covered by the MAS, then these measures violate that agreement for a further reason, since they are alleged to continue beyond 1 April 2001. Having found

and convince us that these changes alter the plain meaning of clause 1 of the MAS. The Panel does not find that they have done so.

that the Panel's competence is not precluded by the MAS in any event, it does not need to consider this argument further.

2. Measures to be eliminated in the course of implementing the results of the *India - Quantitative Restrictions* dispute and abusive splitting of a dispute

- 7.136 India invokes further arguments in support of its contention that the Panel should not consider the claims referred to it, because the MOUs are an inherent part of the import licensing scheme which India is required to eliminate as of 1 April 2001 as a result of the ruling of the *India Quantitative Restrictions* dispute and of its mutually agreed solution with the EC.
- 7.137 India argues that it would be pointless for this Panel to "add" new grounds of violations concerning them. This argument relates to the Article III of GATT and TRIMs allegations not dealt with or referred to in either *India Quantitative Restrictions* or the MAS. India also argues that a complainant may not abusively "split" a dispute into successive actions instead of raising all the issues relating to a single matter in one proceeding.
- 7.138 The Panel has already concluded that the matter before it is different to the matter before the *India-QR* panel and the matter dealt with in the MAS. Thus this is not, in the Panel's view, a case of a new claim being expressly directed against a distinct measure previously addressed under another provision. The remaining underlying assumption in both of these arguments appears to be that all issues relating to measures which may need to be eliminated as a result of a specific proceeding should all have been dealt with at once and a separate dispute should not be engaged in where the issues at stake are arguably already disposed of through the outcome of the initial disputes.

(a) Abusive splitting

- 7.139 India argues that it should not be possible to "split" a dispute into separate successive cases. Although India seems to invoke an autonomous "principle" of prohibition of "abusive splitting" of disputes, it has not explained any source for this specific principle. Rather, its argument appears to rest essentially on the basis that this would be an abusive exercise of the right to bring complaints before the DSB.
- 7.140 Such a principle could be sought to be argued to fit within general principles of due process or some equitable principle such as good faith or abuse of rights. While such notions might be of relevance to WTO dispute settlement in some circumstances, the Panel is not convinced that the conditions which might justify such a decision are present in this instance. This is not a case where it is alleged that a complainant knowing all the facts about a measure, chooses to deliberately burden a respondent with a range of separate actions dealing with distinct WTO provisions. Here it is not in dispute that new measures were introduced in 1997. These measures themselves contain a range of provisions specific to the auto sector and constitute a distinct "auto licensing policy". These measures may have significantly affected the legal and commercial implications of the indigenization and trade balancing undertakings made by signatories to earlier MOUs.
- 7.141 In such circumstances, the Panel recalls the terms of Article 3.7 of the DSU, according to which "Before bringing a case, a Member shall exercise its judgment as to whether action under these procedures would be fruitful". The DSU thus entrusts Members with the responsibility of being essentially self-regulating in deciding to bring a case to the attention of the DSB. In this instance, as noted above, the specific measures brought before this Panel, although they have some relation with the matter ruled on in QRs are in themselves distinct measures concerning a specific industrial sector. In addition, while India disputes the significance and overall distortive effects brought about by the 1997 changes, it has never presented any evidence that the complainants were acting in bad faith or

contrary to the spirit of Article 3.7 in coming to a different view about the 1997 measures. Consequently the Panel does not find any evidence to support an allegation of abusive splitting, even if such a principle could ever allow a panel to make a discretionary determination not to pursue its express mandate.

- 7.142 In view of the above factual finding the Panel does not rule on whether such a power could in theory exist.
- (b) Unnecessary litigation
- 7.143 This argument seems to rely on the notion that matters which have not been expressly ruled on, (in this case, in India's view, the claims based on provisions other than Article XI of GATT), should not be re-considered on the basis that it would be, in its view, "pointless" to examine these additional claims in relation to measures which are required to be eliminated in any case.
- 7.144 The assumption seems to be that if full compliance with previous rulings incidentally requires the elimination of part or all of the measures considered in the subsequent dispute, then these distinct claims should not be examined by the second panel.
- 7.145 The argument seems to assume a certain interpretation of what is required in terms of implementation of the previous panel ruling. As outlined below, however, this Panel takes the view that this assessment is not for it to perform. Therefore, the question of whether the measures before the *India Quantitative Restrictions* panel need to be brought into compliance to a degree which would make the issues before this Panel without practical significance, is not one which can be part of this Panel's mandate.
- 7.146 In any event the Panel does not believe that the facts support an assertion that the new claims can have no practical significance. This argument by India relates to claimed violations that the Panel has concluded were <u>not</u> ruled on by the *India Quantitative Restrictions* panel. Even if the proper implementation of the rulings in that case may involve the elimination of measures in a way that will incidentally dispose of other violations also, this cannot be presumed in all instances. For example, if Public Notice No. 60 and the MOUs were to involve violations of Article III, it cannot be presumed that these violations would necessarily disappear with the disappearance of the import licensing scheme in implementation of the *India Quantitative Restrictions* rulings. There may be many ways for India to implement the results of the *India Quantitative Restrictions* ruling in order to eliminate the import restrictions maintained for balance-of-payments purposes on different products, and depending on the structure of the specific measures at stake, this may or may not entail the elimination of other violations of the Agreements.
- 7.147 In concluding its consideration of these arguments relating to measures that India may be "required to eliminate" as a result of the *India Quantitative Restrictions* disputes, the Panel wishes to stress that it does not intend, in these proceedings, to draw any conclusions on what the implementation of the rulings adopted in the *Quantitative Restrictions* case with the United States and of the MAS with the European Communities may or may not require. This would be a matter for a compliance panel to establish, if so requested. This Panel's mandate, however, is quite distinct from that of a compliance panel under Article 21.5.
- 7.148 It was never argued that the measures as they existed at the dates of request for establishment of the Panel were taken in order to comply with the rulings and recommendations of the DSB in the *India Quantitative Restrictions* dispute. On the contrary, at the time of their adoption, they supported and contributed to the continued application of the import licensing scheme which India

was subsequently required to bring into conformity with the WTO Agreements under the *India – Ouantitative Restrictions* case.

7.149 Nor do subsequent events alter this distinction between this Panel's jurisdiction and any possible compliance panel's jurisdiction. It is true that, in the parties' own description of the facts, the evolution of the measures in the course of these proceedings is directly linked to actions taken with a view to implementing the *India* – *Quantitative Restrictions* rulings. Nevertheless, the Panel does not see this question as affecting its jurisdiction to consider the measures as they existed as of the date of its establishment. A compliance panel might, at most, if so requested, consider these measures as of the date for compliance with the *India* – *Quantitative Restrictions* rulings. Any potential for overlap between views expressed under this Panel's mandate and views which might be expressed by a hypothetical compliance panel under *India* – *Quantitative Restrictions* may be a systemic issue of significance, but it is not for this Panel to resolve. The potential ambit of a compliance panel's findings does not in any way diminish this Panel's obligation to comply with its express mandate from the DSB to consider the measures before it.

3. Conclusion

7.150 In examining each of India's arguments, the Panel has given careful consideration to the concerns expressed by India concerning the possible reopening of the same disputes between the same parties. It is apparent to the Panel that the measures before it have operated in direct relation to the discretionary import licensing system which India has committed itself to eliminate in accordance with the recommendations of the DSB and its agreed solution with the European Communities. Nonetheless, the Panel is of the view that this dispute raises distinct legal issues from those resolved through the *India – Quantitative Restrictions* disputes, and that it must consider these issues. In view of the foregoing analysis, the Panel rejects each of India's arguments to the effect that it does not have jurisdiction or should exercise some discretion not to consider the matter before it. The Panel now turns to a consideration of the substantive issues.

D. ORDER OF EXAMINATION OF THE CLAIMS

7.151 Both complainants have made claims that the measures in issue are in violation of both GATT 1994 and the TRIMs Agreement (Article 2). The Panel first wishes to consider whether it is desirable to consider these claims in any particular order.

7.152 A number of factors should be considered in that regard. The Panel recalls the Appellate Body's comment in *European Communities – Regime for the Importation, Sale and Distribution of Bananas* (hereafter "*EC – Bananas III*") suggesting that a panel would normally be expected to examine the more specific agreement before the more general, where two Agreements apply simultaneously.³⁷³ The Panel also recalls that it is permitted to apply judicial economy in considering

Nor is there any reason to presume that any panel subsequently examining an issue considered by a previous panel would not take the outcome of such ruling into consideration, to the extent relevant.

³⁷³ See Report of the Appellate Body in *EC – Bananas III*, WT/DS27/AB/R, adopted on 25 September 1997, para. 204 (DSR 1997:II, 591): "Although Article X:3(a) of the GATT 1994 and Article 1.3 of the Licensing Agreement both apply, the Panel, in our view, should have applied the Licensing Agreement first, since this Agreement deals specifically, and in detail, with the administration of import licensing procedures. If the Panel had done so, then there would have been no need for it to address the alleged inconsistency with Article X:3(a) of the GATT 1994".

matters before it, so that "a panel need only address those claims which must be addressed in order to resolve the matter in issue in the dispute". 374

- 7.153 The Panel has also taken into account the fact that two complaints are effectively being addressed in this report, the scopes of which are not in all respects identical. In considering these questions of order and judicial economy, the Panel has been mindful of its obligations under Article 9 of the DSU in preserving the rights of all parties where multiple complaints are involved.
- 7.154 Where the order of analysis of claims is concerned, it is important to consider if a particular order is compelled by principles of valid interpretative methodology, which, if not followed, might constitute an error of law.³⁷⁵
- 7.155 In circumstances other than where a proper application of one provision might be hindered without prior consideration of other issues, the adoption by a panel of a particular order of examination of discrete claims would rarely lead to any errors of law. A panel would always be required to consider, *inter alia*, contextual aspects of interpretation of each of the relevant provisions in accordance with customary rules of interpretation, and to take account of the principle of effectiveness in such interpretation, whichever agreement it chooses to examine first. Here the Panel notes the comments of the Appellate Body in US FSC concerning the United States' argument that the Panel erred by failing to begin its examination of the European Communities' claim under Article 3.1(a) of the *SCM Agreement* with footnote 59 of that Agreement. In the Appellate Body's view,

"it was not a legal error for the Panel to begin its examination of whether the FSC measure involves export *subsidies* by examining the general definition of a "*subsidy*" that is applicable to export *subsidies* in Article 3.1(a). In any event, whether the examination begins with the general definition of a "subsidy" in Article 1.1 or with footnote 59, we believe that the outcome of the European Communities' claim under Article 3.1(a) would be the same. The appropriate meaning of both provisions can be established and can be given effect, irrespective of whether the examination of the claim of the European Communities under Article 3.1(a) begins with Article 1.1 or with footnote 59. 376

7.156 In this case, the Panel is dealing with separate claims to the effect that a particular factual situation falls foul of both GATT 1994 provisions and provisions of the TRIMs Agreement. Previous

³⁷⁶ Report of the Appellate Body in *US – FSC*, WT/DS108/AB/R, adopted on 20 March 2000, para. 89.

Report of the Appellate Body in *US – Wool Shirts and Blouses*, WT/DS33/AB/R, adopted on 23 May 1997, p.19 (DSR 1997:I, 323). The Appellate Body noted more generally in that paragraph that "[n]othing in [Article 11 of the DSU] or in previous GATT practice requires a panel to examine all legal claims made by the complaining party. Previous GATT 1947 and WTO panels have frequently addressed only those issues that such panels considered necessary for the resolution of the matter between the parties, and have declined to decide other issues. (...) Given the explicit aim of dispute settlement that permeates the DSU, we do not consider that Article 3.2 of the DSU is meant to encourage either panels or the Appellate Body to "make law" by clarifying existing provisions of the *WTO Agreement* outside the context of resolving a particular dispute. A panel need only address those claims which must be addressed in order to resolve the matter in issue in the dispute".

³⁷⁵ An example of a situation where a proper order of examination of different elements of a certain provision can be identified would be Article XX of GATT 1994, as highlighted by the Report of the Appellate Body in *US – Shrimp*, WT/DS58/AB/R, adopted on 6 November 1998, para. 120.

panels confronted with concurrent claims concerning these two agreements have had diverse approaches to the choice of order of analysis of such claims.³⁷⁷

7.157 As a general matter, even if there was some guiding principle to the effect that a specific covered Agreement might appropriately be examined before a general one where both may apply to the same measure, it might be difficult to characterize the TRIMs Agreement as necessarily more "specific" than the relevant GATT provisions. Although the TRIMS Agreement "has an autonomous legal existence", independent from the relevant GATT provisions, as noted by the *Indonesia – Autos* panel, 378 the substance of its obligations refers directly to Articles III and XI of the GATT, and clarifies their meaning, *inter alia*, through an Illustrative list. On one view, it simply provides additional guidance as to the identification of certain measures considered to be inconsistent with Articles III:4 and XI:1 of the GATT 1994. On the other hand, the TRIMs Agreement also introduces rights and obligations that are specific to it, through its notification mechanism and related provisions. An interpretative question also arises in relation to the TRIMs Agreement as to whether a complainant must separately prove that the measure in issue is a "trade-related investment measure". For either of these reasons, the TRIMs Agreement might be arguably more specific in that it provides additional rules concerning the specific measures it covers. ³⁷⁹ The Panel is therefore not convinced that, as a general matter, the TRIMs Agreement could inherently be characterized as more specific than the relevant GATT provisions.

7.158 In some circumstances, there may be a practical significance in determining a particular order of examination of claims based on the TRIMS and GATT 1994.³⁸⁰ One example might be where the

³⁷⁷ One panel decided to examine claims under Article III:4 and Article 2.1 of the TRIMs Agreement together, but having found a violation of Article III:4, did not consider it necessary to make a finding on Article 2.1 of the TRIMS Agreement (See Panel Report on *EC – Bananas*, WT/DS27/R, para. 7.168).

Another panel decided to examine first the claims under the TRIMs Agreement, "since the TRIMs Agreement is more specific than Article III:4 as far as the claims under consideration are concerned". In doing this, the panel was seeking to follow the approach endorsed by the Appellate Body in *EC – Bananas III* and referred to above that claims on the more specific agreement should be examined before the more general (See Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 23 July 1998, para. 14.63, DSR 1998:VI, 2201).

A third and subsequent panel, however, was "not persuaded that the TRIMS Agreement can be properly characterized as more specific than Article III:4 in respect of the claims raised by the complainants in the present case". That panel chose to examine first the claims under the GATT 1994, following the parties' order of presentation (See Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.63.)

³⁷⁸ Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, para. 14.63 (DSR 1998:VI, 2201).

³⁷⁹ To say, for instance, that the TRIMs Agreement is more specific because it contains a specific criterion of the presence or absence of a trade-related investment measure depends upon whether that is a distinct criterion and whether the lack of such a criterion in Articles III and XI of GATT 1994 makes these provisions more general as opposed to merely having a broader range of coverage on the same criteria. The only practical difference and potential advantage in looking at the TRIMs agreement first in this instance seems to be the possible utilization of the Illustrative List, to the extent that it would be relevant to the claims at issue and may facilitate the identification of a violation of Articles III:4 or XI:1 of GATT 1994.

Determining order of analysis is different to the question of resolution of conflict between provisions. The latter entails the use of conflict resolution rules or interpretative techniques. The Panel notes that both GATT 1994 and the TRIMs Agreement are part of Annex 1A of the WTO Agreement. A general interpretative note to Annex 1A provides that "[i]n the event of a conflict between a provision of the General Agreement on Tariffs and Trade 1994 and a provision of another agreement in Annex 1A (...) the provision of the other agreement shall prevail to the extent of the conflict." No conflict has been alleged to exist in this instance, in the absence of any issue relating to any TRIMs which would have been notified under the TRIMs Agreement.

temporary cover offered to notified TRIMs would be of relevance. In this instance, India has not claimed to have notified the measures at issue under the TRIMs Agreement. This issue therefore does not arise here. Indeed, in response to a question from the Panel, India has indicated that "the TRIMs Agreement (...) no longer entails rights and obligations in respect of India that are different from those under the GATT", and has encouraged the Panel to refrain from examining the TRIMs Agreement.³⁸¹ For the purposes of this case, therefore, there appears to be, in that respect, no particular reason to start our examination on any particular order. Nor does it find that the end result would be affected by either determination of order of analysis.

- 7.159 Both complainants have also addressed their claims under GATT 1994 prior to their claims under the TRIMS Agreement. Their responses to specific questions from the Panel on the proper order of examination of their claims suggest that both would be in agreement with an approach leading to an examination of the GATT 1994 claims first.³⁸²
- 7.160 In these circumstances, the Panel does not see that there is anything improper in either addressing the GATT claims or the TRIMs claims first.
- 7.161 The order selected for examination of the claims may also have an impact on the potential to apply judicial economy. It seems that an examination of the GATT provisions in this case would be likely to make it unnecessary to address the TRIMs claims, but not vice-versa. If a violation of the GATT claims was found, it would be justifiable to refrain from examining the TRIMs claims under the principle of judicial economy. Even if no violation was found under the GATT claims, that also seems an efficient starting point since it would be difficult to imagine that if no violation has been found of Articles III or XI, a violation could be found of Article 2 of the TRIMs Agreement, which refers to the same provisions. Conversely, if no violation of the TRIMs Agreement were found, this would not necessarily preclude the existence of a violation of GATT Articles III:4 or XI:1 because the scope of the GATT provisions is arguably broader if India's argument was accepted that there is a need to prove that a measure is an investment measure and its assertion that this is not the case with the measures before this Panel.
- 7.162 For all of these reasons the Panel will therefore examine the GATT claims first.
- 7.163 Violations of Articles III and XI have been alleged with regard to two of the conditions foreseen under Public Notice No. 60 and in the MOUs, namely "indigenization" and "trade balancing". The Panel will address claims relating to these two conditions in turn.
- E. CONSISTENCY OF THE INDIGENIZATION CONDITION WITH THE GATT 1994
- 7.164 Paragraph 3 (iii) of Public Notice No. 60 provides that the MOUs shall be based on a number of parameters, including:

"Indigenization of components up to a minimum level of 50% in the third year or earlier from the date of clearance of first import consignment of CKD/SKD kits/components and 70% in the 5th year or earlier. Once the MOU signing firm has reached an indigenization level of 70%, there will be no need for further import licences from DGFT. Consequently, as and when the firms achieve 70% indigenization, they would go outside the ambit of the MOU automatically.

³⁸¹ Response to Question 128 of the Panel.

Response to Question 128 of the Fanci.

382 See the response of the EC to Question 99 of the Panel, and the response of the US to Question 84 of the Panel.

However, they will discharge the export obligation corresponding to the imports made by them till that time."

7.165 The standard MOU annexed to Public Notice No. 60 reflects this obligation by requiring:

"that the party shall achieve indigenization of components up to a minimum level of 50% in the third year or earlier from the date of clearance of first import consignment of CKD/SKD kits and 70% in the fifth year of earlier. Once the party has reached an indigenization level of 70% there will be no need for further import licence from DGTF. However, the party shall discharge the export obligation corresponding to the imports made by them till that time.

That the party intends to achieve the following levels of indigenization of their product, year-wise:

Year

Percentage (%) of indigenization

The party shall aggressively pursue and achieve as soon as possible the development of the local supply base and increased local content, since the same will allow a higher level of indigenization."

- 7.166 It is not disputed that MOUs have been signed with most major car manufacturers in India in accordance with these terms, including the "indigenization" condition.
- 7.167 The European Communities and United States claim that this obligation is inconsistent with Article III:4 of the GATT 1994. The United States also claims that the measure is inconsistent with Article XI:1 of GATT 1994.
- 7.168 The Panel does not believe that it is required as a matter of proper interpretation to consider one of these claims prior to the other as they are distinct Articles and the proper application of one does not require a prior determination of the other. The Panel will consider first the claims concerning an alleged violation of Article III:4, since this claim is common to both complainants. This would provide the greatest opportunity to at least consider the application of the judicial economy principle, since any finding which might be first made on the basis of Article XI, would not have addressed the claim of the European Communities.

1. Claims under Article III:4 of GATT 1994

- 7.169 In accordance with the terms of Article 3.2 of the DSU, the Panel is required to clarify the existing provisions of the covered agreements in accordance with customary rules of interpretation of public international law. As has been recalled by the Appellate Body, these rules are embodied in particular in Articles 31 and 32 of the Vienna Convention on the Law of Treaties. Accordingly, this Panel is required to interpret the text of the agreements by way of reference to the "the ordinary meaning of the terms of the treaty in their context and in the light of its object and purpose".
- 7.170 Article III:1 provides in relevant part that:

"[t]he contracting parties recognize that internal taxes and other internal charges, and laws, regulations and requirements affecting the internal sale, offering for sale, purchase, transportation, distribution or use of the products (...) should not be applied

to imported or domestic products so as to afford protection to domestic production". $^{383}\,$

7.171 Article III:4 provides in relevant part:

"The products of the territory of any contracting party imported into the territory of any other contracting party shall be accorded treatment no less favourable than that accorded to like products of national origin in respect of all laws, regulations and requirements affecting their internal sale, offering for sale, purchase, transportation, distribution or use."

7.172 To determine whether the Indian measure is consistent with Article III:4, it is necessary to examine whether (1) imported products and domestic products are like products; (2) the measures constitute a "law, regulation or requirement"; (3) they affect the internal sale, offering for sale, purchase, transportation, distribution or use; and (4) imported products are accorded less favourable treatment than the treatment accorded to like domestic products.³⁸⁴

(a) Like products

7.173 This claim deals with the treatment of certain specified imported automotive parts and components and their domestic equivalent.

7.174 The Panel notes that the only factor of distinction under the "indigenization" condition between products which contribute to fulfilment of the condition and products which do not, is the origin of the product as either imported or domestic. India has not disputed the likeness of the relevant automotive parts and components of domestic or foreign origin for the purposes of Article III:4 of the GATT 1994. Origin being the sole criterion distinguishing the products, it is correct to treat such products as like products within the meaning of Article III:4.

7.175 The panel in *Canada - Autos* reached the same conclusion in similar conditions. More recently, the compliance panel in US - FSC (Article 21.5) noted that:

"the distinction made between imported and domestic products in the Act's foreign Articles/labour limitation concerning the limitation on fair market value attributable to 'articles' is solely and explicitly based on origin. We do not believe that the mere fact that a good has US origin renders it "unlike" an imported good."³⁸⁶

³⁸⁶ WT/DS108/RW circulated on 20 August 2001.

 $^{^{383}}$ The Appellate Body has recently clarified the contextual role of Article III:1 in the interpretation of Article III:4:

[&]quot;the "general principle" set forth in Article III:1 "informs" the rest of Article III and acts "as a guide to understanding and interpreting the specific obligations contained" in the other paragraphs of Article III, including paragraph 4. Thus, in our view, Article III:1 has particular contextual significance in interpreting Article III:4, as it sets forth the "general principle" pursued by that provision." Report of the Appellate Body on *European Communities – Measures Affecting Asbestos and Asbestos-Containing Products*, WT/DS135/AB/R, adopted on 5 April 2001, para. 93.

Report of the Appellate Body in *Korea – Various Measures on Beef*, WT/DS161/AB/R, WT/DS169/AB/R, adopted on 10 January 2001, para. 133.

³⁸⁵ Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.74,

- 7.176 The Panel therefore concludes that automotive parts and components of domestic and foreign origin are like products within the meaning of Article III:4 of GATT 1994.
- (b) "Laws, regulations or requirements"
- 7.177 While the Panel is at this stage considering the indigenization condition, it does not arise in a vacuum. It emanates from Public Notice No. 60 and the MOUs entered thereunder. The Panel thus needs to look at these measures as identified in our terms of reference to consider whether they constitute "laws, regulations or requirements".
- 7.178 Public Notice No. 60 and the MOUs operate together, so that in combination, the obligations actually accrue: the proper application of Public Notice No. 60 required MOUs to be signed by an entity that wished to obtain licenses. The legal obligations on individual firms therefore resulted from the combination of both instruments: the MOUs were signed in accordance with Public Notice No. 60, and once signed, they embodied for the manufacturers the obligations foreseen in the Public Notice No. 60.
- 7.179 The Panel has therefore chosen to examine these legal instruments together as the combined foundation for the indigenization obligation. If between the two instruments, it is appropriate to conclude that the indigenization condition is an element of certain "laws, regulations or requirements", then this should be enough for this part of the Article III analysis. The situation may be different at the stage of considering any recommendations and rulings. The provision calling for the conditions being challenged in this dispute is contained in Public Notice No. 60 itself. This might mean that this measure in itself might be in violation to the extent that it requires action inconsistent with GATT obligations. The European Communities has asked for a separate ruling in that regard. The US has not. The European Communities request will be considered separately at a later stage. In the meantime, because both instruments contain the indigenization and trade balancing conditions at issue, and in light of the close relationship between Public Notice No. 60 and the signature of the MOUs, the Panel will first complete its analysis of the two conditions in issue taking both instruments into account.
- 7.180 The United States argues that the measures constitute "at least 'regulations' or 'requirements', as those terms in Article III:4 are ordinarily understood". The European Communities considers the measures to be "requirements" within the meaning of Article III:4. Thus both complainants consider that Public Notice No. 60 and the MOUs at least impose "requirements" under Article III:4. The Panel turns to a consideration of this term.
- (i) The notion of "requirement" within the meaning of Article III:4
- 7.181 An ordinary meaning of the term "requirement", as articulated in the New Shorter Oxford Dictionary, is "Something called for or demanded; a condition which must be complied with". The *Canada FIRA* panel further suggested that there must be a distinction between "regulations" and "requirements" and that requirements could not be assumed to mean the same, i.e. "mandatory rules applying across the board".
- 7.182 The European Communities argues that "it is firmly established that Government action need not be compulsory in order to qualify as a "requirement" for the purposes of GATT Article III:4". It acknowledges that "Public Notice No. 60 does not impose upon the joint-ventures a legal obligation to conclude an MOU with the Indian Government. Nonetheless, the conclusion of an MOU is a

³⁸⁷ First Submission, para 62.

³⁸⁸ First Submission, para 46.

necessary condition for obtaining an advantage: the grant of licenses for SKD and CKD kits. Moreover, once it is signed, the MOU is binding upon the signatory and legally enforceable under the FTDR." The United States argues similarly that once the MOUs are entered into, the commitments in them become part of the conditions under which the MOU signatories are permitted to receive import licenses. These conditions are to be enforced through the import licensing scheme and also as a legal contract between the Government of India and the manufacturer. The United States argues that the fact that the firms could have chosen not to sign such contracts does not affect the analysis: "the term "requirement", in its ordinary meaning, encompasses such preconditions to obtaining a benefit from the Government".

7.183 In previous instances, the term "requirement" has been interpreted to encompass commitments entered into on a voluntary basis by individual firms as a condition to obtaining an advantage. ³⁹¹ Under GATT 1947, in *EEC – Parts and Components*, the panel noted that:

"Article III:4 refers to 'all laws, regulations or requirements affecting the internal sale, offering for sale, purchase, transportation or use'. The Panel considered that the comprehensive coverage of 'all laws, regulations or requirements affecting' (emphasis added) the imported sale, etc. of imported products suggests that not only requirements which an enterprise is legally bound to carry out, such as those examined by the 'FIRA panel' (BISD 30S/140, 158), but also those which an enterprise voluntarily accepts in order to obtain an advantage from the government constitute 'requirements' within the meaning of that provision."³⁹²

7.184 GATT jurisprudence thus suggests two distinct situations which would satisfy the term "requirement" in Article III:4:

- (i) obligations which an enterprise is "legally bound to carry out";
- (ii) those which an enterprise voluntarily accepts in order to obtain an advantage from the government

7.185 This interpretation was confirmed in the panel report on *Canada – Autos*) under the WTO. In that instance, "letters of undertaking" submitted by certain firms at the request of the Canadian Government were considered to be "requirements". That panel stated in particular that:

³⁸⁹ First Submission, para. 67.

³⁹⁰ First Submission, para 68.

See Canada – FIRA: "the Panel first examined whether the purchase undertakings are to be considered "laws, regulations or requirements" within the meaning of Article III:4. As both parties had agreed that the FIRA and the FIR Regulations – whilst providing for the possibility of written undertakings – did not make their submission obligatory, the question remained whether the undertakings given in individual cases are to be considered "requirements" within the meaning of Article III:4. In this respect the Panel noted that Section 9(c) of the Act refers to "any written undertakings ... relating to the proposed or actual investment given by any party thereto conditional upon the allowance of the investment" and that Section 21 of the Act states that "where a person who has given a written undertaking ... fails or refuses to comply with such undertaking" a court order may be made "directing that person to comply with the undertaking". The Panel further noted that written purchase undertakings – leaving aside the manner in which they may have been arrived at (voluntary submission, encouragement, negotiation, etc.) once they were accepted, became part of the conditions under which the investment proposals were approved, in which case compliance could be legally enforced. The Panel therefore found that the word "requirement" as used in Article III:4 could be considered a proper description of existing undertakings."

³⁹² Panel Report on *EEC – Parts and Components*, para .5.21, BISD 37S/132.

"We note that it has not been contested in this dispute that, as stated by previous GATT and WTO panel and appellate body reports, Article III:4 applies not only to mandatory measures but also to conditions that an enterprise accepts in order to receive an advantage, including in cases where the advantage is in the form of a benefit with respect to the conditions of importation of a product. The fact that compliance with the CVA requirements is not mandatory but a condition which must be met in order to obtain an advantage consisting of the right to import certain products duty-free therefore does not preclude application of Article III:4."³⁹³

- 7.186 This Panel supports and adopts that interpretation.
- (ii) The indigenization condition as a "requirement"

7.187 The Panel recalls that the indigenization condition is contained in two distinct instruments: Public Notice No. 60 and the MOUs signed by manufacturers under the terms of Public Notice No. 60. The Panel will consider both of these instruments in determining whether the indigenization condition can be considered to constitute a "requirement within the meaning of Article III:4.

7.188 Public Notice No. 60, a governmental measure adopted under the authority of the EXIM policy, clearly requires that an MOU must be signed in order to gain the right to apply for an import license: import licenses will only be granted to manufacturers who sign an MOU containing certain conditions, including indigenization and trade balancing. Under Public Notice No. 60, the signing of an MOU is therefore in itself a condition to obtaining a license. In other words, acceptance of the indigenization condition was necessary under Public Notice No. 60 in order to obtain the right to import the restricted items addressed in Public Notice No. 60. Thus, the indigenization condition, as contained in Public Notice No. 60, constitutes a condition to the granting of an advantage, namely, in this instance, the right to import the restricted kits and components. It therefore constitutes a requirement within the meaning of Article III:4. 394

7.189 The MOUs themselves also contain the same conditions, including the indigenization condition, whose acceptance as legal obligations by the signatories was necessary in order to obtain the right to import the restricted kits and components under license. They were thus entered into in order to gain an advantage, consisting in the right to import these products. In and of itself, this leads to the conclusion that they constitute "requirements" as per the interpretation from *Canada –Autos* that this Panel has endorsed.

7.190 In addition, manufacturers are expected to comply with the terms of the MOUs they have signed. Once signed, the MOUs became binding and enforceable, first under Public Notice No. 60 itself, and also under the FTDR Act and under general principles of contract law. Prior to 1 April 2001, failure to comply with these conditions could lead to the denial of an import license. The MOUs have thus been binding and enforceable since their signature. A binding enforceable condition seems to fall squarely within the ordinary meaning of the word "requirement", in particular as "a condition which must be complied with".

Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, page 10.73

³⁹⁵ New Oxford English Dictionary, as cited above.

modified by the Appellate Body Report, para. 10.73.

394 We recall in this regard that in the context of this analysis, it is not the WTO-consistency or inconsistency of the advantage granted that is at stake. In other words, the fact that these products were restricted and subject to import licensing is not what is at stake in this determination. In this regard, previous jurisprudence has clarified that the advantage to be gained might be of very diverse natures.

- 7.191 The enforceability of the measure in itself, independently of the means actually used or not to enforce it, is a sufficient basis for a measure to constitute a requirement under Article III:4. As far as a manufacturer having entered into an MOU is concerned, the commitments have existed since they were entered into and would affect its commercial behaviour. This further supports the conclusion that the "indigenization" condition as contained in the MOUs themselves also constitutes a "requirement" within the meaning of Article III:4.
- 7.192 The Panel is supported in this view by the jurisprudence³⁹⁶ and by the fact that the TRIMs agreement expressly refers to mere enforceability in the context of the introductory paragraph of the Illustrative List, Item 1. Where the jurisprudence is concerned, the Panel recalls that the panel in *US Malt Beverages* found the mandatory price affirmation laws in Massachusetts and Rhode Island inconsistent with Article III:4 because they accorded imported beer and wine less favourable treatment than the like domestic products even though the States concerned might not have been using their police powers to enforce the legislation. The panel noted that this did not change the fact that the measures were mandatory legislation that may influence the decision of economic operators.
- 7.193 Therefore, the Panel finds that the indigenization condition, as contained in Public Notice No. 60 and in the MOUs signed thereunder, constitutes a "requirement" within the meaning of Article III:4.
- 7.194 India has confirmed that the MOUs have remained binding and enforceable since then, even though the means through which enforcement might be pursued may have changed and may currently not have been decided on. India has argued that there is a fundamental difference between a requirement that has to be fulfilled to obtain an import license under the Indian trade laws and a requirement under private contract law with the Government of India, which the Government is free to enforce or not. In the view of India, the MOUs have become, since 1 April 2001, private contractual obligations that amount to discretionary legislation, and thus cannot be challenged as such under WTO dispute settlement.³⁹⁷ This argument relates to an alleged distinction between the situation pre- and post-April 2001. The Panel recalls in this respect its earlier conclusion that it would address arguments relating to any changes in the measures after 1 April 2001 when considering what recommendations may appropriately be made to the DSB.
- (c) ... affecting the internal sale, offering for sale, purchase (...) of the products
- 7.195 Having determined that the indigenization condition is a "requirement", the Panel next examines whether the Indian measures affect the internal sale, offering for sale, purchase or use of the imported products within the meaning of Article III:4.

 $^{^{396}}$ The Canada-FIRA panel, in considering an argument by Canada that the undertakings were "private contractual" arrangements, found that:

[&]quot;The Panel carefully examined the Canadian view that the purchase undertakings should be considered as private contractual obligations of particular foreign investors vis-à-vis the Canadian government. The Panel recognized that investors might have an economic advantage in assuming purchase undertakings, taking into account the other conditions under which the investment was permitted. The Panel felt, however, that even if this was so, private contractual obligations entered into by investors should not adversely affect the rights which contracting parties, including contracting parties not involved in the dispute, possess under Article III:4 of the General Agreement and which they can exercise on behalf of their exporters. This applies in particular to the rights deriving from the national treatment principle, which – as stated in Article III:1 – is aimed at preventing the use of internal measures 'so as to afford protection to domestic production'." (paras 5.4 to 5.6).

7.196 Under GATT and WTO jurisprudence, the term "affecting" has consistently been defined broadly. In particular, it has been well established that it "implies a measure that has "an effect on" and this indicates a broad scope of application". This term therefore goes beyond laws and regulations which *directly* govern the conditions of sale or purchase to cover also any laws or regulations which might adversely modify the conditions of competition between domestic and imported products.

7.197 The Indian measure at issue requires the MOU signatories to commit to achieving a level of indigenization of components up to a minimum level of 50% in the third year or earlier and 70% in the fifth year or earlier, in order to obtain import licenses. To meet the indigenization requirement, car manufacturers must purchase Indian parts and components rather than imported goods. This provides an incentive to purchase local products. Such a requirement "modifies the conditions of competition between the domestic and imported products" and therefore affects the internal sale, offering for sale, purchase and use of imported parts and components in the Indian market within the meaning of Article III:4 of the GATT 1994.

7.198 The Panel also notes that the indigenization requirement affects not only SKD and CKD kits, but also any imported parts or components. This is because indigenization is required over all automotive parts and components. Any required components that are sourced from imports would not count towards fulfilling the level of indigenization requirement that increases over a period of time.

(d) Whether imported products are accorded less favourable treatment

7.199 Finally, it is necessary to determine whether the measure is such as to accord less favourable treatment to the imported goods. In determining whether imported products are treated less favourably than domestic products, the Panel is obliged to examine whether the Indian measure modifies the conditions of competition in the Indian market to the detriment of imported products. The Appellate Body stated in *Korea—Beef* that "[a]ccording treatment 'no less favourable' means...according conditions of competition no less favourable to the imported product than to the like domestic product."³⁹⁹

7.200 In *Japan—Alcoholic Beverages*, the Appellate Body stated:

"The broad and fundamental purpose of Article III is to avoid protectionism in the application of internal tax and regulatory measures. More specifically, the purpose of Article III "is to ensure that internal measures 'not be applied to imported or domestic products so as to afford protection to domestic production". Toward this end, Article III obliges Members of the WTO to provide equality of competitive conditions for imported products in relation to domestic products. [T]he intention of the drafters of the Agreement was clearly to treat the imported products in the same way as the like domestic products once they had been cleared through customs. Otherwise indirect protection could be given".

³⁹⁸ See *Italy – Agricultural Machinery*, BISD 7S/60, para. 12. See also the Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.80.

³⁹⁹ Report of the Appellate Body in *Korea – Various Measures on Beef*, WT/DS161/AB/R, WT/DS169/AB/R, adopted on 10 January 2001, para. 135.

⁴⁰⁰United States - Section 337 of the Tariff Act of 1930, BISD 36S/345, para. 5.10.

⁴⁰¹United States - Taxes on Petroleum and Certain Imported Substances, BISD 34S/136, para. 5.1.9; Japan - Customs Duties, Taxes and Labelling Practices on Imported Wines and Alcoholic Beverages, BISD 34S/83, para. 5.5(b).

7.201 As noted previously, the very nature of the indigenization requirement generates an incentive to purchase and use domestic products and hence creates a disincentive to use like imported products. This requirement is more than likely to have some effect on manufacturers' choices as to the origin of parts and components to be used in manufacturing automotive vehicles, since they need to take into account the requirement to use a certain proportion of products of domestic origin. Car manufacturers are required to purchase a certain amount of parts and components that are of Indian origin for the simple reason that this is the only way to meet the indigenization requirement. This amount increases over a period of time and hence so too does the disincentive to the use of imported components. Car manufacturers are thus not free to choose to purchase imported parts and components over domestic parts and components in excess of a certain proportion. In these circumstances, imported products cannot compete on an equal footing with Indian-origin parts and components because the indigenization requirement explicitly sets out the percentage of domestic parts and components that must be used. 402

7.202 Such a requirement clearly modifies the conditions of competition of domestic and imported parts and components in the Indian market in favour of domestic products. The Panel therefore finds that the indigenization requirement foreseen in Public Notice No. 60 and incorporated in the MOUs signed thereunder, by requiring manufacturers to use specified percentages of parts and components of domestic origin is a requirement affecting the internal sale of like imported products, and which affords these imported products less favourable treatment than that accorded to like products of national origin.

7.203 The Panel also recalls in particular the conclusions of the Panel on Canada – Autos, in examining a "Canadian Value Added" requirement (CVA) that "the CVA requirements accord less favourable treatment within the meaning of Article III:4 to imported parts, materials and non-permanent equipment than to like domestic products because, by conferring an advantage upon the use of domestic products but not upon the use of imported products, they adversely affect the equality of competitive conditions of imported products in relation to like domestic products". 403

(e) Conclusion

7.204 The Panel thus finds the "indigenization" condition, as contained in Public Notice No. 60 and in the MOUs entered into thereunder, is in violation of Article III:4 of GATT 1994 as at the date of its establishment.

7.205 This conclusion is consistent with the fact that the TRIMs Agreement Illustrative List identifies measures which require "the purchase or use by an enterprise of products of domestic origin or from any domestic source, whether specified in terms of particular products, in terms of volume or value of products, or in terms of a proportion of volume or value of its local production" as being TRIMs inconsistent with Article III:4 of GATT 1994. In this instance, MOU signatory manufacturers are required to use a certain proportion of products of domestic origin in their local production.

⁴⁰² A GATT panel on *EEC – Parts and Components* made a finding of violation of Article III:4 involving relatively similar circumstances: "The Panel noted that the EEC made the *grant of an advantage*, namely the suspension of proceedings under the anti-circumvention provision, *dependent on undertakings to limit the use of parts or materials of Japanese origin without imposing similar limitations on the use of like products of EEC or other origins* hence dependent on undertakings to accord treatment to imported products less favourable than that accorded to like products of national origin in respect of their internal use".

⁴⁰³ Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.85.

2. Claim under Article XI

7.206 The United States has claimed that the indigenization requirement is also inconsistent with Article XI:1 of GATT 1994. The Panel has found this requirement to be in violation of Article III:4 of the GATT 1994.

7.207 The Panel recalls that, under the terms of Article 11 of the DSU, it is only required to address "those issues which must be addressed in order to resolve the matter in issue in the dispute". This implies that if a particular measure is found to be in violation of a particular provision, the Panel does not necessarily need to go on to consider whether the same measure is also inconsistent with another provision of the GATT.

7.208 In this instance, the Panel has found the indigenization condition in its entirety to be inconsistent with Article III:4. In these circumstances, the Panel does not see any need to address in addition whether the same condition is also inconsistent with Article XI:1 of the GATT. It therefore exercises judicial economy with respect to that claim.

F. GATT CLAIMS CONCERNING THE TRADE BALANCING CONDITION

1. Factual aspects of the measure

7.209 Subparagraphs 3(i) through (iv) of Public Notice No. 60 set out four obligations which an MOU imposes on a manufacturing company that is a signatory. Subparagraph 3(iv) contains the "trade balancing" obligation.

7.210 MOU signatories must achieve "broad trade balancing of foreign exchange over the entire period of the MOU", in terms of balancing between the actual CIF value of imports of CKD/SKD kits/components and the FOB value of exports of cars and auto components over that period. Other parts of the MOU indicate the time-frame for that obligation. While a signatory has a trade balancing obligation over the period of the MOU, there is a two-year moratorium during which the firm does not need to fulfill that commitment. The period of execution of the export obligation therefore begins from the third year of commencement of production. However, imports made during the moratorium count towards the firm's total export obligation under the MOU.

7.211 Two other aspects of the alleged administration of this measure are of particular relevance to the Panel's deliberations and will be addressed at a later stage of analysis. The first is that, although there is no express provision to that effect in Public Notice No. 60, it has been confirmed by India that purchases by an MOU signatory within India of previously imported CKD/SKD kits/components that were subject to a license requirement on their importation, also count towards the signatory's export balancing requirement.⁴⁰⁵

7.212 The second is the assertion by India that the trade balancing obligation has only ever applied in respect of products that were subject to discretionary import licensing. India has indicated that since 1 April 2001, the import restrictions for SKD/CKD kits and car components have been abolished and no licenses are required. As from that date, the signatories of existing MOUs would continue to be required to discharge the export obligations corresponding to the imports made by them

 $^{^{404}}$ See Report of the Appellate Body in US-Wool Shirts and Blouses, WT/DS33/AB/R, adopted on 23 May 1997, p.18 as cited above (DSR 1997:I, 323).

 $^{^{405}}$ There is no dispute between the parties as to this assertion and we address its legal significance separately below.

before that date. They would no longer incur any *new* export obligations as a result of the further importation of SKD/CKD kits since these are no longer restricted. 406

7.213 There is a factual dispute between the European Communities and India on this aspect of the scope of the export obligation. The European Communities argues that under the terms of the MOUs and of Public Notice No. 60, the requirement applies to "imports of kits/components" generally, i.e. not just those kits and components which are or were subject to licensing. The European Communities also argues that the trade balancing requirement, irrespective of the scope of the products covered, would continue to apply so long as the signatories have not achieved the 70% indigenization level.

2. Order of analysis of the claims

7.214 The Panel recalls that in considering the order of examination of the claims concerning the indigenization condition, it decided to first examine the claim of violation of Article III:4 which was common to both complainants. On the trade balancing issue, the Panel again wishes to follow the same approach, namely to seek to determine whether common ground between the claims can be identified to assist in determining the order of their examination. This seem to the Panel the best means of allowing for the possible exercise of judicial economy while respecting its obligation under Article 9 of the DSU with regard to the preservation of the rights and obligations of parties in cases involving multiple complaints.

7.215 Both complainants assert that Articles III and XI respectively apply to at least some aspects of the measure. However their claims differ somewhat in their coverage:

- the European Communities merely makes a claim of violation of Article III:4 in relation to its impact upon purchases made on the domestic market. It considers all other aspects of the measure to fall under Article XI.
- The United States makes a claim of violation of Article III:4 in relation to both purchases on the domestic market and in relation to imports made directly by the signatories. While it thus takes a broader view than the European Communities as to the relevance of Article III, it also makes a distinct comprehensive claim of violation based on Article XI:1.⁴⁰⁸
- 7.216 While the claim that the trade balancing condition is inconsistent with Article XI is common to both complainants, the claim of inconsistency with Article III:4 argued by the United States involves aspects of that condition which the European Communities does not argue to be inconsistent with Article III.4. The scope of the United States's claim under Article III:4 is therefore much broader than that of the European Communities. In this instance, because both the European Communities' and the United States' claims seek to bring the entire measure within Article XI and because the European Communities addresses a wider range of effects under that Article than under Article III, the Panel has resolved to begin its analysis of trade balancing under Article XI.

⁴⁰⁶ It is not clear, despite a question put to all three parties on the issue, what the time-frame is for the execution of outstanding export requirements. Nothing turns on this from a legal point of view, although it affects the degree of commercial significance of the residual obligation.

⁴⁰⁷ The Panel will address this factual question and its legal ramifications below in the context of analysing the scope of the EC's claim. See below, section 4 b.

The United States appears to have ultimately argued in the alternative. It has stated in its responses to the Panel's questions that the measure might be covered by one or the other provision, but in any case it should be found contrary to at least one of the two.

3. Relationship between Articles III and XI

7.217 While the Panel has resolved to begin its analysis with Article XI, the parties have addressed the relative scope of Articles III and XI and their relationship to each other, in light of India's arguments concerning these issues. In light of this, the Panel felt it appropriate to begin by outlining its general interpretative approach to this Article and these submissions.

7.218 India has argued that the trade balancing condition cannot fall within the purview of Article XI. It asserts that Article XI only applies to border measures. It considers that this measure is not a border measure because it does not relate to the "process" of importation. It suggests that it should be considered as an internal measure and hence be analysed under Article III. This argument will be addressed in due course in our analysis under Article XI.

7.219 More generally, India has warned of the systemic implications of applying Article XI to other than "border" measures, as to the effect on the respective scopes of Articles III:4 and XI:1 of GATT. In particular, there may be a concern that too broad an interpretation of Article XI might lead to it inappropriately covering ground intended for Article III. Others might assert a contrary concern with a broad interpretation of Article III:4.

7.220 In that regard the Panel notes that Articles III and XI seem to have been considered to have distinct scopes of application. In *Canada –FIRA* a panel stated that:

"The Panel shares the view of Canada that the General Agreement distinguishes between measures affecting the 'importation' of products, which are regulated in Article XI:1, and those affecting 'imported products', which are dealt with in Article III. If Article XI:1 were interpreted broadly to cover also internal requirements, Article III would be partly superfluous."

- 7.221 Only in the very specific circumstance of state trading enterprises involving a monopoly over both importation and distribution of goods has a blurring of the traditional distinction between measures affecting imported products and measures affecting importation been acknowledged.⁴¹⁰
- 7.222 The Panel feels that it is vital that the task be approached solely through an application of the customary rules of interpretation of public international law as required by Article 3.2 of the DSU. This should occur without any presumption as to some preordained or systemic balance between the two Articles. The customary rules provide sufficient mechanisms to ensure an appropriate outcome that should deal with such concerns, as they require consideration of ordinary meaning in context and in the light of object and purpose of the treaty. In this regard, context includes a reading of each Article in relation to other potentially relevant provisions and an analysis, where necessary, of any differences in terminology. The principle of effectiveness would also apply to prevent reducing any provision to inutility.
- 7.223 While other provisions in the WTO Agreement may usefully be considered as part of the context which informs the meaning of a given provision, the scope of that provision should not be

.

⁴⁰⁹ Panel Report, L/5504, adopted on 7 February 1987, para. 5.14.

⁴¹⁰ See the Panel Report on *Korea – Various Measures on Beef*, WT/DS161/R, WT/DS169/R, adopted on 10 January 2001, as modified by the Appellate Body Report, para. 766: "Based on the panel findings in the *Canada – Marketing Agencies (1988)* case, the Panel considers that to the extent that LPMO fully controls both the importation and distribution of its 30 per cent share of Korean beef quota, the distinction normally made in the GATT between restrictions affecting the importation of products (i.e. border measures) and restrictions affecting imported products (i.e. internal measures) loses much of its significance."

assumed *a priori* to vary depending on the mere presence of other provisions which may have some relevance to the situation: the contours of a provision should flow from its terms, as read in context alongside the other provisions of the agreement. The Panel is also mindful of the fact that different aspects of a particular measure may legitimately be covered by distinct provisions of the WTO Agreements.

7.224 For all the foregoing reasons, it therefore cannot be excluded *a priori* that different aspects of a measure may affect the competitive opportunities of imports in different ways, making them fall within the scope either of Article III (where competitive opportunities on the domestic market are affected) or of Article XI (where the opportunities for importation itself, i.e. entering the market, are affected), or even that there may be, in perhaps exceptional circumstances, a potential for overlap between the two provisions, as was suggested in the case of state trading. Any analysis of the applicability of either Article III:4 or XI:1 should thus be based on the principles within Article 3.2 of the DSU. This is how the Panel will proceed in its examination.

4. Claims under Article XI:1

- 7.225 Both the United States and the European Communities argue that the trade balancing requirement is inconsistent to Article XI:1 of GATT. Although the substance of their claims is essentially similar, they differ somewhat in that the European Communities initially made its claim subject to some specific conditions. This will need to be considered as a preliminary matter.
- (a) Scope of the European Communities' claim
- 7.226 The European Communities' request for establishment of a panel indicated the following:

"The measures concerned are contained in:

- Public Notice No. 60 (PN 97-02) of the Indian Ministry of Commerce, effective 12 December 1997; and
- the Memorandums of Understanding (MOUs) entered into by certain manufacturers of automobiles with the Government of India pursuant to Public Notice No. 60.

The above measures require manufacturers of automobiles to sign an MOU as a condition for obtaining licenses for importing automotive products that are currently subject to import restrictions. According to the Indian authorities, the MOUs are "binding" and "enforceable" instruments, which shall remain valid after the date when those restrictions are eliminated. The MOUs require (i) to establish "actual production facilities" in India; (ii) to make a minimum investment; (iii) to achieve a certain level of "indigenisation"; and (iv) to export a certain amount of automotive products.

The European Communities considers that the requirements imposed by the above measures are in violation of Articles III:4 and XI:1 of GATT and of Article 2.1 of the TRIMs Agreement."

- 7.227 In its First Submission, however, the European Communities indicated that it was making the claim of violation of Article XI:1 concerning the trade balancing obligation only in so far as:
- prior to 1 April 2001, the MOUs required the "trade balancing" of imports of "components" other than chassis and bodies; and

- the MOUs would remain binding and enforceable after 1 April 2001, both with respect to passenger cars and components therefor.
- 7.228 In the course of the proceedings, the European Communities further clarified the scope of its claims, and as was noted above, a disagreement arose and persisted between the European Communities and India as to the scope of application of the trade balancing condition. The Panel will address this issue first.
- (i) The scope of the trade balancing obligation
- 7.229 The European Communities contends that because the plain meaning of Public Notice No. 60 and the MOUs does not limit the reference in the trade balancing obligation to imports subject to licensing, it remains concerned that accrued export obligations might include calculations based on non-restricted imports.
- 7.230 India argues that the regime has never been interpreted or administered in that way. It asks the European Communities to rely on its assurances to that effect. In response to questions from the Panel as to the way this distinction was effectively administered, India explained that ongoing calculation of the export obligation was based on the actual imports made under the import licenses themselves. In response to similar questions, the European Communities was not able to provide any evidence of calculations made contrary to India's assertions. It relied instead on the plain meaning of the documents concerned.
- 7.231 The Panel notes that the terms of Public Notice No. 60, as reflected in the standard MOU, refer to an obligation to balance "the actual CIF value of imports of CKD/SKD kits/components" (subparagraph 3(iv)). It is true that it is not expressly mentioned in this phrase that this refers exclusively to *restricted* kits and components. Nevertheless, these terms should be considered in their proper context. Paragraph 2 of Public Notice No. 60 provides as follows: "Pursuant to the above, imports of components for motor vehicles in CKD/SKD form, *which is restricted for import under the current Export Import Policy*, shall be allowed against a license (...)" (emphasis added). This suggests that in the context of Public Notice No. 60, it could be argued that where "imports of CKD/SKD kits/components" are referred to, the reference may be to those "components for motor vehicles in CKD/SKD form" defined in paragraph 2 of the Public Notice No. 60, i.e. the restricted items under the EXIM Policy. The Panel is not persuaded that the ordinary meaning of these words, in this context, would encompass any non-restricted components. At most, the plain meaning of the expression, in context, appears ambiguous.
- 7.232 In addition, as noted above the European Communities has adduced no additional evidence to suggest that the reading it suggests is indeed that which prevails, while India has provided an indication of how it assesses the obligation which suggests a direct reliance on the licenses themselves.
- 7.233 In these circumstances, the European Communities has not proven on balance that the trade balancing requirement accrues in relation to non-restricted imports. A mere reference to a plain meaning which, read in context, is debatable, without reference to any evidence of relevant domestic Indian interpretations or reference to evidence from any signatory in support of the contention and in the face of India's clear assertions as to methodology to the contrary, is not sufficient to support a factual finding of that nature. There is thus no basis for any specific ruling on the consistency with Article XI of the trade balancing condition as it allegedly applies to non-restricted imports as requested by the European Communities and the Panel declines to do so.

7.234 In ruling on this factual question, the Panel wishes to point out that it has no jurisdiction to make any binding inter-party determination as to the actual level of the export obligation accrued by any signatory under individual MOUs. It is merely saying that it has not been demonstrated, for the purposes of this proceeding, that this condition applies to the importation of non-restricted parts or components. The Panel's ruling in this regard cannot be determinative of what interpretation might be otherwise made of the scope of this condition in the MOUs on an inter-party basis.

(ii) The continued enforceability of the MOUs after 1 April 2001

7.235 The European Communities also indicated in its First Submission that it was making a claim of violation of Article XI in relation to the trade balancing only to the extent that the MOUs remained binding and enforceable after 1 April 2001. The European Communities specifically argued that the trade balancing requirement would continue to apply so long as the MOU signatories have not achieved the 70% indigenization level. The Panel has previously noted India's confirmation that the MOUs did remain binding and enforceable after that date. More specifically, India confirmed that it did not intend to release signatories from obligations incurred in relation to imports made before 1 April and that the indigenization condition remained in place.

7.236 In response to questions from the Panel, the European Communities further explained the qualified submission it had made at that stage. It confirmed its view that the violation it was referring to would have existed as of the date of establishment of the Panel already, but that as a factual matter, since India had been otherwise authorized to maintain quantitative restrictions on these products beyond that date, there would have been little practical interest in challenging the measures earlier unless India intended to maintain them after 1 April 2001. The European Communities further clarified in the course of the proceedings that it was nonetheless generally requesting the Panel to make findings that the measures were in violation of the WTO Agreements as of the date of establishment of the Panel, and also that they subsequently remained in violation of the same provisions.

7.237 The Panel finds the European Communities' explanations of the scope of its claim under Article XI to have been somewhat confusing at the early stages of the proceeding. All Nevertheless, the European Communities referred to this claim in its request for establishment of a panel and has clarified that it is essentially requesting the Panel to rule on the measures as they existed as of the date of its request for establishment of a panel and as they, in its view, continued to exist subsequently. In the context of the Panel request and terms of reference, the statements as a whole lead the Panel to conclude that the claim presented by the European Communities under Article XI concerning trade balancing is appropriately within its terms of reference.

7.238 The Panel understands the European Communities' articulation of its claim in its First Submission as meaning that it wished to pursue this particular claim only if, as a factual matter, the MOUs remained binding and enforceable beyond 1 April 2001, rather than as a suggestion that the Panel's legal assessment of the claim should be on the basis exclusively of the situation as of 1 April 2001. The European Communities confirmed that any violations it was requesting the Panel to make were violations that would have existed at the time of establishment of this Panel.

⁴¹¹ See First Submission, paras. 11 ff.

to the post-April situation. The Panel is on balance satisfied that the explanations provided by the European Communities in the course of the proceedings have clarified that they intended the Panel to examine a claim of violation which would have existed as of its date of establishment, which is appropriate, although the motivation for the European Communities' request, which is not for a Panel to pass judgment upon, may have rested principally on the expectation of the effects of a finding after the 1 April 2001. The DSU allows Members to exercise their own judgment as to when a case is worth pursuing.

7.239 In light of the foregoing, the Panel considers it appropriate to rule on the EC's claim on the basis of the measures as they existed as of the date of their request for establishment of this Panel, in accordance with the approach determined in section B above. Any implications of the events of 1 April 2001 in relation to the continuation of these violations will be addressed, as determined in the same section, at a later stage.

(b) Arguments of the parties

7.240 The United States argues that starting in the fourth year, each MOU imposed a quantitative limitation on imports through the trade balancing obligation. It asserts that the trade balancing requirement restricts imports because it limits the value of an MOU signatory's imports to the value of the signatory's exports (which, pursuant to paragraph III, clause (vi), the MOU signatory was required to specify at the time of signing). Because there would obviously be limitations on the amount of exports which a car manufacturer might be able or willing to make, it argues that the trade balancing obligation itself restricts imports.

7.241 In addition, the United States notes that the obligation is clearly enforceable. The Government of India had confirmed that denial of an import license was effectively mandatory if the trade balancing obligation was not met. A manufacturer's failure to comply with an MOU obligation could lead to loss of import privileges or to confiscation of the goods concerned pursuant to various provisions of the FTDR Act and regulations thereunder. It appeared that after April 1, 2001, these additional provisions would be the instruments through which India would enforce Public Notice No. 60 and the MOUs (and thus prevent SKD/CKD kits/components from being brought into India to compete with domestic parts and components). For all of these reasons, it considered that the trade balancing obligation in Public Notice No. 60 and the MOUs imposed import restrictions as such, and was therefore inconsistent with Article XI:1 of the GATT 1994.

7.242 The United States further refers to Item 2(a) of the Illustrative List of the TRIMs Agreement to support its argument, asserting that the MOUs restricted importation to an amount related to the value of locally produced goods that a manufacturer exports, as foreseen by that item, and the MOUs were "enforceable". The Untied States argued that India had confirmed both points. The trade balancing requirement thus fell within the scope of this paragraph, and consequently it was "inconsistent with the obligation of general elimination of quantitative restrictions provided for in paragraph 1 of Article XI of GATT 1994".

7.243 The European Communities argues that the trade balancing obligation stipulated in the MOUs is inconsistent with GATT Article XI:1 in that it restricts imports of passenger cars, and of components therefor, by the signatories of the MOUs. In the view of the EC, it does so because it places a maximum limit on the value of the imports that the signatories are authorised to make, equal to the value of their exports. In practice, there were limits to the amount of exports which a signatory might be able or willing to make, related both to its manufacturing capacity in India and to the demand for its products in foreign markets. Thus, by limiting the amount of a signatory's imports to that of its exports, the trade balancing obligation "restricted" the amount of imports. The European Communities also refers to Item 2(a) in the TRIMs Illustrative List to support its argument under Article XI:1.

⁴¹³ India's First Submission, para. 14, and *Replies by India to Questions Posed by Japan*, G/TRIMS/W/15, circulated 30 October 1998, response to Question 24; Exhibit US-5: "CKD/SKD kits imports would be allowed with reference to the extent of export obligation fulfilled in the previous year." As discussed above, the MOUs were also "necessary to obtain an advantage".

⁴¹⁴ US response to Question 82 of the Panel.

7.244 As noted above, India first argues that the trade balancing obligation does not fall within the purview of Article XI because it is not a border measure and does not relate to the "process" of importation. It also argues that in any case, after 1 April 2001, the execution of outstanding export obligations will in no way be tied to imports and therefore will be a straight export requirement, which is not in itself prohibited by any provision of GATT 1994. This second aspect of India's defense, which relates to a distinction between the measures as they applied prior to 1 April 2001 and after that date, will be addressed as necessary in our section on Recommendations.

- (c) Scope of Article XI:1
- (i) The notion of "measures" within the meaning of Article XI:1
- 7.245 Article XI:1 provides that:

"No prohibition or restriction other than duties, taxes or other charges, whether made effective through quotas, import or export licenses or other measures, shall be instituted or maintained by any [Member] on the importation of any product of the territory of any other [Member] or on the exportation or sale for export of any product destined for the territory of any other [Member]."

7.246 Article XI:1 refers to restrictions "made effective through quotas, import or export licenses or other measures". This formulation, which includes a "broad residual category" of "other measures", suggests a broad scope to the types of measures which can be considered to fall within the meaning of Article XI:1.

7.247 Past jurisprudence supports such a broad interpretation. The Panel recalls in particular the conclusion of the panel in *Japan – Semi-conductors* that

"Article XI:1, unlike other provisions of the General Agreement, did not refer to laws or regulations but more broadly to measures. This wording indicated clearly that any measure instituted or maintained by a contracting party which restricted the exportation or sale for export of products was covered by this provision, irrespective of the legal status of the measure" 416

7.248 Such a broad interpretation of the notion of "measure" has also been subsequently endorsed in the context of an analysis of that notion under Article XXIII:1 (b),⁴¹⁷ and more generally in the context of WTO dispute settlement.⁴¹⁸

⁴¹⁵ As described in the Panel Report on *Argentina – Hides Leather*, WT/DS155/R, adopted on 16 February 2001, para. 11.17.

February 2001, para. 11.17.

Ale Report of the Panel on *Japan – Semi-conductors*, adopted on 4 May 1988, BISD35S/116, para. 106.

Ale Panel Report on *Japan – Film*, WT/DS44/R, adopted on 22 April 1998, para. 10.45 (DSR 1998:IV, 1179).

⁴¹⁸ See Report of the Appellate Body in *Guatemala – Cement I*, WT/DS60/AB/R, adopted on 25 November 1998, footnote 47: "In the practice established under the GATT 1947, a "measure" may be any act of a Member, whether or not legally binding, and it can include even non-binding administrative guidance by a government (see *Japan – Semi-conductors*, adopted 4 May 1988, BISD 35S/116). A measure can also be an omission or a failure to act on the part of a Member (see, for example, *India – Patents (US)*, WT/DS50/R and WT/DS50/AB/R, adopted 16 January 1998, and also *India – Patents (EC)*, WT/DS79/R, adopted 22 September 1998)." Interestingly, this analysis appears to encompass together analyses of the notion of "measure" as contained in different provisions of the GATT 1994, including Article XI:1.

- 7.249 Thus, under past jurisprudence, a network of non-binding incentive measures has been considered to constitute "measures" and a restriction within the meaning of Article XI:1 of GATT. Such a broad interpretation is consistent both with the express terms of the provision, which are all-encompassing, and also more generally with its context, including other provisions defining the type of measures which can be validly be submitted to dispute settlement.
- 7.250 It is notable in this respect that the more specific terms of Article III:4 ("laws, regulations or requirements"), as noted in our analysis of the indigenization, has been broadly interpreted to cover measures which are not mandatory but are either binding and enforceable or necessary in order to obtain an advantage. The Panel sees no reason to interpret the more general notion of "measure" under Article XI:1 more restrictively, especially as these two provisions, as the *Superfund* panel stated, have "essentially the same rationale, namely to protect expectations of the contracting parties as to the competitive relationship between their products and those of the other contracting parties".
- 7.251 In this instance, India does not seem to dispute that Public Notice No. 60 constitutes a governmental measure which could properly be the object of dispute settlement. It is less clear, however, whether it accepts that the MOUs as such can constitute such "measures", including within the meaning of Article XI:1 of the GATT 1994.
- 7.252 As previously noted, the MOUs have been signed between the Government of India and automotive manufacturers, in accordance with the terms of Public Notice No. 60, a government measure which provides the framework and basis for these MOUs. As noted above in the context of our analysis of the indigenization condition, the MOUs constitute binding and enforceable contracts, whose signature with all the specified conditions, including trade balancing, was necessary in order to obtain an advantage, namely the right to import restricted auto kits and components. The acceptance of this particular condition was thus in itself necessary to obtain the advantage sought (namely, the right to import restricted items). Although they were not mandatory in the sense of being directly applicable to any manufacturer outside of that manufacturer's acceptance of it, a manufacturer wishing to import restricted kits and components had no choice but to sign an MOU. In addition, the terms of these MOUs, including the trade balancing condition, were dictated by a pro-forma required by a governmental measure, Public Notice No. 60, which clearly set out the requirements to be accepted by car manufacturers for importation of restricted kits and components. The MOUs emanate directly from a government measure, Public Notice No. 60, which requires car manufacturers to execute an MOU and comply with the terms contained therein to be able to import the kits and components.
- 7.253 For these reasons, the Panel finds that the trade balancing condition, as contained in both Public Notice No. 60 and the MOUs constitutes a "measure" within the meaning of Article XI:1.
- (ii) "Restriction on importation" within the meaning of Article XI:1 and "border measures"
- 7.254 The restrictions envisaged by Article XI:1 are exclusively restrictions "on importation". Since India argues that the measure at issue cannot fall under Article XI because it is not a "border measure", it is necessary to analyze first whether the phrase "restriction.... on importation" suggests such a limitation on the scope of Article XI:1. This argument is considered first, before considering more generally whether the type of obligation at issue can be said to constitute such a "restriction".

⁴¹⁹ The Panel is further comforted in this analysis by the fact that the Illustrative List of the TRIMs Agreement, in its introductory paragraph to item 2 concerning TRIMs that are inconsistent with Article XI:1, refers to "those which are mandatory or enforceable under domestic law or under administrative rulings, or compliance with which is necessary to obtain an advantage". This reference also suggests an endorsement, in the context of recourse to Article XI:1 for the purposes of the TRIMs Agreement, of a broad notion of "measures".

7.255 In its second submission, in support of its argument that the trade balancing obligation was an internal measure not subject to Article XI, India argued that

"Obviously, the provisions of the MOUs on trade balancing do not affect the process of entering products into India's customs territory and can also for this reason not constitute restrictions on importation within the meaning of Article XI:1". 420

7.256 As indicated above, the Panel will address any such contentions through an application of the customary rules of interpretation of public international law.

7.257 The Panel turns therefore to consider the ordinary meaning of the phrase "restriction...on importation". An ordinary meaning of the term "on", relevant to a description of the relationship which should exist between the measure and the importation of the product, includes "with respect to", "in connection, association or activity with or with regard to". In the context of Article XI:1, the expression "restriction... on importation" may thus be appropriately read as meaning a restriction "with regard to" or "in connection with" the importation of the product. On a plain reading, this would not necessarily be limited to measures which directly relate to the "process" of importation. It might also encompass measures which otherwise relate to other aspects of the importation of the product. Thus, at this stage of our analysis, the Panel does not find that the ordinary meaning of the words supports India's contentions.

7.258 India also makes contextual arguments in support of its contentions in pointing out the difference between the use of the term "importation" alone, in Article XI:1, as compared to "exportation or sale for export". It appears from these terms that Article XI does not intend to deal directly with measures relating to the "sale for importation", but this otherwise suggests no limitation on the *manner* in which a given measure might affect or relate to "importation".

7.259 Nor would a contextual comparison with other Articles, such as Article III, support India's contentions. The use of the term "importation" in Article XI, rather than "imports", or "imported products", clearly suggests that what is targeted in Article XI:1 is exclusively those restrictions which relate to the importation itself, and not to *already* imported products. This does not, however, support a conclusion that the only manner in which a measure can be a restriction "*on importation*" is by being a "border measure", or a measure relating to "the process of importation", as described by India.

7.260 Contextual and purposive interpretation support this initial conclusion by the Panel. The text of the Ad note to Article III clarifies that some measures applied "at the point or time of importation" for imported products are nonetheless to be considered as internal measures under Article III. This would appear to confirm that the application of the measure "at the point of importation" is not necessarily the decisive criterion. If that note requires a particular measure to be considered as an internal measure, it would mean that application at the point of importation would not be decisive in determining the scope of Article XI:1. 422

⁴²⁰ Footnote 8, para 20, Second Submission. See also Annex 1 to second oral statement.

⁴²¹ Webster's New Encyclopedic Dictionary, 1994 ed.

⁴²² The Panel's conclusions are not dependant on the Ad Note as it could also be read to the contrary to argue that the expression "at the point or time of importation" reflects a description of what measures would naturally fall within the scope of Article XI, some of them then falling "out" of the scope of the provision by express derogation in certain circumstances. Such a view would contend that such derogation would not have been necessary if it had been evident that border application of the measure was not sufficient to make it fall under Article XI.

- 7.261 Purposive interpretation suggests rather that it is the nature of the measure as a restriction *in relation to importation* which is the key factor to consider in determining whether a measure may properly fall within the scope of Article XI:1.
- 7.262 Even if the trade balancing requirement cannot necessarily be considered to relate to the actual "process" of importation, or to constitute a "border" measure, this would therefore not be a sufficient reason to conclude that it cannot come within the scope of Article XI:1.
- 7.263 However, it still must be determined how it might constitute a "restriction.... on importation" within the meaning of that provision. A rejection of India's suggested test for the conditions of applicability of the provision does not imply that the claimants' contentions are made out. The Panel turns to address those contentions.
- (d) The trade balancing condition as a "restriction ... on importation"
- (i) The notion of "restriction ... on importation"
- 7.264 Although the title of Article XI refers to the elimination of "quantitative restrictions", the text of the provision makes no distinction between different types of restrictions on importation. On the contrary, the words "No prohibitions or restrictions ... whether made effective through quotas, import or export licenses or other measures" (emphases added) suggest an intention to cover any type of measures restricting the entry of goods into the territory of a Member, other than those specifically excluded, namely, duties, taxes or other charges. As was noted by the India Quantitative Restrictions panel,

"the text of Article XI:1 is very broad in scope, providing for a general ban on import or export restrictions or prohibitions 'other than duties, taxes or other charges'. As was noted by the panel in *Japan - Trade in Semi-conductors*, the wording of Article XI:1 is comprehensive: it applies 'to all measures instituted or maintained by a [Member] prohibiting or restricting the importation, exportation, or sale for export of products other than measures that take the form of duties, taxes or other charges. The scope of the term 'restriction' is also broad, as seen in its ordinary meaning, which is 'a limitation on action, a limiting condition or regulation'."

- 7.265 This Panel endorses the ordinary meaning of the term "restriction" as identified by the *India Quantitative Restrictions* panel and its view as to the generally broad scope of the prohibition expressed in Article XI:1. As a result, it can be concluded that any form of *limitation* imposed on, or in relation to importation constitutes a restriction on importation within the meaning of Article XI:1.
- 7.266 As far as the trade balancing obligation is concerned, it is a condition placed on importation of the product. It results both from the signature of the MOU (whereby the principle of the obligation is agreed to, as part of the conditions to gain the right to import the restricted products) and from the actual importation of products (which determines the "quantum" of the export obligation). The MOUs are signed "on the basis of" projections regarding indigenization and trade balancing. After entering into the MOU, signatories had the opportunity to apply for licenses whenever they wished to

⁴²³ Panel Report on *Japan – Semi-conductors*, BISD 35S/116, para. 104.

⁴²⁴ Panel Report on *India – Quantitative Restrictions*, WT/DS90/R, para. 5.128.

What is perhaps not entirely clear in the operation of the MOUs is the extent to which these export projections by the manufacturers might "bind" them to the extent of preventing them from applying for licenses over the period of the MOU *beyond* the initial projections. Nothing turns on this from a legal perspective in terms of the Panel's analysis.

import kits or components subject to restrictions. It seems that the licenses themselves, before 1 April 2001, made a reference to the trade balancing obligation.

7.267 Both complainants describe the measure as a "limitation" on imports, and highlight the practical threshold imposed by the export requirement on the value of imports that can be made, which, they assert, amounts to a restriction.

7.268 The trade balancing condition does not set an absolute numerical limit on the amount of imports that can be made. It does, however, limit the value of imports that can be made to the value of exports that the signatory intends to make over the life of the MOU. If all signatories could at all times have an unlimited desire and ability to export, this obligation would be unlikely to have any impact upon import decisions. That is not a realistic scenario, however and was not contended for by India. In reality, therefore, the limit on imports set by this condition is induced by the practical threshold that a signatory will impose on itself as a result of the obligation to satisfy a corresponding export commitment. The amount of imports is therefore linked to a certain amount of anticipated exports. The more a signatory would be concerned about its ability to export profitably at significant levels, the more it would be induced by the trade balancing obligation to limit its imports of the relevant products.

7.269 The question of whether this form of measure can appropriately be described as a restriction on importation turns on the issue of whether Article XI can be considered to cover situations where products are technically allowed into the market without an express formal quantitative restriction, but are only allowed under certain conditions which make the importation more onerous than if the condition had not existed, thus generating a disincentive to import.

7.270 On a plain reading, it is clear that a "restriction" need not be a blanket prohibition or a precise numerical limit. Indeed, the term "restriction" cannot mean merely "prohibitions" on importation, since Article XI:1 expressly covers both "prohibition or restriction". Furthermore, the Panel considers that the expression "*limiting condition*" used by the *India – Quantitative Restrictions* panel to define the term "restriction" and which this Panel endorses, is helpful in identifying the scope of the notion in the context of the facts before it. That phrase suggests the need to identify not merely a condition placed on importation, but a condition that is limiting, i.e. that has a limiting effect. In the context of Article XI, that limiting effect must be on importation itself.

7.271 The Panel believes that a substance over form approach should be taken to the analysis of the facts in the context of this test. Such an approach is consistent with that taken by the panel on *Japan-Semi-conductors*. The panel examined a series of actions taken by the Government of Japan after concluding an Agreement on trade in semi-conductors with the United States. These included requests which the Japanese Government addressed to Japanese producers and exporters of semi-conductors not to export semi-conductors at prices below company-specific costs to contracting parties other than the United States. There was also a statutory requirement for exporters to submit information on export prices and systematic monitoring of company and product-specific costs and export prices by the Government. This was backed up with the use of supply and demand forecasts to impress on manufacturers the need to align their production to appropriate levels. The panel concluded that:

"the complex of measures exhibited the rationale as well as the essential elements of a formal system of export control. The only distinction in this case was the absence of formal legally binding obligations in respect of exportation or sale for export of semi-conductors. However, the Panel concluded that this amounted to a difference in form

⁴²⁶ See Panel Report cited above in footnote 410.

rather than substance because the measures were operated in a manner equivalent to mandatory requirements. The Panel concluded that the complex of measures constituted a coherent system restricting the sale for export of monitored semi-conductors at prices below company-specific costs to markets other that the United States, inconsistent with Article XI:1."

7.272 This finding suggests that measures which involve no formal restriction but rather a network of strong suggestions can fall within the scope of Article XI:1. It is true that in that instance, exports were being restricted, rather than imports, and the panel referred in its final conclusion to a restriction on "the sale for export". This Panel acknowledges that this phrase has no matching phrase under Article XI:1 with regard to imports. However, the rest of the *Semi-conductors* panel's conclusions suggest that the restrictions found by the panel also concerned exportation (e.g. "The Panel considered that the complex of measures exhibited the rationale as well as the essential elements of a formal system of *export control*. The only distinction in this case was the absence of formal legally binding obligations in respect of *exportation or sale for export* of semi-conductors."). This report thus seems to support a reading of Article XI:1 encompassing import limitations made effective through disincentives to importation, without a formal numerical limit on imports.

7.273 The same panel also found that "export licensing practices by Japan, leading to delays of up to three months in the issuing of licences for semi-conductors destined for contracting parties other than the United States, had been non-automatic and constituted restrictions on the exportation of such products inconsistent with Article XI:1." This finding suggests that a measure that does not preclude any exportation but rather makes it more burdensome can also amount to a restriction on exportation. *Mutatis mutandis*, the same reasoning could apply to a restriction on importation.

7.274 The Panel Report on *EEC – Programme of Minimum Import Prices, Licences and Surety Deposits for Certain Processed Fruits and Vegetables* also made a finding on Article XI:1 which is of interest in identifying its scope. In that case, the panel examined a minimum import price and associated security system for tomato concentrate. The United States was arguing that the system prohibited importation of goods below a certain price and was, therefore, a restriction within the meaning of Article XI on the importation of these goods. The EC, on the contrary, was arguing that the system was a non-tariff barrier measure and that, in principle, imports of tomato concentrates into the Community were allowed, but not below the minimum price. The panel found that "the minimum price system, as enforced by the additional security, was a restriction "other than duties, taxes or other charges" within the meaning of Article XI:1".⁴²⁸

7.275 That report was adopted by the CONTRACTING PARTIES and cited by the *Japan - Semi-conductors* panel as having decided that "the import regulation allowing the import of a product in principle, but not below a minimum price level, constituted a restriction on importation

-

⁴²⁷ *Ibid*, para. 117.

There was a dissenting opinion on this point. One panelist (out of a panel of five), however, "considered that the minimum import price system, as enforced by the additional security, could well be applied in a way which would qualify it as a "restriction "other than duties, taxes or other charges" within the meaning of Article XI:1. However, having noted the explanation given with respect to the functioning of the system, this member considered that importation of tomato concentrates at a price lower than the minimum price could still be carried out by exporters who had an interest in doing so. He further considered that the system operated in a way to levy an additional charge which raised the price of tomato concentrate imported at a price lower than the minimum price. Therefore, he concluded that the minimum import price system was not being enforced in a manner which would qualify it as a restriction within the meaning of Article XI." This suggests that, for that panelist, the measure operated so as to levy an additional *charge*, rather than as an import restriction (i.e. it was possible to import below the minimum price, but with an additional charge). BISD 25S/68.

within the meaning of Article XI:1". 429 The Panel also notes, in this regard, the observation made by the panel in *Argentina –Hides and Leather* that "[t]here can be no doubt, in [its] view, that the disciplines of Article XI:1 extend to restrictions of a *de facto* nature".

7.276 As noted above, the *India – Quantitative Restrictions* panel endorsed the broad scope of the provision articulated by *Japan – Semi-conductors* in finding that a discretionary import licensing scheme, where licenses were not granted automatically but rather on "unspecified" merits, was contrary to Article XI:1. This Panel agrees with and adopts these interpretations. For reasons outlined above, the Panel does not consider that it is a separate requirement of Article XI that a measure can be described as a border measure. It is the impact of a measure by way of a "restriction ...on importation" that counts, not the physical place of its application.

(ii) Analysis of the trade balancing condition

7.277 With regard to the trade balancing condition, the Panel finds that as at the date of its establishment, there would necessarily have been a practical threshold to the amount of exports that each manufacturer could expect to make, which in turn would determine the amount of imports that could be made. This amounts to an import restriction. The degree of effective restriction which would result from this condition may vary from signatory to signatory depending on its own projections, its output, or specific market conditions, but a manufacturer is in no instance free to import, without commercial constraint, as many kits and components as it wishes without regard to its export opportunities and obligations.

7.278 The Panel therefore finds that the trade balancing condition contained in Public Notice No. 60 and in the MOUs signed thereunder, by limiting the amount of imports through linking them to an export commitment, acts as a restriction on importation, contrary to the terms of Article XI:1. With respect to the European Communities' argument that the MOU signatories that have yet to achieve the 70% indigenization requirement would continue to incur export obligations after 1 April 2001, the Panel notes that no evidence was presented to show that any such new export obligations have in fact accrued.

7.279 The Panel is comforted in this finding by the fact that it appears consistent with Item 2(a) of the Illustrative list of the TRIMs Agreement which suggests that measures linking the amount of imports to a certain quantity or value of exports can constitute restrictions on importation within the meaning of Article XI:1. The Illustrative List thus provides that:

"TRIMS that are inconsistent with the obligation of general obligation of elimination of quantitative restrictions provided for in paragraph 1 of Article XI of GATT 1994 include those which are mandatory or enforceable under domestic law or under administrative rulings, or compliance with which is necessary to obtain an advantage, and which restrict:

(a) the importation by an enterprise of products used in or related to its local production, generally or to an amount related to the volume or value of local production that it exports."⁴³¹

⁴²⁹ Panel Report on *Japan – Semi-conductors*, BISD 35S/116, para. 105.

⁴³⁰ Panel Report on *Argentina – Hides and Leather*, WT/DS155/R, adopted on 16 February 2001, para. 11.17.

⁴³¹ The Illustrative List of the TRIMs Agreement is referred to in its Article 2.2 and contained in the Annex to that Agreement.

7.280 In particular, the Panel notes that this item does not limit the linkage to past exports.

7.281 Nevertheless, the Panel is not ruling on whether the specific type of measure under consideration here is necessarily the precise type of measure envisaged in the Illustrative List or on the extent to which this list may operate as an aid to interpretation of Article XI itself. The Panel also notes that to fall within the terms of item 2(a), the measures in question may in any case need to be characterized as measures that "restrict" imports in certain ways. The essence of our analysis has been to consider the proper meaning of a similar term under Article XI:1 itself. Nonetheless, this item is at least consistent with the finding that a measure linking imports to a certain amount of exports constitutes a restriction on importation within the meaning of Article XI.1.

5. Balance-of-payments defense

7.282 To the extent that a violation of Article XI would be found, India is claiming a defense under Article XVIII:B, although India also indicates that it will not be applying any measures for balance-of-payments purposes after 1 April 2001.

7.283 India argues that for the purposes of the present proceedings, its balance-of-payments situation should be assessed as of the dates of each request for the establishment of this Panel. This is not disputed by the complainants.

7.284 India has not presented any evidence as to its balance of payments situation as at the relevant dates. Instead, India considers that the burden of proof is on the complainants to establish that its measures are not justified on balance-of-payments grounds. It cites the *India – Quantitative Restrictions* report as support for its position, as well as the Appellate Body's ruling on burden of proof in the context of the *US – Wool Shirts and Blouses* case under the Textiles and Clothing Agreement (ATC).

7.285 The Panel does not agree with India's contentions as to burden of proof. In both of these cases cited by India, the rule applied was the general rule on burden of proof as consistently affirmed by the Appellate Body and panels. That rule is to the effect that a party claiming a violation must assert and prove its claim, and that a party asserting a defense is required to prove that the conditions for invoking that defense are met. In the *India – Quantitative Restrictions* case, while the panel found that the United States had the burden of proving its own claim that Article XVIII:11 had been violated by India, it also clearly held that India had the burden of proving its own defense under Article XVIII:B. Similarly, in the *US – Wool Shirts and Blouses* case, India, as the complainant, had the burden of proving its own claim that Article 6 of the ATC was violated by the United States. Thus in both of these cases, the party asserting a particular claim or defense was required to prove its assertions.

7.286 In this instance, the United States and European Communities asserted their claims that India had violated Articles III and XI of GATT 1994 and Article 2 of TRIMs. They made no representations concerning Article XVIII:B. It is India who invoked Article XVIII:B as a defense to any violations of Article XI which the Panel might find. Therefore, it is for India to assert this defense.

7.287 India suggests that applying the burden of proof to a party claiming a balance of payments defense but not a party applying a safeguard measure would be systemically unfair to developing countries. The Panel does not agree that this is an accurate policy consideration that might inform a purposive interpretation on this issue. Safeguards actions may be taken by any country. More importantly, areas of coverage such as safeguards, anti-dumping and countervailing duties require domestic analysis where there is an onus on applicants and where affected foreign interests may be

represented and may use local courts to challenge bureaucratic determinations. It is only where the foreign interest alleges through the WTO that the procedures and principles were not properly followed that the burden then lies on it. There is no such domestic process with balance of payments issues. Thus the Panel does not find the analogy to be relevant.

- 7.288 Thus the Panel holds that the burden is on India in relation to this defense. To successfully assert this defense it must at a minimum, present a *prima facie* case that these measures can be considered to be maintained under Article XVIII:B.
- 7.289 Article XVIII:B foresees the possibility for developing country Members to apply certain measures to safeguard their external financial position and to ensure a level of reserves adequate for their level of development. Article XVIII:9 lists the substantive conditions which should be met in order to apply such measures. In addition, Article XVIII:B and the Understanding on Balance-of-payments provisions require the notification of such measures to the BOPs Committee.
- 7.290 In support of its assertion that its measures were justified under Article XVIII:B, India invokes the notification which it submitted to the BOPs Committee in 1997. It asserts that no new notification was required in respect of Public Notice No. 60 because the measures it contained were not significant changes to the measures within the terms of Article XVIII:B.
- 7.291 Other than that assertion, India has presented no evidence of any discussion in the BOPs Committee subsequent to the rulings of the panel and Appellate Body in the *India Quantitative Restrictions* dispute. It has also not explained how any of the substantive conditions foreseen by Article XVIII:B might be fulfilled. It has also presented no evidence whatsoever concerning its actual balance of payments during the period which it itself has defined as the relevant time of examination for this Panel.
- 7.292 In the light of the foregoing, the Panel finds that India has failed to make a *prima facie* case that its measures were justified under Article XVIII:B.
- 7.293 The Panel also notes that the complainants have, in response to India's defense, presented some arguments and figures suggesting that India's balance-of-payments situation was not such as to justify the application of balance-of-payments measures under Article XVIII:B. India has also not provided evidence to refute those contentions.
- 7.294 India has also indicated that it would expect the Panel to consult with the IMF in determining India's balance-of-payments situation as of the dates of each claimant's request for establishment of this Panel. The Panel does not rule on whether consultation with the IMF is compulsory or not before the final factual resolution by a panel of a balance-of-payments matter, where there is conflicting evidence presented. Whatever the proper view as to this question, such a consultation could not be used as a total substitute for asserting and providing a prima facie case as to a defense under Article XVIII:B, and in the absence of any indication of how the measures might fall within the terms foreseen in that provision. It is clear that a panel's fact finding mandate should not be utilised so as to make out a *prima facie* case where that is not achieved by the relevant party. At an appropriate

⁴³² The Panel recalls in this regard the Appellate Body's observation that "Article 13 of the DSU and Article 11.2 of the SPS agreement suggest that panels have a significant investigative authority. However, this authority cannot be used by a panel to rule in favour of a complaining party which has not established a prima facie case of inconsistency based on specific legal claims asserted by it. A panel is entitled to seek information and advice from experts and from any other relevant source it chooses, pursuant to Article 13 of the DSU and, in an SPS case, Article 11.2 of the SPS Agreement, to help it understand and evaluate the evidence submitted and the arguments made by the parties, but not to make the case for a complaining party". (Report of the Appellate Body, *Japan – Agricultural Products II*, WT/DS76/AB/R, adopted on 19 March 1999, para. 129).

stage in proceedings, consultation of appropriate international experts or authorities could be helpful in establishing whether one of the specific situations foreseen in Article XVIII:B applied to India's situation. As stated by the *India – Quantitative Restrictions* panel, such consultation could "assist in assessing the claims submitted" to the Panel. However, the arguments presented did not even lead the Panel to that point.

6. Claims under Article III:4

7.295 In light of the Panel's findings that the trade balancing condition contained in Public Notice No. 60 and the MOUs is inconsistent with Article XI:1, it wishes to apply the principle of judicial economy and does not consider it necessary in this instance, to separately consider the United States' general claim that the trade balancing condition is inconsistent also with Article III:4 of the GATT 1994.

7.296 Nevertheless, the Panel wishes to make one observation in that regard owing to India's general claim that the measure should be considered under Article III and the implication that the provisions are necessarily mutually exclusive. In support of the conclusion that while Articles III:4 and XI:1 deal with "imported products" and "importation" respectively, but without thereby prejudging what the outcome of such an Article III:4 examination might have been in this instance, the Panel wishes to note that it sees merit in the proposition that there may be circumstances in which specific measures may have a range of effects. In appropriate circumstances they may have an impact both in relation to the conditions of importation of a product and in respect of the competitive conditions of imported products on the internal market within the meaning of Article III:4. This is also in keeping with the well established notion that different aspects of the same measure may be covered by different provisions of the covered Agreements.

7.297 While the Panel refrains from further consideration of the broader application of Article III:4 to the same features dealt with in the Article XI analysis, there is one distinct element of the trade balancing measure that was not factually in dispute between the parties and which was alleged by the complainants to constitute a distinct aspect of violation of Article III:4. This was the export obligation that India acknowledged to be incurred in relation to previously imported products acquired by manufacturers directly on the Indian market. This specific aspect was not considered in the examination under Article XI:1. Because it is a discrete condition attached to the administration of the trade balancing obligation and because it was addressed by the complainants in their Article XI analysis, the Panel feels it is appropriate to make a separate ruling on it.

(a) Purchases of restricted imported kits and components on the domestic market

7.298 Both the United States and European Communities argue that the trade balancing condition is inconsistent with Article III:4 in that it requires MOU signatories who purchase restricted kits and components on the Indian market to count the value of these purchases towards their trade balancing obligations. India has confirmed that,

"if an MOU signatory purchased a component that was subject to import restrictions in India from either a trading company or another MOU signatory that had imported

⁴³³ The Panel notes that the TRIMS Agreement Illustrative List envisages measures relating to export requirements both in the context of Article XI:1, as noted above in the context of our analysis under Article XI:1, and in the context of Article III:4 of the GATT 1994, by listing as inconsistent with that provision measures which require "that an enterprise's purchases or use of imported products be limited to an amount related to the volume or value of local products that it exports" TRIMS Illustrative List, Item 1 (b).

such a component on the basis of an import license, the value of such components would be taken into account for purposes of the neutralization requirement."⁴³⁴

7.299 The Panel recalls that Article III:4 provides that:

"The products of the territory of any [Member] imported into the territory of any other contracting party shall be accorded treatment no less favourable than that accorded to like products of national origin in respect of all laws, regulations and requirements affecting their internal sale, offering for sale, purchase, transportation distribution or use."

7.300 As previously noted, for a violation of Article III:4 to be established, it is necessary to examine whether (1) imported products and domestic products are like products; (2) the measures constitute a "law, regulation or requirement"; (3) they affect the internal sale, offering for sale, purchase, transportation, distribution or use; and (4) imported products are accorded less favourable treatment than the treatment accorded to like domestic products. The Panel will consider these points successively.

(i) Like products

7.301 In this instance, the products at stake are the imported kits and components to which the trade balancing requirement applies. India asserts that these products are those that are/were subject to import licensing, since these are the only ones for which the trade balancing requirement applies. The Panel has already separately addressed the factual disagreement on this issue and concluded that the European Communities have not demonstrated that trade balancing applied in the past, or would still apply, to non-restricted items.

7.302 For the purpose of this analysis, the Panel will therefore consider only those products that were subject to restriction, i.e. kits and certain listed components. As was the case with the analysis under Article III:4 concerning the indigenization requirement, the only distinguishing factor between imported kits and components and kits and components of domestic origin would be their origin. Such differences in origin would not alone be such as to make products unlike. The Panel also recalls that it need not, for the purpose of this analysis, demonstrate actual trade in the products concerned to establish the likeness of the products at stake. The Panel therefore concludes that kits and components of imported and domestic origin are like products within the meaning of Article III:4. 435

(ii) Laws, regulations or requirements

7.303 The Panel recalls its earlier conclusion that indigenization conditions in Public Notice No. 60 and the MOUs signed thereunder are "requirements" within the meaning of Article III:4 of GATT 1994. This conclusion is equally applicable to the trade balancing, obligation which is, like the indigenization requirement, one of the conditions provided for in Public Notice No. 60 and to be

⁴³⁴ Response to Question 47 (b) of the Panel.

⁴³⁵ See for example the Panel Report on *Indonesia – Autos*, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, adopted on 23 July 1998, para. 14.113 (DSR 1998:VI, 2201) ("... an origin-based distinction in respect of internal taxes suffices in itself to violate Article III:2, without the need to demonstrate the existence of actually traded like products.")

See also the Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.74.

See also the Compliance Panel Report on US-FSC (Article 21.5), WT/DS108/RW, circulated on 20 August 2001, para. 8.133.

accepted by MOU signatories as a condition for obtaining the advantages of a license. Thus the Panel concludes that it is a requirement.

(iii) affecting internal sale, use, ...

7.304 Having established that the trade balancing condition foreseen in Public Notice No. 60 and the MOUs signed thereunder is a requirement within the meaning of Article III:4, the Panel must now consider whether this requirement is one "affecting the internal sale, offering for sale, purchase, use or distribution" of the products at issue, in that it requires MOU signatories purchasing restricted kits and components within India to count their value towards their export obligation.

7.305 The Panel recalls that the ordinary meaning of the term "affecting" has been understood to imply "a measure that has "an effect on" and this indicates a broad scope of application". The disciplines of Article III:4 thus govern a broad range of measures, i.e. not only measures which are directly intended to regulate the product, but in addition, any law, regulation or requirement which "has an effect" on either the internal purchase, offering for sale etc. of the product. The fact that a provision is not necessarily primarily aimed at regulating the offering for sale or use of the product on the domestic market is thus not an obstacle to its "affecting" them. Nonetheless, it applies only to situations where either internal sale, or purchase (...) or use are affected.

7.306 The fact that the measure applies only to imported products need not, in itself, an obstacle to its falling within the purview of Article III. For example, an internal tax, or a product standard conditioning the sale of the imported but not of the like domestic product, could nonetheless "affect" the conditions of the imported product on the market and could be a source of less favorable treatment. Similarly, the fact that a requirement is imposed as a condition on importation is not necessarily in itself an obstacle to its falling within the scope of Article III:4. As

7.307 In the present instance, an MOU signatory choosing to purchase an imported kit or component within the Indian market will incur an additional export obligation of a value at least equivalent to that of the product, whenever the product has been imported subject to restrictions (which is normally the case for these products). On the other hand, were it to purchase a locally-made kit or component, the manufacturer would not incur such an additional export obligation. The purchase of the imported good thus carries with it an additional burden not incurred upon the purchase of a like domestic product. This obviously affects the competitive conditions of the imported product on the Indian market and more specifically, affects the conditions of internal offering for sale or purchase of these products.

 437 Article III:1 refers to the application of measures "to imported or domestic products", which suggests that application to both is not necessary.

⁴³⁶ Panel Report on *Canada – Autos*, WT/DS139/R, WT/DS142/R, adopted on 19 June 2000, as modified by the Appellate Body Report, para. 10.80.

⁴³⁸ Thus, the "advantage" to be obtained could consist in a right to import a product. See for instance, the Report of the second GATT panel on *EC – Bananas II* as cited and endorsed in *EC – Bananas III*, WT/DS27/R/USA, adopted on 25 September 1997, as modified by the Appellate Body Report, para. 4.385 (DSR 1997:II, 943:

[&]quot;The Panel further noted that previous panels had found consistently that this obligation applies to any requirement imposed by a contracting party, including requirements 'which an enterprise voluntarily accepts to obtain an advantage from the government.' In the view of the Panel, a requirement to purchase a domestic product in order to obtain the right to import a product at a lower rate of duty under a tariff quota is therefore a requirement affecting the purchase of a product within the meaning of Article III:4."

- (iv) Whether the measure affords less favourable treatment to imported products
- 7.308 By requiring that the purchaser of an imported kit or component take on an additional obligation to export cars or components of equal value when such domestic purchases occur, the trade balancing requirement creates a disincentive to the purchase of these products, and consequently makes them more difficult to dispose of on the internal market. This element of the trade balancing obligation therefore distinctly accords less favorable treatment to these imported products than to like products of domestic origin, within the meaning of Article III:4 of GATT 1994.
- 7.309 The Panel therefore finds that, by requiring that MOU signatories count towards their export obligation under the MOU any purchases of imported restricted kits and components on the Indian market, the trade balancing condition is in violation of Article III:4 of the GATT 1994.

7. Ruling on the consistency of Public Notice No. 60 with the GATT 1994

- 7.310 To date the Panel has considered the claims concerning the indigenization and trade balancing successively, examining each time both Public Notice No. 60 and the MOUs together. The Panel previously indicated that it thought this to be the appropriate approach, since both instruments contain the two specific obligations that are the measures in issue. Indeed, it is only through the existence of Public Notice No. 60 that manufacturers were induced to sign the MOUs. While the European Communities has not objected to such an approach, it has also requested the Panel to make a separate ruling that Public Notice No. 60 itself is inconsistent with the GATT and TRIMs Agreement.
- 7.311 This request is not simply a further claim about an identified measure that could be the subject of a decision to apply judicial economy. It is instead a distinct claim as to a more restrictively defined measure, albeit still relating to the two conditions of indigenization and trade balancing. As such, the Panel feels bound to rule on it.
- 7.312 In order to consider this separate claim, the Panel will address successively the claims of inconsistency with Articles III:4 and XI of the GATT 1994.
- (a) Consistency of Public Notice No. 60 with Article III:4 of GATT 1994
- 7.313 In addressing whether Public Notice No. 60 is consistent with Article III:4 of the GATT 1994, the Panel notes that the following elements must be considered: whether the imported products and domestic products are "like products"; whether Public Notice No. 60 is a "law, regulation, or requirement affecting the internal sale, offering for sale, purchase...or use"; whether the imported products are accorded "less favourable treatment" than that accorded to like domestic products. In the previous analysis, the Panel found that automotive parts and components of domestic and foreign origin are like products within the meaning of Article III:4.
- 7.314 In respect of whether Public Notice No. 60 constitutes a "requirement" under Article III:4, the Panel first noted that commitments entered into on a voluntary basis by individual firms to obtain an advantage constituted a "requirement". In this instance, the Panel noted that auto manufacturers were required to accept the indigenization condition set out in Public Notice No. 60 to obtain an import license to be able to import the restricted items contained in Public Notice No. 60. The Panel went

⁴³⁹ In response to the Panel's question whether the European Communities was seeking a distinct ruling on Public Notice No. 60, separate from whether any MOU has been executed, the European Communities replied that it was seeking a distinct ruling. The European Communities argued that "[t]o the extent that the MOUs impose obligations that are inconsistent with the GATT and the TRIMs Agreement, a requirement to enter into an MOU is also inconsistent with those agreements, regardless of whether any manufacturer actually concludes a MOU." See EC's Response to Question 88.

on to conclude that the indigenization condition contained in Public Notice No. 60 is a condition to the granting of an advantage, i.e. the right to import restricted kits and components, and therefore constituted a requirement within the meaning of Article III:4.

- 7.315 The Panel subsequently found that the indigenization condition affects the internal sale, offering for sale, purchase and use of the imported parts and components in the Indian market and that, by requiring auto manufacturers to use a certain percentage of domestic products, it affects the internal sale of like imported products which affords less favourable treatment to the imported products.
- 7.316 The Panel is mindful that Public Notice No. 60 does not, in and of itself, impose a "direct" obligation on individual manufacturers as it is the execution of the MOUs that imposes such obligations. However, the Panel notes that it is not necessary for actual MOUs to have been signed under Public Notice No. 60 in order for the indigenization condition, as contained in Public Notice No. 60, to afford less favourable treatment to imported products within the meaning of Article III:4. The Panel recalls in particular, in this respect, that the purpose of Article III:4 is to protect the competitive opportunities of imported products on the domestic markets. It is a well established principle under WTO jurisprudence that Article III of the GATT 1994 is to provide equality of competitive conditions for imported products in relation to domestic products. That is, Article III requires Members to provide equality of competitive conditions for imported products in relation to domestic products. These competitive conditions are affected even in the absence of actual trade flows, wherever the conditions afforded to imported products are such as to affect their competitive opportunities on the market. 440 In this instance, Public Notice No. 60 imposes the acceptance of the indigenization condition on any manufacturer wishing to import restricted kits or components. This in itself adversely affects the competitive opportunities of all automotive parts and components, as far as the indigenization condition is concerned because it can be assumed that certain car manufacturers will wish to import restricted goods and will therefore choose to accept the indigenization condition.
- 7.317 In light of the foregoing, the Panel finds that Public Notice No. 60, independent of whether the MOUs have been executed, is a requirement that is inconsistent with Article III:4 of the GATT 1994.
- (b) Consistency of Public Notice No. 60 with Article XI:1 of the GATT 1994 in so far as it contains the trade balancing condition
- 7.318 In determining whether Public Notice No. 60 is inconsistent with Article XI:1 of the GATT 1994, the Panel recalls its earlier analysis of the trade balancing condition as contained in the previous section.
- 7.319 First, it recalls its conclusion that Public Notice No. 60, as a governmental measure requiring manufacturers to accept certain conditions in order to be allowed to import restricted automotive kits and components, constituted a "measure" within the meaning of Article XI:1. This conclusion remains relevant to this analysis and the Panel confirms its earlier conclusion in this respect.
- 7.320 Second, in order to establish whether Public Notice No. 60, in itself, can be considered to be inconsistent with Article XI:1, it has to be established that it constitutes a "restriction ... on importation" within the meaning of that provision. The Panel recalls in this respect its earlier

⁴⁴⁰ The Appellate Body stated that "Article III protects expectations not of any particular trade volume but rather of the equal competitive relationship between imported and domestic products." See Appellate Body Report in *Japan – Alcoholic Beverages II*, WT/DS8/R, WT/DS10/R, WT/DS11/R, adopted on 1 November 1996, p. 16 (DSR 1996:I, 97).

conclusion that the trade balancing condition, as contained both in Public Notice No. 60 and in the MOUs signed thereunder, constituted a restriction on importation contrary to Article XI:1 in that it effectively limits the amount of imports that a manufacturer may make by linking imports to commitment to undertake a certain amount of exports. Under such circumstance, an importer is not free to import as many restricted kits or components as he otherwise might so long as there is a finite limit to the amount of possible exports.

- 7.321 While the Panel, in reaching its earlier conclusion, analyzed both instruments containing the trade balancing condition, the conclusion that this condition is inconsistent with Article XI:1 is not dependent on the signature by individual manufacturers of MOUs as such condition contained in Public Notice No. 60 is a measure that restricts the import of kits or components.
- 7.322 The Panel therefore concludes that Public Notice No. 60 in itself, to the extent that it requires the acceptance of the trade balancing condition in order to gain the advantage of importing the restricted products, imposes a restriction on imports and is inconsistent with Article XI:1 of the GATT 1994.
- G. CLAIMS UNDER THE TRIMS AGREEMENT
- 7.323 Both complainants argue that the indigenization and neutralization conditions are in violation of Article 2 of the TRIMs Agreement, which prohibits the use of trade-related investment measures inconsistent with Articles III or XI of GATT 1994.
- 7.324 However, having found that these measures are respectively in violation of Articles III:4 and XI:1 of the GATT, the Panel applies the principle of judicial economy and finds that it is not necessary to consider separately whether they are also inconsistent with the provisions of the TRIMs Agreement.

VIII. CONCLUSIONS AND RECOMMENDATIONS

- A. CONCLUSIONS
- 8.1 In light of the foregoing findings, the Panel concludes that:
 - (a) India acted inconsistently with its obligations under Article III:4 of the GATT 1994 by imposing on automotive manufacturers, under the terms of Public Notice No. 60 and the MOUs signed thereunder, an obligation to use a certain proportion of local parts and components in the manufacture of cars and automotive vehicles ("indigenization" condition);
 - (b) India acted inconsistently with its obligations under Article XI of the GATT 1994 by imposing on automotive manufacturers, under the terms of Public Notice No. 60 and the MOUs signed thereunder, an obligation to balance any importation of certain kits and components with exports of equivalent value ("trade balancing" condition);
 - (c) India acted inconsistently with its obligations under Article III:4 of the GATT 1994 by imposing, in the context of the trade balancing condition under the terms of Public Notice No. 60 and the MOUs signed thereunder, an obligation to offset the amount of any purchases of previously imported restricted kits and components on the Indian market, by exports of equivalent value;

- 8.2 The Panel recalls that it has examined the foregoing measures as of the date of the complainants' requests for establishment of a panel. However, the Panel also recalls that a number of arguments were exchanged in the course of the proceedings concerning events which took place during the Panel's proceedings and their potential impact on the rulings or recommendations to be made by the Panel.
- 8.3 The Panel had earlier concluded that events subsequent to the requests for establishment of the Panel had not removed the Panel's initial competence to consider the validity of the measures as at the date of its establishment. It had also noted that separate consideration would be given to the issue whether the events which took place subsequently, including on or after 1 April 2001, might have affected the existence of any violations identified and that it would consider whether those events affect the nature or range of any recommendations it may make to the DSB in accordance with Article 19.1 of the DSU. This issue will be addressed now.
- B. CONSEQUENCES OF EVENTS HAVING TAKEN PLACE IN THE COURSE OF THE PROCEEDINGS

1. Presentation of the issue: arguments of the parties

8.4 India principally argued that the Panel could not rule on the measures at issue because they had already been the subject of a DSB ruling and a mutually agreed solution. It also argued that any future measures which India might take as of after 1 April would be outside of this Panel's terms of reference. In addition, India made a number of arguments to the effect that the situation in respect of the measures at issue had changed after 1 April 2001 and that in any event, India's measures as applied after 1 April 2001, in the absence of any import licensing restrictions, could not be in violation of the GATT 1994 or of the TRIMs Agreement.

8.5 India thus argued generally that "Public Notice No. 60 is no longer applicable because the import restrictions it was to administer no longer exist. In the absence of an import restriction on cars, this notice cannot be implemented and serves no purpose. It is therefore not clear to India how the European Communities and the United States can claim that the Public Notice as such is capable of violating WTO law." As for Public Notice No. 60 itself, in its first submission (prior to April 2001), India indicated that Public Notice No. 60 would no longer apply as of 1 April 2001 and explained what the consequences of this absence of further application of Public Notice No. 60 would be: no new MOUs would be required and no licenses would be required for the importation of previously restricted kits and components. In its second submission, India confirmed that Public Notice would no longer be applicable in the absence of import restrictions of India confirmed that Public Notice would no longer be applicable in the absence of import restrictions In its response to a concern raised by the United States that the notice appeared to still remain in effect, India explained in its second submission that "a new public notice rescinding Public Notice No. 60 is (...) not required to ensure the consistency of Public Notice No. 60 with Articles III:4 and XI:1 of the GATT. The Panel was not advised by India until the interim review stage that Public Notice had, nonetheless, been repealed in September 2001.

8.6 It is clear that India's arguments relating to the elimination of its import licensing regime were relevant to a number of its legal arguments, including its principal arguments concerning the inadmissibility of the claims, as having already been the object of dispute settlement procedures. Nonetheless, India also envisaged the possibility that the Panel might conclude that it was competent

⁴⁴¹ Second Submission by India, para. 12.

⁴⁴² Response to Question 2(b) by the Panel.

⁴⁴³ Second submission by India, para. 14.

⁴⁴⁴ See document annexed to the Oral Statement of India at the interim review meeting.

to examine the matter before it and examine "the operation of Public Notice No. 60 and the trade balancing provisions under India's former licensing regime". 445

8.7 In this respect, India requested that:

If the Panel were to conclude that the operation of Public Notice No. 60 and the trade balancing provisions under India's former licensing regime was inconsistent with the GATT, it should – following the practice of other panels – note that the licensing regime had been abolished on 1 April 2001 and that it was *therefore* not necessary for the Panel to recommend to the DSB in accordance with Article 19:1 of the DSU that it request India to bring these measures into conformity with the GATT (emphasis added). 446

8.8 In addition, India suggested in its first submission that it was quite possible that a number of MOU signatory manufacturers may have reached, or be close to reaching, the required level of indigenization in manufacturing. At the second substantive meeting, India presented the results of a survey following a meeting with "the most important car manufacturers from the EC, Japan and the United States to verify their performance under the MOUs" which led to written confirmation by these signatories that they had reached the required levels of indigenization, except for one of the surveyed companies, who remained below the 70% threshold for one car model. 447

8.9 India thus also requested that:

"If the Panel were to rule that the indigenization requirements in the MOUs are inconsistent with Article III:4, it should find that, according to the evidence submitted by the parties, there remained as of May 2001 only one company that was still bound by the indigenization requirement with respect to one car model. India therefore requested the Panel to limit any recommendation under Article 19:1 of the DSU to the indigenization requirement that remains to be performed by that particular company with respect to that model."

- 8.10 Both complainants, to the contrary, generally considered that the disappearance of India's licensing system did not affect their legal claims and that the measures, as they would remain in place after 1 April 2001, would continue to be inconsistent with the provisions invoked by them. They also confirmed the nature of their requests for findings in light of these elements, which suggested an expectation that the Panel would take them into account.
- 8.11 The United States thus requested the Panel to find that the measures at issue in this dispute, the indigenization and the trade balancing requirements, imposed by Public Notice No. 60 and the MOUs, were inconsistent with Article III:4 and XI:1 of GATT 1994 and Articles 2.1. and 2.2 of the TRIMs Agreement. The United States requested this Panel to make findings to this effect, and to recommend that India bring all such measures into conformity with its obligations. In response to

⁴⁴⁵ See para. 3.7 above.

⁴⁴⁶ Para. 3.8 of the report.

⁴⁴⁷ Statement of India at the second meeting, para. 29 and Annex II to that statement.

⁴⁴⁸ Para. 3.8 of the report.

⁴⁴⁹ See also Questions 1 and 17 of the Panel.

⁴⁵⁰ See para 3.1. The United States also considered it would be appropriate for the Panel to rule on Public Notice No. 60 given that the requirements were still in place through the MOUs. The United States said it was not requesting the Panel to make findings on Public Notice No. 60 that were separate from its findings on the MOUs executed thereunder. The United States was, however, requesting that the Panel's findings

questions from the Panel, the United States confirmed that it was "requesting the Panel to make findings with respect to the indigenization requirement and the trade balancing requirement as such. Those requirements were in existence at the time the Panel was established, and remain in existence today. There is therefore no distinction to be drawn between the measures as they existed at the time of the Panel establishment and now."⁴⁵¹ The United States also indicated in response to a question from the Panel that "the Panel must of course consider the arguments that India has advanced about the relevance of the situation after April 1, 2001, since India has made those arguments a central part of its defense (...)".⁴⁵²

- 8.12 The European Communities requested the Panel to find that the trade balancing and indigenization requirements contained in Public Notice No. 60 and in the MOUs concluded thereunder were inconsistent with Articles III:4 and XI:1 of the GATT 1994 and Article 2.1 of the TRIMs Agreement as of the date of establishment of this Panel and had remained so after 1 April 2001. It clarified that the measures applied after 1 April 2001 continued to be inconsistent with the GATT 1994 and the TRIMs Agreement because they were the same as the measures applied before that date. Furthermore, the European Communities requested the Panel to find that, at the time when the Panel was established, Public Notice No. 60 was inconsistent with Article XI:1 and III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.
- 8.13 The issue of the impact of certain events subsequent to the request for establishment of the Panel was therefore clearly raised before the Panel, and it appears from the terms of their arguments and requests for findings that this issue was relevant in the findings of violation which the Panel might make with respect to the measures as they were applied as of its establishment. In these circumstances, the Panel found it necessary to determine whether it was still appropriate to make a recommendation notwithstanding India's assertion that the legal situation had fundamentally changed and that there may be nothing left to bring into conformity.

2. Approach of the Panel

- 8.14 Article 19.1 of the DSU provides that "[w]here a panel or the Appellate Body concludes that a measure is inconsistent with a covered agreement, it shall recommend that the Member concerned bring the measure into conformity with that Agreement."
- 8.15 This provision thus envisages a situation where a violation *is* in existence. Indeed, this formulation reflects the usual situation which panels encounter, examining a matter which does not significantly evolve in the course of the proceedings. It is only natural in such circumstances for a panel, following a finding that a measure is inconsistent with a covered agreement, to simply recommend, without the need for any further substantive analysis, that this measure be brought into conformity with that Agreement.
- 8.16 In this instance, however, a number of arguments have been presented, as was noted, to suggest that certain events having occurred in the course of the proceedings fundamentally affect the existence or persistence of the alleged violations. In fact, issues relating to the elimination of India's import licensing regime and its particular impact on the measures at issue in this dispute were discussed before this Panel from the very first stages of the proceedings. As noted above, India itself

encompass both Public Notice No. 60 and the individual MOUs (see para 3.2). See also para. 38 of the First Submission of the United States to the Panel.

⁴⁵¹ Response to Question 2 (a) from the Panel.

⁴⁵² Response to Question 72 from the Panel para. 10.

⁴⁵³ See Second Oral Statement para. 41.

⁴⁵⁴ Para. 3.5. See also Question 87 of the Panel.

had suggested in its requests for findings that, should the Panel examine the measures and find them to be inconsistent with India's obligations under the GATT 1994, these factors be taken into account in the context of the Panel's recommendations under Article 19.1, either to preclude the need for such recommendations or to limit their scope. The complainants, on the other hand, have expected the Panel to make certain findings and recommendations implying that the events having taken place in the course of the proceedings have been taken into account.

- 8.17 The determination the panel therefore needs to make is whether it is appropriate for it, in the circumstances of the case, to take these developments into account and determine in this light whether it still should make a recommendation under Article 19.1, and, if so, whether the content of this recommendation should be affected by these developments.
- 8.18 In most instances of dispute settlement under the WTO, the measures at issue are not affected, or argued to be affected, by events having occurred during the course of the proceedings in relation to the measures at issue. Some panels have, however, had to consider whether to take into account amendments made to the measures in the course of the procedure. While some panels have declined to examine subsequent amendments to the measures and their conformity with the agreements, at least one panel has taken into account the most recent evolutions of the measures where this allowed it to find the "current" measures no longer to be in violation to the measures at issue are not affected, or argued to be affected, by events having occurred during the course of the proceedings in relation to the measures at issue are not affected, or argued to be affected, by events having occurred during the course of the proceedings in relation to the measures at issue are not affected, or argued to be affected, by events having occurred during the course of the proceedings in relation to the measures at issue are not affected, or argued to be affected, by events having occurred during the course of the proceedings in relation to the measures at issue are not affected, or argued to the measures at issue are not affected, or argued to the measures at issue are not affected, by events having occurred to the measures at issue are not affected, by events at issue are not affected, or argued to the measures at issue are not affected, or argued to the measures at issue are not affected, by events at issue are not affected, by events at issue are not affected.
- 8.19 In situations where the withdrawal of the entire measure in dispute has taken place, it may understandably be possible for the Panel, assuming that it nonetheless had examined the consistency of the relevant measure, to conclude that that there is no recommendation left to make under Article 19.1. However, the situation arising in this instance is quite distinct. While it is not disputed that some evolutions have taken place in the more general context of India's import restrictions, the parties are in strong disagreement as to the effect of these evolutions on the specific measures at issue in this dispute.
- 8.20 It should first be stressed that the issue arising in this instance is *not* whether the Panel might be entitled to examine the WTO-consistency of any new measures which might have been taken by India after the beginning of these proceedings. Such subsequent measures have been clearly stated not to be within this Panel's terms of reference. None of the parties contended that the measures themselves were no longer those that were within the Panel's initial terms of reference. The issue is

⁴⁵⁵ The Panel notes, as a general matter, that if a respondent could make changes of any degree to a measure subject to challenge and then always be able to successfully argue that the Panel's report can have no ultimate normative effect because the changes are distinct measures from those originally envisaged, this could entirely frustrate the dispute settlement system.

hereunder were revoked during the panel process. Citing previous GATT and WTO panel reports, the panel proceeded to make findings on the National Car program. In *US – Section 337*, the challenged measure was amended during the panel process but the panel refused to take into account such amendments. See also *Argentina – Textiles*, Report of the Appellate Body, para. 64.

457 See under the GATT, the *Thailand – Cigarettes* case: the panel took into account, in drawing its

See under the GATT, the *Thailand – Cigarettes* case: the panel took into account, in drawing its conclusions as to the compatibility of the measure submitted to it, an amendment in the measure on the basis of which it was able to conclude that the measure was no longer in violation of the relevant GATT provision:

"the Panel observed that the new Thai measure, by eliminating business and municipal taxes on cigarettes, removed the internal taxes imposed on imported cigarettes in excess of those applied to domestic cigarettes. The Panel noted that (...) the Tobacco Act continued to enable the executive authorities to levy the discriminatory taxes. However, the Panel (...) found that the possibility that the Tobacco Act might be applied contrary to Article III:2 was, by itself, not sufficient to make it inconsistent with the General Agreement".

The panel concluded that "the current regulations relating to the excise, business and municipal taxes on cigarettes are consistent with Thailand's obligations under Article III".

limited solely to the question of whether, as argued by the respondent, certain events subsequent to the Panel's establishment are such as to affect the continued relevance of the Panel's initial findings with regard to measures clearly within its terms of reference. This raises the issue of whether they should be considered, in this light, before the Panel can make appropriate recommendations as to the need for India to bring its measures into conformity with the GATT 1994. 458

8.21 In approaching this question, the Panel considers it appropriate to seek guidance in the definition of the mandate given to it by the DSB under the DSU. The Panel recalls in particular the terms of Article 11 of the DSU, which describes the function of panels as follows:

"The function of panels is to assist the DSB in discharging its responsibilities under this Understanding and the covered agreements. Accordingly, a panel should make an objective assessment of the matter before it, including an objective assessment of the facts of the case and the applicability of and conformity with the relevant covered agreements, and make such other findings as will assist the DSB in making the recommendations or in giving the rulings provided for in the covered agreements. (emphasis added and final sentence omitted)"

8.22 The Panel also recalls its own terms of reference in this case:

"To examine, in the light of the relevant provisions cited by the United States in WT/DS175/4 and by the European Communities in WT/DS146/4, the matters referred to the DSB by the United States and the European Communities in those documents and to *make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements*. (emphasis added)"⁴⁵⁹

- 8.23 The Panel believes that it has an important responsibility to the DSB to "assist [it] in making the recommendations or giving the rulings" provided for under the covered agreements, and, more generally, to "assist it in discharging its responsibilities under this Understanding and the covered agreements" as foreseen in Article 11 of the DSU. As recalled by the Appellate Body, the aim of dispute settlement is "to resolve the matter at issue and 'to secure a positive solution to a dispute". 460
- 8.24 The responsibilities of a panel under the DSU include the requirement under Article 19.1 of the DSU that "[w]here a panel or the Appellate Body concludes that a measure is inconsistent with a covered agreement, it shall recommend that the Member concerned bring the measure into conformity with that Agreement".
- 8.25 If only as a matter of logic, there can be no sense in making such a recommendation if a Panel is of the view that the violation at issue has ceased to exist when its recommendation is being made. 461

⁴⁶⁰ Australia – Salmon, Report of the Appellate Body, para. 223.

⁴⁵⁸ It should be stressed in this respect that none of the issues raised here relate to any alleged "worsening" of the situation likely to generate additional violations resulting from the measures at stake, but rather to whether the situation as examined has lost part or all of its legal or practical relevance.

⁴⁵⁹ See para. 1.4.

⁴⁶¹ This was recalled by the Appellate Body in its report on *US – Certain EC Products*, where it observed that "there is an obvious inconsistency between the finding of the Panel that 'the 3 March Measure is no longer in existence' and the subsequent recommendation of the Panel that the DSB request that the United States bring its measure into conformity with its WTO obligations. The Panel erred in recommending that the DSB request the United States to bring into conformity with its WTO obligations a measure which the Panel has found no longer exists" (*US – Certain EC Products*, Report of the Appellate Body, WT/DS165/AB/R, adopted on 10 January 2001, para. 81).

The Panel does not believe that Articles 11 and 19 of the DSU should be interpreted to demand that a panel must make a formalistic statement that a measure needs to be brought into compliance when it is faced with factual and legal arguments that this is no longer the case and must do so without being entitled to resolve those contentions.

- 8.26 Indeed, were these issues not taken into account at all in the Panel's assessment of the matter, there may be a significant risk that the resulting ruling could remain of very uncertain value even as of the date of release of the Panel's report, as well as potentially be internally inconsistent in making a recommendation to bring into conformity measures alleged to have ceased to have an effect.
- 8.27 In the circumstances of the case, the respondent itself has requested that the events which occurred in the course of the proceedings be taken into account in the context of making the recommendations under Article 19.1. The complainants, while they had not specifically requested the Panel to consider these issues at this stage of its analysis, did make requests for findings which called for these subsequent events to be taken into account, in requesting the Panel to find that the measures had "remained" in violation subsequent to 1 April.
- 8.28 In light of the foregoing, the Panel felt that it would not be making an "objective assessment of the matter before it", or assisting the DSB in discharging its responsibilities under the DSU in accordance with Article 11 of the DSU, had it chosen not to address the impact of events having taken place in the course of the proceedings, in assessing the appropriateness of making a recommendation under Article 19.1 of the DSU.
- 8.29 This is, in the Panel's view, an entirely distinct question from the issue of *how* India might appropriately remedy this situation and bring its measures into conformity in the future. The Panel does not seek here to engage in such an analysis. Any future issues arising as to whether India has complied with any recommendations resulting from the adoption of this report would be for a compliance panel to assess.
- 8.30 It should be highlighted in concluding this section that the decision taken by this Panel to proceed in this way in the particular circumstances of this case is in no way intended to imply that panels have a general duty to systematically re-evaluate the existence of any violations identified before proceeding with making their recommendations under Article 19.1. This Panel is simply responding to the particular arguments placed before it, where the parties disagree as to the implications of subsequent events on the Panel's power to make recommendations and rulings. The principal aim of the Panel in proceeding in this manner is to discharge its duty in the most efficient way towards resolving the matter at issue in this dispute.

3. Analysis

- 8.31 The parties agree that as from 1 April 2001 India removed its discretionary import licensing regime on the products at issue and that Public Notice No. 60 and the MOUs were originally introduced and operated in the context of this licensing system. This was obviously a significant change in the situation affecting the conditions of importation of these previously restricted products. In particular, India clarified that no new MOUs would be required to be signed and no import licenses would be required for future non-restricted imports.
- 8.32 However, the question to consider is not whether these changes were significant *per se*, but rather, what effect they may have on the foregoing legal and factual analysis of the two measures before the Panel, namely the indigenization and trade balancing obligations, and consequently, on the recommendations the Panel might make under Article 19.1 of the DSU.

- 8.33 India raises essentially two lines of argument:
 - (a) the general import licensing framework under which the measures operated had disappeared and that, as a result, there could be no further violation of the relevant provisions; more specifically, India argued that one of the instruments containing the challenged obligations (i.e. Public Notice No. 60), first "ceased to operate" and then was formally repealed; it also argued that after 1 April 2001, the MOUs signed under Public Notice No. 60 became mere private contracts;
 - (b) as a factual matter, all signatories but one had fulfilled their indigenization obligations and the Panel's recommendation in that regard could therefore be limited to that signatory who had not yet reached required indigenization level. This is a separate issue from the consequences of the repeal of Public Notice No. 60 as such. The Panel will however also address it here, since it is raised as a fact of potential relevance to the scope of the Panel's recommendations.
- 8.34 The Panel notes that after 1 April, the situation appears to differ somewhat between the two instruments in which the violating measures are to be found, i.e. Public Notice No. 60 and the MOUs signed thereunder. It was alleged that Public Notice No. 60 no longer applied, and then that it was repealed. It is not alleged that the MOUs, as such, have ceased to apply. Indeed, it has been confirmed that while no new MOUs will be required and that previously restricted products could now be freely imported without licenses, manufacturers are expected to fulfill their indigenization requirements and outstanding export obligations under existing MOUs. However, India has also presented specific arguments to the effect that the nature of the MOUs as mere private contracts since 1 April has also affected how they might be considered as measures for the purposes of WTO dispute settlement.
- 8.35 As in its principal findings, the Panel will address successively the two challenged measures, i.e. the indigenization and trade balancing provisions, and consider within these sections the different issues at stake.
- (a) Indigenization provisions
- 8.36 The Panel recalls that the indigenization provisions, as contained in both Public Notice No. 60 and the MOUs signed thereunder, were found to be inconsistent with the provisions of Article III:4 of the GATT.
- (i) The cessation of the application of Public Notice No. 60 and the MOUs as private contracts
- 8.37 It is not in dispute that the indigenization requirements contained in MOUs already signed by individual manufacturers in accordance with Public Notice No. 60 have remained under the same terms as of 1 April 2001 or any other date. India confirmed that, even after the elimination of import restrictions and licensing requirements on the products subject to Public Notice No. 60, it did not intend to release these manufacturers from their legal obligations under those terms. Nevertheless, because no new MOUs would be signed, there would be no new violation in respect of any person not a signatory as of that date.
- 8.38 However, India considers that since 1 April 2001, the MOUs have become enforceable merely as private contracts. It argues that there is a fundamental difference between a requirement that has to be fulfilled to obtain an import license under the Indian trade laws and a requirement under private contract law with the Government of India, which the Government is free to enforce or not. It

has thus argued that India is free to waive or enforce them, and that they have therefore become discretionary measures which cannot be challenged as such under WTO dispute settlement. 462

- 8.39 India's argument in this regard concentrates on the means of enforcement rather than the existence and enforceability of the primary obligation itself. The fact that the Indian government may have some "discretion" in seeking the actual enforcement of the requirements, in that it could choose to pursue or not pursue its legal rights under the MOUs and enabling legislation, does not fundamentally alter the fact that the obligation already exists. The measure's status as a requirement as at the date of the Panel's establishment does not fundamentally alter, simply because a change has been made in the background circumstances which may then alter the potential means of enforcement.
- 8.40 A distinction must be made between the binding character and enforceability of the commitment, and its actual enforcement. As far as a manufacturer having entered into an MOU is concerned, the commitments already exist and affect commercial behaviour, whether or not the Indian Government ultimately would choose to seek their enforcement through specific administrative or judicial action in the event of breach.⁴⁶³
- 8.41 The Panel is supported in this view by the fact that the Illustrative List of the TRIMs agreement expressly refers to mere enforceability in the context of defining measures which may fall within the scope of Article III:4 as well as Article XI:1.
- 8.42 This view is also supported by the jurisprudence. This Panel has previously alluded to US Malt Beverages where a panel found the mandatory price affirmation laws in Massachusetts and Rhode Island inconsistent with Article III:4 because they accorded imported beer and wine less favourable treatment than the like domestic products even though the States concerned might not have been using their police powers to enforce the legislation. The panel noted that this did not change the fact that the measures were mandatory legislation which may influence the decision of economic operators.
- 8.43 Changing the historical inducement for entering and then complying with the relevant commitments from the potential denial of import licensing under Public Notice No. 60 to inducement for compliance with existing commitments through the potential for contractual or penalty-based litigation, does not change the legal and historical basis upon which the relevant companies came to adopt the obligations in the first place.
- 8.44 The Panel therefore cannot accept India's argument that the legal nature of the MOUs would have been altered after 1 April so as to make them discretionary measures not subject to review under WTO dispute settlement. On the contrary, the MOUs in place after 1 April are those very same legal

The Canada – FIRA panel, in considering an argument by Canada that the undertakings were "private contractual" arrangements, found that:

"The Panel carefully examined the Canadian view that the purchase undertakings should be considered as private contractual obligations of particular foreign investors vis-à-vis the Canadian government. The Panel recognized that investors might have an economic advantage in assuming purchase undertakings, taking into account the other conditions under which the investment was permitted. The Panel felt, however, that even if this was so, private contractual obligations entered into by investors should not adversely affect the rights which contracting parties, including contracting parties not involved in the dispute, possess under Article III:4 of the General Agreement and which they can exercise on behalf of their exporters. This applies in particular to the rights deriving from the national treatment principle, which – as stated in Article III:1 – is aimed at preventing the use of internal measures 'so as to afford protection to domestic production'." (paras 5.4 to 5.6).

⁴⁶² Second Submission, para. 30.

instruments signed by manufacturers prior to 1 April 2001 and that it is only by virtue of this prior commitment that they remain binding after 1 April 2001, despite the disappearance of the supporting government measure which served to induce their signature. Suppressing the advantage that served to induce the manufacturers to sign the MOUs does not fundamentally alter the analysis of the commitments they undertook.

- 8.45 Any signatories who have not attained the stipulated levels are still required to do so. Thus they would still be induced to favour domestic products if they are concerned to comply with their contractual commitments to the government of India. As indicated above, the only potential change relates to the means by which any enforcement of a breach of that obligation would occur, and that change alone does not alter the nature of the commitment entered into. Hence, the Panel concludes that the indigenization obligations contained in the MOUs have not been essentially altered by the events of 1 April 2001.
- 8.46 The indigenization requirements as they stand in existing MOUs thus remain in violation of Article III:4. The cessation or repeal of Public Notice No. 60 has not suppressed this inconsistency. Public Notice No. 60 served to induce the signature of these MOUs, and these instruments contain the effective commitments of manufacturers in relation to indigenization and trade balancing. India has clearly indicated that these MOUs remain in existence. It has also clearly indicated that it was not its intention to release signatories from their remaining obligations thereunder, including the indigenization requirement generally.⁴⁶⁴
- 8.47 In light of the foregoing, the Panel concludes that the indigenization conditions contained in Public Notice No. 60 and in the MOUs, as they have continued to exist and apply after 1 April 2001, have remained in violation of the relevant GATT provisions.
- (ii) Achievement of the required levels of indigenization by MOU signatories
- 8.48 India informed the Panel in the course of the proceedings that among the "most important manufacturers" surveyed in April-May 2001, only one remained below the indigenization threshold with regard to one car model. On that basis, India requested the Panel to "limit any recommendations under Article 19:1 of the DSU to the indigenization requirement that remains to be performed by that particular company with respect to that model". 465
- 8.49 However, the Panel notes that India indicated that the survey on which this assessment is based concerned "the most important manufacturers". It is therefore not clear that the figures given would concern all signatories and that it could be legitimately assumed on the basis of that survey alone that among all those manufacturers having signed the MOUs, this would be the only one still required to fulfil an indigenization requirement.

Para. 3.8, cited above.

⁴⁶⁴ The Panel also notes in this regard that signatories who might have reached the required level, at whatever date, would have, in India's own description, "moved out of the ambit of the MOU" in that regard regardless of the events of 1 April 2001: to the extent that manufacturers would have reached the 70% indigenization level, their situation would arguably not have been fundamentally altered by the continued application or not of Public Notice No. 60: Moving "out of the ambit of the MOU altogether" as described by India, arguably would have the same practical consequences as the general suppression of the import licensing regime, regardless of the continuation of Public Notice No. 60. We note however that there is a persisting disagreement on this issue among the parties, since both complainants remain concerned that it is unclear what the consequences of subsequently "slipping" below the required level may be. The Panel does not find it necessary, for the purpose of its analysis, to rule on this particular issue.

- 8.50 Indeed, in the course of the interim review meeting, the European Communities suggested the name of one company that was not covered by the survey. The European Communities also noted during the interim review meeting that the survey provided by India reflected the manufacturers' own assessment of their fulfilment of their obligations, which may not preclude Indian authorities from reaching a different conclusion. The European Communities also recalls that it is, in its view, still unclear what the effects of falling below the 70% threshold in the future might be. 466
- 8.51 On the basis of the elements before it, the Panel is not in a position to conclude that it has been established that only one signatory remains who needs to fulfill the indigenization requirement and would not be able to limit its recommendation to the obligation to be performed by one of the signatories only. Indeed, it is not in a position to assess the exact modalities under which existing MOU signatories still have to discharge their indigenization obligations. Nor is it, in its view necessary to attempt such an assessment for the purposes of this analysis. In India's own admission, at least one manufacturer yet has to achieve the required level of indigenization and it can thus not be concluded that this issue has lost any practical relevance so as to preclude the Panel from making recommendations with regard to the indigenization requirements as contained in the MOUs.
- (b) Trade balancing provisions
- 8.52 The Panel recalls its earlier finding that the trade balancing obligations, as contained in Public Notice No. 60 and in the MOUs signed thereunder, and as applied as of the date of establishment of this Panel, were in violation of Articles XI:1 and III:4 of the GATT 1994.
- (i) Accrued obligations as independent from imports
- 8.53 India has argued that outstanding trade balancing obligations accrued under the MOUs before 1 April 2001 cannot be in violation of its obligations under GATT 1994 or the TRIMs Agreement, since these obligations will be executed independently of any new imports. In India's view, these obligations will therefore not be inconsistent with India's obligations under the WTO, since the WTO does not generally prohibit export obligations as such. In particular, India argues that they can no longer be in violation of Article XI:1, because they will be executed independently of any new imports.
- 8.54 The Panel notes in the first instance that this argument concerning the fact that these accrued obligations will be executed independently of any new imports appears to be addressed towards the Panel's finding of violation of Article XI:1, and would not be of relevance to its finding on the basis of Article III:4, which was not tied to a link between imports and the amount of exports required.
- 8.55 This argument draws a distinction between the conditions in which the obligations were entered into and accrued in the first place and the circumstances in which they will be executed. In this instance, it is not disputed that any remaining export obligations accrued in relation to restricted imports prior to 1 April 2001. While they might be actually executed independently of the conditions of any *future* imports, this would in fact already have always been the case. Future restricted imports would have generated *additional* export obligations. They would not as such have affected previously accrued obligations.
- 8.56 The fact that, at the time of *execution* of any outstanding export obligations, there will be no future import restrictions does not alter the fact that these particular obligations were *accrued* in relation to imports, in the same way as any other export obligations accrued under the MOUs. They

⁴⁶⁶ Oral Statement at the interim review meeting, para. 12. This concern is also shared by the United States.

were certainly not accrued independently of any imports and they are to be executed under the same terms as were found to be inconsistent with the provisions of both Articles XI:1 and Article III:4 of the GATT 1994. These residual obligations are therefore not different in that respect from any other accrued obligations under the MOUs. Signatories continue to be bound to the execution of conditions which were found to be inconsistent with the provisions of Articles III:4 and XI:1 of the GATT 1994.

- 8.57 The Panel notes that the events of 1 April 2001 did not alter the nature of any export obligation *previously* accrued under the MOUs. The Panel also notes that no evidence was presented to show that any new export obligations have in fact accrued for any imports taking place *after* 1 April 2001.
- 8.58 The essential issue here is that the obligation foreseen in Public Notice No. 60 and reflected in the individual MOUs, which was found to be inconsistent, continues to be binding and to produce effects on those signatories who have not yet fully discharged their export obligations. This issue does not relate to whether any *past execution of* trade balancing obligations might be required to be "undone" or otherwise called into question, but merely to establishing whether the measure previously found to be in violation of two of the GATT provisions continues to have an existence today, so that the Panel would be justified in making a recommendation that this measure be brought into conformity with the relevant agreement as of today.⁴⁶⁷
- (ii) Cessation of the application of Public Notice No. 60 and the MOUs as private contracts
- 8.59 India's argument concerning the alleged changed nature of the MOUs, as constituting mere private contracts, has already been addressed in the previous section. The Panel refers to its analysis under that section to draw the same conclusion in the context of the trade balancing provisions, namely that the MOUs in place after 1 April are those very same legal instruments signed by manufacturers prior to April 2001 and that it is only by virtue of this prior commitment that they remain binding after 1 April 2001, despite the disappearance of the supporting government measure which served to induce their signature.
- 8.60 With regard to the cessation or repeal of Public Notice No. 60, the Panel notes that Public Notice No. 36 (RE-2000) which provides that Public Notice No. 60 "stands repealed", also indicates that "however, export obligation incurred by the MOU signatories in respect of imports made up to 31/3/2001 shall be fulfilled by them within the stipulated period unless extended by the Government for good and sufficient reasons." It thus appears that the repeal of Public Notice No. 60 has in any case not altered the continued existence of export obligations accrued under existing MOUs.
- 8.61 In light of the foregoing, the Panel concludes that the trade balancing conditions contained in Public Notice No. 60 and in the MOUs, as they have continued to exist and apply after 1 April 2001, have remained in violation of the relevant GATT provisions.
- 8.62 As to the question of any future enforcement of the current obligations of MOU signatories, the Panel takes note of and welcomes India's statement that any future measures it may take to enforce the MOUs will not involve any inconsistencies with its WTO obligations. However, the Panel also

The Panel also notes that it does not seek to engage, in this analysis, in an assessment of the exact levels of any currently existing trade balancing obligations to be fulfilled. Indeed, none of the parties have provided any relevant evidence in that regard. While this is an issue which may be relevant in relation to the implementation of any recommendations the DSB might adopt in this case, such an analysis is not required in this context.

 $^{^{468}}$ Public Notice No. 36 (RE-2000) / 1997-2002, 4th September 2001, as annexed by India to its Oral Statement during the interim review meeting, with the indication "To be published in the Gazette of India Extraordinary".

recalls its earlier conclusion that such future actions would in any case not be part of its terms of reference.

8.63 Finally, the Panel recalls that India has asserted that it did not apply any balance-of-payments measures after 1 April 2001, so that any defense which it might have been able to establish for the period prior to that date would in any case not have justified the subsequent continuation of the violation by India of Article XI:1 of the GATT 1994.

C. RECOMMENDATIONS

- 8.64 In accordance with the terms of Article 3.8 of the DSU, the Panel notes that the violations by India of its obligations under the GATT 1994 is considered *prima facie* to constitute a case of nullification and impairment of benefits under that Agreement, and that this has not been rebutted.
- 8.65 The Panel consequently recommends that the DSB request India to bring its measures into conformity with its obligations under the WTO Agreements.

TABLE 1

Auto components licensing policy

The cabinet decision on the new policy for import of cars in CKD and SKD condition was announced by the principal information officer who acts as the spokesperson for the cabinet subsequently. The DGFT released a notice on 12 December, 1997 to announce the import licensing policy for CKD and SKD imports in the auto industry. Essentially, the public notice requires minimum foreign investment of \$50 mm for joint ventures. Further, indigenization up to 70 percent is required by the fifth year. Last, foreign exchange outflow over the five year period of the MOU must be balanced by corresponding inflow from export of cars and auto components subsequently, relaxations have been made in specific cases. Interpretation of the policy too has been simplified. However, the basic Public Notice of 12.12.97 continuous as before without amendment.

Subject: Export and Import Policy April, 1997 – March 2002 – Policy relating to import of CKD/SKD kits/components by Joint Venture Car manufacturer companies under MOU to be signed with the Government of India.

- In exercise of the powers conferred under Paragraph 4.11 of the Export and Import Policy, 1997-2002 as amended from time to time, Director General of Foreign Trade hereby draws attention to the above subject and the parameters stipulated in the year 1995 for import of CKD/SKD kits/components by the Joint Venture Motor Vehicle manufacturer companies under Memorandum of Understanding to be signed by them with the Government of India. These parameters have now been reviewed in the light of the changed circumstances and joint venture motor vehicle manufacturing companies (both existing and new) are required now to sign a fresh MOU with the Government of India as per the revised parameters.
- 2. Pursuant to the above, import of components for motor vehicle in CKD/SKD form, which is restricted for import under the current Export-Import Policy, shall be allowed for importation against a licence and such a licence will be issued only to joint venture automobile manufacturing companies. Thus, all joint venture manufacturers shall enter in to an MOU DGFT for import CKD/SKD kits/components.
- 3. The MOU shall be based on the following parameters:-
 - (i) Establishment of actual production facilities for manufacture of cars and not for mere assembly of imported kits/components.
 - (ii) A minimum foreign equity of US\$50 Million to be brought in by the foreign partner within the first three years of start of operations, if the Joint Venture involves majority foreign equity ownership. However, this condition will apply to new Joint Venture companies only.
 - (iii) Indigenization of components up to a minimum level of 50% in the third year or earlier from the date of clearance of first import consignment of CKD/SKD kits/components and 70% in the 5th year or earlier. Once the MOU signing firm has reached an indigenization level of 70%, there will be no need for further import licences from DGFT. Consequently, as and when the firms achieve 70% indigenization, they would go outside the ambit of the MOU automatically. However, they will discharge the export obligation corresponding to the imports made by them till that time.
 - (iv) Regarding export obligation, the firms entering into MOU would achieve broad trade balancing of foreign exchange over the entire period of the MOU in terms of balancing between the actual CIF value or imports of CKD/SKD kits/components and the FOB value of exports of cars and auto components over the said period. The period of export obligation would commence from the third year of commencement of production. The date of commencement of production would be deemed to be the date of the first release of consignment from factory after payment of excise duty, but there would be a moratorium to two years from this particular date of commencement of production during which the firm need not fulfill any export commitment. However, from the third year onwards, (effective from date of release of first consignment), the MOU signing firm would have an export obligation equivalent to the CIF value of imports made by them till that time for the remainder of the MOU period till they complete the entire export obligation. From 4th year onwards, the value of import of CKD/SKD may be regulated with reference to the export obligation fulfilled in the previous years as per the MOU. The export commitment would be met by export of cars as well as auto components. This export obligation will be over and above the EPCG related export obligation.
- 4. The MOU Scheme would be enforced through the import licensing mechanism and MOU signing firms would be granted import licences by DGFT based on above parameters.

- 5. To monitor the progress in respect of the elements stipulated above all the Joint Ventures would submit annual reports to the GFT on the parameters outlined above and a Joint Annual Review of the progress made in respect of these parameters would be undertaken by Ministry of Commerce, DIPP and Department of Revenue.
- 6. These revised guidelines will apply to all existing and future entrants into this sector.
- 7. By way of exception to the foregoing, companies intending to set up manufacturing units under foreign collaboration for light or heavy commercial vehicles, tractors, earthmoving equipments etc. will not be required to enter into any MOU. Their requests for CKD/SKD imports shall be considered by the Special Licensing Committee on merits on an annual basis.
- 8. A standard format for MOU is enclosed as appendix to this Public Notice and MOU is required to be signed as per this format.

TABLE 2

Memorandum of Understanding (MOU)

I. This MOU has been made on day of 1997-98

between

Government of India acting through the Director-General of Foreign Trade (hereinafter referred to as the DGFT) Udyog Bhawan, New Delhi,

and

M/s.(hereinafter referred to as the party which expression shall be deemed to include their executors, successors, administrators and assignee).

- II. Whereas the party has sought issue of an import licence to cover import of CKD/SKD parts for manufacture of cars.
- III. And whereas the party shall do the following by way of implementation of the Joint Venture:
- (i) that the party shall make an investment of Rs. in this joint venture with equity share of US\$ (Rs....) of M/s...., the foreign partner, which will be in freely convertible currency as per the time frame mentioned hereunder:

Year Foreign Collaborator's
Equity Contribution in freely
convertible currency.
(Rs. in crores)
(US\$ millions)

Total investment (RS. in crores) (US\$ in millions)

- (ii) That a minimum foreign equity of US\$ 50 million to be brought in by the foreign partner within the first three years of start of operations, if the joint venture involves majority foreign equity ownership.
- (iii) that the party shall establish actual manufacturing facilities, and not mere assembly facilities, in India to produce cars. The following will be production volumes year-wise:

Year Production Volumes (Numbers)

(iv) that the party shall achieve indigenization of components up to a minimum level of 50% in the third year or earlier from the date of clearance of first import consignment of CKD/SKD kits and 70% in the fifth year or earlier. Once the party has reached an indigenization level of 70% there will be no need for further import licence from DGTF. However, the party shall discharge the export obligation corresponding to the imports made by them till that time.

That the party intends to achieve the following levels of indigenization of their product, year-wise:

Year Percentage (%) of indigenization

The party shall aggressively pursue and achieve as soon as possible the development of the local supply base and increased local content, since the same will allow a higher level of indigenization.

(v) that the party intends to import the following number of kits with CIF value as indicated in the first five years:

Year No. of Kits CIF value (Rs. in crores) (US\$ in millions)

(vi) That the party shall achieve a broad trade balancing of foreign exchange over the entire period of the MOU in terms of balancing between the actual CIF value of imports of CKD/SKD/components and the FOB value of export of cars and auto components over the said period. The period of export obligation would commence from the third year of commencement of production. The date of commencement of production would be deemed to be the date of the first release of consignment from factory after payment of excise duty, but where would be a moratorium of two years from this particular date of commencement of production during which the firm need not fulfill any export commitment. However, from the third year onwards (effective from date of release of first consignment), the MOU signing firm would have an export obligation equivalent to the CIF value of imports made by them till that time for the remainder of the MOU period till they complete the entire export obligation. From 4th year onwards the value of imports of CKD/SKD may be regulated with reference to the export obligation fulfilled in the previous year as per the MOU. The export commitment would be met by export of only cars as well as auto components,. This export obligation will be over and above the EPCG related export obligation.

That the party intends to achieve export of cars and auto components as under year-wise:

Year Exports (RS. crores) Exports (US\$ in million)

- IV. To monitor the progress in respect of the elements stipulated above the party shall submit annual reports to the DGFT on the parameters outlined above and further licences will be issued to the party on the basis of an annual report of the progress made in relation to these parameters.
- V. The MOU Scheme would be enforced through the import licensing mechanism and MOU signing firms would be granted import licences by DGFT based on the progress made in respect of the parameters mentioned at para. III above.

Managing Director	Director General of Foreign Trade
On behalf of	On behalf of Govt. of India
M/s	M/s
Witnesses:	Witnesses:
1.	1.
2.	2.

WORLD TRADE

ORGANIZATION

WT/DS146/R/Corr.1 WT/DS175/R/Corr.1

10 January 2002

(02-0094)

Original: English

INDIA — MEASURES AFFECTING THE AUTOMOTIVE SECTOR

Report of the Panel

Corrigendum

Paragraph 6.18

Should read as follows:

"In footnote 414, the Panel added the relevant reference to the European Communities' argument as requested by it."

Paragraph 7.21

The last sentence should read as follows:

"The European Communities, similarly, requested the Panel to find that the trade balancing requirements contained in Public Notice No. 60 and in the MOUs concluded thereunder were inconsistent with Articles III and XI of the GATT 1994 and Article 2 of the TRIMs Agreement and that the indigenization requirements contained in these measures are inconsistent with Article III of the GATT 1994 and with Article 2 of the TRIMs Agreement. 308 "

Paragraph 7.86

The second and third sentences should reads as follows:

"Nor did that panel expressly consider the earlier MOUs applicable to restricted kits and components at the time of establishment of that panel. The *India – Quantitative Restrictions* panel based its finding of violation in relation to the discretionary import licensing on the more general observation that licenses were granted on the basis of "unspecified 'merits".

Paragraph 7.211

The second sentence should read as follows:

"The first is that, although there is no express provision to that effect in Public Notice No. 60, it has been argued by the complainants and confirmed by India that purchases by an MOU signatory within India of previously imported CKD/SKD kits/components that were subject to a license requirement on their importation, also count towards the signatory's export balancing requirement.⁴⁰⁵ "

WT/DS146/R/Corr.1 WT/DS175/R/Corr.1 Page 2

Paragraph 7.230

The last sentence should read as follows:

"It indicated that evidence concerning calculations performed in the past and signatories' annual report under the MOUs were not available to it, and relied on the plain meaning of the documents concerned. 410 bis

^{410 bis} See EC's response to Question 94 of the Panel."

Paragraph 7.234

The last sentence should be replaced with the following:

"The Panel's ruling does not seek to determine, or prejudge, any determination which might be made in other contexts relating to the extent of specifications under individual MOUs."

Paragraph 7.287

The fourth sentence should read as follows:

"More importantly, areas of coverage such as safeguards, anti-dumping and countervailing duties require domestic analysis where affected foreign interests may be represented and may use local courts to challenge bureaucratic determinations."

Footnote 414

Footnote 414 should read as follows:

"See First Submission of the European Communities, para. 74. See also US response to Question 82 of the Panel."

Downloaded from worldcourts.com. Use is subject to terms and conditions. See worldcourts.com/terms.htm

WORLD TRADE

ORGANIZATION

WT/DS146/R/Corr.1 WT/DS175/R/Corr.1

10 January 2002

(02-0094)

Original: English

INDIA — MEASURES AFFECTING THE AUTOMOTIVE SECTOR

Report of the Panel

Corrigendum

Paragraph 6.18

Should read as follows:

"In footnote 414, the Panel added the relevant reference to the European Communities' argument as requested by it."

Paragraph 7.21

The last sentence should read as follows:

"The European Communities, similarly, requested the Panel to find that the trade balancing requirements contained in Public Notice No. 60 and in the MOUs concluded thereunder were inconsistent with Articles III and XI of the GATT 1994 and Article 2 of the TRIMs Agreement and that the indigenization requirements contained in these measures are inconsistent with Article III of the GATT 1994 and with Article 2 of the TRIMs Agreement. 308 "

Paragraph 7.86

The second and third sentences should reads as follows:

"Nor did that panel expressly consider the earlier MOUs applicable to restricted kits and components at the time of establishment of that panel. The *India – Quantitative Restrictions* panel based its finding of violation in relation to the discretionary import licensing on the more general observation that licenses were granted on the basis of "unspecified 'merits".

Paragraph 7.211

The second sentence should read as follows:

"The first is that, although there is no express provision to that effect in Public Notice No. 60, it has been argued by the complainants and confirmed by India that purchases by an MOU signatory within India of previously imported CKD/SKD kits/components that were subject to a license requirement on their importation, also count towards the signatory's export balancing requirement.⁴⁰⁵ "

WT/DS146/R/Corr.1 WT/DS175/R/Corr.1 Page 2

Paragraph 7.230

The last sentence should read as follows:

"It indicated that evidence concerning calculations performed in the past and signatories' annual report under the MOUs were not available to it, and relied on the plain meaning of the documents concerned. 410 bis

^{410 bis} See EC's response to Question 94 of the Panel."

Paragraph 7.234

The last sentence should be replaced with the following:

"The Panel's ruling does not seek to determine, or prejudge, any determination which might be made in other contexts relating to the extent of specifications under individual MOUs."

Paragraph 7.287

The fourth sentence should read as follows:

"More importantly, areas of coverage such as safeguards, anti-dumping and countervailing duties require domestic analysis where affected foreign interests may be represented and may use local courts to challenge bureaucratic determinations."

Footnote 414

Footnote 414 should read as follows:

"See First Submission of the European Communities, para. 74. See also US response to Question 82 of the Panel."

Downloaded from worldcourts.com. Use is subject to terms and conditions. See worldcourts.com/terms.htm